INTERSECTION COMPLEXES AND UNRAMIFIED L-FACTORS

YIANNIS SAKELLARIDIS AND JONATHAN WANG

ABSTRACT. Let X be an affine spherical variety, possibly singular, and L^+X its arc space. The intersection complex of L^+X , or rather of its finite-dimensional formal models, is conjectured to be related to special values of local unramified L-functions. Such relationships were previously established in [BFGM02] for the affine closure of the quotient of a reductive group by the unipotent radical of a parabolic, and in [BNS16] for toric varieties and L-monoids. In this paper, we compute this intersection complex for the large class of those spherical G-varieties whose dual group is equal to \check{G} , and the stalks of its nearby cycles on the horospherical degeneration of X. We formulate the answer in terms of a Kashiwara crystal, which conjecturally corresponds to a finite-dimensional \check{G} -representation determined by the set of B-invariant valuations on X. We prove the latter conjecture in many cases. Under the sheaf–function dictionary, our calculations give a formula for the Plancherel density of the IC function of L^+X as a ratio of local L-values for a large class of spherical varieties.

Contents

1.	Introduction	2
1.1.	Arc spaces and their IC functions	2
1.2.	Asymptotics and L -functions	6
1.3.	Zastava spaces and the main theorems in terms of sheaves	9
1.4.	Reduction to canonical affine closure	12
1.5.	Semi-infinite orbits and dimension estimates	13
1.6.	Organization of the paper	14
1.7.	Index of notation	14
1.8.	Acknowledgments	18
2.	Spherical varieties and their arc spaces	18
2.1.	Spherical varieties	18
2.2.	Affine degeneration, and assumptions in positive characteristic	20
2.3.	The formal loop space	22
3.	Models for the arc space	25
3.1.	Global model	25
3.2.	Toric case	27
3.3.	Zastava model	29
3.4.	Graded factorization property	30
3.5.	Global-to-Zastava yoga	30
3.6.	Stratification of the Zastava model	32
3.7.	Relation to the affine Grassmannian	32
3.8.	Theorem of Grinberg–Kazhdan, Drinfeld	34
4.	Compactification of the Zastava model	35
4.1.	1 1	35
4.2.	Stratification	37
4.3.	The central fiber	38

4.4. Dimension of central fibers	40
4.5. Comparison of IC complexes	41
5. Global Hecke action and closure relations	45
5.1. Main results of this section	45
5.2. Hecke action on global model	47
5.3. Reduction to $\mathfrak{c}_{X^{\bullet}} = \mathbb{N}^{\mathcal{D}}$	49
5.4. Open Zastava	51
5.5. Proof of Theorem 5.1.1	52
5.6. Closure relations and components in the global model	58
5.7. Closure relations and components in the Zastava model	61
6. Stratified semi-smallness	62
6.1. Upper bounds on dimension	63
6.2. Connected components of open Zastava	64
6.3. Stratified semi-smallness	65
6.4. Euler product	66
6.5. Critical dimension	69
7. The crystal of a spherical variety	70
7.1. The crystal \mathfrak{B}_X	70
7.2. Reduction to Levi	74
7.3. Lowering operators via hyperplane intersections	77
8. Nearby cycles	80
8.1. Principal degeneration	80
8.2. Grinberg–Kazhdan theorem in families	81
8.3. Results on nearby cycles	83
8.4. Contraction principle	86
9. Function-theoretic corollaries	87
9.1. Pushforward of the basic function	87
9.2. Asymptotics and the basic function	88
9.3. Euler factors of global integrals	88
Appendix A. Properties of the global stratification	92
A.1. The factorizable space of formal loops	92
A.2. Multi-point orbits	93
A.3. Proof of Lemma 3.1.6	94
A.4. Generic-Hecke modifications	95
A.5. Proof of Theorem 5.1.5	98
References	101

1. Introduction

1.1. Arc spaces and their IC functions. Let \mathbb{F} be a finite field, G a connected reductive group over \mathbb{F} , and X an affine spherical G-variety. The formal arc space L^+X is the infinite-dimensional scheme that represents the functor $R \mapsto X(R[\![t]\!])$, and it is singular, if X is. However, its singularities at "generic" \mathbb{F} -points (namely, arcs $\mathrm{Spec}\,\mathbb{F}[\![t]\!] \to X$ which generically lie in the smooth locus X^{sm}) are of finite type, according to the theorem of Grinberg–Kazhdan and Drinfeld, and this allows one to define an IC function [BNS16], that is, the function that should correspond under Frobenius trace to the "intersection complex" of \mathbb{L}^+X . This is a function Φ_0 on $X(\mathfrak{o}) \cap X^{\mathrm{sm}}(F)$ (where $\mathfrak{o} = \mathbb{F}[\![t]\!]$ and $F = \mathbb{F}(\!(t)\!)$), and it was conjectured in

[Sak12] (before the rigorous definition of this function was available) that it is related to special values of *L*-functions. Such a relation was established in [BNS16] for toric varieties and certain group embeddings, termed *L*-monoids, generalizing the local unramified Godement–Jacquet theory. One goal of the present paper is to prove such relations, for a different and broad class of spherical varieties.

In order to compute the IC function, we need to work with finite-dimensional global models of the arc space, or rather, the arc space over the algebraic closure k of \mathbb{F} ; from now on, we base change to k, without changing notation. However, for motivational purposes, let us here pretend that L^+X were already finite-dimensional, and the intersection complex on it were defined as a $\overline{\mathbb{Q}}_\ell$ -valued constructible (derived) sheaf, for some prime ℓ different from the characteristic of \mathbb{F} . We would normalize such a sheaf to be constant in degree zero over the arc space of the smooth locus $\mathsf{L}^+X^\mathrm{sm}$, and we would normalize the intersection complex of a substratum $S \subset \mathsf{L}^+X$ of codimension d to be $\overline{\mathbb{Q}}_\ell(\frac{d}{2})[d]$ on its smooth locus, where $\overline{\mathbb{Q}}_\ell(\frac{1}{2})$ denotes a chosen square root of the cyclotomic Tate twist.

The IC function is computed via some version of a "Satake transform" for the spherical variety — we will recover the original function from its Satake transform in Corollary 1.2.1. Let $B \supset N$ be a Borel subgroup of G, and its unipotent radical, and consider the invariant-theoretic quotient $X/\!\!/N = \operatorname{Spec} k[X]^N$, which is an affine embedding of a quotient T_X of the Cartan T = B/N. In this paper, we restrict ourselves to varieties where B acts freely on an open subset X° of X, so let us already make this assumption for notational simplicity. Then $T_X = T$, and we fix a base point to identify T as the open orbit in $X/\!\!/N$. In fact, our assumption on X is stronger, requiring that the (Gaitsgory–Nadler) dual group of X, G_X , is equal to the Langlands dual group of G — this condition is equivalent to the following:

B acts freely on X° , and for every simple root α , if P_{α} is the parabolic (1.1) generated by B and the root space of $-\alpha$, the stabilizer of a point in the open P_{α} -orbit $X^{\circ}P_{\alpha}$ is a torus (necessarily one-dimensional).

The pushforward map $\pi: X \to X/\!\!/ N$ induces a map between arc spaces. Under our assumptions, the "generic" L⁺T-orbits on the arc space of $X/\!\!/ N$, that is, those corresponding to arcs whose generic fiber lands in the open T-orbit, are naturally parametrized by a strictly convex (i.e., not containing non-trivial subgroups) submonoid $\mathfrak{c}_X \subset \check{\Lambda}$ of the cocharacter group of T, with $\check{\lambda} \in \mathfrak{c}_X$ corresponding to the image $t^{\check{\lambda}} := \check{\lambda}(t)$ of a uniformizer (acting on a fixed base point of $X/\!\!/ N$). The pushforward $\pi_! \mathrm{IC}_{\mathsf{L}^+ X}$ of the IC sheaf is $\mathsf{L}^+ T$ -equivariant, and under Frobenius trace translates to a $T(\mathfrak{o})$ -invariant function on $X/\!\!/ N(\mathfrak{o}) \cap T(F)$, which will be denoted by $\pi_! \Phi_0$. Explicitly,

$$\pi_!\Phi_0(a) = \int_{N(F)} \Phi_0(an) dn,$$

is a Radon transform on the spherical variety, i.e., the integral of the IC function Φ_0 over *generic horocycles*, that is, over the fibers of the map $X(\mathfrak{o}) \to X/\!\!/ N(\mathfrak{o})$, where the Haar measure on N(F) is so that $dn(N(\mathfrak{o})) = 1$.

This integral is really a finite sum, hence makes sense over $\overline{\mathbb{Q}}_{\ell}$, but let us for simplicity choose an isomorphism $\overline{\mathbb{Q}}_{\ell} \simeq \mathbb{C}$, such that the geometric Frobenius morphism Fr acts on the chosen half-Tate twist $\overline{\mathbb{Q}}_{\ell}(\frac{1}{2})$ by $q^{\frac{1}{2}}$. Our results and conjectures are best expressed under the assumption that X carries a G-eigen-volume form; we will assume this for the rest of the

¹To be precise, we are referring to the modification of the Gaitsgory–Nadler dual group described in [SV17], because the Gaitsgory–Nadler dual group would be \check{G} even if the stabilizers were *normalizers* of tori. This small distinction is important, and such cases (for example, $O_n\backslash GL_n$) are *not* expected to be directly related to L-functions, and not included in the present paper.

introduction. The absolute value of the volume form is a G(F)-eigenmeasure on the open orbit $X^{\bullet}(F)$, whose eigencharacter we will denote by η . (We will not impose this assumption in the rest of the paper, but see Remark 5.4.4.) Then, one should consider the following normalized form of the integral above, analogous to the standard normalization of the Satake isomorphism:

(1.2)
$$(\eta \delta)^{\frac{1}{2}}(a)\pi_!\Phi_0(a) = (\eta \delta)^{\frac{1}{2}}(a)\int_{N(F)}\Phi_0(an)dn,$$

where $\delta = |e^{2\rho_G}|$ is the modular character of the Borel subgroup.²

Ideally, we would like to prove a conjecture such as the following. To formulate it, recall that $T(\mathfrak{o})$ -orbits on T(F) are parametrized by Galois-fixed (that is, Fr-fixed) elements of Λ .

Conjecture 1.1.1. There is a symplectic representation ρ_X of the L-group, $\rho_X: {}^LG = \check{G} \rtimes {}^LG$ $\langle \operatorname{Fr} \rangle \to \operatorname{GL}(V_X)$, and a LT -stable polarization $V_X = V_X^+ \oplus V_X^-$, such that the multiset \mathfrak{B}^+ of \check{T} -weights of V_X^+ belongs to \mathfrak{c}_X , and the pushforward of the IC function satisfies:

$$(1.3) \qquad (\eta \delta)^{\frac{1}{2}} \cdot \pi_! \Phi_0 = (tr_{\check{T}}(\operatorname{Fr}, \operatorname{Sym}^{\bullet}(\check{\mathfrak{n}}(1))))^{-1} \cdot tr_{\check{T}}(\operatorname{Fr}, \operatorname{Sym}^{\bullet}(V_X^+)).$$

Here, for a representation V of ${}^LT = \check{T} \rtimes \langle \operatorname{Fr} \rangle$, the expression $tr_{\check{T}}(\operatorname{Fr}, V)$ denotes the function on $\check{\Lambda}^{\operatorname{Fr}}$ whose value on $\check{\lambda}$ is equal to the trace of geometric Frobenius Fr on the $(\check{T},\check{\lambda})$ -eigenspace

From the point of view of number theory, the spherical varieties satisfying our assumption $\check{G}_X = \check{G}$ give, in some sense, the most interesting periods, because they are associated to L-values at the center of the critical strip. Indeed, one should always be able to choose the eigencharacter η such that the Frobenius morphism acts on V_X^+ by permuting elements of a basis and scaling by $q^{\frac{1}{2}}$. For example, when G is split and the colors (see below) of X are all defined over \mathbb{F} , this permutation action should be trivial, and (1.3) should read:

(1.4)
$$(\eta \delta)^{\frac{1}{2}} \pi_! \Phi_0 = \frac{\prod_{\check{\alpha} \in \check{\Phi}^+} (1 - q^{-1} e^{\check{\alpha}})}{\prod_{\check{\lambda} \in \mathfrak{B}^+} (1 - q^{-\frac{1}{2}} e^{\check{\lambda}})},$$

where \mathfrak{B}^+ is the multiset of weights, as in the conjecture, and $e^{\check{\lambda}}$ denotes the characteristic function of the $T(\mathfrak{o})$ -orbit of t^{λ} .

We will explain the relationship of this conjecture to various conjectures of arithmetic and geometric origin below. We do not quite prove the conjecture in all cases, but we determine the weights of ρ_X (in terms of X) and in the cases where ρ_X is minuscule, we prove the conjecture (Corollary 7.1.12). It is helpful to distinguish between the special case when $X = \overline{H \setminus G}^{\text{aff}}$ is the affine closure of a homogeneous quasiaffine variety, and the general case. In the special case, let us also assume, at first, that the monoid \mathfrak{c}_X is freely generated with a basis $\check{\nu}_1,\ldots,\check{\nu}_r$, so $X/\!\!/N$ may be identified with \mathbb{A}^r . This condition can always be achieved by passing to an abelian cover of $H\backslash G$ and taking its affine closure, see §5.3. In that case, the generators $\check{\nu}_i$ are the valuations associated to the *colors*, that is, the prime B-stable divisors of $H\backslash G$.

Theorem 1.1.2 (See §9.1). Assume that $X = \overline{H \backslash G}^{aff}$ satisfies the conditions above $(T_X = T, \check{G}_X = \check{G} \text{ and } \mathfrak{c}_X \cong \mathbb{N}^r \text{ is free})$. Then there is a $(\check{T} \rtimes \langle \operatorname{Fr} \rangle)$ -representation V_X^+ satisfying:

- (i) the Ť-weights of V_X⁺ belong to c_X − 0;
 (ii) the set of weights of V_X⁺ ⊕ (V_X⁺)* (without multiplicities) equals the set of weights of a \check{G} -representation ρ_X ;

²We use additive notation for the character group $\check{\Lambda}$ of \check{T} , so $e^{\check{\nu}}$ will denote the actual morphism $\check{T} \to \mathbb{G}_m$ corresponding to $\check{\nu} \in \check{\Lambda}$.

- (iii) the dimensions of the weight spaces of $V_X^+ \oplus (V_X^+)^*$ are invariant under the Weyl group W of G;
- (iv) the weight spaces in V_X^+ of the basis elements $\check{\nu}_1, \ldots, \check{\nu}_r$ of \mathfrak{c}_X have multiplicity one,
- (v) under the Frobenius action, we have

$$(1.5) V_X^+ = \bigoplus_{\mathfrak{B}^+} \overline{\mathbb{Q}}_{\ell}(\frac{1}{2}),$$

for some permutation action of Fr on the multiset \mathfrak{B}^+ of weights, compatible with its action on \check{T} ,

such that the pushforward $\pi_!\Phi_0$ of the IC function³ satisfies the formula (1.3) above.

In fact, we show more: we endow the multiset $\mathfrak{B} := \mathfrak{B}^+ \sqcup (-\mathfrak{B}^+)$ of weights of V_X with the structure of a Kashiwara crystal, see Theorem 1.3.2 below, and show that, if Conjecture 1.1.1 holds (equivalently, if the crystal is the one corresponding to the crystal basis of a finite-dimensional \check{G} -module), then ρ_X must be the direct sum of the irreducible \check{G} -modules with highest weights in $\check{\Lambda}^+ \cap W\{\check{\nu}_1,\ldots,\check{\nu}_r\}$, each with multiplicity one (see Remark 7.1.11). In other words, the highest weights of ρ_X are the dominant coweights that are Weyl translates of the basis elements $\check{\nu}_1,\ldots,\check{\nu}_r$.

As we already mentioned, Theorem 1.1.2 implies Conjecture 1.1.1 when ρ_X is minuscule. In particular, when $H\backslash G$ is itself affine (equivalently, H is reductive: see [Lun73, Ric77]), we observe that ρ_X always has multiplicity-free weight spaces (Corollary 7.3.4). In this case Conjecture 1.1.1 was previously proved by [Sak13, Theorem 7.2.1], under some additional assumptions, and Theorem 1.1.2 gives a geometric interpretation of this result.

For an example when $H\backslash G$ is not affine:

Example 1.1.3. Let X^{\bullet} = the quotient of SL_2^n by the unipotent subgroup

$$H_0 = \left\{ \left(\begin{array}{cc} 1 \\ x_1 & 1 \end{array} \right) \times \left(\begin{array}{cc} 1 \\ x_2 & 1 \end{array} \right) \times \cdots \times \left(\begin{array}{cc} 1 \\ x_n & 1 \end{array} \right) \middle| x_1 + x_2 + \cdots + x_n = 0 \right\},$$

under the action of G = the quotient of $\mathbb{G}_m \times \mathrm{SL}_2^n$ by the diagonal copy of μ_2 , where $a \in \mathbb{G}_m$ acts as left multiplication by $\binom{a^{-1}}{a}$. Let X be the affine closure of X^{\bullet} . Denoting by \check{m} the identity cocharacter of \mathbb{G}_m , the monoid \mathfrak{c}_X is freely generated by the coweights

$$\frac{\check{\alpha}_1 + \check{\alpha}_2 + \dots + \check{\alpha}_n - \check{m}}{2}$$

and

$$\frac{-\check{\alpha}_1 - \dots - \check{\alpha}_{i-1} + \check{\alpha}_i - \dots - \check{\alpha}_n + \check{m}}{2}, \ i = 1, \dots, n,$$

see Remark 2.1.3. These are minuscule weights of $\check{G} = \operatorname{GL}_2^n/\{(z_i)_i \in \mathbb{G}_m^n | \prod_i z_i = 1\}$, and in that case Conjecture 1.1.1 holds, confirming an expectation of [Sak12, §4.5].

For the general case, X contains an open G-orbit $X^{\bullet} = H \backslash G$, which we will assume to satisfy the conditions of Theorem 1.1.2, except perhaps for the freeness of $\mathfrak{c}_{X^{\bullet}}$. In that case, the free monoid $\mathbb{N}^{\mathcal{D}}$, where \mathcal{D} denotes the set of colors, maps through the valuation map to $\mathfrak{c}_{X^{\bullet}}$, and \mathfrak{c}_{X} is generated by its image and a minimal set $\mathcal{D}^{G}_{\mathrm{sat}}(X) = \{\check{\theta}_{1}, \ldots, \check{\theta}_{d}\}$ of distinct antidominant elements of the cocharacter group $\check{\Lambda}$ of T. For each $\check{\theta}_{i}$, we let $V^{\check{\theta}_{i}}$ be the irreducible module of \check{G} with lowest weight $\check{\theta}_{i}$, and assume that the eigencharacter η of the G-eigenmeasure on X(F) is of the form $|e^{\mathfrak{h}}|$ for some algebraic character $\mathfrak{h} \in \Lambda^{W}$.

This doesn't imply minuscule, so we didn't prove Conjecture

Did [SaSph] prove existence of representation vs virtual rep?

³Under the assumptions of the theorem, the restriction of the eigencharacter η to the colors $\check{\nu}_i$ is uniquely determined, see Remark 5.4.4.

Theorem 1.1.4 (See §9.1). In the setting above, if $V_{X\bullet}^+$ denotes the \check{T} -representation for \overline{X}^{\bullet} described in Theorem 1.1.2,⁴ then the pushforward $(\eta\delta)^{\frac{1}{2}} \cdot \pi_! \Phi_0$ of the IC function for L^+X is given by (1.3) with

$$(1.6) V_X^+ = V_{X^{\bullet}}^+ \oplus \bigoplus_i V^{\check{\theta}_i} \left(\frac{\left\langle \mathfrak{h} + 2\rho_G, \check{\theta}_i \right\rangle}{2} \right).$$

Notice that the set $\mathcal{D}_{\text{sat}}^G(X)$ is stable under the Frobenius morphism. The action of Frobenius on the sum of $V^{\check{\theta}_i}$'s is the one obtained by identifying the crystal basis of this space with a set of subvarieties of the affine Grassmannian (see Section 7), and considering the Frobenius action on those.

The reader should compare the passage from Theorem 1.1.2 to Theorem 1.1.4 to the passage from a reductive group G to an L-monoid X in [BNS16, BNS17, Theorem 4.1]. While the result is similar, however, the straightforward proof of [BNS16] uses the monoid structure on X in a crucial way, and cannot be used here.

In §1.3 below we will describe the sheaf-theoretic statements of these theorems, in the setting of appropriate finite type models, the *Zastava spaces* for X and $X/\!\!/N$. Before we do that, let us relate the results above to conjectures in number theory and geometry.

1.2. **Asymptotics and** L-functions. The Radon transform $\pi_!\Phi_0$ of the IC function (also known as "basic function") under the map $X \to X/\!\!/N$ admits various interpretations in terms of harmonic analysis, and, in particular, allows us to compute the Plancherel density of the basic function,

(1.7)
$$\|\Phi_0\|^2 = \int_{\tilde{T}^1/W} \Omega(\chi) d\chi,$$

that is, the decomposition of its norm in the space $L^2(X^{\bullet}(F))$ (with respect to the fixed eigenmeasure) in terms of seminorms $\| \bullet \|_{\chi}$ that factor through eigenquotients for the action of the unramified Hecke algebra. The variable $\chi \in \check{T}^1$ here denotes an unramified unitary character of T(F) (an element of the maximal compact subgroup of the complex dual Cartan), which modulo the action of W represents the Satake parameter of such an eigenquotient, and $\Omega(\chi) = \|\Phi_0\|_{\chi}^2$.

The passage from $\pi_!\Phi_0$ to the Plancherel formula (1.7) is achieved through the theory of asymptotics, which is of geometric interest because of its relation to nearby cycles. For an affine spherical variety X over a field k in characteristic zero, one can define its horospherical degeneration (or asymptotic cone) X_{\emptyset} , by passing to the associated graded of the coordinate ring k[X] as a G-module. A similar degeneration exists under some assumptions in positive characteristic, see §8.1. Under our current assumptions, its open G-orbit X_{\emptyset}^{\bullet} is isomorphic to $N^-\backslash G$, where we use N^- to denote the unipotent radical of a Borel B^- opposite to B — an expository choice without mathematical significance, which we will use to identify the abstract Cartan T = B/N as an automorphism group of X_{\emptyset} by identifying it with the torus $B^- \cap B$ (the latter acting "on the left" on $N^-\backslash G$).

The theory of asymptotics states that there is a canonical morphism

$$e_{\emptyset}^*: C^{\infty}(X^{\bullet}(F)) \to C^{\infty}(X_{\emptyset}^{\bullet}(F))$$

which describes the behavior of any function "at infinity", see [SV17, §5]. Of interest to us is that the spaces $X/\!\!/N$ and $X_{\emptyset}/\!\!/N$ are canonically identified, and the corresponding pushforwards

⁴See §5.3 for a reduction to the case where $\mathfrak{c}_{X} \bullet$ is free.

 $\pi_!$ and $\pi_{\emptyset!}$ satisfy

$$\pi_! = \pi_{\emptyset!} \circ e_{\emptyset}^*$$

(for appropriate functions, in order to ensure convergence), see [SV17, Proposition 5.4.6].

Inverting the Radon transform, (1.8) leads to a calculation of the asymptotics of the basic function. This, in turn, leads to a calculation of the basic function itself, as a function on $X^{\bullet}(F)/G(\mathfrak{o})$. To express both, we note that there is a natural parametrization of the coset space $X^{\bullet}_{\emptyset}(F)/G(\mathfrak{o})$ by the Frobenius-stable elements of the coweight lattice $\check{\Lambda}$ (simply assigning $\check{\theta} \in \check{\Lambda}^{\operatorname{Fr}}$ to the orbit of $N^-t^{\check{\theta}}$), and of the coset space $X^{\bullet}(F)/G(\mathfrak{o})$ by the Frobenius-stable elements of its antidominant submonoid $\check{\Lambda}^-$ — this is explained in Theorem 2.3.5, in the geometric setting, from which we will borrow the notation $x_0t^{\check{\theta}}$ to represent the orbit corresponding to $\check{\theta} \in \check{\Lambda}^{-,\operatorname{Fr}}$. Then we have:

Corollary 1.2.1 (See §9.2). In the setting of Theorem 1.1.4, we have

$$(1.9) (\eta \delta)^{\frac{1}{2}}(a)e_{\emptyset}^*\Phi_0(N^-aG(\mathfrak{o})) = tr_{\check{T}}(\operatorname{Fr}, \operatorname{Sym}^{\bullet}(\check{\mathfrak{n}}))^{-1} \cdot tr_{\check{T}}(\operatorname{Fr}, \operatorname{Sym}^{\bullet}(V_X^+)),$$

and

$$(1.10) \qquad (\eta \delta)^{\frac{1}{2}}(a)\Phi_0(x_0 a G(\mathfrak{o})) = tr_{\tilde{T}}(\operatorname{Fr}, \operatorname{Sym}^{\bullet}(\check{\mathfrak{n}}))^{-1} \cdot tr_{\tilde{T}}(\operatorname{Fr}, \operatorname{Sym}^{\bullet}(V_X^+))|_{\check{\Lambda}^-, \operatorname{Fr}},$$

where we use the notation $tr_{\tilde{T}}(\cdots)$, as in Conjecture 1.1.1, to represent a $T(\mathfrak{o})$ -invariant function on T(F).

For example, in the setting of (1.4), the right hand side reads

$$\frac{\prod_{\check{\alpha}\in\check{\Phi}^+}(1-e^{\check{\alpha}})}{\prod_{\check{\lambda}\in\mathfrak{R}^+}(1-q^{-\frac{1}{2}}e^{\check{\lambda}})}.$$

Finally, the Plancherel decomposition of Φ_0 coincides⁵ up to the factor $|W|^{-1}$ with that of $e_{\emptyset}^*\Phi_0$, which by Mellin transform on X_{\emptyset}^{\bullet} (with respect to the action of T) gives:

Here $\mathfrak{B} = \mathfrak{B}^+ \sqcup (-\mathfrak{B}^+)$, and $L(\chi, V_X, 0)$ denotes a local unramified L-factor, while the density $\frac{d\chi}{L(\chi, \tilde{\mathfrak{g}}/\tilde{\mathfrak{f}}, 0)}$ is the unramified Plancherel measure for G.

The importance of (1.11) for arithmetic is that, according to the generalized Ichino–Ikeda conjecture of [SV17], the quotient of the Plancherel density of Φ_0 by the Plancherel measure of G is related to the local Euler factor of the "X-period integral" of automorphic forms. A provable case of such an Euler factorization is related to Example 1.1.3:

Example 1.2.2. Let G, X be as in Example 1.1.3, over the function field $\mathbb{k} = \mathbb{F}(C)$ of a curve C, and let $\Phi = \prod_v \Phi_v$ be a smooth, factorizable function of moderate growth on the adelic points of the open G-orbit X^{\bullet} , such that the support of Φ_v has compact closure in $X(F_v)$, and for almost every v the function Φ_v is equal to the IC function of X. Let $\Theta_{\Phi}(g) = \sum_{\gamma \in X^{\bullet}(\mathbb{k})} \Phi(\gamma g)$ be the corresponding theta series, a function on the adelic points of G.

Let ϕ be a cusp form, belonging to a cuspidal automorphic representation $\pi = \bigotimes_{v}' \pi_{v}$, and let $W_{\phi}(g) = \prod_{v} W_{\phi,v}(g_{v}) = \int_{N^{-}(\mathbb{R}) \setminus N^{-}(\mathbb{A})} \phi(ng) \psi(n) dn$ be the Whittaker function of ϕ (where

 $^{{}^5\}mathrm{See}$ the proof of Proposition 9.3.1.

 N^- is the lower triangular subgroup), with a chosen Euler factorization with $W_{\phi,v}(1) = 1$ for almost all v. It can be proven by the usual unfolding argument that the pairing

$$\int_{G(\mathbb{k})\backslash G(\mathbb{A})} \phi(g)\Theta_{\Phi}(g)dg$$

is convergent if the central character of ϕ is "large" enough (i.e., its restriction to \mathbb{G}_m has absolute value $|\bullet|^s$ for some $s \gg 0$), and equal to the Euler product of zeta integrals

$$\int_{H_0\backslash G(\mathbb{K}_v)} W_{\phi,v}(g_v) \Phi_v(g_v) dg_v.$$

Our Theorem 1.1.2 implies that almost every Euler factor is equal to the local unramified L-factor $L(\pi_v, \otimes, 1 - \frac{n}{2})$, where \otimes is the tensor product representation of \check{G} . We explain this in §9.3.

Such global applications are beyond the main focus of this paper. Of more immediate interest here is the relation of the asymptotics map e_{\emptyset}^* to nearby cycles: Since X_{\emptyset} is obtained by degenerating the coordinate ring of X, there is an associated $\mathbb{G}_m \times G$ -equivariant Rees family $\mathfrak{X} \to \mathbb{A}^1$ (depending, really, on the choice of a strictly dominant cocharacter into T), whose general fiber is isomorphic to X, and whose special fiber is isomorphic to X_{\emptyset} . This also induces a family of arc spaces, or loop spaces $\mathsf{L}_{\mathbb{A}^1}\mathfrak{X} \to \mathbb{A}^1$ where $\mathsf{L}_{\mathbb{A}^1}\mathfrak{X}$ denotes the family of fiberwise loop spaces, not the loop space of \mathfrak{X} . In the context of an appropriate sheaf theory, to be denoted by D, this would give rise to a nearby cycles map:

$$\Psi: D(\mathsf{L}X) \to D(\mathsf{L}X_\emptyset),$$

whose Frobenius trace is expected to recover the asymptotics map e_{\emptyset}^* .

After replacing $\mathsf{L}^+X, \mathsf{L}^+X_\emptyset, \mathsf{L}^+(X/\!\!/N)$ by finite type models $\mathcal{Y}, \mathcal{Y}_\emptyset, \mathcal{A}$, respectively (see §1.3 below), we prove:

Theorem 1.2.3 (see Theorem 8.3.8). Let X be an affine spherical variety such that B acts freely on X° . Then the following triangle of functors commutes up to natural isomorphism:

$$D_c^b(\mathcal{Y}) \xrightarrow{\Psi} D_c^b(\mathcal{Y}_{\emptyset})$$

$$\downarrow^{\pi_{\emptyset!}}$$

$$D_c^b(\mathcal{A})$$

The function-theoretic asymptotics map e_{\emptyset}^* satisfies the same commutative triangle (1.8), which suggests that nearby cycles Ψ is in a suitable sense the geometrization of the asymptotics map. This resembles an analogous result [BFO12, Corollary 6.2] in the setting of character sheaves. In the case where X is a reductive group, the nearby cycles of the IC complex of (finite type models of) L^+X has been computed by S. Schieder [Sch18, Sch16, Sch15].

In Theorem 8.3.6, we compute the intersection complex of the global model \mathcal{M}_X of the arc space, for the spherical varieties in (1.1), at the level of Grothendieck groups. We do this by relating the nearby cycles complex to $\pi_! \mathrm{IC}_{\mathcal{Y}}$, in a way that corresponds to the known relation (1.8) between asymptotics and Radon transforms.

Finally, we explain how a formula like (1.11) relates to recent conjectures of Ben-Zvi–Sakellaridis–Venkatesh [BZSV]: According to those conjectures, the formula should follow by applying Frobenius traces on the *endomorphism ring* $\operatorname{End}(\operatorname{IC}_{L^+X})$, where the endomorphism ring is taken in the derived sense, in the dg-category of derived constructible sheaves on

 LX/L^+G . (The proper definition of the intersection complex on the arc space remains conjectural, in the singular setting.) This conjecture identifies (ignoring cohomological grading)

(1.12)
$$\operatorname{End}(\operatorname{IC}_{\mathsf{L}^+X_{\mathsf{L}}}) \simeq \overline{\mathbb{Q}}_{\ell}[V_X]^{\check{G}}$$

for some symplectic representation V_X as above. We hope that our results can be related to this conjecture in both directions: Namely, that by relating our results to such endomorphism rings, one can upgrade the \check{T} -structure of Theorem 1.1.2 to a \check{G} -structure, proving Conjecture 1.1.1; and, vice versa, one can make progress towards the conjecture of [BZSV] by utilizing our results.

The conjectures of [BZSV], and the problems addressed by the present paper, can be formulated for more general affine spherical varieties, without the assumption that $\check{G}_X = \check{G}$. As already mentioned, from the point of view of number theory, the latter case is perhaps the most interesting one, as it corresponds to central values of L-functions. In the general case, the vector space V_X appearing in the conjectural relation (1.12) needs to be replaced by a Hamiltonian manifold living over the quotient $\check{G}_X \setminus \check{G}$.

1.3. Zastava spaces and the main theorems in terms of sheaves. From now on, we work over the algebraic closure k of the finite field \mathbb{F} , or over an algebraically closed field k in characteristic zero. When K is defined over a finite field \mathbb{F} , we will keep track of Weil structures on our sheaves, which will always have the form of half-integral Tate twists, where, as mentioned, $(\frac{1}{2})$ denotes a fixed square root of the cyclotomic twist. The intersection complex of a d-dimensional scheme over k will be understood to have stalks $\mathbb{Q}_{\ell}(\frac{d}{2})[d]$ over the smooth locus.

In order to replace the arc space by a model of finite type, we fix a smooth projective curve C over k (or \mathbb{F}). For an algebraic stack \mathcal{X} and an open substack $\mathcal{X}^{\circ} \subset \mathcal{X}$, we will use

$$\operatorname{Maps}_{\operatorname{gen}}(C, \mathfrak{X} \supset \mathfrak{X}^{\circ})$$

to denote the prestack that assigns to a test scheme S the groupoid of maps $C \times S \to \mathcal{X}$ such that the open locus of points sent to \mathcal{X}° maps surjectively to S. Equivalently, these are the maps such that for every geometric point $\bar{s} \to S$, the restricted map $C \times \bar{s} \to \mathcal{X}$ generically lands in \mathcal{X}° . Since C is smooth, Maps_{gen} $(C, \mathcal{X} \supset \mathcal{X}^{\circ})$ is an open substack of the prestack Maps (C, \mathcal{X}) .

Given an affine spherical G-variety X with open G-orbit X^{\bullet} and open B-orbit X° , we consider the following two models for the arc space of X:

• the Artin stack

$$\mathcal{M} = \mathcal{M}_X = \operatorname{Maps}_{gen}(C, X/G \supset X^{\bullet}/G),$$

that we will simply refer to as "the global model";

• the stack

$$\mathcal{Y} = \mathcal{Y}_X = \mathrm{Maps}_{\mathrm{gen}}(C, X/B \supset X^{\circ}/B)$$

that we will refer to as "the Zastava model". In our setting $(X^{\circ} \cong B)$, this turns out to be a scheme. Such a model is often referred to as the "local model" for reasons that have to do with factorization structures, but since this can create confusion with the genuinely local arc space, we will avoid such terminology.

For a discussion of why these are indeed formal models of the arc space (in the formal neighborhoods of suitable points), see Theorem 3.8.2, Lemma 3.5.4. Note that the choice of a Borel subgroup is immaterial, since X/B can also be written as $(X \times \mathcal{B})/G$, where \mathcal{B} is the flag variety, with $X^{\circ}/B = (X \times \mathcal{B})^{\bullet}/G$, where $(X \times \mathcal{B})^{\bullet}$ is the open orbit under the diagonal G-action.

We will also let \mathcal{A} denote the analog of these models for the toric variety $X/\!\!/N$, that is,

$$\mathcal{A} = \operatorname{Maps}_{\text{gen}}(C, (X/\!\!/ N)/T) \supset (X^{\circ}/\!\!/ N)/T),$$

and notice that under our assumptions $X^{\circ}/\!\!/N \simeq T$. Fixing such an identification, for every $\chi \in \mathfrak{c}_X^{\vee}$ (the subset of Λ_X , the character group of T, of those elements that are ≥ 0 on \mathfrak{c}_X), the corresponding map $X/\!\!/N \to \mathbb{G}_a$ gives rise to a morphism $\mathcal{A} \to \operatorname{Maps}_{\mathrm{gen}}(C, \mathbb{G}_a/\mathbb{G}_m \supset \mathbb{G}_m/\mathbb{G}_m) = \operatorname{Sym} C$, the scheme of effective divisors on the curve. Thus, \mathcal{A} can be thought of as the scheme of \mathfrak{c}_X -valued divisors. This scheme is well understood [BNS16]: its normalization is a disjoint union of partially symmetrized powers $C^{\mathfrak{P}}$ of the curve, indexed by formal \mathbb{N} -linear combinations $\mathfrak{P} \in \operatorname{Sym}^{\infty}(\operatorname{Prim}(\mathfrak{c}_X))$ of the primitive elements of \mathfrak{c}_X (see §3.1.4).

The sheaf-theoretic analog of Theorems 1.1.2 and 1.1.4 is a statement about the pushforward of the IC sheaf under

$$\pi: \mathcal{Y} \to \mathcal{A}$$
.

We will only compute $\pi_! IC_{\S}$ in the Grothendieck group of sheaves on \mathcal{A} (see Corollary 4.5.9), which is enough to determine the trace of Frobenius on stalks. The reason we do not compute the pushforward in the DG category (although this can be done in principle) is related to the fact that the map π is not proper. However, one can compactify π by considering the compactified Zastava space

$$\overline{\mathcal{Y}} = \operatorname{Maps}_{\operatorname{gen}}(C, (\overline{X/N})/T \supset X^{\circ}/B),$$

where $\overline{X/N}$ stands for the stack $(X \times \overline{N \setminus G})/G$, where $\overline{N \setminus G} = \operatorname{Spec} k[N \setminus G]$ is the affine closure of $N \setminus G$.

The difference between $\overline{\mathcal{Y}}$ and \mathcal{Y} will account for the factor of $\prod_{\check{\alpha}\in\check{\Phi}^+}(1-q^{-1}e^{\check{\alpha}})$ in (1.3). The number theory-minded reader will recognize in this factor, in the case of $G=\mathrm{SL}_2$, the Euler factor of the quotient between Eisenstein series obtained by summing over integral points of $N\backslash\mathrm{SL}_2$, versus integral points of $\overline{N}\backslash\mathrm{SL}_2=\mathbb{A}^2$. More generally, this is the factor that relates the "naive" and "compactified" Eisenstein series of [BG02], [BFGM02].

In Proposition 4.1.1 and Theorem 6.3.4 we prove:

Theorem 1.3.1. The map $\bar{\pi}: \bar{\mathcal{Y}} \to \mathcal{A}$ is proper and stratified semi-small.

This is one of the key technical results of this paper, because it allows us to get our hands on the pushforward of the intersection complex, without having a description of the complex itself. The assumption that $\check{G}_X = \check{G}$ is critical for the theorem: the analogous statement for the usual Finkelberg–Mirković Zastava space ([FM99, BFGM02]) is far from true.

The condition of being stratified semi-small is a condition on "smallness" of fibers, relative to a fixed stratification which, in this case, is the natural stratification of \mathcal{A} by strata of the form

$$\iota^{\mathfrak{P}}: \mathring{C}^{\mathfrak{P}} \hookrightarrow \mathcal{A},$$

where \mathfrak{c}_X -valued divisors take a fixed set of values. (Here, $\mathring{C}^{\mathfrak{P}}$ denotes the open "disjoint" locus in a certain product of symmetric powers of the curve, corresponding to divisors of the form $\sum_{\check{\lambda} \in \mathfrak{c}_X} \sum_{i=1}^{N_{\check{\lambda}}} (v_i) \check{\mu}$ with all $v_i \in |C|$ distinct; we will denote by $\bar{\iota}^{\mathfrak{P}}$ the natural compactification.)

By the decomposition theorem, stratified semi-smallness ensures that $\bar{\pi}_! IC_{\overline{y}}$ is a direct sum of irreducible *perverse* sheaves. By a factorization property of \overline{y} , this easily implies an expression for $\bar{\pi}_! IC_{\overline{y}}$ of the form

$$(1.13) \qquad \bar{\pi}_{!}(\mathrm{IC}_{\overline{\mathcal{Y}}}) \cong \bigoplus_{\mathfrak{P}} \left(\bigotimes_{\check{\lambda}} \mathrm{Sym}^{N_{\check{\lambda}}}(V_{X,\check{\lambda}}) \right) \otimes \bar{\iota}_{!}^{\mathfrak{P}}(\mathrm{IC}_{C^{\mathfrak{P}}}),$$

⁶One makes the usual modification to the definition when [G, G] is not simply connected.

(see Proposition 6.4.1), where $V_{X,\check{\lambda}}$ are the contributions of diagonal strata $\iota^{[\check{\lambda}]}:C\hookrightarrow\mathcal{A}$, corresponding to divisors supported at one point. Moreover, perversity and an estimate of dimensions imply that these contributions all come from the top degree cohomology of those fibers of the map $\mathcal{Y}\to\mathcal{A}$ which (over diagonal strata) are of "maximal possible dimension" in terms of the semi-smallness inequality. (We will discuss this dimension calculus in detail below.) These fibers, over points on the diagonal stratum $\iota^{[\check{\lambda}]}(C)$ are called the central fibers, and denoted by $\mathsf{Y}^{\check{\lambda}}$, with the point not appearing explicitly in the notation.

Therefore, to complete the calculation and prove Conjecture 1.1.1, we would need to count, for every $\check{\lambda} \in \mathfrak{c}_X$, the irreducible components of the central fiber $\mathsf{Y}^{\check{\lambda}}$ which achieve the maximal dimension, and show that their cardinality is equal to the dimension of the $\check{\lambda}$ -weight space in V_X^+ , which is "half" of a symplectic \check{G} -representation ρ_X . Note that \mathfrak{c}_X is strictly convex, and the weights of ρ_X will have the property that they belong to either $\mathfrak{c}_X - 0$ or $-\mathfrak{c}_X - 0$. Thus, the monoid \mathfrak{c}_X determines which "half" of ρ_X to consider.

There is a reduction from the general case of Theorem 1.1.4 to the special case of Theorem 1.1.2, that is, when $X = \overline{X^{\bullet}}^{\text{aff}}$ is the affine closure of its open orbit (Theorem 5.1.5), and the monoid \mathfrak{c}_X is free. This reduction uses the action of the Hecke algebra, and resembles the calculation of the IC sheaf of reductive L-monoids in [BNS16], but is much harder. This reduction will be discussed in §1.4 below.

Hence for now let us focus on the case $X = \overline{X^{\bullet}}^{\text{aff}}$, assuming that \mathfrak{c}_X is free. In that case, the maximal possible dimension for the central fiber $Y^{\check{\lambda}}$ will be called the *critical dimension*, and is equal to

$$\frac{1}{2}(\operatorname{len}(\check{\lambda}) - 1),$$

where len($\check{\lambda}$) := $\sum_i m_i$, for $\check{\lambda} \in \mathfrak{c}_X$ written uniquely as $\sum_i m_i \check{\nu}_i$ in terms of our basis of \mathfrak{c}_X . Conjecturally, the set of irreducible components of $\mathsf{Y}^{\check{\lambda}}$ of critical dimension, ranging over all $\check{\lambda} \in \mathfrak{c}_X$, should correspond to the subset of the crystal basis of ρ_X with weights in \mathfrak{c}_X .

We do not go quite as far in general, but we show that these irreducible components give rise to the aforementioned crystal, in the sense of Kashiwara [Kas93], over the Langlands dual Lie algebra $\check{\mathfrak{g}}$. Namely, let $\mathfrak{B}_{X^{\bullet}}^+$ denote the set of irreducible components of the central fibers of critical dimension, so $\mathfrak{B}_{X^{\bullet}}^+$ corresponds to a basis of $V_{X^{\bullet}}^+ := \bigoplus_{\mathfrak{c}_X} V_{X^{\bullet},\check{\lambda}}$. Formally define $\mathfrak{B}_{X^{\bullet}}^-$ to be the "negatives" of $\mathfrak{B}_{X^{\bullet}}^+$, so $\mathfrak{B}_{X^{\bullet}}^-$ corresponds to a basis of the dual space $(V_{X^{\bullet}}^+)^*$. Let $\mathfrak{B}_{X^{\bullet}}^+ = \mathfrak{B}_{X^{\bullet}}^+ \sqcup \mathfrak{B}_{X^{\bullet}}^-$. We prove in Theorems 7.1.5 and 7.1.9:

Theorem 1.3.2. Let $X = \overline{X^{\bullet}}^{\text{aff}}$ satisfy the assumptions of Theorem 1.1.2. The set $\mathfrak{B}_{X^{\bullet}}$ has the structure of a semi-normal, self-dual crystal over $\check{\mathfrak{g}}$ such that the weights have the properties described in Theorem 1.1.2.

We conjecture:

Conjecture 1.3.3. The crystal $\mathfrak{B}_{X^{\bullet}}$ is isomorphic to the unique crystal basis of a finite-dimensional \check{G} -module V_X .

This would imply Conjecture 1.1.1.

Theorem 1.3.2 endows $V_X^+ \oplus (V_X^+)^*$, as we have defined it, with an action of an SL₂-triple corresponding to every simple root of \check{G} . These actions imply that the dimensions of the weight spaces are invariant under the Weyl group of G, which provides a kind of "functional equation" for $\pi_!\Phi_0$. This functional equation can be seen as a geometric analog of the functional equation of the Casselman–Shalika method [Cas80, CS80, Sak13]. The content of Conjecture 1.3.3 is to

show that these SL₂-triples satisfy the Weyl relations: $[e_{\alpha}, f_{\beta}] = 0$ for simple roots $\alpha \neq \beta$. (The Weyl relations imply the Serre relations by [CG97, Corollary 4.3.2].)

The construction of the action of the SL_2 corresponding to a simple root α of G goes as follows: we factor $X \to X/\!\!/N$ through $X \to X/\!\!/N_{P_\alpha} \to X/\!\!/N$, where P_α is the sub-minimal parabolic corresponding to α and N_{P_α} is its unipotent radical. Then the GIT quotient $X_\alpha := X/\!\!/N_{P_\alpha}$ is a spherical variety for the Levi factor M_α . But now X_α is (usually) larger than the affine closure of its homogeneous part X_α^{\bullet} . The irreducible components of Y_X of critical dimension (i.e., elements of $\mathfrak{B}_{X^{\bullet}}^+$) will either go to irreducible components

- (i) of $Y_{X_{\bullet}^{\bullet}}$ of critical dimension or
- (ii) of $Y_{X_{\alpha}} Y_{X_{\alpha}^{\bullet}}$ (not necessarily of critical dimension).

While the fibers of $\mathsf{Y}_X \to \mathsf{Y}_{X_\alpha}$ are not necessarily irreducible, we show that the irreducible components of different relevant fibers can be canonically identified. Then we define the SL_2 -action by analyzing the two cases above in the base Y_{X_α} .

Under our assumptions, X_{α}^{\bullet} is a torus torsor over $\mathbb{G}_m \backslash \mathrm{PGL}_2$ and case (i) is an easy calculation. Meanwhile our study of non-canonical affine embeddings using Hecke actions shows that in case (ii) we always get a Mirković–Vilonen cycle (i.e., irreducible component of the intersection of a semi-infinite orbit with a L⁺G-orbit in the affine Grassmannian). The crystal structure on these cycles was constructed by [BG01].

To check the Weyl/Serre relations, one can similarly reduce to a spherical variety $X_{\alpha,\beta}^{\bullet}$ for a Levi of semisimple rank two. There are only a handful such varieties (up to center) satisfying our assumptions — a small subset of the spherical (wonderful) varieties of rank two classified by Wasserman [Was96].

However, checking the Weyl/Serre relations, even in a few cases, "by hand" does not seem to be easy, and we do not have a conceptual proof of them; therefore, we refrain from attempting such a verification.

The remainder of the introduction will be devoted to describing the two most important elements in the proofs of the theorems above.

1.4. Reduction to canonical affine closure. We give an overview of how to reduce the case of an arbitrary affine X with $X^{\bullet} = H \setminus G$ to the canonical affine closure $X^{\operatorname{can}} = \overline{H \setminus G}^{\operatorname{aff}}$.

There is a canonical map $X^{\operatorname{can}} \to X$, which induces an inclusion $X^{\operatorname{can}}(\mathfrak{o}) \cap X^{\bullet}(F) \subset X(\mathfrak{o}) \cap X^{\bullet}(F)$ of $G(\mathfrak{o})$ -stable spaces. Of course, all points of $X^{\bullet}(F)$ are G(F)-translates of points in $X^{\operatorname{can}}(\mathfrak{o}) \cap X^{\bullet}(F)$. It is a fact that if $\check{\theta} \in \check{\Lambda}^-$ is antidominant and belongs to the monoid \mathfrak{c}_X , then the action of the double coset $G(\mathfrak{o})t^{\check{\theta}}G(\mathfrak{o})$ preserves $X(\mathfrak{o}) \cap X^{\bullet}(F)$. The idea for what follows is that we can obtain $X(\mathfrak{o}) \cap X^{\bullet}(F)$ by acting on $X^{\operatorname{can}}(\mathfrak{o}) \cap X^{\bullet}(F)$ by $G(\mathfrak{o})t^{\check{\theta}}G(\mathfrak{o})$ for $\check{\theta} \in \mathfrak{c}_X^- := \check{\Lambda}^- \cap \mathfrak{c}_X$.

The Zastava model \mathcal{Y} lives over Bun_B and does not carry a Hecke action. Thus, to model the G(F)-action on $X(\mathfrak{o}) \cap X^{\bullet}(F)$ we must use the global model $\mathcal{M} = \mathcal{M}_X$, which lives over Bun_G . The canonical map $\mathcal{M}_{X^{\operatorname{can}}} \to \mathcal{M}_X$ is a closed embedding. For $\check{\theta} \in \mathfrak{c}_X^- = 0$, let $\mathcal{H}_{G,C}^{\check{\theta}}$ denote the Hecke stack over $\operatorname{Bun}_G \times C$ with fibers isomorphic to $\overline{\operatorname{Gr}}_G^{\check{\theta}}$, the closure of the L^+G -orbit in the affine Grassmannian corresponding to $\check{\theta}$. In reality, we need a symmetrized (multi-point) version of the Hecke stack, but we only describe the case where there is one point on the curve in this introduction for simplicity. There is a well-defined map

(1.14)
$$\mathcal{M}_{X^{\operatorname{can}}} \underset{\operatorname{Bun}_{G}}{\times} \mathcal{H}_{G,C}^{\check{\theta}} \to \mathcal{M}_{X}$$

modeling the action of Hecke operators, and we show (Theorem 5.1.1) that this map is birational onto its image. If we allow multiple points above, then the images of the corresponding Hecke actions, with $\check{\theta}$ varying, stratify \mathfrak{M}_X .

Under the assumptions of the previous subsection, we show that $\mathcal{M}_{X^{\operatorname{can}}}$ is irreducible (Corollary 5.6.4), so the study of $\operatorname{IC}_{\mathcal{M}_X}$ reduces to the study of the Hecke action on $\operatorname{IC}_{\mathcal{M}_X \operatorname{can}}$ and the determination of which of the strata above form irreducible components of \mathcal{M}_X . For the latter, we need to understand the closure relations among the different strata (Proposition 5.6.1). When $\mathfrak{c}_{X^{\bullet}} = \mathbb{N}^r$, the stratum corresponding to $\check{\theta}$ is contained in the closure of the stratum corresponding to $\check{\theta}'$ if and only if $\check{\theta} - \check{\theta}' \in \mathfrak{c}_{X^{\bullet}}$ (more generally, the closure relations are determined by the colors of X).

1.5. Semi-infinite orbits and dimension estimates. There is another way to understand the central fibers $\mathsf{Y}^{\check{\lambda}}$: as subsets of the affine Grassmannian of G. Let us fix the point $v \in C$ that we take central fibers with respect to. Then a k-point of $\mathsf{Y}^{\check{\lambda}}$ is a map $C \to X/B$ such that C - v is sent to $X^{\circ}/B = \mathsf{pt}$. Restricting to the completed local ring \mathfrak{o}_v at v gives a map $\mathsf{Y}^{\check{\lambda}} \to \mathsf{L} X^{\circ}/\mathsf{L}^+B$. If we fix a base point $x_0 \in X^{\circ}(k)$ to identify $X^{\circ} \cong B$, we get a map $\mathsf{Y}^{\check{\lambda}} \to \mathsf{Gr}_B$ and this turns out to be a closed embedding. The reduced image of the components of Gr_B in Gr_G are the semi-infinite orbits $\mathsf{S}^{\check{\lambda}}(k) = N(F)t^{\check{\lambda}}G(\mathfrak{o})/G(\mathfrak{o})$. After passing to reduced schemes we get identifications $\mathsf{Y}^{\check{\lambda}}_{\mathrm{red}} = (\mathsf{S}^{\check{\lambda}} \times_{\mathsf{L}X/\mathsf{L}^+G} \mathsf{L}^+X/\mathsf{L}^+G)_{\mathrm{red}}$ and $\overline{\mathsf{Y}}^{\check{\lambda}}_{\mathrm{red}} = (\overline{\mathsf{S}}^{\check{\lambda}} \times_{\mathsf{L}X/\mathsf{L}^+G} \mathsf{L}^+X/\mathsf{L}^+G)_{\mathrm{red}}$ (see Lemma 4.3.2).

Semi-infinite orbits have an important meaning for the geometric Satake equivalence [MV07]: the fundamental classes of the irreducible components of the intersection $S^{\check{\lambda}} \cap \overline{\mathrm{Gr}}_G^{\check{\theta}}$, the $\mathit{Mirkovi\acute{c}-Vilonen\ cycles}$, are in bijection with the "canonical basis" for the $\check{\lambda}$ -eigenspace of the irreducible \check{G} -module of lowest weight $\check{\theta}$.

Our analysis of the central fibers $Y^{\check{\lambda}}$ is founded upon the following argument from [MV07, §3]. The boundary $\overline{S}^{\check{\lambda}} - S^{\check{\lambda}} = \bigcup_{\check{\nu} < \check{\lambda}} S^{\check{\nu}}$ is a hyperplane section for some projective embedding of Gr_G . Hence any closed subscheme of Gr_G which intersects $\overline{S}^{\check{\lambda}}$, also intersects its boundary in codimension one (unless already contained in the boundary). By inductively "cutting" by these hyperplanes, we prove:

Theorem 1.5.1. Let $X = X^{\operatorname{can}}$ be as in Theorem 1.1.2. Let b be an irreducible component of the central fiber $Y^{\check{\lambda}}$. Then

- dim b $\leq \frac{1}{2}(\operatorname{len}(\lambda) 1)$,
- for a basis element $\check{\nu}_i$ of \mathfrak{c}_X (corresponding to a color), $\mathsf{Y}^{\check{\nu}_i} = \mathsf{pt}$,
- the inequality is an equality only if there is a sequence $\alpha_1, \ldots, \alpha_d$ of simple roots (with repetitions) such that $\overline{b} \cap S^{\check{\lambda} \check{\alpha}_1 \cdots \check{\alpha}_j}$ is of dimension $\dim b j$ (hence, also of critical dimension), and $\check{\lambda} \sum_{i=1}^d \check{\alpha}_i = \check{\nu}$ for a color $\check{\nu}$.

The operation of hyperplane "cutting" can almost be thought of as the lowering operator for some SL_2 -triple; unfortunately it is not quite precise enough, see Proposition 7.3.1.

If $X \neq X^{\operatorname{can}}$, then we also show that if **b** is an irreducible component of $\mathsf{Y}_X^{\check{\lambda}}$ of critical dimension that is not contained in $\mathsf{Y}_{X^{\operatorname{can}}}^{\check{\lambda}}$, then $\check{\lambda}$ must be a weight of $V^{\check{\theta}_i}$ for one of the $\check{\theta}_i$ appearing in Theorem 1.1.4, and **b** is birational to a Mirković–Vilonen cycle in $\mathsf{S}^{\check{\lambda}} \cap \operatorname{Gr}_G^{\check{\theta}_i}$. The latter correspondence comes from the Hecke action (1.14).

Let us comment on how the above relates to Theorems 1.3.1 and 1.1.4. Under our assumption that $\mathfrak{c}_{X^{\bullet}} = \mathbb{N}^r$, the space $\mathcal{Y}_{X^{\operatorname{can}}}^{\check{\lambda}}$ is irreducible. Then the dimension estimate in Theorem 1.5.1, together with a factorization property of \mathcal{Y}_X , implies the "stratified semi-smallness" condition.

The irreducible components of $\mathsf{Y}_{X^{\mathrm{can}}}$ of critical dimension correspond to a basis of $V_{X^{\bullet}}^+$. The other irreducible components of $\mathsf{Y}_X - \mathsf{Y}_{X^{\mathrm{can}}}$ of critical dimension correspond to Mirković–Vilonen cycles in $\mathrm{Gr}_G^{\check{\theta}_i}$, and these provide a basis for $V^{\check{\theta}_i}$ in Theorem 1.1.4 by geometric Satake.

1.6. Organization of the paper. In §2 we briskly review the salient combinatorics of spherical varieties and the classification of $G(\mathfrak{o})$ -orbits of the loop space of X^{\bullet} . In §3 we introduce the global and Zastava models for the arc space of X and their stratifications, explain why they are indeed models, and prove some foundational properties. In §4, we introduce the compactification of the Zastava model and define the central fibers of (compactified and non-compactified) Zastava models. Then we perform the comparison between $\pi_! IC_{y}$ and $\bar{\pi}_! IC_{\overline{y}}$ that accounts for the "numerator" in the Euler factor (1.4).

Sections 5 and 6 are the technical heart of this paper. In §5 we establish the closure relations for the global model \mathcal{M}_X and determine its irreducible components. This involves a study of the G-Hecke action on the global model, which also reduces the problem to the canonical affine closure, as explained earlier. In §6, we analyze the geometry of the central fiber and prove the crucial dimension estimates using the Mirković–Vilonen boundary hyperplanes of semi-infinite orbits. This allows us to prove Theorem 1.3.1.

In §7, we prove the aforementioned results on crystals. In §8, we combine the results of the preceding sections to compute the nearby cycles of the IC complex on the global model using a well-known contraction principle. Here we establish that the nearby cycles functor does indeed correspond to the asymptotics map under the sheaf–function dictionary.

In Appendix A, we collect various technical results concerning the stratification of the global model, some of which use the notion of generic-Hecke modification from [GN10].

1.7. Index of notation.

In general, we will use calligraphic letters \mathcal{D}, \mathcal{V} to denote standard combinatorial objects associated to spherical varieties in the literature, script letters $\mathcal{M}, \mathcal{Y}, \mathcal{F}$ for algebraic stacks and sheaves, sans serif letters Y, S for (ind-)schemes that are subspaces of certain loop spaces with respect to a fixed point $v \in |C|$. The following table contains most of the notation used in this paper, except for notation defined and used locally.

- k an algebraically closed field. The characteristic of k can be zero or positive, but in the latter case we will impose some restrictions on our spherical varieties (see §2.2), to ensure that their geometry is similar to that in characteristic zero.
- \mathbb{F} , Fr At some points in this paper, k is the algebraic closure of a finite field \mathbb{F} , and then Fr denotes the geometric Frobenius morphism.
 - pt Spec k.
 - C a connected smooth projective curve over k.
- Sym C, $C^{(n)}$ the scheme of effective divisors on (=symmetric powers of) C, and the component of divisors of degree n.
 - $\mathring{C}^n, \mathring{C}^{(n)}$ the open subsets of distinct *n*-tuples of points, resp. multiplicity free divisors of degree n, on the curve.
 - $\mathring{\prod}$, $\mathring{\times}$ for schemes living over any partially symmetrized powers of the curve, the restriction of their Cartesian product over the multiplicity-free locus.
 - $\mathbb{k} = k(C)$ the field of rational functions on C.

|C| = C(k) the set of closed points of C.

 \mathfrak{o}_v for $v \in |C|$, it denotes the completion of the local ring at v.

 F_v the fraction field of \mathfrak{o}_v . By choosing a local coordinate t we have a non-canonical isomorphism $\mathfrak{o}_v \cong k[\![t]\!] =: \mathfrak{o}$ and $F_v \cong k(\!(t)\!) =: F$. We sometimes implicitly make this identification when the choice of local coordinate is irrelevant.

 \mathbb{N} the monoid of non-negative integers.

G a connected reductive group over k.

T the (abstract) Cartan of G, i.e., the reductive quotient of any Borel subgroup. We sometimes fix a splitting $T \hookrightarrow B \hookrightarrow G$ of the abstract Cartan into a Borel subgroup.

 \mathcal{B} the flag variety of Borel subgroups of G.

W the (abstract) Weyl group of G.

 $s_{\alpha} \in W$ for a simple root α , the corresponding reflection.

 $\check{\Lambda}_G$ (Λ_G) the coweight (resp. weight) lattice of T. The index G will often be omitted.

 $\check{\Lambda}_{G}^{+}$ (Λ_{G}^{+} , $\check{\Lambda}_{G}^{-}$) The monoid of dominant coweights (resp., dominant weights, antidominant coweights).

 $\check{\Lambda}_G^{\mathrm{pos}}$ The monoid generated by the non-negative *integral* span of the positive $(\Lambda_G^{\mathrm{pos}} \subset \Lambda_G)$ coroots (resp. roots) in $\check{\Lambda}_G$.

 $\check{\Delta}_G$ (Δ_G) the set of simple coroots (resp. roots) of G.

 $2\check{\rho}_G \in \check{\Lambda}_G$ the sum of the positive coroots (roots) of G

 $(2\rho_G\in\Lambda_G)$

 $\check{\lambda} \geq \check{\mu}$ For $\check{\lambda}, \check{\mu} \in \check{\Lambda}_G$, this means that $\check{\lambda} - \check{\mu} \in \check{\Lambda}_G^{\mathrm{pos}}$.

 \check{G} the Langlands dual group of G over $\overline{\mathbb{Q}}_{\ell}$, i.e., \check{G} is the connected reductive group where the weights, roots of \check{G} equal the coweights, coroots of G, etc.

 $V^{\check{\lambda}}$ for $\check{\lambda} \in \check{\Lambda}_G$ either dominant or antidominant, this denotes the irreducible \check{G} -module over $\overline{\mathbb{Q}}_{\ell}$ with highest (resp. lowest) weight $\check{\lambda}$. Similarly, V^{λ} denotes the irreducible G-module over k for $\lambda \in \Lambda_G^+$.

variety will mean a reduced, finite type k-scheme (not necessarily irreducible).

 \mathbb{Z}/H for a stack \mathbb{Z} with an action of an algebraic group H, this will denote the quotient stack.

 $Z/\!\!/H$ for an affine variety Z over k, and a group H acting on it, the invariant-theoretic quotient Spec $k[Z]^H$.

 $\mathfrak{X} \times^G \mathfrak{Y}$ if $\mathfrak{X}, \mathfrak{Y}$ are stacks with (right) G-actions, we will use this to denote the stack quotient $(\mathfrak{X} \times \mathfrak{Y})/G$ by the diagonal action.

 L^+X , $\mathsf{L}X$ the formal arc and loop spaces of a scheme X (see §2.3).

 $D_c^b(\mathfrak{Z})$ for an algebraic stack \mathfrak{Z} , this is the derived category of bounded constructible $\overline{\mathbb{Q}}_{\ell}$ -complexes on \mathfrak{Z} .

'sheaf' means a complex of sheaves. All functors between sheaves are derived functors.

 $P(\mathcal{Z}) \subset D_c^b(\mathcal{Z})$ when \mathcal{Z} is locally of finite type over k, this is the abelian category of perverse sheaves.

 ${}^{p}D^{\leq 0}$, ${}^{p}D^{\geq 0}$ the subcategories with respect to the perverse t-structure.

the direct sum of the intersection cohomology complexes of all irreducible components of \mathcal{Z} . When working over a finite field, we will normalize this sheaf to be pure of weight zero.

an affine G-spherical variety over k. (See §2.1 for notions pertaining to spherical varieties.)

the open B-orbit, for a fixed choice of Borel subgroup B.

 $x_0 \in X^{\circ}(k)$ a fixed base point.

the stabilizer of x_0 .

the open G-orbit $H \setminus G$.

 X^{can} the "canonical" affine embedding $\operatorname{Spec} k[X^{\bullet}]$ of X^{\bullet} .

the (abstract) Cartan of X, that is, the quotient by which the abstact Cartan of G acts on X°/N , where N is the unipotent radical of B.

 $\Lambda_X, \check{\Lambda}_X$ the character and cocharacter groups of T_X . Our assumptions on X will identify $\check{\Lambda}_X$ with $\check{\Lambda}_G$, so it will often just be denoted by $\check{\Lambda}$.

the dual group of X; it has a canonical maximal torus isomorphic to the dual of T_X .

the cone of invariant valuations of X; equivalently, the antidominant chamber of the dual group of X.

 $\mathfrak{c}_X^{\vee} \subset \Lambda_X \\ (\mathfrak{c}_X \subset \check{\Lambda}_X)$ the monoid of weights of T_X on $k[X/\!/N]$ (resp., its dual monoid).

 $\mathcal{C}_0 = \mathcal{C}_0(X) \subset \mathfrak{t}_X$ the cone spanned by \mathfrak{c}_X , inside of the vector space spanned by $\mathring{\Lambda}_X$.

 \mathfrak{c}_X^- the intersection of \mathfrak{c}_X with the cone $\mathcal V$ of invariant valuations. $\mathcal D(X)$ the set of irreducible B-stable divisors in X.

the set of colors, i.e., irreducible B-stable divisors which are not G-stable; equivalently, this can be identified with $\mathcal{D}(X^{\bullet})$.

for a simple root α , the set of colors D of X^{\bullet} such that $DP_{\alpha} \supset X^{\circ}$, where P_{α} is the parabolic generated by B and the root space $\mathfrak{g}_{-\alpha}$.

 $\varrho_X(D) = \check{\nu}_D$ for $D \in \mathcal{D}(X)$, the associated B-invariant valuation, restricted to the group of nonzero B-eigenfunctions: $\check{\nu}_D: k(X)^{(B)} \to \mathbb{Z}$, and understood as a functional on the character group $\Lambda_X = k(X)^{(B)}/k^{\times}$.

the monoid generated by the $\check{\nu}_D$, $D \in \mathcal{D}$.

for $\check{\lambda}, \check{\mu} \in \check{\Lambda}_X$, this means that $\check{\mu} - \check{\lambda} \in \mathfrak{c}_X^{\mathcal{D}}$.

 $\mathcal{D}^G_{\mathrm{sat}}(X)$ the set of those primitive (=indecomposable) elements in \mathfrak{c}_X^- that are minimal with respect to the \leq partial order.

the moduli stack of G-bundles, resp. B-bundles, on C. $\operatorname{Bun}_G, \operatorname{Bun}_B$

> \mathcal{M}_X the "global model" of generic maps from a curve to X/G; it lives over Bun_G . The restriction of such a map, defined over k, to the formal neighborhood of a point $v \in |C|$ gives rise to a well-defined "valuation", that is, an element of $(X^{\bullet}(F_v) \cap X(\mathfrak{o}_v))/G(\mathfrak{o}_v)$.

See Section 3 for the various models of the arc space. For any model, when X is understood, the index will be omitted.

the "Zastava model" of generic maps from a curve to X/B; it lives over Bun_B. The restriction of such a map, defined over k, to the formal neighborhood of a point $v \in |C|$ gives rise to a well-defined element of $(X^{\circ}(F_v) \cap X(\mathfrak{o}_v))/B(\mathfrak{o}_v)$

Drinfeld's compactification of Bun_B , see §4.1.

 \overline{y}_X the compactified Zastava model (Section 4).

 \mathcal{A} the global/Zastava model for the T_X -space $X/\!\!/N$.

 $\pi, \bar{\pi}$ the natural maps $\pi: \mathcal{Y}_X \to \mathcal{A}, \ \bar{\pi}: \overline{\mathcal{Y}}_X \to \mathcal{A}$ (extending π).

 $X^{\bullet}(F)_{G,\check{\theta}}, \mathsf{L}^{\check{\theta}}X$ the $G(\mathfrak{o})$ -orbit on $X^{\bullet}(F)$ parametrized by $\check{\theta} \in \mathcal{V} \cap \check{\Lambda}_X$ (Theorem 2.3.5), and the corresponding stratum of the loop space. When $\check{\theta} \in \mathfrak{c}_X^-$, these belong to $X(\mathfrak{o})$, resp. the arc space L^+X .

 $\operatorname{Sym}^{\infty}(S)$ the set of multisets in elements of a set S; equivalently, the free monoid $\bigoplus_{S} \mathbb{N}$ in the elements of S.

 $C^{\mathfrak{P}}, \mathring{C}^{\mathfrak{P}}$ for a multiset $\mathfrak{P} = \sum_{s \in S} N_s[s]$, the partially symmetrized power $\prod_S C^{(N_s)}$ of the curve, and its disjoint locus $\prod_{s} \mathring{C}^{(N_s)}$.

for $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - \{0\})$, the stratum of $\mathfrak M$ containing those maps whose multiset of nontrivial valuations (as elements of $\mathfrak{c}_X^- - \{0\}$) is equal to $\check{\Theta}$. When $\check{\Theta} = \{\check{\theta}\}\$ is a singleton, we will write $\mathcal{M}^{\check{\theta}}$.

the connected component of $\mathcal A$ of maps with total valuation $\check\lambda\in\mathfrak c_X\subset$ $\check{\Lambda}_X = T_X(F)/T_X(\mathfrak{o}).$

the preimage of $\mathcal{A}^{\check{\lambda}}$ in \mathcal{Y} , resp. in $\overline{\mathcal{Y}}$. They live over strata $\operatorname{Bun}_{B}^{-\check{\lambda}}$, $\overline{\operatorname{Bun}_{B}^{-\check{\lambda}}}$ of Bun_B , resp. \overline{Bun}_B .

the fiber products of $\mathcal{Y}^{\check{\lambda}}, \overline{\mathcal{Y}}^{\check{\lambda}}$ with $\mathcal{M}^{\check{\Theta}}$ over \mathcal{M} .

for $D \in \mathbb{N}^{\mathcal{D}}$, a certain connected/irreducible component of the "open Zastava" space $\mathcal{Y}_{X\bullet} = \mathcal{Y}^{?,0}$, defined in §5.4. The question mark? corresponds to the valuation $\varrho_X(D)$.

denotes the closure of the stratum $\mathcal{M}^{\check{\Theta}}$. Note that $\overline{\mathcal{Y}}^{\check{\lambda}}$, $\overline{\mathcal{Y}}^{\check{\lambda},\check{\Theta}}$, in contrast, are not closures of strata, but strata of the compactified Zastava space. In the case of the global model, there is no room for confusion, so we allow ourselves this notation, for typographical reasons.

 $C_{\check{\nu}}, {}_{\check{\nu}}\overline{\mathrm{Bun}}_{B}^{\check{\lambda}}, {}_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda}}$ for $\check{\nu} \in \check{\Lambda}_G^{\text{pos}}$, the partially symmetrized power $C^{\mathfrak{P}}$ when $\check{\nu}$ is thought of as a multiset \mathfrak{P} in the simple coroots, a stratum of $\overline{\operatorname{Bun}}_B^{\check{\lambda}}$, and a stratum of $\overline{\mathbb{P}}_{\check{\nu}}^{\check{\lambda}}$, isomorphic, respectively, to $C_{\check{\nu}} \times \operatorname{Bun}_B^{\check{\lambda}+\check{\nu}} \hookrightarrow \overline{\operatorname{Bun}}_B^{\check{\lambda}}$ and $C_{\check{\nu}} \times \mathring{\mathbb{P}}^{\check{\lambda}-\check{\nu}}$

 Gr_G, Gr_B the affine Grassmannian of G, resp. B.

 $\operatorname{Gr}_{G,\operatorname{Sym}C}$, the Beilinson-Drinfeld affine Grassmannians, living over Sym C (see §3.7). $\operatorname{Gr}_{B,\operatorname{Sym} C}$

 $\operatorname{Gr}_G^{\check{\theta}}, \, \overline{\operatorname{Gr}}_G^{\check{\theta}}$ for $\check{\theta} \in \check{\Lambda}_G^-$, the L⁺G-orbit in the affine Grassmannian containing the class of $t^{\check{\theta}}$, and its closure.

 $\begin{array}{ll} \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}} & \text{for } \check{\Theta} \in \mathrm{Sym}^{\infty}(\check{\Lambda}_{G}^{-}-0), \text{ the multi-point version of } \overline{\mathrm{Gr}}^{\check{\theta}}, \text{ see } \S 5.2.1. \\ \mathsf{S}^{\check{\lambda}}, \overline{\mathsf{S}}^{\check{\lambda}} & \text{the "semi-infinite" } \mathsf{L}N\text{-orbit of } t^{\check{\lambda}} \text{ in } \mathrm{Gr}_{G}, \text{ and its closure.} \end{array}$

the central fibers of $\mathcal{Y}^{\check{\lambda}}, \overline{\mathcal{Y}}^{\check{\lambda}}$, living over a point of the diagonal stratum $C \hookrightarrow$ $\mathcal{A}^{\check{\lambda}}$. They can be identified as subspaces of $\mathsf{S}^{\check{\lambda}}$, $\overline{\mathsf{S}}^{\check{\lambda}}$ (see §4.3). From §4.3 onwards, we will implicitly consider only the underlying reduced structure on all central fibers.

 $\mathfrak{B}_{X,\check{\lambda}}$ for $\check{\lambda} \in \mathfrak{c}_X$, the set of all irreducible components of critical dimension of the central fiber $\mathsf{Y}^{\check{\lambda}}$, see Proposition 6.5.1.

 $\mathfrak{B}_{X}^{+},\mathfrak{B}_{X}^{-}$ the union of all $\mathfrak{B}_{X,\check{\lambda}},\,\check{\lambda}\in\mathfrak{c}_{X},$ and the "crystal of X" (§7.1.4).

1.8. Acknowledgments. We thank R. Bezrukavnikov, V. Drinfeld, T. Feng, M. Finkelberg, V. Ginzburg, F. Knop and D. Gaitsgory for helpful comments and conversations. We thank the anonymous referee for helpful suggestions. We thank the Institute for Advanced Study for its hospitality during the academic year 2017–2018, during which part of our work was conducted. Y.S. was supported by NSF grants DMS-1801429 and DMS-1939672, and by a stipend to the IAS from the Charles Simonyi Endowment. J.W. was supported by NSF grant DMS-1803173.

2. Spherical varieties and their arc spaces

2.1. **Spherical varieties.** A spherical variety over k is a normal variety with an open B-orbit. Let X be an affine G-spherical variety over k. Let X° denote the open B-orbit. We choose and fix a point $x_0 \in X^{\circ}(k)$ and let H denote its stabilizer. Let $X^{\bullet} = H \setminus G$ denote the open G-orbit.

The quotient X°/N has an action of the universal Cartan T = B/N, and is a torsor for a quotient torus $T \to T_X$. By our choice of base point, we can identify this torsor with T_X . In the rest of this paper, we will assume that our spherical variety satisfies $T_X = T$; however, for now we proceed with general definitions.

All important combinatorial invariants of the spherical variety live in the rational vector space \mathfrak{t}_X^* spanned by the character group Λ_X of this torus, or in the dual vector space \mathfrak{t}_X , containing the dual lattice $\check{\Lambda}_X$. By "lattice points", below, we will mean points belonging to one of these lattices. The spaces $\mathfrak{t}_X, \mathfrak{t}_X^*$ are the root and coroot space for the dual group \check{G}_X of X. The antidominant Weyl chamber for \check{G}_X in \mathfrak{t}_X is denoted by \mathcal{V} in the theory of spherical varieties, because it coincides with the so-called cone of G-invariant valuations, see [Kno91]. Up to this point, all data depend only on the open G-orbit X^{\bullet} , not on its affine embedding X.

The affine embedding X of X° defines an affine toric embedding $X/\!\!/N$ of T_X , described by the cone $C_0(X) \subset \mathfrak{t}_X$ whose lattice points are all cocharacters $\check{\Lambda}$ into T_X such that $\lim_{t\to 0} t^{\check{\Lambda}} \in X/\!\!/N$. We will denote by \mathfrak{c}_X the monoid of lattice points $\check{\Lambda}_X \cap C_0(X) \subset \mathfrak{t}_X$, and by \mathfrak{c}_X^- its intersection with the cone \mathcal{V} of invariant valuations. The cone $C_0(X) \subset \mathfrak{t}_X$ has a canonical set of generators $\check{\nu}_D$, the valuations associated to all B-stable divisors $D \subset X$. The set of all irreducible B-stable divisors in X will be denoted by $\mathcal{D}(X)$, and by "valuation associated" we mean the restriction of the corresponding valuation to the group of nonzero B-eigenfunctions: $\check{\nu}_D : k(X)^{(B)} \to \mathbb{Z}$, which factors through the character group Λ_X and hence can be identified with an element of $\check{\Lambda}_X \subset \mathfrak{t}_X$. The map $\mathcal{D}(X) \ni D \mapsto \check{\nu}_D \in \check{\Lambda}_X$ will be denoted by ϱ_X . Inside of $\mathcal{D}(X)$ there is a distinguished subset \mathcal{D} , depending only on the open G-orbit X^{\bullet} , which consists of the closures of B-stable divisors in X^{\bullet} ; those are called *colors*. We will often abuse language and write "colors" for the images of \mathcal{D} in $\check{\Lambda}_X$.

We remark that the valuation map ϱ_X may fail to be injective, but this can only happen when two colors have the same image. If this is the case for $X^{\bullet} = H \backslash G$, there is always a torus covering of it such that all colors have distinct valuations (see §5.3); for example, $GL_1 \backslash PGL_2$ has two colors with valuation $\frac{\alpha}{2}$, but their preimages in $GL_1 \backslash GL_2$ (where GL_1 is embedded

⁷The Gaitsgory–Nadler dual group was defined in a Tannakian way in [GN10], but not completely identified in all cases. A combinatorial description of a dual group (presumably the same) was consequently afforded by Knop and Schalke [KS17]. The invariants that we present here are those of Knop and Schalke, which match standard invariants of the theory of spherical varieties. For this paper, however, this distinction between constructions of the dual group is immaterial, as we impose the condition that $T_X = T$ and "all simple roots of G are spherical roots of type T", which implies that in both versions of the dual group, $\check{G}_X = \check{G}$.

as the general linear group of a one-dimensional subspace) induce different valuations. In any case, colors give rise to a map $\mathbb{N}^{\mathcal{D}} \to \mathfrak{c}_X$, whose image we denote by $\mathfrak{c}_X^{\mathcal{D}}$.

We define an ordering \leq on $\check{\Lambda}_X$ by postulating that $\check{\lambda} \leq \check{\lambda}'$ if $\check{\lambda}' - \check{\lambda}$ can be written as a non-negative *integral* combination of the valuations $\check{\nu}_D$, with $D \in \mathcal{D}$, i.e., if $\check{\lambda}' - \check{\lambda} \in \mathfrak{c}_X^{\mathcal{D}}$. We use the symbol \leq for the ordering on $\check{\Lambda}_G$ defined by the positive coroots of G, that is, $\check{\lambda} \leq \check{\lambda}'$ iff $\check{\lambda}' - \check{\lambda}$ is a sum of positive coroots. Notice that the ordering \leq is not defined simply in terms of the cone spanned by the $\check{\nu}_D$'s: there can be non-comparable lattice points in this cone (and similarly for the ordering \leq). As we will see later, this ordering describes the closure relations on the global model of the arc space of X.

2.1.1. Spherical roots of type T. From Section 3 onwards we assume that $T_X = T$, equivalently, B acts simply transitively on X° . From Section 5 on we assume, further, that all simple roots of G are spherical roots of type T. Let us explain what this means: For a simple root α of G, let $P_{\alpha} \supset B$ denote the corresponding parabolic of semisimple rank one. The quotient $P_{\alpha}/\Re(P_{\alpha})$ by its radical is isomorphic to PGL_2 , and the invariant-theoretic (or geometric) quotient $X^{\circ}P_{\alpha}/\Re(P_{\alpha})$ is a spherical variety for PGL_2 . In characteristic zero, over an algebraically closed field, those belong to one of the following types, see [Kno95, Lemma 3.2]:

- a point: $PGL_2 \backslash PGL_2$;
- type $T: \mathbb{G}_m \backslash PGL_2;$
- type $N: \mathcal{N}(\mathbb{G}_m)\backslash \mathrm{PGL}_2$, where \mathcal{N} denotes normalizer;
- type $U: S \backslash PGL_2$, where $N \subset S \subset B$.

In positive characteristic there are some more cases, investigated by Knop in [Kno14].

Our assumption from Section 5 onwards is that for every simple root α , this PGL₂-spherical variety is isomorphic to $\mathbb{G}_m \backslash \mathrm{PGL}_2$. Our assumptions imply that the stabilizer in P_{α} of a point on the open orbit is isomorphic to \mathbb{G}_m , and that there are precisely two colors $D_{\alpha}^+, D_{\alpha}^-$ contained in $X^{\circ}P_{\alpha}$ (the \pm labeling is arbitrary). We will denote the set of these two elements by $\mathcal{D}(\alpha) \subset \mathcal{D}$; notice that these sets are not disjoint as α varies. Moreover, the associated valuations satisfy (see [Lun97, §3.4]):

$$\check{\nu}_{D_{\alpha}^{+}} = -s_{\alpha}\check{\nu}_{D_{\alpha}^{-}},$$

$$\check{\nu}_{D_{\alpha}^{+}} + \check{\nu}_{D_{\alpha}^{-}} = \check{\alpha},$$

and finally an element $D \in \mathcal{D}$ belongs to $\mathcal{D}(\alpha)$ iff $\langle \alpha, \check{\nu}_D \rangle > 0$ (in which case $\langle \alpha, \check{\nu}_D \rangle = 1$, by the above).

Remark 2.1.2. Our assumptions above are over the algebraically closed field k. Over a finite field \mathbb{F} , the Galois group acts on the set of colors, compatibly with its action on the set of simple roots. If, for example, G is split, each set $\mathcal{D}(\alpha)$ is preserved by Frobenius, and the stabilizer of a point on the open P_{α} -orbit is a form of \mathbb{G}_m . If that form is split, Frobenius acts trivially on $\mathcal{D}(\alpha)$; if not, it permutes the two colors.

Remark 2.1.3. A very straightforward way to compute the valuations $\check{\nu}_{D_{\alpha}^{\pm}}$ in any example is the following: If all simple roots are spherical roots of type T and $T_X = T$, the stabilizer S of a point in the open P_{α} -orbit on X is a subgroup isomorphic (over the algebraic closure) to \mathbb{G}_m . Choose a Borel subgroup $B \subset P_{\alpha}$ containing S. An isomorphism $\mathbb{G}_m \simeq S$ gives rise to a cocharacter $\check{\nu}: \mathbb{G}_m \to B \to T$, and we can choose this isomorphism so that $\langle \alpha, \check{\nu} \rangle = 1$. There are two choices for B, and they correspond to the valuations $\check{\nu}_{D_{\alpha}^{\pm}}$.

- 2.2. Affine degeneration, and assumptions in positive characteristic. In characteristic zero, there is a well-known affine family degenerating X to a horospherical variety [Pop86], [GN10, §5.1], which in turn is related to its degeneration to the normal bundle of orbits in a smooth toroidal (e.g., a "wonderful") compactification of X, [Bri07], [SV17, §2.5]. These constructions sometimes fail in positive characteristic, and therefore the statements of this subsection should be considered as assumptions in positive characteristic. These assumptions will be imposed on X for the remainder of the paper. Notice that the Luna-Vust theory of spherical embeddings holds in arbitrary characteristic over an algebraically closed field by [Kno91].
- 2.2.1. We define a filtration \mathcal{F}_{λ} on k[X] for $\lambda \in \Lambda_X$ by letting \mathcal{F}_{λ} consist of all $f \in k[X]$ such that every highest weight $\mu \in \Lambda_X$ of the rational G-module generated by f satisfies $\langle \lambda - \mu, \mathcal{V} \rangle < 0$. The affine degeneration \mathscr{X} is the affine variety defined to be the spectrum of the Rees algebra associated to the filtration above:

$$k[\mathscr{X}] = \bigoplus_{\lambda \in \Lambda_X} \mathcal{F}_{\lambda} \otimes e^{\lambda} \subset k[X \times T_X]$$

where $e^{\lambda} \in k[T_X]$ denotes the character corresponding to $\lambda \in \Lambda_X$. This family is naturally

equipped with an action of the product $G \times T_X$. Note that $k[\mathcal{X}]$ contains $\bigoplus_{\langle \lambda, \mathcal{V} \rangle \leq 0} ke^{\lambda}$. Define $\overline{T_{X,ss}}$ to be the spectrum of $\bigoplus_{\langle \lambda, \mathcal{V} \rangle \leq 0} ke^{\lambda}$. This is an affine toric variety with open orbit isomorphic to the quotient $T_{X,ss}$ of T_X by the subtorus $\mathcal{Z}(X)^0$ (the "connected center of X") generated by cocharacters in $\mathcal{V} \cap (-\mathcal{V})$.

In summary, we get a $G \times T_X$ -equivariant map

$$\mathscr{X} \to \overline{T_{X,ss}},$$

which we consider as our affine family of degenerations.

2.2.2. For $a \in \overline{T_{X,ss}}(k)$, let X_a denote the fiber of $\mathscr{X} \to \overline{T_{X,ss}}$ over a. From the definition of the filtration \mathcal{F}_{λ} , it follows that $k[X_a]^N = k[X]^N$. The following "multiplicity-free" characterization of spherical varieties (in arbitrary characteristic) implies that each fiber X_a is spherical.

Theorem 2.2.3 ([VK78], [Tim11, Theorem 25.1]). A normal quasi-affine variety X is spherical if and only if the non-zero weight spaces of $k[X]^N$ are all 1-dimensional.

Since the torus T_X is determined by $k[X]^N$, we have a family of spherical varieties X_a with associated torus $T_{X_a} \cong T_X$.

We also have $k[\mathscr{X}]^N = \bigoplus_{\langle \lambda-\mu, \mathcal{V} \rangle \leq 0} k[X]_{\mu}^{(B)} \otimes e^{\lambda}$ where $k[X]_{\mu}^{(B)}$ is the 1-dimensional *B*-eigenspace of weight μ . This gives a canonical isomorphism

$$\mathscr{X}/\!\!/N \cong X/\!\!/N \times \overline{T_{X,ss}}.$$

Define \mathscr{X}° to be the preimage of $T_X \times \overline{T_{X,ss}} \subset \mathscr{X}/\!\!/ N$ under the projection $\mathscr{X} \to \mathscr{X}/\!\!/ N$. Equivalently, \mathscr{X}° is the union of the open B-orbits of the fibers X_a .

It will be convenient to lift the isomorphism (2.1) to \mathscr{X}° .

Proposition 2.2.4. There is a (non-canonical) B-equivariant isomorphism $\mathscr{X}^{\circ} \cong X^{\circ} \times \overline{T_{X,ss}}$ over $T_{X,ss}$, compatible with (2.1).

We will comment on the proof of this proposition in conjunction with Theorem 2.2.5 below. The affine degeneration is closely related to compactifications of the spherical variety. Namely, let $\mathscr{X}^{\bullet} \subset \mathscr{X}$ denote the union of the G-translates of \mathscr{X}° . (It is the open subvariety which specializes over each fiber X_a to the open G-orbit.) The quotient $\mathscr{X}^{\bullet}/T_X$ turns out to be a proper embedding of $X^{\bullet}/\mathcal{Z}(X)^0$. For applications, one is interested in proper embeddings of

 X^{\bullet} itself, and preferably smooth ones. Without getting into the details of the construction (we point the reader to [Kno91]), we formulate the following result on the existence and local structure of "smooth toroidal compactifications":

Define $P(X) \supset B$ to be the parabolic subgroup of G equal to $\{g \in G \mid X^{\circ} \cdot g = X^{\circ}\}$, and let $N_{P(X)}$ denote its unipotent radical. Recall that we have fixed a base point $x_0 \in X^{\circ}$, for convenience.

Theorem 2.2.5. There is a proper, smooth, G-equivariant embedding $X^{\bullet} \hookrightarrow \overline{X}$, a Levi⁸ $L \subset P(X) \subset G$, and a smooth toric embedding $\overline{T_X}$ of the quotient T_X of L, such that the action map

$$L \times N_{P(X)} \ni (l, u) \mapsto x_0 l u$$

descends to T_X and extends to an open embedding

$$\overline{T_X} \times N_{P(X)} \hookrightarrow \overline{X},$$

whose image is the union of all open B-orbits on \overline{X} . Moreover, the support of the fan describing the toric variety $\overline{T_X}$ is the cone $\mathcal V$ of invariant valuations.

Remark 2.2.6. Note that the embedding that we denoted above by $\overline{T_{X,ss}}$ is associated to the image of the *opposite* cone $-\mathcal{V}$ modulo $\mathcal{V} \cap (-\mathcal{V})$. Thus, there is a map from $\overline{T_X}$ to $\overline{T_{X,ss}}$ only after inversion in T_X . We allow ourselves this notational flaw, since $\overline{T_X}$ will not be used beyond this subsection and the next.

Theorem 2.2.5 is the local structure theorem of Brion–Luna–Vust [BLV86, Théorème 3.5], applied to smooth toroidal compactifications [Kno91]. We outline its proof due to Knop [Kno94, Theorem 2.3], which also applies to Proposition 2.2.4. The main issue is how to choose the pair (x_0, L) appropriately. Let \mathcal{B} denote the flag variety of G, and assume for the moment that no choice of B, x_0 has been made. One considers triples $(x, B, \chi) \in X \times \mathcal{B} \times \mathfrak{t}_X^*$ such that x lies in the open B-orbit. Out of these data one constructs elements in the cotangent bundle T^*X^{\bullet} as follows: If χ is the differential of a character (also to be denoted by $\chi \in \Lambda_X$), the corresponding cotangent vector is the logarithmic differential $\frac{df_X}{f_X}\Big|_x$ at x of a rational B-eigenfunction with eigencharacter χ , and for general χ we extend this construction by linearity. This gives rise to a map

$$(X \times \mathcal{B})^{\bullet} \times \mathfrak{t}_{X}^{*} \to T^{*}X^{\bullet}$$

(where the bullet denotes "open G-orbit") with dense image. We can also apply this construction to the family \mathscr{X}^{\bullet} , obtaining vectors in the relative cotangent bundle over $\overline{T_{X,\mathrm{ss}}}$. Composing with the moment map $T^*X^{\bullet} \to \mathfrak{g}^*$, Knop shows that if we choose a sufficiently generic, semisimple vector ξ in the image, and take $L = \mathrm{Stab}_G(\xi)$ and x_0 a point of a triple (x_0, B, χ) in its fiber, Theorem 2.2.5 is satisfied. The same argument applies to prove Proposition 2.2.4, as follows: Instead of fixing x_0 , fix first just the Borel subgroup B, and consider all triples (x, B, χ) over ξ , now with $x \in \mathscr{X}^{\circ}$ (defined with respect to this Borel). The set of these triples is a T_X -equivariant section of the map $\mathscr{X} \to \mathscr{X}/\!\!/ N \cong X/\!\!/ N \times \overline{T_{X,\mathrm{ss}}}$ over $T_X \times \overline{T_{X,\mathrm{ss}}}$, where T_X acts diagonally on the latter.

⁸Of course, there is no distinguished Levi in P(X) abstractly, but the choice of base point x_0 sometimes imposes restrictions on the Levi subgroups that work; for example, the derived subgroup of L must stabilize x_0 . In any case, the assumptions on X in the main body of this paper, that B acts simply transitively on X° and V = the antidominant cone, imply that any $L \subset P(X) = B$ satisfies the Local Structure Theorem.

2.3. The formal loop space. For any k-scheme X, define the space of formal arcs by

$$\mathsf{L}^+ X(R) = X(R[[t]])$$

for a test ring R. It is well-known (cf. [KV04, Proposition 1.2.1]) that L^+X is representable by a scheme (of infinite type), which is the projective limit of the schemes L_n^+X , $n \in \mathbb{N}$, representing the spaces of n-arcs $\mathsf{L}_n^+X(R) = X(R[t]/t^n)$. If X is of finite type over k, then so is each L_n^+X . If X is smooth over k, then each L_n^+X is smooth over k and L^+X is formally smooth over k. If X is affine, then so are L_n^+X and L^+X .

Define the formal loop space $\mathsf{L}X(R) = X(R((t)))$. If X is affine, then $\mathsf{L}X$ is representable by an ind-affine ind-scheme, and we have a closed embedding $\mathsf{L}^+X \hookrightarrow \mathsf{L}X$.

2.3.1. Let X be an affine spherical G-variety. Define

$$\mathsf{L}^{\bullet}X := \mathsf{L}X - \mathsf{L}(X - X^{\bullet}),$$

which admits an open embedding into LX. The G-action on X induces a natural action of L^+G on LX and L^+X , $L^{\bullet}X$ are stable under this action.

Example 2.3.2. Let $X = \mathbb{A}^1$ with the scaling \mathbb{G}_m -action. Then $\mathsf{L}^+X = \mathbb{A}^\infty$, where we consider $\mathbb{A}^\infty = \operatorname{Spec} k[a_0, a_1, \dots]$ as the coefficients of infinite Taylor series, and the ind-scheme $\mathsf{L}X = \text{``lim''}$ Spec $k[a_{-m}, a_{-m+1}, \dots]$ considered as the coefficients of Laurent series. Let $X^\bullet = \mathbb{A}^1 - \{0\}$. Then $\mathsf{L}^\bullet X = \mathsf{L}X - \{0\}$ so $\mathsf{L}^+X \cap \mathsf{L}^\bullet X = \mathbb{A}^\infty - \{0\}$, whereas $\mathsf{L}^+(X^\bullet) = \mathbb{G}_m \times \mathbb{A}^\infty$.

Remark 2.3.3. The k-points of $L^{\bullet}X$ are in bijection with $X^{\bullet}(k((t)))$. Since X^{\bullet} is in general not affine, however, $X^{\bullet}(k((t)))$ does not always have an ind-scheme structure. Even when X^{\bullet} is affine, $L(X^{\bullet})$ may not be isomorphic to $L^{\bullet}X$, despite having the same sets of k-points.

2.3.4. Orbits on the formal loop space. For ease of notation, let $\mathfrak{o} = k[\![t]\!]$ and $F = k(\!(t)\!)$, so $\mathsf{L}^+G(k) = G(\mathfrak{o})$, $\mathsf{L}^\bullet X(k) = X^\bullet(F)$. We review the decomposition of $G(\mathfrak{o})$ -orbits on $X^\bullet(F)$ due to [LV83]. We present the reformulation of this result found in [GN10, Theorem 3.3.1].

Let X be an affine spherical variety, and pick a pair (x_0, L) as in Theorem 2.2.5 to write $X^{\circ} = T_X \times N_{P(X)}$ for its open Borel orbit. For a cocharacter $\check{\theta} \in \check{\Lambda}_X$ we let $t^{\check{\theta}} \in T_X(F)$ denote the image of the uniformizer $t \in \mathfrak{o}$ under the map $\check{\theta} : \mathbb{G}_m \to T_X$. Let $X^{\bullet}(F)/G(\mathfrak{o})$ denote the set of equivalence classes of $G(\mathfrak{o})$ -orbits in $X^{\bullet}(F)$.

Theorem 2.3.5 ([LV83], [GN10, Theorem 3.3.1]). (i) The map

(2.2)
$$\check{\theta} \mapsto X^{\bullet}(F)_{G:\check{\theta}} := x_0 \cdot t^{\check{\theta}}G(\mathfrak{o})$$

is a bijection of sets $V \cap \check{\Lambda}_X \cong X^{\bullet}(F)/G(\mathfrak{o})$ (which is independent of the choice of (x_0, L)).

(ii) This bijection restricts to a bijection $\mathfrak{c}_X^- \cong (X(\mathfrak{o}) \cap X^{\bullet}(F))/G(\mathfrak{o})$.

The theorem can be viewed as a generalization of the Cartan decomposition. Define $\mathsf{L}^\theta X$ to be the $\mathsf{L}^+ G$ -orbit of $x_0 t^{\check{\theta}} \in \mathsf{L}^\bullet X$.

Remark 2.3.6. Observe that if $\check{\theta} \in \mathfrak{c}_X^-$, then in particular $\check{\theta} \in \Lambda_X^-$ is antidominant (as a weight for the dual group \check{G}_X of X). Therefore, $w\check{\theta} - \check{\theta} \in \Lambda_X^{\mathrm{pos}} \subset \mathfrak{c}_X$ for any $w \in W_X$, and hence $w\check{\theta} \in \mathfrak{c}_X$. We suggest that the monoid $\mathfrak{c}_X^- = \mathcal{C}_0(X) \cap \mathcal{V} \cap \check{\Lambda}_X$ should be thought of as the set of W_X -orbits in $\check{\Lambda}_X$ that are entirely contained in the cone $\mathcal{C}_0(X)$.

Proposition 2.3.7. If k has positive characteristic, assume that the stabilizer of B acting on $x_0 \in X^{\circ}$ is a smooth subgroup. For any $\check{\theta} \in \mathcal{V} \cap \check{\Lambda}_X$, the L^+G -orbit $\mathsf{L}^{\check{\theta}}X$ is a formally smooth k-scheme with k-points equal to $X^{\bullet}(F)_{G:\check{\theta}}$, and $\mathsf{L}^{\check{\theta}}X$ is open in its closure in $\mathsf{L}X$.

Therefore, the collection of $\mathsf{L}^{\check{\theta}}X$, $\check{\theta} \in \mathcal{V} \cap \check{\Lambda}_X$, form a stratification of $\mathsf{L}^{\bullet}X$, in the sense that the strata are disjoint and contain all the k-points.

Proof. Let $S \subset L^+G$ denote the stabilizer subgroup of $x_0t^{\check{\theta}} \in LX(k)$. A priori, the L^+G -orbit $\mathsf{L}^{\check{\theta}}X$ is defined as the fpqc sheaf quotient L^+G/S . For $n\in\mathbb{N}$, let S_n denote the scheme-theoretic image of S under the projection $L^+G \to L_n^+G$. For m > n, the transition map

$$\mathsf{L}_m^+ G/S_m \to \mathsf{L}_n^+ G/S_n = \mathsf{L}_m^+ G/(S_n \cdot \ker(\mathsf{L}_m^+ G \to \mathsf{L}_n^+ G))$$

is smooth affine since $\ker(\mathsf{L}_m^+G\to\mathsf{L}_n^+G)$ is smooth and unipotent. The schemes L_n^+G/S_n are also smooth since $\mathsf{L}_n^+ G$ is smooth. Therefore $\mathsf{L}^{\dot{\theta}} X \cong \mathsf{L}^+ G/S \cong \varprojlim \mathsf{L}_n^+ G/S_n$ is representable by a formally smooth scheme.

The fact that $\mathsf{L}^{\dot{\theta}}X \to \mathsf{L}X$ is open in its closure will follow from Lemma 2.3.10 below.

2.3.8. Recall the family \mathscr{X} introduced in §2.2. We define a map of formal loop spaces

$$(2.3) i_{\check{\theta}}: \mathsf{L}X \to \mathsf{L}\mathscr{X}$$

as follows: By definition, we have a map $p: X \times T_X \to \mathscr{X}$. For a test ring R and $\gamma \in X(R((t)))$, define $i_{\check{\theta}}(\gamma) = p(\gamma, t^{-\check{\theta}}) \in \mathscr{X}(R((t)))$, where we are considering $t^{-\check{\theta}} \in T_X(R((t)))$.

Lemma 2.3.9. The point $i_{\check{a}}(x_0t^{\check{b}}) \in L\mathscr{X}(k)$ is contained in $L^+(\mathscr{X}^{\circ})$.

Proof. By Proposition 2.2.4 (and the ensuing discussion), there exists a non-canonical section $s: T_X \times \overline{T_{X,ss}} \to \mathscr{X}^{\circ}$ such that $s(1,1) = x_0$. If we choose a splitting $T \hookrightarrow G$ contained in the Levi L from Theorem 2.2.5, then the section s is $T \times T_X$ -equivariant, where T_X acts diagonally on $T_X \times \overline{T_{X,ss}}$. Thus on k((t))-points, we have $i_{\check{\theta}}(x_0 t^{\check{\theta}}) = p(x_0 t^{\check{\theta}}, t^{-\check{\theta}}) = s(t^{\check{\theta}} \cdot t^{-\check{\theta}}, t^{-\check{\theta}}) = s(1, t^{-\check{\theta}}),$ which lies in $\mathscr{X}^{\circ}(k[t])$ since $-\check{\theta} \in -\mathcal{V}$.

The map $i_{\check{\theta}}$ is LG-equivariant. As a consequence of the lemma, we see that the L⁺G-orbit of $i_{\check{\theta}}(x_0t^{\check{\theta}})$ is contained in $\mathsf{L}^+(\mathscr{X}^{\bullet})$. Thus $i_{\check{\theta}}$ restricts to a map $\mathsf{L}^{\check{\theta}}X \to \mathsf{L}^+(\mathscr{X}^{\bullet})$. We have a closed embedding $\mathsf{L}^+\mathscr{X} \hookrightarrow \mathsf{L}\mathscr{X}$ and an open embedding $\mathsf{L}^+(\mathscr{X}^{\bullet}) \hookrightarrow \mathsf{L}^+\mathscr{X}$.

Lemma 2.3.10. Assume that the stabilizer of B acting on $x_0 \in X^{\circ}$ is a smooth subgroup. Then the map $i_{\check{\theta}}$ induces an isomorphism

$$\mathsf{L}^{\check{\theta}}X \overset{\sim}{\to} \mathsf{L}X \underset{i_{\check{\theta}},\mathsf{L}\mathscr{X}}{\times} \mathsf{L}^{+}(\mathscr{X}^{\bullet}).$$

Proof. Since $i_{\check{\theta}}: \mathsf{L}X \to \mathsf{L}\mathscr{X}$ is injective on S-points, it suffices to show that $\mathsf{L}X \times_{i_{\check{\theta}},\mathsf{L}\mathscr{X}} \mathsf{L}^+(\mathscr{X}^{\bullet})$ is the L^+G -orbit of $i_{\check{\theta}}(x_0t^{\check{\theta}}) \in \mathsf{L}^+(\mathscr{X}^{\bullet})(k)$.

Let $\mathcal{B} = G/B$ denote the flag variety, and let $(\mathcal{X} \times \mathcal{B})^{\bullet} \subset \mathcal{X} \times \mathcal{B}$ denote the G-stable open subvariety consisting of all points (x, \tilde{B}) where $x \in \mathcal{X}$ in the open \tilde{B} -orbit. We have a smooth surjection $(\mathscr{X} \times \mathcal{B})^{\bullet} \to \mathscr{X}^{\bullet}$ by first projection. On the other hand, Proposition 2.2.4 gives a non-canonical G-equivariant isomorphism $\mathscr{X}^{\circ} \times^B G \cong (X^{\circ} \times^B G) \times \overline{T_{X,ss}}$. The action map and second projection induce an isomorphism $\mathscr{X}^{\circ} \times^{B} G \cong (\mathscr{X} \times \mathcal{B})^{\bullet}$ and similarly for $X^{\circ} \times^{B} G$. Thus there is a G-equivariant isomorphism

$$(\mathscr{X} \times \mathcal{B})^{\bullet} \cong (X \times \mathcal{B})^{\bullet} \times \overline{T_{X,ss}}$$

over the base $\overline{T_{X,ss}}$. The choice of base point $(x_0,B) \in (X \times \mathcal{B})^{\bullet}$ gives an isomorphism $G/\operatorname{Stab}_B(x_0) \cong (X \times \mathcal{B})^{\bullet}$. Thus we have a G-equivariant smooth surjection $G \times \overline{T_{X,ss}} \to \mathscr{X}^{\bullet}$ which induces a surjection

(2.4)
$$\mathsf{L}^+(G \times \overline{T_{X,\mathrm{ss}}}) \to \mathsf{L}^+(\mathscr{X}^{\bullet})$$

because $\operatorname{Stab}_B(x_0)$ is assumed to be smooth. Since the composition $LX \stackrel{i_{\check{\theta}}}{\to} L\mathscr{X} \to L\overline{T_{X,\operatorname{ss}}}$ sends everything to the image of $t^{-\check{\theta}}$ in $L\overline{T_{X,\operatorname{ss}}}(k)$, we deduce that (2.4) induces a surjection $L^+G \to LX \times_{i_{\check{\theta}},L\mathscr{X}} L^+(\mathscr{X}^{\bullet})$. Thus the latter fiber product is a single L^+G -orbit.

2.3.11. In this subsection we will use properties of toroidal compactifications of X. We refer the reader to [SV17, §2.3-2.5], [Kno91], [GN10, §8] for an overview.

Let \overline{X} denote a complete, smooth toroidal embedding of $X^{\bullet} = H \setminus G$ (the embedding is not simple if $N_G(H)/H$ is not finite). For any $\check{\theta} \in \mathcal{V} \cap \check{\Lambda}_X$, the point $x_0 \cdot t^{\check{\theta}} \in X^{\bullet}(F) \subset \overline{X}(F)$ defines an \mathfrak{o} -point of \overline{X} by the valuative criterion of properness. In particular, we can take the limit as $t \to 0$ to get a point $x_{\check{\theta}} := \lim_{t \to 0} x_0 \cdot t^{\check{\theta}} \in \overline{X}(k)$. Let $Z \subset \overline{X}$ denote the G-orbit of $x_{\check{\theta}}$.

Let Δ_X denote the set of normalized spherical roots of X. Equivalently, Δ_X is the set of simple coroots of \check{G}_X . Let $I \subset \Delta_X$ denote the set of spherical roots σ such that $\langle \sigma, \check{\theta} \rangle = 0$. In the language of [SV17, §2.3.6], the orbit Z "belongs to I-infinity". Let $X_I^{\bullet} = H_I \setminus G$ denote the open G-orbit on the normal bundle $N_Z \overline{X}$ of Z in \overline{X} ; this is called a boundary degeneration of X. As the notation suggests, X_I^{\bullet} and the conjugacy class of H_I depend only on the subset I and not $\check{\theta}$ (see [SV17, Proposition 2.5.3]). We call H_I a "satellite" of H, following the terminology of [BM20].

Lemma 2.3.12. Assume that B acts simply transitively on X° . Then the satellite subgroup $H_I \subset G$ is connected and smooth, for any subset $I \subset \Delta_X$.

Proof. We have $X^{\circ} \cong X_{I}^{\circ} \cong B$. Let H_{I}^{0} denote the reduced identity component of H_{J} . Then $H_{I}^{0}\backslash G \to H_{I}\backslash G$ is a finite covering of spherical varieties sending the open B-orbit of $H_{I}^{0}\backslash G$ to X_{I}° . Since B acts on $X_{I}^{\circ} \cong X^{\circ}$ with trivial stabilizer, the covering must be an isomorphism. \square

Corollary 2.3.13. Assume that B acts simply transitively on X° . Let $\check{\theta} \in \mathcal{V} \cap \check{\Lambda}_X$. Then the stabilizer of the group ind-scheme LH acting on $t^{\check{\theta}} \in \operatorname{Gr}_G$ is a connected scheme.

Proof. The cocharacter $-\check{\theta} \in (-\mathcal{V})$ extends to a map $\mathbb{A}^1 \to \overline{T_{X,\mathrm{ss}}}$. Let $\mathfrak{X} := \mathscr{X} \times_{\overline{T_{X,\mathrm{ss}}},-\check{\theta}} \mathbb{A}^1$. Then $\mathfrak{X} \to \mathbb{A}^1$ is an affine family with preimage over \mathbb{G}_m isomorphic to $X \times \mathbb{G}_m$. By [SV17, Proposition 2.5.3], the open G-orbit of the fiber over 0 is isomorphic to X_I^{\bullet} . Let \mathfrak{X}^{\bullet} denote the union of the open G-orbits on each fiber, so \mathfrak{X}^{\bullet} is a family over \mathbb{A}^1 degenerating X^{\bullet} to X_I^{\bullet} .

The map $i_{\tilde{\theta}}$ from (2.3) factors through an embedding

$$\widetilde{i}_{\check{ heta}}:\mathsf{L}X\hookrightarrow\mathsf{L}\mathfrak{X}\underset{\mathsf{L}\mathbb{A}^1}{ imes}$$
 pt

where $\mathsf{pt} \to \mathsf{L}\mathbb{A}^1$ corresponds to $t \in k((t))$. Lemma 2.3.9 implies that $\gamma := \tilde{i}_{\check{\theta}}(x_0 t^{\check{\theta}}) \in \mathsf{L}^+(\mathfrak{X}^{\bullet})$. Observe that $\mathsf{Stab}_{\mathsf{L}H}(t^{\check{\theta}}, \mathsf{Gr}_G) = \mathsf{L}H \cap t^{\check{\theta}}(\mathsf{L}^+G)t^{-\check{\theta}}$ conjugates to

$$t^{-\check{\theta}}(\mathsf{L}H)t^{\check{\theta}}\cap\mathsf{L}^+G=\operatorname{Stab}_{\mathsf{L}^+G}(x_0\cdot t^{\check{\theta}},\mathsf{L}X)=\operatorname{Stab}_{\mathsf{L}^+G}(\gamma,\mathsf{L}^+(\mathfrak{X}^\bullet)).$$

Let $\mathfrak{J} \subset G \times \mathfrak{X}^{\bullet}$ denote the inertia group scheme consisting of all $(g,x) \in G \times \mathfrak{X}^{\bullet}$ such that gx = x. The fibers of $\mathfrak{J} \to \mathfrak{X}^{\bullet}$ are all conjugate to either H or H_I . Since \mathfrak{X}^{\bullet} is smooth over \mathbb{A}^1 (cf. [GN10, Corollary 5.2.3]), we have H and H_I are of the same dimension, and they are smooth by Lemma 2.3.12. Thus $\mathfrak{J} \to \mathfrak{X}^{\bullet}$ is a smooth morphism.

Consider γ as an arc Spec $k[\![t]\!] \to \mathfrak{X}^{\bullet}$. For $n \in \mathbb{N}$, let $\gamma_n : \mathsf{D}_n := \mathrm{Spec}(k[t]/t^n) \to \mathfrak{X}^{\bullet}$ denote the corresponding n-jet and $\mathfrak{J}_n := \mathfrak{J} \times_{\mathfrak{X}^{\bullet}, \gamma_n} \mathsf{D}_n$ the fiber product, which is a smooth group scheme over \mathfrak{D}_n . Then

$$\operatorname{Stab}_{\mathsf{L}^+G}(\gamma,\mathsf{L}^+(\mathfrak{X}^{\bullet})) \cong \varprojlim_n \operatorname{Sect}(\mathsf{D}_n,\mathfrak{J}_n)$$

where $\operatorname{Sect}(\mathsf{D}_n,\mathfrak{J}_n)$ is the scheme of maps $\mathsf{D}_n\to\mathfrak{J}_n$ over D_n . By smoothness of $\mathfrak{J}\to\mathfrak{X}^{\bullet}$, the transition maps $\operatorname{Sect}(\mathsf{D}_m,\mathfrak{J}_m)\to\operatorname{Sect}(\mathsf{D}_n,\mathfrak{J}_n)$ are surjective for m>n. The transition

maps are also group homomorphisms with unipotent kernels, so they have contractible fibers. Therefore $\operatorname{Stab}_{\mathsf{L}^+G}(\gamma,\mathsf{L}^+(\mathfrak{X}^{\bullet}))$ contracts to $\operatorname{Sect}(\mathsf{D}_0,\mathfrak{J}_0)=\operatorname{Stab}_G(\gamma(0),X_1^{\bullet}),$ which is conjugate to the subgroup H_I . Since H_I is connected by Lemma 2.3.12, we are done.

3. Models for the arc space

In this section we define two models (in the sense of Grinberg-Kazhdan) for the arc space of X, both of which were already introduced in [GN10] and go back to ideas of Drinfeld. We call these the global and Zastava models (the term 'global' refers to the fact that the model depends on the curve C). The global model \mathcal{M}_X is crucial because it allows us to model the $G(\mathfrak{o})$ -orbits of $X(\mathfrak{o})$, something which cannot be done directly via the Zastava model. On the other hand, the Zastava model \mathcal{Y}_X is more suitable for finite type calculations.

Various incarnations of these constructions have been used in [FM99, FFKM99, BG02, BFGM02, BFG06, ABB+05]. To place our work in this context, we remark that when X = $\overline{N\backslash G}^{\mathrm{aff}}$ we have $\mathfrak{M}_X=\overline{\mathrm{Bun}}_N$ is Drinfeld's compactification of Bun_N and \mathfrak{Y}_X is "the" Zastava

While Gaitsgory-Nadler define the global and Zastava models for any affine X, in order to avoid various technical difficulties they faced (such as the existence of twisted strata, which are related to the existence of disconnected stabilizer subgroups) we make the following simplifying assumption:

Starting from §3.3, we assume for the rest of the paper that B acts simply transitively on X° . If $\check{G}_X = \check{G}$ and X has no spherical roots of type N, then the assumption above always holds.

Remark 3.0.1. The assumption that B acts simply transitively on X° implies that H must be connected, by Lemma 2.3.12.

3.1. Global model. Gaitsgory-Nadler [GN10] define certain stacks of meromorphic quasimaps from $C \longrightarrow X/G$ to model $X^{\bullet}(F)$, the loop space of X^{\bullet} . Our global model \mathcal{M}_X is the substack consisting of those quasimaps that extend to regular maps $C \to X/G$.

3.1.1. Definition. Define the stack

$$\mathcal{M}_X := \mathrm{Maps}_{\mathrm{gen}}(C, X/G \supset X^{\bullet}/G)$$

to be the open substack of Maps(C, X/G) representing maps generically landing in $X^{\bullet}/G =$ $(H\backslash G)/G=H\backslash \mathsf{pt}.$

An S-point of \mathcal{M}_X is a map $f: C \times S \to X/G$, which is equivalent to the datum (\mathcal{P}_G, σ) where

- \mathcal{P}_G is a G-bundle on $C \times S$ and
- $\sigma: C \times S \to X \times^G \mathcal{P}_G$ is a section over $C \times S$ such that $\sigma|_{\operatorname{Spec} k(C) \times S}$ lands in $X^{\bullet} \times^G \mathcal{P}_G = H \backslash \mathcal{P}_G$ and gives $\mathcal{P}_G|_{\operatorname{Spec} k(C) \times S}$ the structure of

We call the preimage $\sigma^{-1}(X^{\bullet} \times^G \mathcal{P}_G) \subset C \times S$ the locus of G-nondegenerate points.

Proposition 3.1.2. The natural map $\mathcal{M}_X \to \operatorname{Bun}_G$ is schematic locally of finite type. In particular, \mathcal{M}_X is an algebraic stack locally of finite type over k.

⁹In the literature, when $X = \overline{N \setminus G}$ regular maps $C \to \overline{N \setminus G}$ are still referred to as quasimaps to $N \setminus G$.

Proof. Since \mathcal{M}_X is an open substack of Maps(C, X/G), it suffices to show the latter is schematic locally of finite type over Bun_G . Let $S \to \mathrm{Bun}_G$ correspond to a G-bundle \mathcal{P}_G on $C \times S$. Then the fiber product $S \times \mathrm{Maps}(C, X/G)$ is isomorphic to the space of sections $C \times S \to X \times^G \mathcal{P}_G$ over $C \times S$. This space is representable by a k-scheme locally of finite type ([FGI⁺05, Theorem 5.23]).

If $X = H \setminus G$ is homogeneous, then $\mathfrak{M}_X = \operatorname{Maps}(C, H \setminus \mathsf{pt}) = \operatorname{Bun}_H$.

3.1.3. Adelic description. For motivational purposes, we give an "adelic" description of the k-points of \mathcal{M}_X . Let \mathbb{A} denote the restricted product $\prod_{v\in |C|}' F_v$ and $\mathbb{O} = \prod_{v\in |C|} \mathfrak{o}_v$. (If $k=\mathbb{F}_q$ then \mathbb{A} is the ring of adeles of the function field $\mathbb{k}=k(C)$.)

The underlying set of meromorphic quasimaps $C \longrightarrow X/G$ can be identified with the set

$$X^{\bullet}(\Bbbk) \overset{G(\Bbbk)}{\times} G(\mathbb{A})/G(\mathbb{O}) = H(\Bbbk) \backslash G(\mathbb{A})/G(\mathbb{O}).$$

Note that the G-action on X induces a map

$$X^{\bullet}(\Bbbk) \overset{G(\Bbbk)}{\times} G(\mathbb{A})/G(\mathbb{O}) \to X^{\bullet}(\mathbb{A})/G(\mathbb{O}) \subset X(\mathbb{A})/G(\mathbb{O}).$$

The underlying set of $\mathfrak{M}_X(k)$ identifies with the preimage of $(X^{\bullet}(\mathbb{A}) \cap X(\mathbb{O}))/G(\mathbb{O})$ under the map above.

Note that the topologies on $X^{\bullet}(\mathbb{A})$ vs. $X(\mathbb{A})$ are different: the fact that the geometric constructions above depend on X can be expressed by saying that we are always using the topology of $X(\mathbb{A})$, not of $X^{\bullet}(\mathbb{A})$.

3.1.4. For any set S, define the set of unordered multisets in S to be the formal direct sum

$$\mathrm{Sym}^\infty(S) := \bigoplus_{\check{\lambda} \in S} \mathbb{N}[\check{\lambda}].$$

An element of $\operatorname{Sym}^{\infty}(\mathsf{S})$ is a formal sum $\mathfrak{P} = \sum_{\check{\lambda} \in \mathsf{S}} N_{\check{\lambda}}[\check{\lambda}]$ where $N_{\check{\lambda}} \geq 0$ are integers and only finitely many are nonzero. For a multiset \mathfrak{P} , define

$$\mathring{C}^{\mathfrak{P}} = \prod_{\check{\lambda} \in S} \mathring{C}^{(N_{\check{\lambda}})}$$

to be the open subscheme of $C^{\mathfrak{P}}:=\prod C^{(N_{\tilde{\lambda}})}$ with all diagonals removed, i.e., the subscheme of multiplicity free divisors. We write $\mathfrak{P}=0$ for the zero element, and we will use the convention $\mathring{C}^0=C^0=\operatorname{pt}$. If we define $|\mathfrak{P}|=\sum N_{\check{\lambda}}$, then there are natural maps $C^{|\mathfrak{P}|}\to C^{\mathfrak{P}}\to C^{(|\mathfrak{P}|)}\subset \operatorname{Sym} C$.

Now consider the case when S = M - 0 for a commutative monoid M. Then $\operatorname{Sym}^{\infty}(M - 0)$ identifies with the set of *partitions* of arbitrary elements in M. Given a partition $\mathfrak{P} \in \operatorname{Sym}^{\infty}(M - 0)$ as above, we define $\deg : \operatorname{Sym}^{\infty}(M - 0) \to M$ by

$$\deg(\mathfrak{P}) := \sum_{\check{\lambda} \in \mathsf{M} = 0} N_{\check{\lambda}} \check{\lambda}$$

with addition taking place in M. There is a natural order on the set $\operatorname{Sym}^{\infty}(\mathsf{M}-0)$: we say that \mathfrak{P} refines \mathfrak{P}' if the difference $\mathfrak{P}-\mathfrak{P}'$ viewed as an element of $\bigoplus_{\check{\lambda}\in\mathsf{M}-0}\mathbb{Z}[\check{\lambda}]$ can be written as a sum of elements of the form $[\check{\lambda}']+[\check{\lambda}'']-[\check{\lambda}]$ with $\check{\lambda}'+\check{\lambda}''=\check{\lambda}$ in M.

Let $\operatorname{Prim}(M)$ be the set of primitive elements of M, i.e., the elements $\check{\lambda} \in M-0$ that cannot be decomposed as a sum $\check{\lambda} = \check{\lambda}_1 + \check{\lambda}_2$ where $\check{\lambda}_1, \check{\lambda}_2 \in M-0$. Then any partition in $\operatorname{Sym}^{\infty}(M-0)$ can be refined to an element in $\operatorname{Sym}^{\infty}(\operatorname{Prim}(M))$.

3.1.5. Stratification of \mathcal{M}_X . We would like to stratify \mathcal{M}_X according to $G(\mathfrak{o}_v)$ -orbits of $X(\mathfrak{o}_v) \cap X^{\bullet}(F_v)$ at each $v \in |C|$, described in Theorem 2.3.5.

Consider the set $\operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$ of partitions in \mathfrak{c}_X^- , as defined in §3.1.4. Let

$$\check{\Theta} = \sum_{\check{\theta} \in \mathfrak{c}_{\mathbf{Y}}^- - 0} N_{\check{\theta}}[\check{\theta}]$$

denote such a partition. We will write $\check{\Theta} = 0$ for the empty partition and $\check{\Theta} = [\check{\theta}]$ for the singleton partition corresponding to a single element $\check{\theta}$.

In §A.3, we define locally closed substacks $\mathcal{M}_X^{\check{\Theta}}$ of \mathcal{M}_X ranging over all partitions $\check{\Theta}$ in \mathfrak{c}_X^- . For simplicity we only describe $\mathcal{M}_X^{\check{\Theta}}$ on k-points below: Such a point consists of a map $f: C \to X/G$ and a formal sum $\sum_{v \in |C|} \check{\theta}_v \cdot v$ satisfying the following conditions:

- $\check{\theta}_v \neq 0$ for finitely many $v \in |C|$,
- for a fixed $\check{\theta} \neq 0$, the cardinality of $\{v \in |C| \mid \check{\theta}_v = \check{\theta}\}$ equals $N_{\check{\theta}}$,
- for each $v \in |C|$ the restriction $f|_{\operatorname{Spec} \mathfrak{o}_v} : \operatorname{Spec} \mathfrak{o}_v \to X/G$ defines a point in $\mathsf{L}^{\check{\theta}_v} X/\mathsf{L}^+G$.

Lemma 3.1.6. The substack $\mathfrak{M}_X^{\check{\Theta}}$ is smooth and locally closed in \mathfrak{M}_X .

We defer the proof of Lemma 3.1.6 to §A.3 of the appendix.

Proposition 3.1.7. Assume $k = \mathbb{C}$. Let S denote the collection of connected components of $\mathfrak{M}_X^{\check{\Theta}}$, ranging over all $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$. Then S is a Whitney¹⁰ stratification of \mathfrak{M}_X .

By a stratification we mean a collection of locally closed substacks that form a disjoint union on k-points and such that the closure of any stratum is a union of strata. The stratification is Whitney if the strata are smooth and every pair of strata satisfies Whitney's condition B. (This only makes sense if the characteristic of k is zero; in positive characteristic, see §3.1.9 below.)

The proof of Proposition 3.1.7 is given in §A.4.8. We call this the *fine stratification* of \mathcal{M}_X . The open stratum $\mathcal{M}_X^0 = \operatorname{Maps}(C, H \backslash G/G)$ identifies with Bun_H . We abbreviate $\mathcal{M}_X^{\check{\theta}} := \mathcal{M}_X^{[\check{\theta}]}$.

Remark 3.1.8. If X is affine and homogeneous, then $C_0(X) \cap \mathcal{V} = 0$ so the stratification is trivial, consisting of the single smooth stratum \mathcal{M}_X itself.

3.1.9. Let $D_{\mathbb{S}}^b(\mathcal{M}_X, \overline{\mathbb{Q}}_\ell)$ denote the subcategory of bounded S-constructible complexes, i.e., the usual cohomology sheaf $H^i(\mathcal{F})|_S$ is a local system of finite rank for all $i \in \mathbb{Z}$ and $S \in \mathbb{S}$. As explained in [BBDG18], the category $D_{\mathbb{S}}^b(\mathcal{M}_X)$ has a perverse t-structure. Let $P_{\mathbb{S}}(\mathcal{M}_X)$ denote the heart of this t-structure, i.e., this is the abelian subcategory of all perverse sheaves that are S-constructible.

In particular, the IC complex of the closure of any stratum $\mathcal{M}_X^{\check{\Theta}}$ is an object of $P_{\mathcal{S}}(\mathcal{M}_X)$. When k has positive characteristic, this is the condition on \mathcal{S} that we need (in place of the Whitney condition). Proposition A.4.11 shows that this condition indeed holds in positive characteristic.

3.2. **Toric case.** If we apply the definitions above to the special case where G is replaced by the torus T_X and X is replaced by the toric variety $X/\!\!/N$, we obtain the space

(3.1)
$$\mathcal{A} = \operatorname{Maps}_{gen}(C, (X//N)/T_X \supset \mathsf{pt})$$

of maps generically landing in $T_X/T_X = pt$.

 $^{^{10}}$ We say that a stratification on an algebraic stack \mathcal{M} locally of finite type is Whitney if the stratification is Whitney after pullback along any (equivalently all) smooth cover of \mathcal{M} by a scheme.

The stack \mathcal{A} has been previously studied in [BNS16, §3] as a model for the formal arc space of the toric variety $X/\!\!/N$ (in particular, \mathcal{A} turns out to be representable by a scheme). We review the relevant properties below.

For any $N \in \mathbb{N}$, we have the Nth symmetric product $C^{(N)}$ of C, which identifies with the Hilbert scheme $\operatorname{Hilb}^N(C)$ parametrizing relative effective divisors in C of degree N. Let $\operatorname{Sym} C$ denote the disjoint union $\bigsqcup_{N \in \mathbb{N}} C^{(N)}$ (where $C^{(0)} = \operatorname{pt}$).

Example 3.2.1. Observe that $\operatorname{Maps}_{\operatorname{gen}}(C, \mathbb{A}^1/\mathbb{G}_m \supset \operatorname{pt})$ sends a test scheme S to the set of relative effective Cartier divisors on $C \times S$, i.e., $\operatorname{Maps}_{\operatorname{gen}}(C, \mathbb{A}^1/\mathbb{G}_m \supset \operatorname{pt}) \cong \operatorname{Sym} C$. Addition of divisors gives $\operatorname{Sym} C$ the structure of a monoid.

Let $\mathfrak{c}_X^{\vee} = \operatorname{Hom}(\mathfrak{c}_X, \mathbb{N})$ denote the monoid dual to \mathfrak{c}_X , so $k[X/\!\!/N]$ is the semigroup algebra of \mathfrak{c}_X^{\vee} . Then there is an isomorphism

$$\mathcal{A} \cong \operatorname{Hom}(\mathfrak{c}_X^{\vee}, \operatorname{Sym} C)$$

where the right hand side represents homomorphisms of monoid objects in the category of schemes (with \mathfrak{c}_X^{\vee} viewed as a discrete scheme). A k-point of $\mathcal A$ is a formal finite sum

$$\sum_{v \in |C|} \check{\lambda}_v \cdot v$$

where $\check{\lambda}_v$ is an element of the dual monoid \mathfrak{c}_X and $\check{\lambda}_v = 0$ for all but finitely many v.

3.2.2. The stratification described in §3.1.5 takes here the following form: For any $\mathfrak{P} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X - 0)$ there is a natural map

$$\mathring{C}^{\mathfrak{P}} \hookrightarrow \mathcal{A}.$$

where the image consists of the k-points $\sum_{v \in |C|} \check{\lambda}_v \cdot v$ such that the unordered multiset of nonzero $\check{\lambda}_v$, counted with multiplicities, coincides with \mathfrak{P} .

Proposition 3.2.3 ([BNS16, Proposition 3.5]).

- (i) The maps $\mathring{\mathcal{C}}^{\mathfrak{P}}\hookrightarrow \mathcal{A}$ are locally closed embeddings, and the collection of such embeddings over all $\mathfrak{P}\in \operatorname{Sym}^{\infty}(\mathfrak{c}_X-0)$ forms a stratification of \mathcal{A} .
- (ii) $\mathring{C}^{\mathfrak{P}'}$ lies in the closure of $\mathring{C}^{\mathfrak{P}}$ if and only if \mathfrak{P} refines \mathfrak{P}' .
- (iii) The irreducible components of \mathcal{A} are in bijection with the closures of $\mathring{C}^{\mathfrak{P}}$ for $\mathfrak{P} \in \operatorname{Sym}^{\infty}(\operatorname{Prim}(\mathfrak{c}_X))$.

Corollary 3.2.4 ([BNS16, Corollary 3.6]). For $\check{\lambda} \in \mathfrak{c}_X$, let

$$\mathcal{A}^{\check{\lambda}}\subset\mathcal{A}$$

denote the subscheme whose k-points consist of all $\sum_{v \in |C|} \check{\lambda}_v \cdot v$ such that $\sum_v \check{\lambda}_v = \check{\lambda}$. Then $\mathcal{A}^{\check{\lambda}}$ is a connected component of \mathcal{A} , and this gives a bijection

$$\pi_0(\mathcal{A}) \cong \mathfrak{c}_X$$
.

Remark 3.2.5. If \mathfrak{c}_X is a free monoid with basis $\{\check{\nu}_i\}$, then for $\check{\lambda} = \sum_i N_i \check{\nu}_i$ we have $\mathcal{A}^{\check{\lambda}} = \prod_i C^{(N_i)}$. The bases for the Zastava spaces in [BFGM02, §2.1] take this form.

3.3. **Zastava model.** For the rest of this paper we assume that B acts simply transitively on X° . This implies that $T_X = T$ and $\Lambda_X = \Lambda_G$, so we will use the notation interchangeably.

We introduce a special case of the model used in [GN10, Part III], which is based on a general pattern pointed out by Drinfeld (see [Dri18, $\S4.2-4.4$]). These are a generalization of the Zastava¹¹ spaces introduced by Finkelberg–Mirković in [FM99, FFKM99, BFGM02], and we will henceforth call them the Zastava model for X.

The Zastava model for X is defined as

$$\mathfrak{Y} = \mathfrak{Y}_X = \mathrm{Maps}_{\mathrm{gen}}(C, X/B \supset \mathsf{pt}),$$

the stack of maps $C \times S \to X/B$ generically landing in $X^{\circ}/B = \mathsf{pt}$.

Applying Maps(C, ?/T) to the natural map $X/N \to X/\!\!/N$ induces a map

$$\pi: \mathcal{Y} \to \mathcal{A}.$$

For $\check{\lambda} \in \mathfrak{c}_X$, let $\mathcal{Y}^{\check{\lambda}}$ denote the preimage of $\mathcal{A}^{\check{\lambda}}$ under π .

We show in Proposition 3.7.2 below that \mathcal{Y} is representable by a scheme locally of finite type over k. This was predicted by Drinfeld [Dri18, Conjecture 4.2.3] in a more general setting.

Example 3.3.1. Let $X = \mathbb{G}_m \backslash \mathrm{GL}_2$ where \mathbb{G}_m is embedded as $(^1_{*})$. Then $\check{G}_X = \check{G} = \mathrm{GL}_2$ and $\check{\Lambda}_X = \check{\Lambda}_G = \mathbb{Z}^2$ with standard basis $\check{\varepsilon}_1, \check{\varepsilon}_2$. The B-orbits on X are the same as \mathbb{G}_m -orbits on $G/B = \mathbb{P}^1$, so there are three orbits: $\mathbb{G}_m, \{0\}, \{\infty\}$. These correspond to X° and two colors $D^+, D^- \subset X$, respectively. We have $\check{\nu}_{D^+} = \check{\varepsilon}_1, \check{\nu}_{D^-} = -\check{\varepsilon}_2$ and $\mathfrak{c}_X = \mathbb{N}^2$ is the free monoid generated by $\check{\nu}_{D^+}, \check{\nu}_{D^-}$. Note that $\check{\nu}_{D^+} + \check{\nu}_{D^-} = \check{\alpha}$ is the simple coroot.

Since X is affine homogeneous, $\mathcal{M}_X = \operatorname{Bun}_H = \operatorname{Bun}_1$ is the moduli stack of line bundles. The Zastava model is $\mathcal{Y} = \operatorname{Maps}_{\operatorname{gen}}(C, \mathbb{G}_m \backslash \operatorname{GL}_2/B \supset \operatorname{pt}) = \operatorname{Maps}_{\operatorname{gen}}(C, \mathbb{G}_m \backslash \mathbb{P}^1 \supset \operatorname{pt})$. This is the stack parametrizing two line bundles $\mathcal{L}, \mathcal{L}'$ on C and a fiberwise injective map of vector bundles $\sigma : \mathcal{L} \hookrightarrow \mathcal{O}_C \oplus \mathcal{L}'$ such that both coordinates $\sigma_1 : \mathcal{L} \hookrightarrow \mathcal{O}_C$ and $\sigma_2 : \mathcal{L} \hookrightarrow \mathcal{L}'$ are generically nonzero. Thus, σ_1, σ_2 are equivalent to two effective Cartier divisors D_1, D_2 which give $\mathcal{L} = \mathcal{O}(-D_1)$ and $\mathcal{L}' = \mathcal{O}(D_2 - D_1)$. The condition that $\sigma = (\sigma_1, \sigma_2)$ is fiberwise injective is equivalent to saying that the supports of D_1, D_2 are disjoint. Therefore, we have an identification

$$\mathcal{Y}_{\mathbb{G}_m \backslash \mathrm{GL}_2} = \mathrm{Sym} \, C \overset{\circ}{\times} \mathrm{Sym} \, C = \bigsqcup_{(n_1, n_2) \in \mathbb{N}^2} C^{(n_1)} \overset{\circ}{\times} C^{(n_2)}.$$

Meanwhile $\mathcal{A}_{\mathbb{G}_m \backslash \mathrm{GL}_2} = \mathrm{Sym}\, C \times \mathrm{Sym}\, C = \bigsqcup_{\mathbb{N}^2} C^{(n_1)} \times C^{(n_2)}$. In this case $\mathcal{Y} \to \mathcal{A}$ is the natural open embedding, and $\mathcal{Y}^{n_1\check{\nu}_{D^+} + n_2\check{\nu}_{D^-}} = C^{(n_1)}\mathring{\times} C^{(n_2)}$ is connected. The map $\mathcal{Y} \to \mathcal{M}_X$ forgetting the B-structure corresponds to $(D_1, D_2) \mapsto \mathcal{O}(D_2 - D_1)$.

Note that for the embedding above of $\mathbb{G}_m \hookrightarrow \mathrm{GL}_2$, the open B-orbit is the orbit of $\begin{pmatrix} 1 & 0 \\ 1 & 1 \end{pmatrix} \in \mathbb{G}_m \backslash \mathrm{GL}_2$. In particular X° does not contain the identity coset. If we conjugate \mathbb{G}_m to the embedding $\begin{pmatrix} 1 & 0 \\ 1-a & a \end{pmatrix}$, then we may take the base point $x_0 = 1$.

Example 3.3.2. In the example above we could instead replace GL_2 by $G = \operatorname{PGL}_2$ and $H = \mathbb{G}_m$ becomes the split torus inside PGL_2 . Then we still have $X = \mathbb{G}_m \backslash \operatorname{PGL}_2$ affine spherical with $\check{G}_X = \check{G} = \operatorname{SL}_2$ and two colors D^+, D^- . However now $\check{\Lambda}_X = \check{\Lambda}_G = \mathbb{Z}^{\frac{\check{\alpha}}{2}}$ and $\check{\nu}_{D^+} = \check{\nu}_{D^-} = \frac{\check{\alpha}}{2}$. The space $\mathcal{Y}_{\mathbb{G}_m \backslash \operatorname{PGL}_2}$ still identifies with $\operatorname{Sym} C \overset{\circ}{\times} \operatorname{Sym} C$. However now $\mathcal{A}_{\mathbb{G}_m \backslash \operatorname{PGL}_2} = \operatorname{Sym} C$ with $\mathcal{A}^{n\frac{\check{\alpha}}{2}} = C^{(n)}$. Then $\mathcal{Y}_{\mathbb{G}_m \backslash \operatorname{PGL}_2}^{n\frac{\check{\alpha}}{2}} = \bigcup_{n_1+n_2=n} C^{(n_1)} \overset{\circ}{\times} C^{(n_2)}$ and the map $\mathcal{Y} \to \mathcal{A}$ corresponds to addition of divisors.

¹¹ "Zastava" is Croatian for "flag".

Note that $\mathcal{M}_X = \operatorname{Bun}_1$ and $C^{(n_1)} \mathring{\times} C^{(n_2)}$ maps to the component $\operatorname{Bun}_1^{n_2-n_1}$ of degree $n_2 - n_1$ line bundles. Thus, while $\mathcal{Y}^{n\frac{\check{\alpha}}{2}}$ is not connected, fixing a connected component of Bun_1 determines the connected component of $\mathcal{Y}^{n\frac{\check{\alpha}}{2}}$.

3.4. **Graded factorization property.** Note that our assumption on X° implies that $X^{\circ}/B = \mathsf{pt}$ is a dense open substack of X/B. In the language of [Dri18, §4.2.1], the stack X/B is pointy. Drinfeld observed that maps from a curve to a pointy stack will have local behavior with respect to C (compared to [Dri18], we are in the special setting where we have a group B acting on X, not just a groupoid).

Let $\check{\lambda} = \check{\lambda}_1 + \check{\lambda}_2$ with $\check{\lambda}_i \in \mathfrak{c}_X$ and let us denote by $\mathcal{A}^{\check{\lambda}_1} \mathring{\times} \mathcal{A}^{\check{\lambda}_2}$ the open subset of the direct product $\mathcal{A}^{\check{\lambda}_1} \times \mathcal{A}^{\check{\lambda}_2}$ consisting of $\alpha_1, \alpha_2 : C \to (X/\!\!/N)/T$ such that the supports of $C - \alpha_1^{-1}(\mathsf{pt})$ and $C - \alpha_2^{-1}(\mathsf{pt})$ are disjoint. We have a natural étale map $\mathcal{A}^{\check{\lambda}_1} \mathring{\times} \mathcal{A}^{\check{\lambda}_2} \to \mathcal{A}^{\check{\lambda}}$.

Proposition 3.4.1. The scheme \forall has the graded factorization property, in the sense that there is a natural isomorphism

$$\mathcal{Y}^{\check{\lambda}}\underset{\mathcal{A}^{\check{\lambda}}}{\times}(\mathcal{A}^{\check{\lambda}_{1}}\overset{\circ}{\times}\mathcal{A}^{\check{\lambda}_{2}})\cong(\mathcal{Y}^{\check{\lambda}_{1}}\times\mathcal{Y}^{\check{\lambda}_{2}})|_{\mathcal{A}^{\check{\lambda}_{1}}\overset{\circ}{\times}\mathcal{A}^{\check{\lambda}_{2}}}.$$

Proof. Let $y_1, y_2: C \times S \to X/B$ be S-points of $\mathcal{Y}^{\check{\lambda}_1}, \mathcal{Y}^{\check{\lambda}_2}$, respectively. Let $U_i = y_i^{-1}(\mathsf{pt}) \subset C \times S$; the condition that $(\pi(y_1), \pi(y_2)) \in \mathcal{A}^{\check{\lambda}_1} \mathring{\times} \mathcal{A}^{\check{\lambda}_2}$ is equivalent to requiring that $U_1 \cup U_2 = C \times S$. Then $y_1|_{U_1 \cap U_2} \cong y_2|_{U_1 \cap U_2}: U_1 \cap U_2 \to \mathsf{pt}$ provides a gluing data for y_1, y_2 on the covering of $C \times S$ by U_1 and U_2 . Since X/B is a stack, the gluing data descends to a map $y: C \times S \to X/B$ that sends $U_1 \cap U_2$ to pt . This defines $y \in \mathcal{Y}^{\check{\lambda}}(S)$. The map in the opposite direction is constructed in the same way and they are mutually inverse.

We will henceforth use the notation $\mathcal{Y}^{\check{\lambda}_1} \overset{\circ}{\times} \mathcal{Y}^{\check{\lambda}_2}$ to denote $(\mathcal{Y}^{\check{\lambda}_1} \times \mathcal{Y}^{\check{\lambda}_2})|_{\mathcal{A}^{\check{\lambda}_1} \overset{\circ}{\times} \mathcal{A}^{\check{\lambda}_2}}$.

3.5. Global-to-Zastava yoga. We have a map $\mathcal{Y} \to \mathcal{M}_X$ by forgetting the *B*-structure. More precisely, we have an open embedding

$$\mathfrak{Y} \hookrightarrow \mathfrak{M}_X \underset{\operatorname{Bun}_G}{\times} \operatorname{Bun}_B.$$

Although the natural map $\operatorname{Bun}_B \to \operatorname{Bun}_G$ is in general not smooth, it is smooth over a large enough open substack: consider T as the Levi quotient of B^- . Let \mathfrak{n}^- denote the Lie algebra of N^- viewed as a T-module. Define the open substack $\operatorname{Bun}_T^r \subset \operatorname{Bun}_T$ to consist of those T-bundles \mathfrak{P}_T for which $H^1(C, V \times^T \mathfrak{P}_T) = 0$, for all T-modules V which appear as subquotients of \mathfrak{n}^- . Let Bun_B^r be the preimage of Bun_T^r under the natural projection $\operatorname{Bun}_B \to \operatorname{Bun}_T$.

For $\check{\mu} \in \check{\Lambda}_T$, let $\mathrm{Bun}_{\bar{L}}^{\check{\mu}}$ denote the corresponding connected component of Bun_T of degree $\check{\mu}$, and let $\mathrm{Bun}_B^{\check{\mu}}$ (resp. $\mathrm{Bun}_B^{\check{\mu},r}$) be its preimage in Bun_B (resp. Bun_B^r). Note that by Riemann–Roch, $\mathrm{Bun}_B^{\check{\mu},r} = \mathrm{Bun}_B^{\check{\mu},r}$ if $\langle \alpha_i, \check{\mu} \rangle > 2g-2$ for all simple roots α_i , where g is the genus of C. We say that $\check{\mu}$ is "large enough" if $\langle \alpha_i, \check{\mu} \rangle > N$ for all simple roots α_i and some $N \gg 0$.

Lemma 3.5.1 ([DS95], [BFGM02, Lemma 3.7], [GN10, Lemma 14.2.1]). The restriction of the map $\operatorname{Bun}_B \to \operatorname{Bun}_G$ to Bun_B^r is smooth. Any open substack $U \subset \operatorname{Bun}_G$ of finite type is contained in the image of $\operatorname{Bun}_B^{-\check{\mu}}$ for $\check{\mu}$ large enough, and the fibers of $\operatorname{Bun}_B^{-\check{\mu}} \to \operatorname{Bun}_G$ over U are geometrically connected.

Under our conventions, the composition $\mathcal{Y}^{\check{\lambda}} \to \operatorname{Bun}_B \to \operatorname{Bun}_T$ lands in the connected component $\operatorname{Bun}_T^{-\check{\lambda}}$.

Corollary 3.5.2. (i) The map $\mathcal{Y}^{\check{\mu}} \to \mathcal{M}_X$ is smooth with geometrically connected fibers (when nonempty) for $\check{\mu}$ large enough.

(ii) Any k-point of \mathfrak{M}_X lies in the image of $\mathcal{Y}^{\check{\mu}}$ for all $\check{\mu}$ in a translate of $\check{\Lambda}_G^{\mathrm{pos}}$.

Proof. We have an open embedding $\mathcal{Y}^{\check{\mu}} \hookrightarrow \mathcal{M}_X \times_{\operatorname{Bun}_G} \operatorname{Bun}_B^{-\check{\mu}}$, so (i) follows from Lemma 3.5.1 by base change. To show (ii), we consider the fiber of $\mathcal{Y} \to \mathcal{M}_X$ on k-points:

A point in $\mathcal{M}_X(k)$ is equivalent to a datum $(\mathcal{P}_G, \sigma : C \to X \times^G \mathcal{P}_G)$. First we show that there exists some point in $\mathcal{Y}(k)$ that maps to f. By [Ste65], there exists an open subscheme $U \subset C$ on which $\mathcal{P}_G|_U$ can be trivialized. If we fix such a trivialization, then $\sigma|_U$ identifies with a section $U \to H \setminus G$. Since $H \setminus G$ is spherical, $\sigma(U)$ intersects the open orbit of a Borel subgroup $gBg^{-1} \subset G$ for some $g \in G(k)$. Then g defines a point in $(G/B)(k) \subset (G/B)(k)$, where k = k(C). Using our fixed trivialization of $\mathcal{P}_G|_U$, we get a section

$$\operatorname{Spec} \mathbb{k} \xrightarrow{g} \operatorname{Spec} \mathbb{k} \times G/B \cong \mathfrak{P}_{G}^{0}|_{\operatorname{Spec} \mathbb{k}} \xrightarrow{G} (G/B) \cong \mathfrak{P}_{G}|_{\operatorname{Spec} \mathbb{k}} \xrightarrow{G} G/B,$$

which extends to a section $C \to \mathcal{P}_G \times^G G/B$ since G/B is proper. The latter is equivalent to giving a B-structure \mathcal{P}_B on \mathcal{P}_G . By construction (\mathcal{P}_B, σ) satisfies the generic condition for it to lie in $\mathcal{Y}(k)$, and (\mathcal{P}_B, σ) maps to $(\mathcal{P}_G, \sigma) \in \mathcal{M}_X(k)$.

Next let us fix an arbitrary lift of (\mathcal{P}_G, σ) to $(\mathcal{P}_B^1, \sigma) \in \mathcal{Y}(k)$. Fix a trivialization of $\mathcal{P}_B^1|_{\operatorname{Spec} \mathbb{k}}$, which also specifies a trivialization $\mathcal{P}_G|_{\operatorname{Spec} \mathbb{k}} \cong \mathcal{P}_G^0|_{\operatorname{Spec} \mathbb{k}}$. With respect to this trivialization, we have a bijection of sets

$$(3.5) H(\mathbb{k})B(\mathbb{k})/B(\mathbb{k}) \xrightarrow{\sim} \{ \text{lift of } (\mathcal{P}_G, \sigma) \text{ to a point in } \mathcal{Y}(k) \}$$

where the map is given by sending $h \in H(\mathbb{k})B(\mathbb{k})/B(\mathbb{k}) \subset (G/B)(\mathbb{k})$ to the section

$$\operatorname{Spec} {\Bbbk} \xrightarrow{h} \operatorname{Spec} {\Bbbk} \times G/B \cong {\mathcal P}_{G}^{0}|_{\operatorname{Spec} {\Bbbk}} \xrightarrow{G} (G/B) \cong {\mathcal P}_{G}|_{\operatorname{Spec} {\Bbbk}} \xrightarrow{G} G/B$$

and uniquely extending to a section $C \to \mathcal{P}_G \times^G G/B$. The point $B \in (G/B)(\mathbb{k})$ is sent under (3.5) to the lift $(\mathcal{P}_B^1, \sigma)$.

We are concerned with the possible degrees of \mathcal{P}_B for lifts $(\mathcal{P}_B, \sigma) \in \mathcal{Y}(k)$ above (\mathcal{P}_G, σ) . For a simple root α of G, let P_{α} denote the corresponding minimal parabolic in G. The image of $H \cap P_{\alpha}$ in $P_{\alpha}/\mathfrak{R}(P_{\alpha}) = \operatorname{PGL}_2$ must contain a subgroup conjugate to $\binom{* \ 0}{* \ *}$ or $\binom{1 \ *}{0 \ 1}$. Thus we can choose an identification $P_{\alpha}/B \cong \mathbb{P}^1$ such that the image of $(H \cap P_{\alpha})(\mathbb{k})$ in $\mathbb{P}^1(\mathbb{k})$ contains $\mathbb{k}^{\times} = k(C)^{\times}$. For $f \in (H \cap P_{\alpha})(\mathbb{k})$, let \bar{f} denote its image in $(P_{\alpha}/B)(\mathbb{k}) \cong \mathbb{P}^1(\mathbb{k})$. For such a function f, let (\mathcal{P}_B, σ) denote the corresponding lift of (\mathcal{P}_G, σ) under the bijection (3.5). If $\bar{f} \in k(C)^{\times}$, then \mathcal{P}_B is of degree $-(\check{\mu}_1 + N\check{\alpha})$, where $-\check{\mu}_1$ is the degree of \mathcal{P}_B^1 and N is the degree of the divisor of zeros of \bar{f} .

By the Riemann–Roch theorem, we have a rational function $\bar{f} \in k(C)^{\times}$ with divisor of zeros of degree N for any $N \gg 0$. Therefore for any $N \gg 0$, there is a lift \mathcal{P}_B of degree $-(\check{\mu}_1 + N\check{\alpha})$. We conclude that there exists a lift $(\mathcal{P}_B, \sigma) \in \mathcal{Y}^{\check{\mu}}(k)$ for any $\check{\mu}$ in a certain translate of $\check{\Lambda}_G^{\text{pos}}$. \square

3.5.3. It is not in general true that the natural map $\mathcal{Y}^{\check{\lambda}} \to \mathcal{M}_X$ is smooth for arbitrary $\check{\lambda}$. However, the following well-known argument (cf. [GN10, Theorem 16.2.1]) shows that any neighborhood of a point in $\mathcal{Y}^{\check{\lambda}}$ is smooth locally isomorphic to a neighborhood of a point in \mathcal{M}_X :

Let $\check{\lambda} \in \mathfrak{c}_X$ be arbitrary. Since $\check{\Lambda}_G^{\mathrm{pos}} \subset \mathfrak{c}_X$, we can always find $\check{\mu} \in \check{\Lambda}_G^{\mathrm{pos}}$ large enough such that $\check{\lambda} + \check{\mu}$ is also large enough. Let $\mathcal{Y}^{\check{\mu},0}$ denote the preimage of $\mathcal{M}_X^0 = \mathrm{Bun}_H$ under the smooth map $\mathcal{Y}^{\check{\mu}} \to \mathcal{M}_X$. Then $\mathcal{Y}^{\check{\mu},0}$ is smooth and the first projection

$$\mathcal{Y}^{\check{\lambda}}\times\mathcal{Y}^{\check{\mu},0}\to\mathcal{Y}^{\check{\lambda}}$$

is smooth. On the other hand, by the graded factorization property (Proposition 3.4.1), there is a natural étale map

$$\mathcal{Y}^{\check{\lambda}} \mathring{\times} \mathcal{Y}^{\check{\mu},0} \to \mathcal{Y}^{\check{\lambda}+\check{\mu}}.$$

We can compose this with the smooth map $\mathcal{Y}^{\check{\lambda}+\check{\mu}} \to \mathcal{M}_X$ to get a smooth map $\mathcal{Y}^{\check{\lambda}} \mathring{\times} \mathcal{Y}^{\check{\mu},0} \to \mathcal{M}_X$. To summarize, we have constructed:

Lemma 3.5.4. For any $\check{\lambda} \in \mathfrak{c}_X$ and any $\check{\mu} \in \check{\Lambda}_G^{\mathrm{pos}}$ large enough, there is a correspondence

$$y^{\check{\lambda}} \leftarrow y^{\check{\lambda}} \mathring{\times} y^{\check{\mu},0} \rightarrow \mathfrak{M}_X$$

where the left arrow is smooth surjective, and the right arrow is smooth.

3.6. Stratification of the Zastava model. We stratify $\mathcal{Y}^{\check{\lambda}}$ according to the fine stratification of \mathcal{M}_X : for a partition $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$ and $\check{\lambda} \in \mathfrak{c}_X$, define the stratum

$$\mathcal{Y}^{\check{\lambda}, \check{\Theta}} := \mathcal{Y}^{\check{\lambda}} \underset{\mathcal{M}_X}{\times} \mathcal{M}_X^{\check{\Theta}}.$$

Abbreviate $\mathcal{Y}^{\check{\lambda},\check{\theta}} := \mathcal{Y}^{\check{\lambda},[\check{\theta}]}$.

Proposition 3.6.1. The stratum $\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ is a smooth locally closed subscheme of $\mathcal{Y}^{\check{\lambda}}$.

Proof. By Lemma 3.5.4, there exists $\check{\mu} \in \check{\Lambda}_X$ such that there is a smooth correspondence

$$y^{\check{\lambda}} \leftarrow y^{\check{\lambda}} \overset{\circ}{\times} y^{\check{\mu},0} \rightarrow \mathfrak{M}_X$$

where the left arrow is surjective. Note that by definition of $\mathcal{M}_X^{\check{\Theta}}$, the preimage of $\mathcal{M}_X^{\check{\Theta}}$ in $\mathcal{Y}^{\check{\lambda},\check{\Theta}}\mathring{\times}\mathcal{Y}^{\check{\mu},0}$ is isomorphic to $\mathcal{Y}^{\check{\lambda},\check{\Theta}}\mathring{\times}\mathcal{Y}^{\check{\mu},0}$ since $\mathcal{Y}^{\check{\mu},0}$ consists of maps $C\to X^{\bullet}/B$ which can only define points in $\mathsf{L}^0X/\mathsf{L}^+G$ upon restriction to \mathfrak{o}_v for any $v\in |C|$. Therefore, we get a smooth correspondence

$$(3.6) \hspace{3cm} \mathcal{Y}^{\check{\lambda}, \check{\Theta}} \leftarrow \mathcal{Y}^{\check{\lambda}, \check{\Theta}} \mathring{\times} \mathcal{Y}^{\check{\mu}, 0} \rightarrow \mathcal{M}_{X}^{\check{\Theta}}$$

where the left arrow is still surjective. Now smoothness of $\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ follows from smoothness of $\mathcal{M}_X^{\check{\Theta}}$ (Lemma 3.1.6).

We call the collection of connected components of $\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ the fine stratification of $\mathcal{Y}^{\check{\lambda}}$. By the smooth correspondence (3.6) above and Proposition 3.1.7, this is a Whitney stratification (in fact the Zastava model is used in the proof of *loc cit*.).

Note that for a fixed $\check{\lambda}$, many of the $\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ are empty.

3.7. Relation to the affine Grassmannian. Let $\operatorname{Gr}_{G,\operatorname{Sym} C} \to \operatorname{Sym} C$ denote the following version of the Beilinson–Drinfeld affine Grassmannian: an S-point consists of a relative effective Cartier divisor $D \subset C \times S$ and a G-bundle \mathcal{P}_G on $C \times S$ together with a trivialization $\mathcal{P}_G|_{C \times S - D} \cong \mathcal{P}_G^0|_{C \times S - D}$ where \mathcal{P}_G^0 is the trivial G-bundle. For any linear algebraic group G, the functor $\operatorname{Gr}_{G,\operatorname{Sym} C}$ is representable by an ind-scheme, ind-of finite type over $\operatorname{Sym} C$ (cf. [BD96], [Zhu17, Theorem 3.1.3]).

In particular, we can consider the ind-scheme $\operatorname{Gr}_{B,\operatorname{Sym} C}$. Let $\operatorname{Gr}_{B,C^{(N)}}$ denote the preimage over $C^{(N)}$.

3.7.1. Choose some $\delta \in \Lambda_X$ that lies on the interior of the cone dual to $C_0(X)$, so $\langle \delta, \check{\lambda} \rangle > 0$ for any nonzero $\check{\lambda} \in C_0(X)$. Let $f_{\delta} \in k[X]^{(B)}$ denote the corresponding δ -eigenfunction. Then f_{δ} induces a map $X /\!\!/ N \to \mathbb{A}^1$, which in turn induces a map $A \to \operatorname{Sym} C$ sending $A^{\check{\lambda}} \to C^{(\langle \delta, \check{\lambda} \rangle)}$. We can map¹²

$$\mathfrak{Y}^{\check{\lambda}} \to \operatorname{Gr}_{BC(\langle \delta, \check{\lambda} \rangle)}$$

¹²The effective Cartier divisor cut out by f_{δ} has the property that the complement of its support in X equals X° . In this guise, the map $\mathcal{Y} \to \operatorname{Sym} C$ we have constructed coincides with the one described in [Dri18, Remark 4.2.6].

as follows: let $y: C \times S \to X/B$ be an S-point of \mathcal{Y} . Let $D \subset C \times S$ be the relative effective divisor corresponding to the image of $\pi(y) \in \mathcal{A}(S)$ under the map $\mathcal{A} \to \operatorname{Sym} C$. Since δ was chosen in the interior of the dual cone of $\mathcal{C}_0(X)$, we have $C \times S - D = y^{-1}(\operatorname{pt}) =: U$. Thus, y defines a B-bundle \mathcal{P}_B on $C \times S$ together with a section $U \to X^\circ \times^B \mathcal{P}_B \cong \mathcal{P}_B$, i.e., a trivialization of $\mathcal{P}_B|_U$. The datum $(D, \mathcal{P}_B, \mathcal{P}_B|_U \cong \mathcal{P}_B^0|_U)$ defines an S-point of $\operatorname{Gr}_{B,\operatorname{Sym} C}$.

Proposition 3.7.2. The map (3.7) is a closed embedding. Moreover, the stack \mathcal{Y} is representable by a scheme locally of finite type over k, and for fixed $\check{\lambda} \in \mathfrak{c}_X$, the scheme $\mathcal{Y}^{\check{\lambda}}$ is of finite type over k. Consequently, the map $\pi : \mathcal{Y} \to \mathcal{A}$ is schematic of finite type.

Proof. First we show that (3.7) is a closed embedding. Fix a map $S \to \operatorname{Gr}_{B,\operatorname{Sym} C}$ corresponding to a pair (D, \mathcal{P}_B) and a trivialization of \mathcal{P}_B away from D. A trivialization of \mathcal{P}_B on $C \times S - D$ is equivalent to a section $\sigma_0 : C \times S - D \to \mathcal{P}_B \cong X^{\circ} \times^B \mathcal{P}_B$. The fiber of $\mathcal{Y} \to \operatorname{Gr}_{B,\operatorname{Sym} C}$ over S parametrizes commutative diagrams

$$C \times S' - D' \xrightarrow{\sigma_0} X^{\circ} \overset{B}{\times} \mathfrak{P}_B$$

$$\downarrow \qquad \qquad \downarrow$$

$$C \times S' \xrightarrow{\sigma} X \overset{B}{\times} \mathfrak{P}_B$$

where S' is an S-scheme, $D' := D \times_S S'$, and σ is a section over $C \times S$. Observe that σ is uniquely determined by σ_0 . By Lemma 3.7.3 below, the condition that σ_0 extends to σ is closed in S.

We have shown that \mathcal{Y} is representable by an ind-scheme ind-closed in $\operatorname{Gr}_{B,\operatorname{Sym}C}$. On the other hand, we have a map $\mathcal{Y} \to \operatorname{Bun}_B$ whose fiber over an S-point $\mathcal{P}_B \in \operatorname{Bun}_B(S)$ is open in the space of sections $C \times S \to X \times^B \mathcal{P}_B$ over $C \times S$. This space is representable by a scheme locally of finite type over k ([FGI⁺05, Theorem 5.23]). Since Bun_B is an algebraic stack, we conclude that \mathcal{Y} is an algebraic stack representable by an ind-scheme. Hence \mathcal{Y} is representable by a scheme. For fixed $\check{\lambda} \in \mathfrak{c}_X$, we now know that $\mathcal{Y}^{\check{\lambda}}$ is a closed subscheme of $\operatorname{Gr}_{B,C(\langle \delta,\check{\lambda}\rangle)}$, which is ind-of finite type. It follows that $\mathcal{Y}^{\check{\lambda}}$ is of finite type over k. The other assertions all follow.

Lemma 3.7.3. Let S be a test scheme and $D \subset C \times S$ a relative effective divisor. Let $\mathfrak X$ be a scheme affine of finite presentation over $C \times S$. Suppose that there exists a section σ : $C \times S - D \to \mathfrak X$ over $C \times S$. Then the functor sending S' to the set of maps $S' \to S$ such that σ extends to a regular map on $C \times S'$ is representable by a closed subscheme of S.

Proof. The map σ is equivalent to a map of $\mathcal{O}_{C\times S}$ -algebras $\mathcal{O}_{X}\to \mathcal{O}_{C\times S-D}$. Given $S'\to S$, the condition that σ extends to $C\times S'$ is equivalent to requiring the image of $\mathcal{O}_{X}\to \mathcal{O}_{C\times S-D}$ to land in $\mathcal{O}_{C\times S'}\subset \mathcal{O}_{C\times S'-D'}$ after base change to S', where $D':=D\times_S S'$. The claim is local in S, so we may assume that \mathcal{O}_{X} is surjected onto by $\operatorname{Sym}_{\mathcal{O}_{C\times S}}(\mathcal{E}^{\vee})$ for some vector bundle \mathcal{E} on $C\times S$. Then we just need the composed $\mathcal{O}_{C\times S}$ -linear map $\mathcal{E}^{\vee}\to \mathcal{O}_{C\times S-D}\to \mathcal{O}_{C\times S-D}/\mathcal{O}_{C\times S}$ to vanish after base change to S'. Since \mathcal{E}^{\vee} is coherent, the image of this map is contained in a submodule $\mathcal{F}=\mathcal{O}(m\cdot D)/\mathcal{O}_{C\times S}$ for some integer $m\geq 0$. The projections $p:C\times S\to S$, $p':C\times S'\to S'$ are proper, and we are considering when an element in $H^0p'_*(\mathcal{E}\otimes_{\mathcal{O}_{C\times S}}\mathcal{F}\otimes_{\mathcal{O}_S}\mathcal{O}_{S'})$ vanishes. Note that $p_*(\mathcal{E}\otimes_{\mathcal{O}_{C\times S}}\mathcal{F})$ is finite locally free as an \mathcal{O}_S -module. By cohomology and base change we are reduced to asking when an element of $p_*(\mathcal{E}\otimes_{\mathcal{O}}\mathcal{F})$ vanishes after base change to S'. This is a closed condition on S.

Remark 3.7.4 (Open curves). The graded factorization property of $\mathcal{Y}^{\check{\lambda}}$ implies that the geometry of $\mathcal{Y}^{\hat{\lambda}}$ is purely local with respect to the curve C. Therefore, we could define

$$\mathcal{Y}(C) = \mathrm{Maps}_{\mathrm{gen}}(C, X/B \supset \mathsf{pt})$$

for any smooth curve C (not necessarily projective), and all the same properties would still hold. For example, in [FM99], [Dri18], the affine curve $C = \mathbb{A}^1$ is used.

3.7.5. Beauville-Laszlo's theorem. Let S be an affine scheme and D a closed affine subscheme of $C \times S$. Denote by \widehat{C}_D the formal completion of $C \times S$ along D and by \widehat{C}'_D the spectrum of the ring of regular functions on \widehat{C}_D (so \widehat{C}_D is an ind-affine formal scheme and \widehat{C}'_D is the corresponding true scheme). Let $\widehat{C}_D^{\circ} := \widehat{C}_D' - D$ denote the open subscheme. There are maps $\widehat{p} : \widehat{C}_D \to C \times S$ and $i : \widehat{C}_D \to \widehat{C}_D'$. We will implicitly use the following fact

in what follows:

Proposition 3.7.6 ([BD96, Proposition 2.12.6]). There exists a unique map $p: \widehat{C}'_D \to C \times S$ such that $\hat{p} = p \circ i$.

To justify that \mathcal{Y} is of local nature, we record the following consequence of the globalized version of Beauville-Laszlo's theorem (cf. [BD96, Theorem 2.12.1], [BL95]). Let S be an affine scheme and $D \subset C \times S$ a relative effective divisor. Proposition 3.7.6 implies that there exists a map $p: \widehat{C}'_D \to C \times S$.

Lemma 3.7.7. Let X be any affine scheme with an action of an algebraic group B such that X/B is pointy, i.e., X has an open B-orbit X° with $X^{\circ}/B = \mathsf{pt}$. Let C, S and D be as above. Then there is a natural equivalence between the following categories:

- (i) the groupoid of (\mathfrak{P}_B, σ) where \mathfrak{P}_B is a B-bundle on $C \times S$ and a section $\sigma : C \times S \to S$ $X \times^B \mathfrak{P}_B$ that sends $C \times S - D$ to $X^{\circ} \times^B \mathfrak{P}_B$,
- (ii) the groupoid of $(\hat{\mathcal{P}}_B, \hat{\sigma})$ where $\hat{\mathcal{P}}_B$ is a B-bundle on \widehat{C}'_D and a section $\hat{\sigma}: \widehat{C}'_D \to X \times^B \hat{\mathcal{P}}_B$ that sends \widehat{C}_D° to $X^{\circ} \times^B \widehat{\mathcal{P}}_B$.

Proof. The functor from (i) to (ii) is just pullback along p. To define the functor from (ii) to (i) we descend along the covering $\widehat{C}'_D \sqcup (C \times S - D) \to C \times S$. The justification for this is Beauville–Laszlo's theorem (cf. [BD96, Theorem 2.12.1]). First, $\hat{\sigma}$ induces a section $\widehat{C}_D^{\circ} \to X^{\circ} \times^B \widehat{\mathcal{P}}_B \cong \widehat{\mathcal{P}}_B$. Then we can "descend" $\widehat{\mathcal{P}}_B$ to a *B*-bundle \mathcal{P}_B on $C \times S$ with a section $C \times S - D \to \mathcal{P}_B$ (which is equivalent to a trivialization of $\mathcal{P}_B|_{C \times S - D}$). The section $\hat{\sigma}$ is equivalent to a map of quasicoherent $\mathcal{O}_{\widehat{C}'_D}$ -algebras $\mathcal{O}_{X \times^B \hat{\mathcal{P}}_B} \to \mathcal{O}_{\widehat{C}'_D}$ since X is affine. Again by [BD96, Theorem 2.12.1], this descends to a map of quasicoherent $\mathcal{O}_{C\times S}$ -algebras $\mathcal{O}_{X \times^B \mathcal{P}_B} \to \mathcal{O}_{C \times S}$ such that the restriction to $C \times S - D$ factors through the trivialization $\mathcal{O}_{X^{\circ} \times^B \mathcal{P}_B} \cong \mathcal{O}_{\mathcal{P}_B} \to \mathcal{O}_{C \times S - D}$. By construction, the two functors are mutually inverse.

3.8. Theorem of Grinberg-Kazhdan, Drinfeld. We now justify why \mathcal{M}_X and \mathcal{Y} are indeed "models" for the formal arc space L^+X . Since \mathfrak{M}_X and \mathfrak{Y} are smooth-locally isomorphic, it suffices to explain the latter.

Definition 3.8.1. A finite type formal model of L^+X at $\gamma_0 \in L^+X(k)$ is the formal completion \hat{Y}_y of a k-scheme of finite type Y at a point $y \in Y$ equipped with an isomorphism of formal schemes

$$\widehat{\mathsf{L}^+X}_{\gamma_0} \simeq \widehat{Y}_y \times \widehat{\mathbb{A}}^\infty,$$

where $\widehat{\mathbb{A}}^{\infty}$ is the product of countably many copies of the formal disk Spf k[t].

Since $X/B \supset pt$ is a pointy stack, Drinfeld's proof [Dri18, §4.2-4.3] of the Grinberg–Kazhdan theorem essentially shows that the scheme \mathcal{Y} (which we have shown is a disjoint union of finite type schemes) explicitly satisfies the following:

Theorem 3.8.2 (Grinberg–Kazhdan, Drinfeld). Fix an arc γ_0 : Spec $k[\![t]\!] \to X$ in $L^+X(k)$ such that $\gamma_0(\operatorname{Spec} k(\!(t)\!)) \subset X^\circ$. Then there exists a point $y \in \mathcal{Y}(k)$ such that the formal completion of \mathcal{Y} at y is a finite type formal model of L^+X at γ_0 .

More precisely, if γ_0 belongs to the stratum $\mathsf{L}^{\check{\theta}}X$, for $\check{\theta} \in \mathfrak{c}_X^-$, we can take y to be the point $t^{\check{\theta}}$ in the central fiber $\mathsf{Y}^{\check{\theta},\check{\theta}}$ over any point $v \in |C|$.

The central fiber $\mathsf{Y}^{\check{\theta}}$ over a point $v \in |C|$ is the fiber of $\mathcal{Y}^{\check{\theta}} \to \mathcal{A}^{\check{\theta}}$ over the "diagonal divisor" $\check{\theta} \cdot v \in \mathcal{A}^{\check{\theta}}(k)$. We define $\mathsf{Y}^{\check{\theta},\check{\theta}} := \mathcal{Y}^{\check{\theta}} \times_{\mathcal{M}_X} \mathcal{M}_X^{\check{\theta}}$. Then $\mathsf{Y}^{\check{\theta}}$ is naturally a subvariety of Gr_B (see §4.3), and $t^{\check{\theta}}$ denotes the corresponding point in Gr_B . The significance of this point will become evident in Corollary 5.5.6(iii).

The statements all follow from the proof of [Dri18, §4]. We also give the same argument, with some notational changes, in the proof of Theorem 8.2.4.

Remark 3.8.3. The point $y: C \to X/B$ can be chosen so that $y^{-1}(\mathsf{pt}) = C - v$ for a single point $v \in |C|$. However it is essential, for the theorem to hold, that \mathcal{Y} contains maps with multiple points of C mapping to $(X - X^{\circ})/B$.

4. Compactification of the Zastava model

The map $\pi : \mathcal{Y} \to \mathcal{A}$ defined in (3.3) is in general not proper, so for example we cannot apply the decomposition theorem. To rectify this, we introduce a compactification.

4.1. Basic properties. Let $\overline{G/N} = \operatorname{Spec} k[G/N]$ denote the canonical affine closure of the quasiaffine variety G/N. For an arbitrary connected reductive group G, Drinfeld's compactification $\overline{\operatorname{Bun}}_B$ is defined¹³ as the closure of Bun_B inside

$$\operatorname{Maps}_{\operatorname{gen}}(C, G \setminus \overline{G/N}/T \supset \operatorname{pt}/B),$$

the stack parametrizing maps $C \to G \setminus \overline{G/N}/T$ that generically land in the open substack $G \setminus (G/N)/T = \operatorname{pt}/B$. (When [G, G] is simply connected, [BG02, Proposition 1.2.3] show that Bun_B is dense in $\operatorname{Maps}_{\operatorname{gen}}(C, G \setminus \overline{G/N}/T \supset \operatorname{pt}/B)$.)

Consider the stack quotient $X \times^G \overline{G/N} := (X \times \overline{G/N})/G$, where G acts anti-diagonally. Then $X/N = X \times^G G/N$ is an open substack of $X \times^G \overline{G/N}$, so we also have the open substack $\mathsf{pt} = X^\circ/B \subset X/B \subset X \times^G \overline{G/N}/T$. Define

$$\overline{\mathcal{Y}} = (\mathcal{M}_X \underset{\operatorname{Bun}_G}{\times} \overline{\operatorname{Bun}}_B)^{\circ} \subset \operatorname{Maps}_{\operatorname{gen}}(C, X \overset{G}{\times} \overline{G/N}/T \supset \operatorname{pt})$$

where the superscript $^{\circ}$ denotes the open substack of

$$\mathfrak{M}_X \underset{\mathrm{Bun}_G}{\times} \overline{\mathrm{Bun}}_B \subset \mathrm{Maps}_{\mathrm{gen}}(C, X \overset{G}{\times} \overline{G/N}/T \supset X^{\bullet}/B)$$

parametrizing maps generically landing in $\operatorname{pt} = X^{\circ}/B$. In particular, $\overline{\mathcal{Y}}$ is an algebraic stack locally of finite type. We can identify $\mathcal{Y} \cong \overline{\mathcal{Y}} \times_{\overline{\operatorname{Bun}}_B} \operatorname{Bun}_B$ as an open substack of $\overline{\mathcal{Y}}$. (If [G,G] is simply connected, the containment in (4.1) is an equality.)

¹³The definition as a closure is only true in characteristic 0. In positive characteristic, see [ABB⁺05, §4.1], [Sch15, §7.2].

There is a natural map from $X \times^G \overline{G/N}$ to

$$(X \times \overline{G/N}) /\!\!/ G = \operatorname{Spec} k [X \times G]^{G \times N}.$$

Since $k[X \times G]^G = k[X]$, we deduce that $(X \times \overline{G/N}) /\!\!/ G = X /\!\!/ N$. Therefore, we have a map $X \times^G \overline{G/N} \to X /\!\!/ N$ extending the natural map $X/N \to X /\!\!/ N$. Applying Maps(C,?/T) to the former, we have constructed a map

$$(4.2) \bar{\pi}: \overline{\mathcal{Y}} \to \mathcal{A}$$

extending $\pi: \mathcal{Y} \to \mathcal{A}$. Let $\overline{\mathcal{Y}}^{\check{\lambda}}$ denote the preimage of the subscheme $\mathcal{A}^{\check{\lambda}}$. The same proof as in Proposition 3.4.1 shows that $\overline{\mathcal{Y}}$ has the graded factorization property. We will see from Lemma 4.1.2 below that $\overline{\mathcal{Y}}^{\check{\lambda}}$ is representable by a scheme of finite type over k.

First, we show that $\bar{\pi}$ is indeed a compactification:

Proposition 4.1.1. The map $\bar{\pi}: \bar{\mathcal{Y}} \to \mathcal{A}$ is proper.

We proceed as in §3.7.1. Choose $\delta \in \Lambda_X$ lying on the interior of the cone dual to $\mathcal{C}_0(X)$. Then δ defines a map $X/\!\!/ N \to \mathbb{A}^1$, which induces a map $A \to \operatorname{Sym} C$. This allows us to consider \mathcal{Y} as a scheme over $\operatorname{Sym} C$. Proposition 3.7.2 gives a closed embedding $\mathcal{Y} \hookrightarrow \operatorname{Gr}_{B,\operatorname{Sym} C}$ over $\operatorname{Sym} C$. We compose this with the natural map $\operatorname{Gr}_{B,\operatorname{Sym} C} \to \operatorname{Gr}_{G,\operatorname{Sym} C}$ to get a map $\mathcal{Y} \to \operatorname{Gr}_{G,\operatorname{Sym} C}$. We can extend this to a map

$$(4.3) \overline{y} \to \operatorname{Gr}_{G,\operatorname{Sym} C}$$

using the same idea as in the definition of (3.7). Namely, let $y: C \times S \to X \times^G \overline{G/N}/T$ be an S-point of $\overline{\mathcal{Y}}$ and let $D \subset C \times S$ denote the divisor it maps to. In particular, y defines a G-bundle \mathcal{P}_G on $C \times S$ with a B-reduction on $C \times S - D$. Since $y(C \times S - D) = X^{\circ}/B = \mathsf{pt}$, the G-bundle in fact admits a trivialization on $C \times S - D$. The data of D, \mathcal{P}_G , and the trivialization defines an S-point of $\mathrm{Gr}_{G,\mathrm{Sym}\,C}$.

Lemma 4.1.2. The map
$$\overline{\mathcal{Y}} \to \operatorname{Gr}_{G,\operatorname{Sym} C} \underset{\operatorname{Sym} C}{\times} \mathcal{A}$$
 is a closed embedding.

Since $\operatorname{Gr}_{G,\operatorname{Sym} C}$ is ind-proper over $\operatorname{Sym} C$ (cf. [Zhu17, Remark 3.1.4]), Proposition 4.1.1 follows from Lemma 4.1.2. We also deduce from the lemma that $\overline{\mathcal{Y}}$ is representable by a scheme, and each $\overline{\mathcal{Y}}^{\check{\lambda}}$ is of finite type.

Proof. The discussion above really defines a map

$$(4.4) \qquad \operatorname{Maps}_{\mathrm{gen}}(C, X \overset{G}{\times} \overline{G/N}/T \supset \mathsf{pt}) \to \operatorname{Gr}_{G,\operatorname{Sym} C} \underset{\operatorname{Sym} C}{\times} \mathcal{A},$$

and $\overline{\mathcal{Y}} \hookrightarrow \operatorname{Maps}_{\operatorname{gen}}(C, X \times^G \overline{G/N}/T \supset \operatorname{pt})$ is a closed embedding. Thus, it suffices to prove that (4.4) is a closed embedding. Fix a test scheme S. Let $\mathcal{P}_G^0, \mathcal{P}_B^0, \mathcal{P}_T^0$ denote the respective trivial bundles on $C \times S$. An S-point of $\operatorname{Gr}_{G,\operatorname{Sym} C} \times_{\operatorname{Sym} C} \mathcal{A}$ consists of the data

$$(\mathcal{P}_G, \mathcal{P}_T, D, \tau, \alpha)$$

with $\mathcal{P}_G \in \operatorname{Bun}_G(S)$, $\mathcal{P}_T \in \operatorname{Bun}_T(S)$, $D \in \operatorname{Sym} C(S)$, a trivialization $\tau : \mathcal{P}_G^0|_{C \times S - D} \cong \mathcal{P}_G|_{C \times S - D}$, and a section $\alpha : C \times S \to (X /\!\!/ N) \times^T \mathcal{P}_T$ such that $(\mathcal{P}_T, \alpha) \in \mathcal{A}(S)$ maps to D. In particular, this means that α induces a trivialization $\mathcal{P}_T^0|_{C \times S - D} \cong \mathcal{P}_T|_{C \times S - D}$. Using the identification $X^\circ \cong B$, we get a section

$$\sigma_0:C\times S-D\to \mathcal{P}_B^0\cong X^\circ\overset{B}{\times}\mathcal{P}_B^0\hookrightarrow X\overset{B}{\times}\mathcal{P}_B^0=X\overset{G}{\times}\mathcal{P}_G^0.$$

The composition $\sigma := \tau \circ \sigma_0$ then defines a section $C \times S - D \to X \times^G \mathcal{P}_G$. On the other hand, the trivial B-bundle also corresponds to a section $\kappa_0: C \times S - D \to \mathcal{P}_G^0 \times^G (G/N)$. Composing κ_0 with the trivialization $\alpha: \mathcal{P}_T^{\bar{0}}|_{C \times S - D} \cong \mathcal{P}_T|_{C \times S - D}$, we get a section

$$\kappa: C \times S - D \to \mathcal{P}_G \overset{G}{\times} \overline{G/N} \overset{T}{\times} \mathcal{P}_T.$$

The datum $(\mathcal{P}_G, \mathcal{P}_T, \sigma, \kappa)$ defines an S-point of Maps $(C, X \times^G \overline{G/N}/T)$ if and only if σ, κ both extend to regular maps on $C \times S$. Therefore, the fiber of our chosen S-point over the map (4.4) parametrizes maps $S' \to S$ such that the base change of σ, κ to S' both extend to $C \times S'$. By Lemma 3.7.3, this fiber is represented by a closed subscheme of S (here the key point is that both X and $\overline{G/N}$ are affine).

Remark 4.1.3. We need the extra factor of \mathcal{A} in Lemma 4.1.2 which was not present in Proposition 3.7.2 because the map $Gr_B \to Gr_G$ is a bijection on k-points but far from an isomorphism of ind-schemes. Since \mathcal{A} embeds into $\operatorname{Gr}_{T,\operatorname{Sym} C}$, the lemma is really embedding \mathcal{Y} into $\operatorname{Gr}_{G,\operatorname{Sym} C} \times_{\operatorname{Sym} C} \operatorname{Gr}_{T,\operatorname{Sym} C}$. The ind-scheme Gr_T is highly non-reduced, while $(\operatorname{Gr}_T)_{\operatorname{red}}$ is a disjoint union of points.

Example 4.1.4. Let $X = \mathbb{G}_m \backslash \mathrm{GL}_2$ as in Example 3.3.1. Then $\overline{\mathcal{Y}} = \mathrm{Sym}\, C \times \mathrm{Sym}\, C$ and $\overline{\mathcal{Y}} \to \mathcal{A}$ is the identity morphism.

4.2. Stratification. We can uniquely write any $\check{\nu} \in \check{\Lambda}_G^{\text{pos}}$ as a sum $\check{\nu} = \sum_{\alpha \in \Delta_G} n_\alpha \check{\alpha}$ where Δ_G is the set of simple coroots and n_α are positive integers. Let $C_{\check{\nu}} := \prod_{\Delta_G} C^{(n_\alpha)}$ denote the corresponding partially symmetrized power of C. Recall that Drinfeld's compactification $\overline{\overline{\mathrm{Bun}}}_B$ of Bun_B has a stratification by defect, where the strata are given by locally closed embeddings

$$\mathfrak{i}_{\check{\nu}}: C_{\check{\nu}} \times \operatorname{Bun}_B^{\check{\mu}+\check{\nu}} \hookrightarrow \overline{\operatorname{Bun}}_B^{\check{\mu}},$$

for $\check{\nu} \in \check{\Lambda}_G^{\text{pos}}$, $\check{\mu} \in \check{\Lambda}_G$, cf. [BFGM02, §1.5, p. 7]. Define the substack $\check{\nu} \overline{\text{Bun}}_B^{\check{\mu}}$ to be the image of the corresponding embedding. We obtain an open substack $\leq \check{\nu} \overline{\overline{\mathrm{Bun}}}_{B}^{\check{\mu}} \subset \overline{\overline{\mathrm{Bun}}}_{B}^{\check{\mu}}$ by taking the union of the strata $_{\check{\nu}'}\overline{\mathrm{Bun}}_B^{\check{\mu}}$ for all $\check{\nu}' \leq \check{\nu}$. Since $\overline{\overline{\mathbb{y}}}^{\check{\lambda}}$ maps to $\overline{\mathrm{Bun}}_B^{-\check{\lambda}}$ for $\check{\lambda} \in \mathfrak{c}_X$, by base change we have locally closed subschemes

$${}_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda}}:=\overline{\mathcal{Y}}^{\check{\lambda}}\underset{\overline{\operatorname{Bun}}_{B}^{-\check{\lambda}}}{\times}{}_{\check{\nu}}\overline{\operatorname{Bun}}_{B}^{-\check{\lambda}}\hookrightarrow\overline{\mathcal{Y}}^{\check{\lambda}}$$

and open subschemes $\leq_{\tilde{\nu}} \overline{\mathcal{Y}}^{\tilde{\lambda}} \hookrightarrow \overline{\mathcal{Y}}^{\tilde{\lambda}}$ defined analogously. Observe that the identification ${}_{\tilde{\nu}}\overline{\mathrm{Bun}}_B^{-\hat{\lambda}} \cong$ $C_{\check{\nu}} \times \operatorname{Bun}_{R}^{\check{\nu}-\check{\lambda}}$ induces an isomorphism

$$_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda}} \cong C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda}-\check{\nu}}.$$

On the other hand, $\overline{\mathcal{Y}}^{\check{\lambda}}$ also maps to \mathcal{M}_X , so we get a locally closed subscheme

$${}_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda},\check{\Theta}}:={}_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda}}\underset{\mathcal{M}_X}{\times}\mathcal{M}_X^{\check{\Theta}}\hookrightarrow{}_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda}}\hookrightarrow\overline{\mathcal{Y}}^{\check{\lambda}}$$

for $\check{\Theta}$ any partition in \mathfrak{c}_X^- (by Lemma 3.1.6). We deduce from (4.5) that there is an isomorphism

$$_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda},\check{\Theta}}\cong C_{\check{\nu}}\times\mathcal{Y}^{\check{\lambda}-\check{\nu},\check{\Theta}}.$$

In particular, Proposition 3.6.1 implies that ${}_{\check{\nu}}\bar{\mathcal{Y}}^{\check{\lambda},\check{\Theta}}$ is smooth. In summary:

Proposition 4.2.1. The collection of locally closed subschemes ${}_{\check{\nu}}\overline{\mathbb{y}}^{\check{\lambda},\check{\Theta}}$, ranging over all $\check{\nu}\in\check{\Lambda}_G^{\mathrm{pos}}$ and partitions $\check{\Theta}\in\mathrm{Sym}^{\infty}(\mathfrak{c}_X^--0)$, forms a smooth stratification of $\overline{\mathbb{y}}^{\check{\lambda}}$.

Note that for fixed $\check{\lambda}$, many of these strata may be empty.

4.2.2. Changing the curve. In this subsection we let C be a smooth curve which is not necessarily proper and we define

$$\mathcal{A}(C) = \operatorname{Hom}(\mathfrak{c}_X^\vee, \operatorname{Sym} C), \quad \forall (C) = \operatorname{Maps}_{\operatorname{gen}}(C, X/B \supset \mathsf{pt})$$

and $\overline{\mathcal{Y}}(C)$ to be the closure of $\mathcal{Y}(C)$ in $\operatorname{Maps}_{\operatorname{gen}}(C, X \times^G \overline{G/N}/T \supset \operatorname{pt})$ to emphasize the curve being used. Here Hom denotes homomorphisms of monoid objects in the category of schemes. Similarly we have $\mathcal{A}^{\check{\lambda}}(C), \mathcal{Y}^{\check{\lambda}}(C), \overline{\mathcal{Y}}^{\check{\lambda}}(C)$. The local nature of $\overline{\mathcal{Y}}(C)$ (in particular Lemma 4.1.2) ensures that $\overline{\mathcal{Y}}^{\check{\lambda}}(C)$ is still a finite type k-scheme.

Let $p:\widetilde{C}\to C$ be an étale map of smooth curves. Let $(\operatorname{Sym}\widetilde{C})_{\operatorname{disj}}\subset\operatorname{Sym}\widetilde{C}$ be the open subset that consists of divisors \widetilde{D} on \widetilde{C} such that any fiber of p contains at most one point of the support of \widetilde{D} . Let $\mathcal{A}(\widetilde{C})_{\operatorname{disj}}$ denote the open subset of $\operatorname{Hom}(\mathfrak{c}_X^\vee,\operatorname{Sym}\widetilde{C})$ consisting of homomorphisms landing in $(\operatorname{Sym}\widetilde{C})_{\operatorname{disj}}$. Then pushforward of divisors defines a map $\mathcal{A}^{\check{\lambda}}(\widetilde{C})\to \mathcal{A}^{\check{\lambda}}(C)$.

Proposition 4.2.3 ([BFG06, Proposition 2.19]). For an étale map $\widetilde{C} \to C$ we have a canonical isomorphism

$$\overline{\mathcal{Y}}^{\check{\lambda}}(C) \underset{\mathcal{A}^{\check{\lambda}}(C)}{\times} \mathcal{A}^{\check{\lambda}}(\widetilde{C})_{\mathrm{disj}} \cong \overline{\mathcal{Y}}^{\check{\lambda}}(\widetilde{C}) \underset{\mathcal{A}^{\check{\lambda}}(\widetilde{C})}{\times} \mathcal{A}^{\check{\lambda}}(\widetilde{C})_{\mathrm{disj}}$$

which preserves the fine stratification.

Note that setting $\widetilde{C} = C \sqcup C$ recovers the graded factorization property.

Proof. Choose $\delta \in \mathfrak{c}_X^{\vee}$ as in §3.7.1. For $(y, \tilde{a}) \in \overline{\mathcal{Y}}^{\check{\lambda}}(C) \times_{\mathcal{A}^{\check{\lambda}}(C)} \mathcal{A}^{\check{\lambda}}(\widetilde{C})_{\mathrm{disj}}$, let \widetilde{D} (resp. D) denote the divisor corresponding to δ paired with \widetilde{a} (resp. $\pi(y) \in \mathcal{A}^{\check{\lambda}}(C)$). Then $p_*\widetilde{D} = D$ and we deduce that there is an isomorphism $\widehat{C}'_D \cong \widehat{\widetilde{C}}'_{\widetilde{D}}$. Lemma 3.7.7, applied to the affine $G \times T$ -scheme $X \times \overline{G/N}$, implies that the point $y \in \overline{\mathcal{Y}}^{\check{\lambda}}(C)$ is equivalent to its restriction $y|_{\widehat{C}'_D}$. Applying the same lemma again shows that $y|_{\widehat{C}'_D}$ is equivalent to a point $\widetilde{y} \in \overline{\mathcal{Y}}^{\check{\lambda}}(\widetilde{C})$ such that $\widetilde{y}(\widetilde{C} - \widetilde{D}) = \mathsf{pt}$. This defines mutually inverse maps in both directions and the compatibility with strata is clear.

Since the diagonal $\delta^{\check{\lambda}}: \widetilde{C} \hookrightarrow \mathcal{A}^{\check{\lambda}}(\widetilde{C})$ sending $\widetilde{v} \mapsto \check{\lambda} \cdot \widetilde{v}$ is contained in $\mathcal{A}^{\check{\lambda}}(\widetilde{C})_{\mathrm{disj}}$, we deduce from the graded factorization property and Proposition 4.2.3 that $\overline{\mathcal{Y}}^{\check{\lambda}}(C)$ is étale-locally isomorphic to $\overline{\mathcal{Y}}^{\check{\lambda}}(\mathbb{A}^1)$ for any smooth curve C.

4.3. The central fiber. Let $\check{\lambda} \in \mathfrak{c}_X$. There is a diagonal map $\delta^{\check{\lambda}} : C \to \mathcal{A}^{\check{\lambda}}$ sending $v \mapsto \check{\lambda} \cdot v$. For a fixed point $v \in |C|$, let us consider $\delta^{\check{\lambda}}_v : v \to \mathcal{A}^{\check{\lambda}}$.

Define the central fiber $\mathsf{Y}^{\check{\lambda}}$ of $\mathsf{Y}^{\check{\lambda}}$ to be the preimage of $\delta_v^{\check{\lambda}}$ under $\pi: \mathsf{Y}^{\check{\lambda}} \to \mathcal{A}^{\check{\lambda}}$. If we take central fibers of the map (3.7), Proposition 3.7.2 implies that we have a closed embedding

 $\mathsf{Y}^{\check{\lambda}} \hookrightarrow \mathrm{Gr}_B$. In fact, $\check{\lambda} \cdot v$ can be considered as a point in Gr_T . If we let $\mathsf{S}^{\check{\lambda}}$ denote the preimage of this point under the projection $Gr_B \to Gr_T$, then we get a closed embedding

$$(4.6) Y^{\check{\lambda}} \hookrightarrow S^{\check{\lambda}}.$$

Note that $S^{\check{\lambda}}$ identifies with the LN-orbit of $t^{\check{\lambda}}$ in Gr_G . The orbits $S^{\check{\lambda}}$ are commonly known as the semi-infinite orbits of Gr_G , and their geometric properties were extensively studied by Mirković-Vilonen in [MV07]. Let \overline{S}^{λ} denote the scheme-theoretic closure of $S^{\check{\lambda}}$ in Gr_G .

We analogously define $\overline{Y}^{\check{\lambda}}$ as the central fiber¹⁴ of $\bar{\pi}: \bar{\mathcal{Y}}^{\check{\lambda}} \to \mathcal{A}^{\check{\lambda}}$, and we deduce from Lemma 4.1.2 that there is a closed embedding $\overline{Y}^{\lambda} \hookrightarrow Gr_G$. From the moduli-theoretic description of \overline{S}^{λ} (see [Zhu17, proof of Proposition 5.3.6]), we see that this factors through a closed

$$(4.7) \overline{Y}^{\check{\lambda}} \hookrightarrow \overline{S}^{\check{\lambda}}.$$

4.3.1. Local description of $Y^{\check{\lambda}}$. Note that the central fiber $Y^{\check{\lambda}}$ intersects the stratum $\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ only if $\check{\Theta} = [\check{\theta}]$ is the singleton partition corresponding to a single $\check{\theta} \in \mathfrak{c}_X^-$. In this case, let $\mathsf{Y}^{\check{\lambda},\check{\theta}}$ denote the intersection $\mathsf{Y}^{\check{\lambda}} \cap \mathcal{Y}^{\check{\lambda},\check{\theta}} = \mathsf{Y}^{\check{\lambda}} \times_{\mathcal{M}_X} \mathcal{M}_X^{\check{\theta}}$. Also let $\overline{\mathsf{Y}}^{\check{\lambda},\check{\theta}} = \overline{\mathsf{Y}}^{\check{\lambda}} \times_{\mathcal{M}_X} \mathcal{M}_X^{\check{\theta}}$. The $\mathsf{L}G$ -action on the base point $x_0 \in \mathsf{L}X(k)$ defines a map $\mathsf{L}G \to \mathsf{L}X$, which induces a map

 $Gr_G \to LX/L^+G$.

Lemma 4.3.2. There are natural isomorphisms

$$(4.8) Y^{\check{\lambda}} \cong S^{\check{\lambda}} \underset{LX/L+G}{\times} L^{+}X/L^{+}G$$

$$(4.9) \overline{\mathsf{Y}}^{\check{\lambda}} \cong \overline{\mathsf{S}}^{\check{\lambda}} \underset{1 \times I + G}{\times} \mathsf{L}^{+} X / \mathsf{L}^{+} G$$

$$\mathit{which induce} \ (\mathsf{Y}^{\check{\lambda},\check{\theta}})_{\mathrm{red}} \cong (\mathsf{S}^{\check{\lambda}} \underset{\mathsf{L}X/\mathsf{L}^+G}{\times} \mathsf{L}^{\check{\theta}}X/\mathsf{L}^+G)_{\mathrm{red}} \ \mathit{and} \ (\overline{\mathsf{Y}}^{\check{\lambda},\check{\theta}})_{\mathrm{red}} \cong (\overline{\mathsf{S}}^{\check{\lambda}} \underset{\mathsf{L}X/\mathsf{L}^+G}{\times} \mathsf{L}^{\check{\theta}}X/\mathsf{L}^+G)_{\mathrm{red}}.$$

Proof. Consider the composition $\operatorname{Gr}_B^{\check{\lambda}} \to \operatorname{Gr}_G \to \mathsf{L}X/\mathsf{L}^+G$. It follows from the definitions that we have an embedding

$$\mathsf{Y}^{\check{\lambda}} \hookrightarrow \mathrm{Gr}_{B}^{\check{\lambda}} \underset{X/I+G}{\times} \mathsf{L}^{+}X/\mathsf{L}^{+}G$$

The map in the reverse direction is defined using Beauville–Laszlo's theorem: for a k-algebra R, an R-point of $Gr_B^{\check{\lambda}}$ consists of a B-bundle $\hat{\mathcal{P}}_B$ on $\operatorname{Spec} R[\![t]\!]$ and a section $\hat{\sigma}_0 : \operatorname{Spec} R(\!(t)\!) \to \hat{\mathcal{P}}_B$. Using the identification $B \cong X^{\circ}$, we can identify $\hat{\sigma}_0$ with a section $\hat{\sigma} : \operatorname{Spec} R((t)) \xrightarrow{\sim} X^{\circ} \times^B$ $\hat{\mathcal{P}}_B$. If the image of $(\hat{\mathcal{P}}_B, \hat{\sigma})$ in $\mathsf{L}X/\mathsf{L}^+G$ belongs to $\mathsf{L}^+X/\mathsf{L}^+G$, then $\hat{\sigma}$ extends to a section Spec $R[t] \to X \times^B \hat{\mathcal{P}}_B$. By Lemma 3.7.7, the pair $(\hat{\mathcal{P}}_B, \hat{\sigma})$ is equivalent to an R-point $y \in \mathcal{Y}$. By construction, $y \in Y^{\lambda}$. The two maps are mutually inverse, so we get (4.8).

The same argument proves (4.9), and the other identifications are by definition.

We included the subscript red above for clarity, but from now on we consider only the underlying reduced structure on all central fibers and omit the subscript since the étale site is insensitive to reduced structures.

¹⁴The open subscheme $Y^{\check{\lambda}}$ does not need to be dense in $\overline{Y}^{\check{\lambda}}$.

Example 4.3.3. Resume the setting of Example 3.3.1. Then $\mathsf{Y}^{n_1\check{\nu}_D++n_2\check{\nu}_D-}$ is empty if both n_1 and n_2 are nonzero. Otherwise it consists of a single point. If we use the embedding $\mathbb{G}_m \hookrightarrow \mathrm{GL}_2$ via $a \mapsto \left(\begin{smallmatrix} 1 \\ 1-a \end{smallmatrix} _a^0 \right)$ and fix the base point $x_0 = 1$, then $\mathsf{Y}^{\check{\nu}_{D^+}}$ corresponds to the point $\left(\begin{smallmatrix} 1 \\ 0 \end{smallmatrix} _1 \end{smallmatrix} \right) t^{\check{\varepsilon}_1} \in \mathrm{Gr}_B$ while $\mathsf{Y}^{\check{\nu}_{D^-}}$ corresponds to $t^{-\check{\varepsilon}_2} \in \mathrm{Gr}_B$.

- 4.3.4. Lemma 4.3.2 implies that the central fibers $\mathsf{Y}^{\check{\lambda}}, \overline{\mathsf{Y}}^{\check{\lambda}}$ can be defined purely locally (in particular, independently of the point $v \in |C|$). We also deduce that $\mathcal{Y}^{\check{\lambda}} \times_{\mathcal{A}^{\check{\lambda}}, \delta^{\check{\lambda}}} C \cong \mathsf{Y}^{\check{\lambda}} \tilde{\times} C$ and $\overline{\mathcal{Y}}^{\check{\lambda}} \times_{\mathcal{A}^{\check{\lambda}}, \delta^{\check{\lambda}}} C \cong \overline{\mathsf{Y}}^{\check{\lambda}} \tilde{\times} C$, where $?\tilde{\times} C := ? \times^{\operatorname{Aut} k[\![t]\!]} C^{\wedge}$ and $C^{\wedge} \to C$ is the Aut $k[\![t]\!]$ -torsor classifying $v \in C$ together with an isomorphism $\mathfrak{o}_v \cong k[\![t]\!]$.
- 4.4. **Dimension of central fibers.** We will now discuss an argument that is critical when estimating dimensions of Zastava spaces and their central fibers. This argument appeared in the proof of [MV07, Theorem 3.2]. The second author thanks M. Finkelberg for explaining this proof to him.

Semi-infinite orbits have a very simple geometric structure, summarized by the following:

Proposition 4.4.1 ([MV07, Proposition 3.1], [Zhu17, Proposition 5.3.6, Corollary 5.3.8]).

- (i) We have a stratification $\overline{S}^{\check{\lambda}} = \bigcup_{\check{\lambda}' < \check{\lambda}} S^{\check{\lambda}'}$.
- (ii) Inside \overline{S}^{λ} , the boundary of $S^{\check{\lambda}}$ is given by a hyperplane section under an embedding of Gr_G in projective space.

In particular, if a projective subvariety meets the semi-infinite orbit $S^{\check{\lambda}}$, it also meets its boundary $\bigcup_{\lambda'<\check{\lambda}} S^{\check{\lambda}'}$. We will use this simple fact several times, in order to estimate the dimensions of central fibers.

For $\check{\lambda} \in \mathfrak{c}_X$, consider the central fiber $\overline{Y}^{\check{\lambda}} \subset \overline{S}^{\check{\lambda}} \subset \operatorname{Gr}_G$. Observe that the defect stratification of $\overline{Y}^{\check{\lambda}}$ (§4.2) gives a stratification of $\overline{Y}^{\check{\lambda}} = \bigcup_{\check{\lambda}' \leq \check{\lambda}} Y^{\check{\lambda}'}$, where $Y^{\check{\lambda}'} = \overline{Y}^{\check{\lambda}} \cap ({}_{\check{\lambda} - \check{\lambda}'} \overline{\mathcal{Y}}^{\check{\lambda}})$. This is compatible with the stratification of $\overline{S}^{\check{\lambda}} = \bigcup_{\check{\lambda}' \leq \check{\lambda}} S^{\check{\lambda}'}$ under the closed embedding $\overline{Y}^{\check{\lambda}} \subset \overline{S}^{\check{\lambda}}$. From Lemma 4.3.2 we have $\overline{Y}^{\check{\lambda}} \cap S^{\check{\lambda}'} = Y^{\check{\lambda}'}$ for $\check{\lambda}' < \check{\lambda}$.

Proposition 4.4.2. Given an irreducible component $\overline{Y} \subset \overline{S}^{\check{\lambda}}$ of the central fiber $\overline{Y}^{\check{\lambda}}$ of $\overline{y}^{\check{\lambda}}$, there is a $\check{\lambda}' \leq \check{\lambda}$ with $\overline{Y} \cap S^{\check{\lambda}'}$ nonempty of dimension zero, and $d := \dim \overline{Y} \leq \langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle$.

Moreover, if $d = \langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle$, then there exists a sequence of simple roots $\alpha_1, \ldots, \alpha_d$ (possibly with repetitions) and subvarieties $Y_j \subset Y^{\check{\lambda} - \check{\alpha}_1 - \cdots - \check{\alpha}_j}$ for $j = 0, \ldots, d$ such that

- $\bullet \ \overline{Y_0} = \overline{Y}$
- Y_j is an irreducible component of $\overline{Y_{j-1}} \cap S^{\check{\lambda}-\check{\alpha}_1-\cdots-\check{\alpha}_j}$ of dimension d-j for $j=1,\ldots,d$.

Proof. By the stratification $\overline{S}^{\check{\lambda}} = \bigcup_{\check{\lambda}' \leq \check{\lambda}} S^{\check{\lambda}}$, there must exist $\check{\lambda}_0 \leq \check{\lambda}$ such that $Y_0 := \overline{Y} \cap S^{\check{\lambda}_0}$ is dense in \overline{Y} . Then $\overline{Y} \subset \overline{S}^{\check{\lambda}_0}$ so we may assume $\check{\lambda} = \check{\lambda}_0$. Write $\partial \overline{S}^{\check{\lambda}} = \overline{S}^{\check{\lambda}} - S^{\check{\lambda}}$ for the hyperplane of Proposition 4.4.1(ii). Since \overline{Y} is a projective subvariety of $\overline{S}^{\check{\lambda}}$, it must meet $\partial \overline{S}^{\check{\lambda}}$. Hence there exists $\check{\lambda}_1 < \check{\lambda}$ such that $\dim(\overline{Y} \cap S^{\check{\lambda}_1}) = d - 1$. Let Y_1 be any irreducible component of $\overline{Y} \cap S^{\check{\lambda}_1}$ and \overline{Y}_1 its closure in $\overline{S}^{\check{\lambda}_1}$. Continuing in this fashion we produce a sequence of coweights $\check{\lambda}_0 = \check{\lambda}, \check{\lambda}_1, \ldots, \check{\lambda}_d$ and irreducible components $Y_j \subset \overline{Y}_{j-1} \cap S^{\check{\lambda}_j}$ for $j = 1, \ldots, d$, such that $\check{\lambda}_j < \check{\lambda}_{j-1}$ and $\dim Y_j = d - j$. Since $\check{\lambda}_j < \check{\lambda}_{j-1}$ implies $\langle \rho_G, \check{\lambda}_{j-1} - \check{\lambda}_j \rangle \geq 1$, we get $\langle \rho_G, \check{\lambda} - \check{\lambda}_d \rangle \geq d$ and $\check{\lambda}' = \check{\lambda}_d$ is the claimed coweight in the proposition statement.

In order for the last inequality to be an equality, we must have $\check{\lambda} = \check{\lambda}_0$ and $\check{\lambda}_j = \check{\lambda}_{j-1} - \check{\alpha}_j$ for some simple coroot $\check{\alpha}_j$, $j = 1, \ldots, d$.

4.5. Comparison of IC complexes. In order to describe the restriction of $IC_{\overline{y}}$ to strata we will need to introduce several (graded) factorization algebras on the collection of $C_{\check{\nu}}$, $\check{\nu} \in \check{\Lambda}_G^{\mathrm{pos}}$. We only state their definitions; we refer the reader to [BG08, Gai] for further context.

Let $\check{\mathfrak{n}}_C = \check{\mathfrak{n}} \otimes_{\overline{\mathbb{Q}}_{\ell}} \overline{\mathbb{Q}}_{\ell C}$ be the constant sheaf of Lie algebras over C. Let $\mathfrak{U}(\check{\mathfrak{n}}_C)^{\check{\nu}}$ denote the following sheaf on $C_{\check{\nu}}$ for $\check{\nu} \in \check{\Lambda}_G^{\mathrm{pos}}$: its *-stalk at a point $\sum \check{\nu}_i \cdot v_i$ with $v_i \in |C|$ distinct is the tensor product $\bigotimes_i U(\check{\mathfrak{n}})^{\check{\nu}_i}$ where the superscript $\check{\nu}_i$ refers to the corresponding weight space of the universal enveloping algebra $U(\check{\mathfrak{n}})$. These stalks glue to a sheaf by means of the co-multiplication map on $U(\check{\mathfrak{n}})$. Let $\mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)^{-\check{\nu}} = \mathbb{D}(\mathfrak{U}(\check{\mathfrak{n}}_C)^{\check{\nu}}) \in \mathcal{D}_c^b(C_{\check{\nu}})$ denote the Verdier dual.

We will also need the Chevalley–Cousin complex $\Upsilon(\check{\mathfrak{n}}_C)$ of $\check{\mathfrak{n}}_C$. Consider the (homological) Chevalley complex $C_{\bullet}(\check{\mathfrak{n}}_C)$ as a sheaf of co-commutative DG co-algebras on C, endowed with a grading by elements of $\check{\Lambda}_G^{\mathrm{pos}}$. By the general procedure of [BD04, §3.4], a sheaf of $\check{\Lambda}_G^{\mathrm{pos}}$ -graded co-commutative DG co-algebras on C is equivalent to a commutative factorization algebra on $\bigcup C_{\check{\nu}}$. We let $\Upsilon(\check{\mathfrak{n}}_C)^{\check{\nu}}$ denote the corresponding complex on $C_{\check{\nu}}$, which can be explicitly constructed via a Cousin complex. Its *-stalk at $\sum \check{\nu}_i \cdot v_i$ with v_i 's disinct is the tensor product $\bigotimes_i C_{\bullet}(\check{\mathfrak{n}})^{\check{\nu}_i}$. We remark that $\Upsilon(\check{\mathfrak{n}}_C)^{\check{\nu}}$ is actually a perverse sheaf, and $\Upsilon(\check{\mathfrak{n}}_C)$ and $\mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)$ are related by a certain Koszul duality.

4.5.1. Let

$$\mathfrak{i}_{\overline{\mathcal{Y}},\check{\nu}}:C_{\check{\nu}}\times\mathcal{Y}^{\check{\lambda}-\check{\nu}}\hookrightarrow\overline{\mathcal{Y}}^{\check{\lambda}}$$

denote the locally closed embedding corresponding to (4.5).

Proposition 4.5.2. For any $\check{\lambda} \in \mathfrak{c}_X$, $\check{\nu} \in \check{\Lambda}_G^{\mathrm{pos}}$, there is an equality

$$[\mathfrak{i}_{\overline{\mathbb{y}},\check{\nu}}^*(\mathrm{IC}_{\overline{\mathbb{y}}\check{\lambda}})] = [\mathfrak{U}^\vee(\check{\mathfrak{n}}_C)^{-\check{\nu}} \boxtimes \mathrm{IC}_{\mathbb{y}\check{\lambda}-\check{\nu}}]$$

in the Grothendieck group of perverse sheaves on $C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda} - \check{\nu}}$.

We will prove the proposition in the course of this subsection. It is based on the following result of [BG08, Proposition 4.4], [BFGM02, Theorem 1.12].

Theorem 4.5.3. There exists a canonical isomorphism

$$\mathfrak{i}_{\check{\nu}}^*(\mathrm{IC}_{\overline{\operatorname{Bun}}_B^{\check{\mu}}}) \cong \mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)^{-\check{\nu}} \boxtimes \mathrm{IC}_{\operatorname{Bun}_B^{\check{\mu}+\check{\nu}}}$$

for any $\check{\nu} \in \check{\Lambda}_G^{\text{pos}}, \, \check{\mu} \in \check{\Lambda}_G$.

Given a map $f: Y \to S$ between finite type algebraic k-stacks, we use the notion of a complex on Y which is universally locally acyclic (ULA) with respect to f, as in [Del77, Définition 2.12]. The general lemma we will use is the following:

Lemma 4.5.4 ([BG02, Lemma 7.1.3]). Consider a Cartesian diagram of finite type algebraic stacks

$$\begin{array}{ccc} Y' & \stackrel{f'}{\longrightarrow} & S' \\ \downarrow g' & & & \downarrow g \\ Y & \stackrel{f}{\longrightarrow} & S \end{array}$$

where S is smooth. Let $j: Y_0 \hookrightarrow Y$ be an open dense substack such that the map $f \circ j: Y_0 \to S$ is smooth. In addition, assume that the complexes IC_Y and $j_!(IC_{Y_0})$ are ULA with respect to the map f.

Denote the closure¹⁵ of $Y_0 \times_S S'$ in Y' by $\overline{Y_0 \times_S S'}$. Then there is a natural isomorphism

$$\operatorname{IC}_{\overline{Y_0}\times_S S'} \cong \operatorname{IC}_{S'} \underset{S}{\boxtimes} \operatorname{IC}_Y := f'^*(\operatorname{IC}_{S'}) \otimes g'^*(\operatorname{IC}_Y)[-\dim S],$$

¹⁵In general it is possible for Y' to have more irreducible components than $\overline{Y_0 \times_S S'}$.

where the left hand side is implicitly extended by zero to Y'.

For a proof of Lemma 4.5.4, see [Wan].

We would like to apply this lemma with $Y = \overline{\operatorname{Bun}}_B$, $Y_0 = \operatorname{Bun}_B$, $S = \operatorname{Bun}_G$, and $S' = \mathcal{M}_X$. Unfortunately the stack $\overline{\operatorname{Bun}}_B$ is "too big" for all of $\operatorname{IC}_{\overline{\operatorname{Bun}}_B}$ to be ULA over Bun_G . However, we do get the ULA property if we restrict to open substacks where the defect is not "too big": Let $j: \operatorname{Bun}_B \hookrightarrow \overline{\operatorname{Bun}}_B$ denote the open embedding.

Proposition 4.5.5 ([Cam19, Cam18]). Fix $\check{\nu} \in \check{\Lambda}_G^{\text{pos}}$. Then for any $\check{\mu} \in \check{\Lambda}_G$ large enough:

- (i) The complexes $IC_{\preceq_{\tilde{\nu}}\overline{Bun}_{\mathcal{P}}^{-\tilde{\mu}}}$ and $\mathfrak{j}_!IC_{Bun_{\mathcal{B}}}|_{\preceq_{\tilde{\nu}}\overline{Bun}_{\mathcal{P}}^{-\tilde{\mu}}}$ are ULA over $Bun_{\mathcal{G}}$.
- (ii) The fiber of $\operatorname{Bun}_B^{-\check{\mu}} \to \operatorname{Bun}_G$ is dense in the fiber of $\leq_{\check{\nu}} \overline{\operatorname{Bun}}_B^{-\check{\mu}} \to \operatorname{Bun}_G$ over any k-point of Bun_G .

Here "large enough" is in the same sense as in Lemma 3.5.1, i.e., deep enough in the dominant chamber $\check{\Lambda}_G^+$.

Proof. The ULA property for IC $_{\leq_{\bar{\nu}}\overline{\mathrm{Bun}_{B}^{-\bar{\mu}}}}$ is [Cam19, Corollary 4.1.1.1]. The ULA property for $\mathfrak{j}_{!}\mathrm{IC}_{\mathrm{Bun}_{B}}|_{\leq_{\bar{\nu}}\overline{\mathrm{Bun}_{B}^{-\bar{\mu}}}}$ can be deduced from the former, cf. [Cam18, §4.3]. We review the salient features of the proof in [Cam19] to deduce (ii).

The key object introduced in [Cam19] is Kontsevich's compactification $\overline{\operatorname{Bun}}_B^K$ of Bun_B , which is a resolution of singularities $\overline{\operatorname{Bun}}_B^K \to \overline{\operatorname{Bun}}_B$ of Drinfeld's compactification. For an affine test scheme S, an S-point of $\overline{\operatorname{Bun}}_B^K$ is a commutative square

$$\begin{array}{ccc} \widetilde{\mathbb{C}} & \longrightarrow & \mathsf{pt}/B \\ \downarrow & & \downarrow \\ C \times S & \stackrel{\mathcal{P}_G}{\longrightarrow} & \mathsf{pt}/G \end{array}$$

where $\widetilde{\mathbb{C}} \to S$ is a flat family of connected nodal projective curves of the same arithmetic genus as C, the map $\widetilde{\mathbb{C}} \to C \times S$ has degree 1, and the induced section $\widetilde{\mathbb{C}} \to \mathcal{P}_G \times^G G/B$ is stable in the sense of [Kon95] over every geometric point of S. Let \mathscr{M}_C denote the moduli stack of proper connected nodal curves $\widetilde{\mathbb{C}}$ equipped with a degree one map to C. Then the natural map $\overline{\mathrm{Bun}}_B^K \to \mathscr{M}_C$ is smooth, and \mathscr{M}_C is smooth ([Cam19, Proposition 2.4.1]) and there is a proper map $\overline{\mathrm{Bun}}_B^K \to \overline{\mathrm{Bun}}_B$ over Bun_G ([Cam19, Proposition 3.2.2]). Inside \mathscr{M}_C we have the open point corresponding to id : $C \to C$ and the complement $\partial \mathscr{M}_C$ is a normal crossings divisor ([Cam19, Proposition 4.4.1]). Evidently the preimage of the open point in $\overline{\mathrm{Bun}}_B^K$ identifies with Bun_B . Let $\leq_{\check{\nu}} \overline{\mathrm{Bun}}_B^{K,-\check{\mu}}$ denote the preimage of $\leq_{\check{\nu}} \overline{\mathrm{Bun}}_B^{-\check{\mu}}$. Then [Cam19, Proposition 4.1.1] shows that for $\check{\nu}$ fixed and $\check{\mu}$ large enough, the map $\leq_{\check{\nu}} \overline{\mathrm{Bun}}_B^{K,-\check{\mu}} \to \mathscr{M}_C \times \mathrm{Bun}_G$ is smooth. Now for any $\mathcal{P}_G \in \mathrm{Bun}_G(k)$, the fiber of $\widetilde{\mathfrak{p}}^K : \leq_{\check{\nu}} \overline{\mathrm{Bun}}_B^{K,-\check{\mu}} \to \mathrm{Bun}_G$ is smooth over \mathscr{M}_C . The preimage over the open point equals $(\widetilde{\mathfrak{p}}^K)^{-1}(\mathcal{P}_G) \cap \mathrm{Bun}_B$ and since $\partial \mathscr{M}_C$ is a normal crossings divisor, $(\widetilde{\mathfrak{p}}^K)^{-1}(\mathcal{P}_G) \cap \mathrm{Bun}_B$ must be dense in $(\widetilde{\mathfrak{p}}^K)^{-1}(\mathcal{P}_G)$. Since $(\widetilde{\mathfrak{p}}^K)^{-1}(\mathcal{P}_G)$ is a resolution of the fiber of $\leq_{\check{\nu}} \overline{\mathrm{Bun}}_B^{\check{\nu}} \to \mathrm{Bun}_G$, we have proved (ii).

Corollary 4.5.6. For any $\check{\lambda} \in \mathfrak{c}_X$, the subscheme $y^{\check{\lambda}}$ is dense in $\overline{y}^{\check{\lambda}}$.

Proof. The subschemes $\leq_{\check{\eta}} \overline{\mathcal{Y}}^{\check{\lambda}} = \overline{\mathcal{Y}}^{\check{\lambda}} \times_{\overline{\operatorname{Bun}}_B} \leq_{\check{\eta}} \overline{\overline{\operatorname{Bun}}_B}$ for $\check{\eta} \in \check{\Lambda}_G^{\operatorname{pos}}$ form an open covering of $\overline{\mathcal{Y}}^{\check{\lambda}}$. Since $\overline{\mathcal{Y}}^{\check{\lambda}}$ is quasicompact, there must exist some $\check{\eta}$ such that $<_{\check{\eta}} \overline{\mathcal{Y}}^{\check{\lambda}} = \overline{\mathcal{Y}}^{\check{\lambda}}$. Fix $\check{\lambda}, \check{\eta}$ as above.

Then we can choose $\check{\mu} \in \check{\Lambda}_G^{\mathrm{pos}}$ such that $\check{\lambda} + \check{\mu} - \check{\nu}$ is large enough, for all $0 \leq \check{\nu} \leq \check{\eta}$, for the purposes of Proposition 4.5.5 and Lemma 3.5.1. Now we may consider $\leq_{\check{\eta}} \overline{\mathcal{Y}}^{\check{\lambda} + \check{\mu}}$ as an open subscheme of $Y' := \mathcal{M}_X \underset{\mathrm{Bun}_G}{\times} \leq_{\check{\eta}} \overline{\mathrm{Bun}}_B^{-\check{\lambda} - \check{\mu}}$. Proposition 4.5.5(ii) implies that $\mathcal{M}_X \underset{\mathrm{Bun}_G}{\times} \mathrm{Bun}_B^{-\check{\lambda} - \check{\mu}}$

is dense in Y', which implies that $\mathcal{Y}^{\check{\lambda}+\check{\mu}}$ is dense in $\overline{\mathcal{Y}}^{\check{\lambda}+\check{\mu}}$. The graded factorization property of $\overline{\mathcal{Y}}$ gives a natural étale map

$$(4.10) \overline{\mathcal{Y}}^{\check{\lambda}} \mathring{\times} \mathcal{Y}^{\check{\mu}} := (\overline{\mathcal{Y}}^{\check{\lambda}} \times \mathcal{Y}^{\check{\mu}})|_{A^{\check{\lambda}} \mathring{\times} A^{\check{\mu}}} \to \overline{\mathcal{Y}}^{\check{\lambda} + \check{\mu}},$$

where $\mathcal{Y}^{\check{\mu}} \hookrightarrow \overline{\mathcal{Y}}^{\check{\mu}}$ is the open embedding. We deduce that $\mathcal{Y}^{\check{\lambda}} \mathring{\times} \mathcal{Y}^{\check{\mu}}$ is dense in $\overline{\mathcal{Y}}^{\check{\lambda}} \mathring{\times} \mathcal{Y}^{\check{\mu}}$, which implies $\mathcal{Y}^{\check{\lambda}}$ is dense in $\overline{\mathcal{Y}}^{\check{\lambda}}$.

Proof of Proposition 4.5.2. Applying Lemma 4.5.4 to the Cartesian square

$$\begin{array}{ccc} Y' & \longrightarrow & \mathcal{M}_X \\ \downarrow & & \downarrow \\ & & \downarrow \\ \leq \check{\eta} \overline{\operatorname{Bun}}_B^{-\check{\lambda} - \check{\mu}} & \longrightarrow & \operatorname{Bun}_G \end{array}$$

we can identify

(4.11)
$$IC_{Y'} \cong IC_{\mathcal{M}_X} \underset{\operatorname{Bun}_G}{\boxtimes} IC_{\leq \bar{\eta} \overline{\operatorname{Bun}}_B^{-\bar{\lambda} - \bar{\mu}}}.$$

In particular, this gives us a description of ${\rm IC}_{\leq \check{\eta}}_{\overline{\mathcal{Y}}^{\check{\lambda}+\check{\mu}}}={\rm IC}_{Y'}|_{\leq \check{\eta}}_{\overline{\mathcal{Y}}^{\check{\lambda}+\check{\mu}}}$. For $0\leq \check{\nu}\leq \check{\eta}$, we have a Cartesian square

$$C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda} + \check{\mu} - \check{\nu}} \xrightarrow{\mathbf{i}_{\overline{\mathcal{Y}}, \check{\nu}}} \overline{\mathcal{Y}}^{\check{\lambda} + \check{\mu}}$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$C_{\check{\nu}} \times \operatorname{Bun}_{B}^{-\check{\lambda} - \check{\mu} + \check{\nu}} \xrightarrow{\mathbf{i}_{\check{\nu}}} \overline{\operatorname{Bun}_{B}}^{-\check{\lambda} - \check{\mu}}$$

Theorem 4.5.3, together with the Cartesian square above and (4.11), allow us to deduce that there exists an isomorphism

$$\mathfrak{i}_{\overline{\mathbb{y}},\check{\nu}}^*(\mathrm{IC}_{\overline{\mathbb{y}}^{\check{\lambda}+\check{\mu}}}) \cong \mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)^{-\check{\nu}} \boxtimes (\mathrm{IC}_{\mathfrak{M}_X} \underset{\mathrm{Bun}_G}{\boxtimes} \mathrm{IC}_{\mathrm{Bun}_B^{-\check{\lambda}-\check{\mu}+\check{\nu}}})|_{\mathbb{y}^{\check{\lambda}+\check{\mu}-\check{\nu}}} = \mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)^{-\check{\nu}} \boxtimes (\mathrm{IC}_{\mathfrak{M}_X}|_{\mathbb{y}^{\check{\lambda}+\check{\mu}-\check{\nu}}}^{!*})$$

on $C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda}+\check{\mu}-\check{\nu}}$. In the last equality we have used the fact that Bun_B is smooth. Since we chose $\check{\lambda}+\check{\mu}-\check{\nu}$ large enough to satisfy Lemma 3.5.1, the map $\mathcal{Y}^{\check{\lambda}+\check{\mu}-\check{\nu}}\to \mathcal{M}_X$ is smooth. Therefore, $\operatorname{IC}_{\mathcal{M}_X}|_{\mathcal{Y}^{\check{\lambda}+\check{\mu}-\check{\nu}}}^{!*}\cong \operatorname{IC}_{\mathcal{Y}^{\check{\lambda}+\check{\mu}-\check{\nu}}}$ and we get a canonical isomorphism

$$(4.12) i_{\overline{y},\check{\nu}}^*(\mathrm{IC}_{\overline{y}^{\check{\lambda}+\check{\mu}}}) \cong \mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)^{\check{\nu}} \boxtimes \mathrm{IC}_{y^{\check{\lambda}+\check{\mu}-\check{\nu}}}.$$

Observe that the following diagram is Cartesian:

$$(C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda} - \check{\nu}}) \overset{\circ}{\times} \mathcal{Y}^{\check{\mu}} \overset{\iota^{\overset{\bullet}{\overline{y}}, \check{\nu}} \times \operatorname{id}}{\longrightarrow} \overline{\mathcal{Y}}^{\check{\lambda}} \overset{\circ}{\times} \mathcal{Y}^{\check{\mu}}$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \qquad \downarrow \qquad \qquad \qquad \downarrow \qquad \qquad \qquad$$

where the left vertical arrow is identity on $C_{\check{\nu}}$ times the map $\mathcal{Y}^{\check{\lambda}-\check{\nu}}\mathring{\times}\mathcal{Y}^{\check{\mu}} \to \mathcal{Y}^{\check{\lambda}+\check{\mu}-\check{\nu}}$ coming from graded factorization. Since the restriction of $\mathrm{IC}_{\overline{\mathcal{Y}}^{\check{\lambda}+\check{\mu}}}$ to $\overline{\mathcal{Y}}^{\check{\lambda}}\mathring{\times}\mathcal{Y}^{\check{\mu}}$ is $\mathrm{IC}_{\overline{\mathcal{Y}}^{\check{\lambda}}}\mathring{\boxtimes}\mathrm{IC}_{\mathcal{Y}^{\check{\mu}}}$, we deduce from the Cartesian square and (4.12) that there is a canonical isomorphism

$$\mathfrak{i}_{\overline{\mathbb{y}},\check{\nu}}^*(\mathrm{IC}_{\overline{\mathbb{y}}\check{\lambda}}) \overset{\circ}{\boxtimes} \mathrm{IC}_{\mathbb{y}\check{\mu}} \cong (\mathfrak{U}^{\vee}(\check{\mathfrak{n}}_{\mathit{C}})^{-\check{\nu}} \boxtimes \mathrm{IC}_{\mathbb{y}\check{\lambda}-\check{\nu}}) \overset{\circ}{\boxtimes} \mathrm{IC}_{\mathbb{y}\check{\mu}}$$

when restricted to $(C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda}-\check{\nu}})\mathring{\times} \mathcal{Y}^{\check{\mu}}$. Lastly, let \mathscr{Y} be a connected component of $\mathcal{Y}^{\check{\mu},0}$ so that the projection $p:(C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda}-\check{\nu}})\mathring{\times} \mathscr{Y} \to C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda}-\check{\nu}}$ is smooth surjective with irreducible fibers. We have constructed an isomorphism

$$p^*(\mathfrak{i}_{\overline{\mathsf{U}}_{\check{\mathsf{U}}}}^*(\mathrm{IC}_{\overline{\mathsf{U}}^{\check{\lambda}}})) \cong p^*(\mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)^{-\check{\nu}} \boxtimes \mathrm{IC}_{\mathsf{U}^{\check{\lambda}-\check{\nu}}}),$$

which implies Proposition 4.5.2 because the functor $p^*[\dim \mathscr{Y}]$ is fully faithful on the category of perverse sheaves ([BBDG18, Proposition 4.2.5]).

4.5.7. Convolution product. The toric variety $X/\!\!/N$ has the natural structure of a commutative algebraic monoid. The multiplication operator on $X/\!\!/N$ induces a finite map

$$m_{\mathcal{A}}: \mathcal{A}^{\check{\lambda}_1} \times \mathcal{A}^{\check{\lambda}_2} \to \mathcal{A}^{\check{\lambda}_1 + \check{\lambda}_2}.$$

If we have sheaves $\mathcal{F}_i \in \mathcal{D}_c^b(\mathcal{A}^{\check{\lambda}_i}), i = 1, 2$, we define their convolution by

$$\mathcal{F}_1 \star \mathcal{F}_2 := m_{\mathcal{A},!}(\mathcal{F}_1 \boxtimes \mathcal{F}_2) \in \mathcal{D}_c^b(\mathcal{A}^{\check{\lambda}_1 + \check{\lambda}_2}).$$

4.5.8. We have a closed embedding $\mathfrak{i}_{\mathcal{A},\check{\nu}}:C_{\check{\nu}}\hookrightarrow\mathcal{A}^{\check{\nu}}$ corresponding to the partition $\sum_i n_i[\check{\alpha}_i]$ of degree $\check{\nu}=\sum_i n_i\check{\alpha}_i$.

Corollary 4.5.9. There is an equality

$$[\pi_!(\mathrm{IC}_{y^{\check{\lambda}}})] = \sum_{\check{\nu} \in \check{\Lambda}^\mathrm{pos}_{\mathcal{I}}} [i_{\mathcal{A},\check{\nu},!}(\Upsilon(\check{\mathfrak{n}}_C)^{\check{\nu}}) \star \bar{\pi}_!(\mathrm{IC}_{\overline{y}^{\check{\lambda}-\check{\nu}}})]$$

in the Grothendieck group of perverse sheaves on $\mathcal{A}^{\check{\lambda}}$.

Proof. Taking the Grothendieck–Cousin complex associated to the stratifications $i_{\overline{y},\check{\nu}}$ and applying Proposition 4.5.2 gives an equality

$$[\operatorname{IC}_{\overline{\mathcal{Y}}^{\check{\lambda}}}] = \sum_{\check{\nu} \in \check{\Lambda}^{\operatorname{pos}}^{\operatorname{pos}}} [\mathfrak{i}_{\overline{\mathcal{Y}},\check{\nu},!} (\mathfrak{U}^{\vee} (\check{\mathfrak{n}}_{\mathit{C}})^{\check{\nu}} \boxtimes \operatorname{IC}_{\mathcal{Y}^{\check{\lambda}-\check{\nu}}})]$$

in the Grothendieck group of perverse sheaves on $\overline{\mathcal{Y}}^{\check{\lambda}}$. Note that $\mathcal{Y}^{\check{\lambda}-\check{\nu}}$ is nonempty for finitely many values of $\check{\nu}$. The composition $\bar{\pi}\circ i_{\overline{\mathcal{Y}},\check{\nu}}:C_{\check{\nu}}\times\mathcal{Y}^{\check{\lambda}-\check{\nu}}\to\mathcal{A}^{\check{\lambda}}$ coincides with the composition $C_{\check{\nu}}\times\mathcal{Y}^{\check{\lambda}-\check{\nu}}\stackrel{i_{\mathcal{A},\check{\nu}}}{\longrightarrow}\mathcal{A}^{\check{\nu}}\times\mathcal{A}^{\check{\lambda}-\check{\nu}}\stackrel{m_{\mathcal{A}}}{\longrightarrow}\mathcal{A}^{\check{\lambda}}$. Therefore, applying $\bar{\pi}_!$ to the equality above, we get the equality

$$[\bar{\pi}_!(\mathrm{IC}_{\overline{\mathcal{Y}}^{\check{\lambda}}})] = \sum_{\check{\nu} \in \check{\Lambda}_C^{\mathrm{pos}}} [\mathfrak{i}_{\mathcal{A},\check{\nu},!}(\mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)^{\check{\nu}}) \star \pi_!(\mathrm{IC}_{\mathcal{Y}^{\check{\lambda}-\check{\nu}}})]$$

in the Grothendieck group of perverse sheaves on $\mathcal{A}^{\check{\lambda}}$. Applying a further convolution by any $\mathfrak{T} \in \mathcal{D}^b_c(C_{\check{\nu}'}), \check{\nu}' \in \check{\Lambda}^{\mathrm{pos}}_G$ gives

$$[\mathfrak{i}_{\mathcal{A},\check{\nu}',!}(\mathfrak{I})\star\bar{\pi}_!(\mathrm{IC}_{\overline{\mathfrak{Y}}^{\check{\lambda}}})]=\sum_{\check{\nu}}[\mathfrak{i}_{\mathcal{A},\check{\nu}+\check{\nu}',!}(\mathfrak{I}\star\mathfrak{U}^\vee(\check{\mathfrak{n}}_C)^{\check{\nu}})\star\pi_!(\mathrm{IC}_{\mathfrak{Y}^{\check{\lambda}-\check{\nu}}})].$$

It is known ([BG08, §6.4]) that for a fixed nonzero $\check{\nu} \in \check{\Lambda}_G^{\mathrm{pos}}$, we have an equality

$$\sum_{\substack{\check{\nu}_1,\check{\nu}_2 \in \check{\Lambda}_G^{\mathrm{pos}} \\ \check{\nu}_1 + \check{\nu}_2 = \check{\nu}}} [\Upsilon(\check{\mathfrak{n}}_C)^{\check{\nu}_1} \star \mathfrak{U}^{\vee}(\check{\mathfrak{n}}_C)^{-\check{\nu}_2}] = 0$$

in the Grothendieck group of perverse sheaves on C_{ν} . The two preceding equalities and induction prove the claim.

5. Global Hecke action and closure relations

For the rest of this paper, assume that $\check{G}_X = \check{G}$ and all simple roots of G are spherical roots of type T. Equivalently, we are assuming that B acts simply transitively on X° and for every simple root α of G, the PGL₂-variety $X^{\circ}P_{\alpha}/\Re(P_{\alpha})$ is isomorphic to $\mathbb{G}_m\backslash\mathrm{PGL}_2$ (over the algebraically closed field k).

As a consequence, $\mathcal{V} \cap \check{\Lambda}_X = \check{\Lambda}_G^-$, the monoid of antidominant coweights of G. Recall from §2.1.1 that the type T assumption also implies that for every simple root α , the open P_{α} -orbit $X^{\circ}P_{\alpha}$ is the union of X° and the open B-orbits of two colors $\mathcal{D}(\alpha) = \{D_{\alpha}^+, D_{\alpha}^-\}$. We will let $\check{\nu}_{\alpha}^{\pm}$ denote the valuation of D_{α}^{\pm} , respectively. Then $\check{\nu}_{\alpha}^+ + \check{\nu}_{\alpha}^- = \check{\alpha}$ and $\langle \alpha, \check{\nu}_{\alpha}^{\pm} \rangle = 1$. We encourage the reader to refer to Examples 3.3.1, 3.3.2, 4.1.4 and 4.3.3.

5.1. Main results of this section. This section is quite technical, and we advise the reader to read the main results listed here, and skip the rest of the section at first reading. Before we introduce the results, let us observe that, so far, we have uniformly treated all affine spherical varieties. However, the classification of spherical varieties is divided into two parts: (i) the classification of homogeneous spherical varieties $H\backslash G$, and (ii) the classification of spherical embeddings $H\backslash G \hookrightarrow X$ (by a spherical embedding we mean a G-equivariant open, dense embedding $H\backslash G \hookrightarrow X$, where X is a normal, and hence spherical, G-variety).

Since X is affine, $X^{\bullet} = H \setminus G$ is quasiaffine and there is a canonical affine embedding

$$X^{\operatorname{can}} := \operatorname{Spec} k[H \backslash G].$$

(The fact that the coordinate ring of $k[H\backslash G]$ is finitely generated follows from the fact that B-eigenspaces are one-dimensional, and the B-character group is finitely generated.) By normality, X^{can} has no divisors that do not meet $H\backslash G$, i.e., $\mathcal{D}(X)$ has no G-stable divisors, and coincides with the set \mathcal{D} of colors. Therefore, the cone $\mathcal{C}_0(H\backslash G) := \mathcal{C}_0(X^{\operatorname{can}})$ is generated by the valuations $\rho_X(\mathcal{D})$ of colors.

For any other affine spherical embedding $H \setminus G \hookrightarrow X$, there is a natural map $X^{\operatorname{can}} \to X$, so X^{can} is universal among affine embeddings of $H \setminus G$.

It turns out that this distinction between the minimal and the general embeddings is important when we consider arc spaces and their global models. As we will recall in Lemma 5.6.6, the map $X^{\rm can} \to X$ induces a closed embedding of mapping stacks

$$\mathcal{M}_{X^{\mathrm{can}}} \hookrightarrow \mathcal{M}_X$$
,

whose image is a union of irreducible components.

We are more interested in the closure of $\mathfrak{M}_{X^{\bullet}}^{0} = \operatorname{Bun}_{H}$ in the former, which we will denote by $\overline{\mathfrak{M}}_{X}^{0}$ (it may or may not be the same as $\mathfrak{M}_{X^{\operatorname{can}}}$ depending on whether the monoid $\mathfrak{c}_{X^{\operatorname{can}}}$ is generated by colors, see Corollary 5.6.4):

$$(5.1) \overline{\mathcal{M}}_X^0 \hookrightarrow \mathcal{M}_{X^{\operatorname{can}}} \hookrightarrow \mathcal{M}_X.$$

The main theme of this section is, in some sense, a reconstruction of \mathcal{M}_X out of suitable Hecke operators acting on $\overline{\mathcal{M}}_X^0$: For any $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$, a multiset of nonzero elements in \mathfrak{c}_X^- , there is a natural proper map

(5.2)
$$\operatorname{act}_{\mathcal{M}} : \overline{\mathcal{M}}_{X}^{0} \times \overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}} \to \mathcal{M}_{X},$$

which corresponds to the action on the "basic stratum" $\overline{\mathcal{M}}_X^0$ by the closed stratum of the affine Grassmannian parametrized by $\check{\Theta}$. (See Proposition–Construction 5.2.3.) Recall that such a multiset $\check{\Theta}$ also parametrizes a stratum $\mathcal{M}_X^{\check{\Theta}}$ of the global mapping space (§3.1.5). The main technical result of this section is the following:

Theorem 5.1.1. For every $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$, the action map (5.2)

- (i) has image equal to the closure of \mathfrak{M}_X^{Θ} , and
- (ii) is birational onto its image.
- (iii) The restriction of $\operatorname{act}_{\mathcal{M}}$ to $\operatorname{act}_{\mathcal{M}}^{-1}(\mathcal{M}_X^{\check{\Theta}}) \cap (\operatorname{Bun}_H \tilde{\times} \operatorname{Gr}_{G,\mathring{C}^{\check{\Theta}}}^{\check{\Theta}}) \to \mathcal{M}_X^{\check{\Theta}}$ is an isomorphism.

The proof of Theorem 5.1.1 will be given in §5.5.

We use this theorem to achieve two goals in this section:

The first goal is to understand irreducible components of \mathcal{M}_X and closure relations among the strata \mathcal{M}_X^{\bullet} . We will introduce the natural generalization to multisets of the order \succeq among elements of the lattice $\check{\Lambda}_X$ (we remind that this order is determined by the monoid of colors, see §2.1), and prove:

Proposition 5.1.2 (See Proposition 5.6.1.). Let $\check{\Theta}, \check{\Theta}' \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X}^{-} - 0)$. The stratum $\mathfrak{M}_{X}^{\check{\Theta}'}$ lies in the closure of $\mathfrak{M}_{X}^{\check{\Theta}}$ if and only if there exists $\check{\Theta}''$ such that $\check{\Theta}$ refines $\check{\Theta}''$ and $\check{\Theta}' \succeq \check{\Theta}''$.

Moreover, observe that the irreducible components of $\overline{\mathcal{M}}_X^0$ are in bijection with connected components of $\mathcal{M}_{X^{\bullet}} = \operatorname{Bun}_H$, i.e., parametrized by $\pi_0(\operatorname{Bun}_H) = \pi_1(H)$. Using the action (5.2) on those components gives us a parametrization of the irreducible components of the closure of each stratum:

Proposition 5.1.3 (See Corollaries 5.5.9 and 5.7.2). For every $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$, the irreducible components in the closure $\overline{\mathcal{M}}_X^{\check{\Theta}}$ of the corresponding stratum are naturally parametrized by $\pi_1(H)$. For any $\check{\lambda} \in \mathfrak{c}_X$, the base change of an irreducible component to $\mathcal{Y}^{\check{\lambda}} \times_{\mathcal{M}_X} \overline{\mathcal{M}}_X^{\check{\Theta}}$ is still irreducible (when nonempty).

The combination of Propositions 5.1.2 and 5.1.3 implies:

Corollary 5.1.4 (See Corollary 5.6.5). There is a natural bijection between the set of irreducible components of \mathcal{M}_X and

$$\pi_1(H) \times \operatorname{Sym}^{\infty}(\mathcal{D}^G_{\operatorname{sat}}(X)),$$

where $\mathcal{D}_{\mathrm{sat}}^G(X)$ denotes the set of primitive elements in \mathfrak{c}_X^- that cannot be decomposed as a sum $\check{\theta} + \check{\nu}_D$ where $\check{\theta} \in \mathfrak{c}_X^- - 0$ and $\check{\nu}_D$ is the valuation attached to a color.

The second goal achieved by Theorem 5.1.1 is to reduce the study of the IC complex of an arbitrary mapping space \mathcal{M}_X to that of the minimal affine embedding:

Theorem 5.1.5. For any $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X}^{-} - 0)$, there is a natural isomorphism

$$\operatorname{IC}_{\overline{\mathcal{M}}_X^0} \star \operatorname{IC}_{\overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}} \cong \operatorname{IC}_{\overline{\mathcal{M}}_X^{\check{\Theta}}}.$$

We point to §A.5 for the notation and the proof. Theorem 5.1.5 and its proof are independent from the rest of this paper; we include it only for conceptual completeness. The proof involves a passage to Zastava model (the extra structure of a flag) and uses the results of the sequel §6. The argument of proof can also be extended to meromorphic quasimaps to verify [GN10, Conjecture 7.3.2] in the case $\check{G}_X = \check{G}$.

5.2. Hecke action on global model. Now we introduce the action map (5.2), as an analog of the action of G(F) on $X^{\bullet}(F)$ for the global model.

The notation is cumbersome because we give a multi-point version of the action, but the idea is simple: in the notation of §3.1.3, the k-points of \mathcal{M}_X correspond to a subset of $H(\mathbb{k})\backslash G(\mathbb{A})/G(\mathbb{O})$, where $\mathbb{O} = \prod_{v \in |C|} \mathfrak{o}_v$. We have a Hecke correspondence

$$H(\Bbbk)\backslash G(\mathbb{A}) \overset{G(\mathbb{O})}{\times} G(\mathbb{A})/G(\mathbb{O}) \to H(\Bbbk)\backslash G(\mathbb{A})/G(\mathbb{O})$$

induced by multiplication in G. We think of $G(\mathbb{A})/G(\mathbb{O})$ as a factorizable version of the affine Grassmannian. Then the "action map" is simply the restriction of the above to a positively graded subset of $G(\mathbb{A})/G(\mathbb{O})$ such that everything maps to $(X^{\bullet}(\mathbb{A}) \cap X(\mathbb{O}))/G(\mathbb{O})$, i.e., the locus of regular maps.

5.2.1. Positively graded subscheme of factorizable affine Grassmannian. Let $\check{\Theta} = \sum_{\check{\theta}} N_{\check{\theta}}[\check{\theta}] \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^{-} - 0)$ be a partition. We have a closed subscheme

$$\overline{\mathrm{Gr}}_{G,C}^{\check{\theta}} = \overline{\mathrm{Gr}}_{G}^{\check{\theta}} \tilde{\times} C := \overline{\mathrm{Gr}}_{G}^{\check{\theta}} \overset{\mathrm{Aut}^{0}(k[\![t]\!])}{\times} \mathrm{Coord}^{0}(C) \subset \mathrm{Gr}_{G,C}$$

where $\operatorname{Aut}^0(k[\![t]\!]) = \operatorname{Spec} k[a_1^{\pm 1}, a_2, \ldots]$ is the group scheme of algebra automorphisms of $k[\![t]\!]$ that preserve the maximal ideal, and $\operatorname{Coord}^0(C) \to C$ is the $\operatorname{Aut}^0(k[\![t]\!])$ -torsor classifying $v \in C$ together with an isomorphism $k[\![t]\!] \cong \mathfrak{o}_v$ sending t to a uniformizer (see §A.1.1 or [Zhu17, (3.1.11)]). Consider the $N_{\tilde{\theta}}$ -fold product $(\overline{\operatorname{Gr}}_{G,C}^{\tilde{\theta}})^{N_{\tilde{\theta}}} \times_{C^{N_{\tilde{\theta}}}} \mathring{C}^{N_{\tilde{\theta}}}$ restricted to the disjoint locus with all diagonals removed. This descends to a subscheme $\overline{\operatorname{Gr}}_{G,\mathring{C}^{(N_{\tilde{\theta}})}}^{N_{\tilde{\theta}}\tilde{\theta}} \subset \operatorname{Gr}_{G,\mathring{C}^{(N_{\tilde{\theta}})}}$. In the notation of §3.1.4, let

$$\overline{\mathrm{Gr}}_{G,\mathring{C}^{\check{\Theta}}}^{\check{\Theta}} := \prod_{\check{\theta}}^{\circ} \overline{\mathrm{Gr}}_{G,\mathring{C}^{(N_{\check{\theta}})}}^{N_{\check{\theta}}\check{\theta}} \subset \mathrm{Gr}_{G,C^{(|\check{\Theta}|)}} \underset{C^{(|\check{\Theta}|)}}{\times} \underset{C^{(|\check{\Theta}|)}}{\times} \mathring{C}^{\check{\Theta}}$$

and let $\overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Phi}}$ denote its closure in $\mathrm{Gr}_{G,C^{(|\check{\Phi}|)}} \times_{C^{(|\check{\Phi}|)}} C^{\check{\Phi}}$. We consider $\overline{\mathrm{Gr}}_{G,C^{\check{\Phi}}}^{\check{\Phi}}$ as a scheme over $C^{(|\check{\Phi}|)}$ with an action of the group scheme $(\mathcal{L}^+G)_{C^{(|\check{\Phi}|)}}$, the multi-point version of the arc space defined in §A.1.

The closure relations of the Beilinson–Drinfeld affine Grassmannian are known (cf. [Zhu17, Proposition 3.1.14]), so we can describe the reduced fiber of $\overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$ over a point of $C^{\check{\Theta}}$ as follows: a point of $C^{\check{\Theta}}$ is the collection $(D^{\check{\theta}})_{\check{\theta}}$, for each $\check{\theta}$, of a degree $N_{\check{\theta}}$ divisor $D^{\check{\theta}} = \sum_{v \in |C|} N_{\check{\theta},v}v$. The reduced fiber of $\overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$ over this point is the scheme

(5.3)
$$\prod_{v \in |C|} \overline{\operatorname{Gr}}_{G,v}^{\sum_{\check{\theta}} N_{\check{\theta},v}\check{\theta}}.$$

5.2.2. Let $\widehat{\mathcal{M}}_X$ denote the stack representing the data of

$$(\sigma, \mathcal{P}_G) \in \mathcal{M}_X, \ D \in \operatorname{Sym} C, \ \text{and a trivialization} \ \mathcal{P}_G|_{\widehat{C}'_D} \cong \mathcal{P}^0_G|_{\widehat{C}'_D}$$

(see §3.7.5 for the definition of \widehat{C}'_D), i.e., a point of \mathcal{M}_X together with infinite G-level structure at points in the support of D. This admits a natural action by \mathcal{L}^+G , and the forgetful map $\mathcal{M}_X \to \mathcal{M}_X \times \operatorname{Sym} C$ is a \mathcal{L}^+G -torsor. Let

$$\mathcal{M}_X \,\tilde{\times} \, \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}} := \widehat{\mathcal{M}}_X \, \underset{\mathrm{Sym} \, C}{\overset{\mathcal{L}^+ G}{\times}} \, \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$$

denote the twisted product over $C^{(|\check{\Theta}|)} \subset \operatorname{Sym} C$.

For an affine test scheme S, an S-point of $\mathcal{M}_X \times \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$ is the data $(\sigma, \mathcal{P}_G, \mathcal{P}'_G, (D^{\check{\theta}})_{\check{\theta}}, \tau)$ where

- $\begin{array}{l} \bullet \ \, \mathbb{P}_G, \mathbb{P}_G' \ \, \text{are} \,\, G\text{-bundles on} \,\, C \times S, \\ \bullet \ \, \sigma : C \times S \to X \times^G \mathbb{P}_G \,\, \text{is a section such that} \,\, (\sigma, \mathbb{P}_G) \in \mathbb{M}_X(S), \end{array}$
- for each $\check{\theta} \in \mathfrak{c}_X^- 0$, we have a degree $N_{\check{\theta}}$ effective Cartier divisor $D^{\check{\theta}} \subset C \times S$,
- $\tau: \mathcal{P}'_G|_{C \times S D} \cong \mathcal{P}_G|_{C \times S D}$ is a trivialization for $D := \sum_{\check{\theta}} D^{\check{\theta}}$

such that after fixing an isomorphism $\mathcal{P}_G|_{\widehat{C}'_D} \cong \mathcal{P}^0_G|_{\widehat{C}'_D}$ (which always exists after flat base change over S), the datum $((D^{\check{\theta}}), \mathfrak{P}'_G|_{\widehat{C}'_D}, \tau)$ defines a point in $\overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$. Here we are implicitly using Beauville–Lazlo's theorem to pass between the global and local descriptions of $Gr_{G,Sym C}$,

Proposition-Construction 5.2.3. For any $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X}^{-} - 0)$ there is a natural proper map

$$\operatorname{act}_{\mathfrak{M}}: \mathfrak{M}_X \times \overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}} \to \mathfrak{M}_X.$$

Proof. Fix an affine test scheme S and $(\sigma, \mathcal{P}_G, \mathcal{P}'_G, (D^{\check{\theta}})_{\check{\theta}}, \tau) \in \mathcal{M}_X \times \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}(S)$. The composition $\tau^{-1} \circ \sigma|_{C \times S - D}$ defines a section $\sigma' : C \times S - D \to X \times^G \mathcal{P}'_G$. We claim that σ' extends to a regular map on $C \times S$. Given this claim, we can define $(\sigma', \mathcal{P}'_G) \in \mathcal{M}_X(S)$ to be the image of $\operatorname{act}_{\mathcal{M}}$. Properness of $\operatorname{act}_{\mathcal{M}}$ follows from properness of $\operatorname{\overline{Gr}}_{G,C^{\Theta}}^{\bullet}$ and Lemma 3.7.3.

We now prove the claim that σ' extends to all of $C \times S$. Since k[X] is a locally finite Gmodule, it is generated as an algebra by some finite dimensional G-submodule $V \subset k[X]$. This induces a G-equivariant embedding of varieties $\varphi: X \hookrightarrow V^*$, where V^* is considered as a right G-module. It suffices to show that the composition $\varphi(\sigma'): C \times S - D \to V^* \times^G \mathcal{P}'_G$ extends. For any weight μ of $V \subset k[X]$, we have $\mu \leq \lambda$ for $\lambda \in \mathfrak{c}_X^{\vee}$. Therefore given $\check{\theta} \in \mathfrak{c}_X^{-}$, we have $\langle \mu, \hat{\theta} \rangle \geq 0$ for all weights μ of V. We deduce that the group homomorphism $G \to \mathrm{GL}(V)$ induces a natural map

$$\overline{\mathrm{Gr}}_{G,C^{\Theta}}^{\bullet} \to \mathcal{L}^{+}\mathrm{End}(V)/\mathcal{L}^{+}\mathrm{GL}(V).$$

Hence τ^{-1} induces a regular map $V^* \times^G \mathcal{P}_G|_{\widehat{C}'_D} \to V^* \times^G \mathcal{P}'_G|_{\widehat{C}'_D}$ and $\varphi(\tau^{-1} \circ \sigma|_{\widehat{C}'_D})$ defines a section $\widehat{C}'_D \to V^* \times^G \mathcal{P}'_G$. By Beauville–Laszlo's theorem (cf. [BD96, Theorem 2.12.1]), this implies that $\varphi(\sigma')$ is defined on all of $C \times S$.

We describe more precisely what $act_{\mathcal{M}}$ is doing on k-points: at a single $v \in |C|$, a k-point of \mathcal{M}_X gives an element of $(X(\mathfrak{o}_v) \cap X^{\bullet}(F_v))/G(\mathfrak{o}_v)$ and the map $\operatorname{act}_{\mathcal{M}}$ corresponds to the natural $G(F_v)$ -action on $X^{\bullet}(F_v)$. For $\check{\mu} \in \mathring{\Lambda}_G^-$, define the set

$$X^{\bullet}(F_v)_{G:\succeq \check{\mu}} = \bigcup_{\check{\mu}' \in \check{\Lambda}_G^-, \check{\mu}' \succeq \check{\mu}} X^{\bullet}(F_v)_{G:\check{\mu}'},$$

in the notation of §2.3.4. In the next subsection we prove a slightly more precise¹⁶ version of [SV17, Lemma 5.5.2]:

Lemma 5.2.4. Let $\check{\mu}, \check{\theta} \in \check{\Lambda}_G^-$. The action map sends

$$X^{\bullet}(F_v)_{G:\succeq\check{\mu}} \overset{G(\mathfrak{o}_v)}{\times} \overline{\mathsf{L}^+G \cdot t^{\check{\theta}} \cdot \mathsf{L}^+G}(k) \to X^{\bullet}(F_v)_{G:\succeq\check{\mu}+\check{\theta}}.$$

5.3. Reduction to $\mathfrak{c}_{X^{\bullet}} = \mathbb{N}^{\mathcal{D}}$. We are interested in applying Hecke actions to $\overline{\mathcal{M}}_{X}^{0}$, the closure of $\mathcal{M}_{X^{\bullet}} = \mathcal{M}_{X}^{0} = \operatorname{Bun}_{H}$ in \mathcal{M}_{X} , since it is the most basic closure of a stratum. On the other hand, in order to determine the stratification of $\overline{\mathcal{M}}_{X}^{0}$ we need a moduli description of this stack. A first guess would be that $\overline{\mathcal{M}}_{X}^{0} = (\mathcal{M}_{X^{\operatorname{can}}})_{\operatorname{red}}$, but this may not be true if $\mathfrak{c}_{X^{\bullet}} := \mathfrak{c}_{X^{\operatorname{can}}}$ is not equal to $\mathfrak{c}_{X}^{\mathcal{D}} = \{\check{\lambda} \succeq 0\}$. In this subsection we explain how to get around this technical issue: we can always replace G by a central extension $G' \to G$ such that if $H \setminus G = H' \setminus G'$ and $X' := \operatorname{Spec} k[H' \setminus G']$, then $\overline{\mathcal{M}}_{X'}^{0} = \mathcal{M}_{X'}$ and $\mathfrak{c}_{X'} = \mathbb{N}^{\mathcal{D}}$ (see Lemma 5.3.3).

This is a generalization of the need to replace $\overline{N\backslash G}^{\rm aff}$ by $\overline{N\backslash \widetilde{G}}^{\rm aff}$ for \widetilde{G} a simply connected cover of G in [ABB⁺05, §4.1], [Sch15, §7.2] to correctly define Drinfeld's compactification of Bun_N for an arbitrary reductive group G.

5.3.1. Let us first assume that k[G] is a UFD. Further, assume that H is connected, as is the case under our assumptions (Remark 3.0.1). Then the preimage of any color D in G is irreducible, and this defines a bijection between $H \times B$ -stable prime divisors in G and colors; moreover, under our UFD assumption, the former are all principal.

Let $k(G)^{(H \times B)}$ denote the $H \times B$ -eigenvectors of k(G). Then, since HB is open dense in G, we have

$$k(G)^{(H \times B)}/k^{\times} = \mathcal{X}(H) \underset{\mathcal{X}(B \cap H)}{\times} \mathcal{X}(B),$$

where $\mathcal{X}(H)$ is the character group of H (so $\mathcal{X}(B) = \Lambda_G$ by definition). The valuation map gives rise to a short exact sequence

$$(5.4) 0 \to \mathcal{X}(G) \to \mathcal{X}(H) \underset{\mathcal{X}(B \cap H)}{\times} \mathcal{X}(B) \to \mathbb{Z}^{\mathcal{D}} \to 0,$$

which sends an element of $k(G)^{(H\times B)}/k^{\times}$ to its divisor, identified with a \mathbb{Z} -linear combination of the colors. (We have used here both the UFD property, and the fact that invertible regular functions on G are multiples of characters.)

The preimage in G of each color D defines a valuation v_{HD} on $k(G)^{\times}$; its restriction to $k(G)^{(H \times B)}$ defines a map

$$\widetilde{\varrho}_{H\backslash G}: \mathcal{D} \to (\mathcal{X}(H) \underset{\mathcal{X}(B\cap H)}{\times} \mathcal{X}(B))^{\vee}.$$

Composing with the second projection to Λ_X gives the usual valuation map $\varrho_{H\backslash G}$. By construction we have $v_{HD}(f_{D'}) = \delta_{D,D'}$ for $D, D' \in \mathcal{D}$, and the sequence dual to (5.4) is

$$0 \to \mathbb{Z}^{\mathcal{D}} \xrightarrow{\widetilde{\varrho}_{H \setminus G}} (\mathcal{X}(H) \underset{\mathcal{X}(B \cap H)}{\times} \mathcal{X}(B))^{\vee} \to \mathcal{X}(G)^{\vee} \to 0.$$

 $^{^{16}}$ In [SV17] the ordering \succeq is defined with respect to the *rational* cone generated by the valuations of colors, whereas we define \succeq with respect to the monoid generated by non-negative *integral* combinations of valuations of colors.

5.3.2. Now we return to the case of arbitrary G. Let

$$1 \to Z \to \widetilde{G} \to G \to 1$$

be a central extension with connected kernel Z such that the derived group $[\widetilde{G}, \widetilde{G}]$ is simply connected (such a \widetilde{G} always exists). Then $k[\widetilde{G}]$ is a UFD ([KKLV89, Proposition 4.6], [Ive76]). We consider $H \setminus G$ as a \widetilde{G} -spherical variety, so $H \setminus G = \widetilde{H} \setminus \widetilde{G}$ where $\widetilde{H} = HZ$ is the preimage of H in \widetilde{G} . Let \widetilde{B} denote the Borel subgroup of \widetilde{G} . If H is connected, then \widetilde{H} is also connected. The colors stay the same, so $\widetilde{\varrho}_{\widetilde{H} \setminus \widetilde{G}}$ induces a short exact sequence

$$(5.5) 0 \to \mathbb{Z}^{\mathcal{D}} \xrightarrow{\widetilde{\varrho}_{\widetilde{H}} \setminus \widetilde{G}} (\mathcal{X}(\widetilde{H}) \underset{\mathcal{X}(\widetilde{B} \cap \widetilde{H})}{\times} \mathcal{X}(\widetilde{B}))^{\vee} \to \mathcal{X}(\widetilde{G})^{\vee} \to 0.$$

Let $\widetilde{H}_{ab} = \widetilde{H}/[\widetilde{H},\widetilde{H}]$, so $\mathcal{X}(\widetilde{H}) = \mathcal{X}(\widetilde{H}_{ab})$. We consider $[\widetilde{H},\widetilde{H}]\backslash \widetilde{G}$ as a spherical variety for the group $G' := \widetilde{H}_{ab} \times \widetilde{G}$, where \widetilde{H}_{ab} acts by left translation and \widetilde{G} acts by right translation. Then $[\widetilde{H},\widetilde{H}]\backslash \widetilde{G} = H'\backslash G'$ where $H' = (\widetilde{H}/Z)^{\text{diag}} = H^{\text{diag}}$ is diagonally embedded in G'. Observe that $G' \to G$ is a central extension with kernel \widetilde{H}_{ab} and

$$H'\backslash G'=[\widetilde{H},\widetilde{H}]\backslash \widetilde{G}\to \widetilde{H}\backslash \widetilde{G}=H\backslash G$$

is a \widetilde{H}_{ab} -torsor. The Borel subgroup B' of G' is $\widetilde{H}_{ab} \times^Z \widetilde{B}$ and $B' \cap H' = ((\widetilde{B} \cap \widetilde{H})/Z)^{\text{diag}}$, so (5.5) is equivalent to a short exact sequence

$$0 \to \mathbb{Z}^{\mathcal{D}} \xrightarrow{\varrho_{H' \setminus G'}} \check{\Lambda}_{H' \setminus G'} = \ker(\mathcal{X}(B') \to \mathcal{X}(B' \cap H'))^{\vee} \to \mathcal{X}(\widetilde{G})^{\vee} \to 0.$$

To summarize:

Lemma 5.3.3. Let $H \setminus G$ be a homogeneous spherical variety with H connected. Then there exists a central extension $G' \to G$ with connected kernel Z' and a spherical subgroup $H' \subset G'$ such that

- (i) [G', G'] is simply connected.
- (ii) The covering $G' \to G$ restricts to an isomorphism $H' \cong H$. In particular, the projection $H' \setminus G' \to H \setminus G$ is a Z'-torsor.
- (iii) The valuation map $\varrho_{H'\setminus G'}$ embeds $\mathbb{Z}^{\mathcal{D}} \hookrightarrow \check{\Lambda}_{H'\setminus G'}$ as a direct summand.

By (iii), we have that $\mathfrak{c}_{H'\setminus G'} = \mathfrak{c}_{H'\setminus G'}^{\mathcal{D}} = \mathbb{N}^{\mathcal{D}}$ so there is no question of integrality. For a similar result, see [Bri07, Lemma 2.1.1].

Example 5.3.4. If we start with $H = \mathbb{G}_m$ the torus inside $G = \operatorname{PGL}_2$, then $G' = \mathbb{G}_m \times \operatorname{GL}_2$ and $H' = \mathbb{G}_m$ where \mathbb{G}_m maps to GL_2 by $\left(\begin{smallmatrix} * & 0 \\ 0 & 1 \end{smallmatrix}\right)$. So $H' \setminus G' = \operatorname{GL}_2$, where \mathbb{G}_m acts by left translations and GL_2 by right translations. This is a \mathbb{G}_m -torsor over $\mathbb{G}_m \setminus \operatorname{GL}_2$ and a $\mathbb{G}_m \times Z(\operatorname{GL}_2)$ -torsor over $\mathbb{G}_m \setminus \operatorname{PGL}_2$.

Proof of Lemma 5.2.4. Let $1 \to Z' \to G' \to G \to 1$ be a central extension as in Lemma 5.3.3. Since $\mathfrak{c}_{H' \setminus G'} = \mathfrak{c}_{H' \setminus G'}^{\mathcal{D}}$, there is no question of integrality and [SV17, Lemma 5.5.2] applied to the G'-variety $H' \setminus G'$ says that the action map sends

$$(H'\backslash G')(F_v)_{\succeq \tilde{\mu}} \overset{G'(\mathfrak{o}_v)}{\times} G'(\mathfrak{o}_v) t^{\tilde{\theta}} G'(\mathfrak{o}_v) \to (H'\backslash G')(F_v)_{\succeq \tilde{\mu} + \tilde{\theta}},$$

where $\tilde{\mu}, \tilde{\theta} \in \check{\Lambda}^-_{H'\backslash G'}$. Since $H'\backslash G' \to H\backslash G = X^{\bullet}$ is a Z'-torsor, it is surjective on F_v -points (and corresponding orbits). Since $\mathfrak{c}^{\mathcal{D}}_{H'\backslash G'}$ maps to $\mathfrak{c}^{\mathcal{D}}_X$, we deduce that the action map sends

$$X^{\bullet}(F_v)_{\succeq \check{\mu}} \overset{G(\mathfrak{o}_v)}{\times} G(\mathfrak{o}_v) t^{\check{\theta}} G(\mathfrak{o}_v) \to X^{\bullet}(F_v)_{\succeq \check{\mu} + \check{\theta}}$$

for $\check{\mu}, \check{\theta} \in \check{\Lambda}_G^-$. The preimage of $\overline{\mathrm{Gr}}_G^{\check{\theta}}(k)$ in $G(F_v)$ consists of the union of $G(\mathfrak{o}_v)t^{\check{\theta}'}G(\mathfrak{o}_v)$ for $\check{\theta}' \in \check{\Lambda}_G^-$, $\check{\theta}' \geq \check{\theta}$. Since $\check{\theta}' \geq \check{\theta}$ implies $\check{\theta}' \succeq \check{\theta}$, we deduce the claim. \square

5.4. **Open Zastava.** Consider $\mathcal{Y}^{?,0} = \mathcal{Y}_{X^{\bullet}} = \mathcal{Y} \times_{\mathcal{M}_X} \mathcal{M}_X^0 = \operatorname{Maps}_{\operatorname{gen}}(C, H \setminus G/B \supset \operatorname{pt})$, which is an open subscheme of \mathcal{Y} . Since $\mathcal{M}_X^0 \cong \operatorname{Bun}_H$ is smooth and $\mathcal{Y}_{X^{\bullet}}$ is smooth locally isomorphic to \mathcal{M}_X^0 by Lemma 3.5.4, the scheme $\mathcal{Y}_{X^{\bullet}}$ is also smooth.

The preimage of the connected component $\mathcal{A}^{\check{\lambda}}$ in $\mathcal{Y}_{X^{\bullet}}$ is by definition $\mathcal{Y}^{\check{\lambda},0}$. Let $G' \to G$ be a central extension as in Lemma 5.3.3, so we have a torsor $H' \backslash G' \to X^{\bullet}$. Let $X' = \operatorname{Spec} k[H' \backslash G']$. Then $\mathfrak{c}_{X'} = \mathbb{N}^{\mathcal{D}}$ and we have an isomorphism of stacks $X'^{\bullet}/B' \cong X^{\bullet}/B$. Hence,

$$\mathcal{Y}_{X^{\bullet}} \cong \mathcal{Y}_{X^{\prime \bullet}} = \operatorname{Maps}_{\operatorname{gen}}(X^{\prime \bullet}/B^{\prime} \supset \operatorname{pt}),$$

and the map $\mathcal{Y}_{X^{\bullet}} \to \mathcal{A}$ factors through $\pi_{X'}: \mathcal{Y}_{X'} \to \mathcal{A}_{X'}$. The base $\mathcal{A}_{X'}$ is a disjoint union of smooth partially symmetrized powers of the curve indexed by $\mathbb{N}^{\mathcal{D}}$. For $D = \sum_{\mathcal{D}} n_{D'} \cdot D' \in \mathbb{N}^{\mathcal{D}}$, we will denote by $\mathcal{Y}_{X^{\bullet}}^{D}$ the preimage of the corresponding component of $\mathcal{A}_{X'}$. We define $\varrho_{X}(D) = \sum_{\mathcal{D}} n_{D'} \check{\nu}_{D'}$ and $\text{len}(D) = \sum_{\mathcal{D}} n_{D'}$. Hence, if $\check{\lambda} = \varrho_{X}(D)$, then $\mathcal{Y}_{X^{\bullet}}^{D} \subset \mathcal{Y}_{X^{\bullet}}^{\check{\lambda}}$. This implies:

Lemma 5.4.1. The stratum $\mathcal{Y}^{\check{\lambda},0}$ is nonempty only if $\check{\lambda} \succeq 0$.

Under our assumption that all simple roots of G are spherical roots of type T, every color $D \in \mathcal{D}$ belongs to $\mathcal{D}(\alpha)$ for some simple root α of G and $X \circ P_{\alpha}/\mathfrak{R}(P_{\alpha}) = \mathbb{G}_m \backslash \mathrm{PGL}_2$.

Lemma 5.4.2. If $\mathcal{D}(\alpha) = \{D_{\alpha}^+, D_{\alpha}^-\}$ for a simple root α , then for any $n^{\pm} \in \mathbb{N}$, there is an isomorphism

$$y_{X\bullet}^{n^+D_{\alpha}^++n^-D_{\alpha}^-} = C^{(n^+)} \mathring{\times} C^{(n^-)}.$$

Proof. We may assume $\mathfrak{c}_{X^{\bullet}} = \mathbb{N}^{\mathcal{D}}$. Then, $\mathbb{Y}_{X^{\bullet}}^{n^+D_{\alpha}^+ + n^-D_{\alpha}^-}$ classifies maps from the curve to X^{\bullet}/B which have zero valuation on every color other than those in $\mathcal{D}(\alpha)$, and therefore is a subscheme of $\operatorname{Maps}_{\mathrm{gen}}(C, (X^{\bullet} - \bigcup_{D \in \mathcal{D} - \mathcal{D}(\alpha)} D)/B \supset \mathsf{pt})$. Since $X^{\circ}P_{\alpha} = \mathbb{G}_m \backslash P_{\alpha}$ is affine, its complement is the union of the colors that it does not intersect. Hence

$$(X^{\bullet} - \bigcup_{D \in \mathcal{D} - \mathcal{D}(\alpha)} D)/B = (X^{\circ} P_{\alpha})/B = \mathbb{G}_m \backslash \mathrm{PGL}_2/B_{\mathrm{PGL}_2} = \mathbb{G}_m \backslash \mathbb{P}^1.$$

By Example 3.3.1, we see that $\operatorname{Maps}_{\operatorname{gen}}(C, \mathbb{G}_m \backslash \mathbb{P}^1 \supset \operatorname{pt}) = \operatorname{Sym} C \overset{\circ}{\times} \operatorname{Sym} C$, and it follows that the components correspond to $\mathbb{N}^{\mathcal{D}(\alpha)}$ in the natural way.

Remark 5.4.3. For a general element $D = \sum_{\mathcal{D}} n_{D'} \cdot D' \in \mathbb{N}^{\mathcal{D}}$, the graded factorization property together with Lemma 5.4.2 imply that there is an open embedding $\mathring{\prod}_{\mathcal{D}} C^{(n_{D'})} \hookrightarrow \mathcal{Y}_{X\bullet}^{\mathcal{D}}$, where the product is over the disjoint locus. We will show in Lemma 6.2.1 that $\mathcal{Y}_{X\bullet}^{\mathcal{D}}$, $D \in \mathbb{N}^{\mathcal{D}}$ are precisely the connected components of $\mathcal{Y}_{X\bullet}$, so the open subscheme above is dense. (This also implies that the $\mathcal{Y}_{X\bullet}^{\mathcal{D}}$ are defined intrinsically and independently of the choice of $G' \to G$.)

Remark 5.4.4. If D is as above, and $\check{\lambda} = \varrho_X(D)$, we can read off the length len(D) directly from $\check{\lambda}$, if X^{\bullet} happens to admit a G-eigen-volume form. Namely, under this assumption there is a character $\gamma \in \Lambda$ such that $\langle \gamma, \check{\nu}_{D'} \rangle = 1$ for every color D', and therefore len $(D) = \langle \gamma, \check{\lambda} \rangle$. Indeed, for such a color D', there is a simple root α such that $D'P_{\alpha}$ contains the open Borel orbit, and then $D'P_{\alpha} \simeq \check{\nu}_{D'}(\mathbb{G}_m)\backslash P_{\alpha}$. For such a homogeneous space to have a P_{α} -eigen-volume form with eigencharacter \mathfrak{h} , we must have that $\langle \mathfrak{h} + 2\rho_{N_{\alpha}}, \check{\nu}_{D'} \rangle = 0$, where $2\rho_{N_{\alpha}}$ is the sum of roots in the unipotent radical of P_{α} . Equivalently, since $\langle \alpha, \check{\nu}_{D'} \rangle = 1$, this reads $\langle \mathfrak{h} + 2\rho_G, \check{\nu}_{D'} \rangle = 1$. Therefore,

(5.7)
$$\operatorname{len}(D) = \langle \mathfrak{h} + 2\rho_G, \check{\lambda} \rangle,$$

which we can unambiguously denote by len($\check{\lambda}$).

5.5. Proof of Theorem 5.1.1.

5.5.1. *Idea of the proof.* Let us give a set-theoretic idea for the proof of Theorem 5.1.1, as well as a guide to new notation we are about to introduce. The discussion here is not rigorous, using sets as avatars for geometric objects.

We have been using the notation $X^{\bullet}(F)_{G:\check{\theta}}$, or $(X^{\bullet}(F)/K)_{G:\check{\theta}}$ to denote the set of points in the K-orbit parametrized by $\check{\theta} \in \mathcal{V} \cap \check{\Lambda}_X$, where $K = G(\mathfrak{o})$. The stratum $\mathcal{M}_X^{\check{\theta}}$ of the global model corresponds to $(X(\mathfrak{o})/K)_{G:\check{\theta}}$, in the sense that it is determined by the condition that the "G-valuation" of maps in this stratum, at points where they fail to land in X^{\bullet} , is given by $\check{\theta}$.

Similarly, $T(\mathfrak{o})N(F)$ -orbits on $X^{\circ}(F)$ are parametrized by $\check{\Lambda}_X$, and let us denote by $X(F)_{B:\check{\lambda}}$ the subset of points which: (1) belong to $X^{\circ}(F)$, and (2) belong to the $T(\mathfrak{o})N(F)$ -orbit parametrized by $\check{\lambda} \in \check{\Lambda}_X$. As with the global model, the *central fibers* of the stratum $\bigvee_X^{\check{\lambda},\check{\theta}}$ of the Zastava space correspond to $(X(\mathfrak{o})/B(\mathfrak{o}))_{G:\check{\theta}}$.

The following facts will be proven about the space $\mathcal{Y}_{X}^{\check{\lambda},\check{\theta}}$:

- It is nonempty only if $\check{\lambda} \succeq \check{\theta}$ (see Lemma 5.4.1 and Corollary 5.5.6). This is essentially a statement about the image of $X(\mathfrak{o})$ in $X/\!\!/N(F)$.
- It contains $\mathcal{Y}_X^{\check{\lambda}-\check{\theta},0}\mathring{\times}\mathcal{Y}_X^{\check{\theta},\check{\theta}}$ as an open dense (see Lemma 5.5.7). This is a geometric statement, and it follows from the constructions that we describe below.

Theorem 5.1.1 says that the geometric analog of the action map

(5.8)
$$\operatorname{act}: \overline{X^{\bullet}(\mathfrak{o})} \times^{K} \overline{Kt^{\check{\theta}}K} \to \overline{X(\mathfrak{o})}_{G \cdot \check{\theta}}$$

is birational and proper. (The closures are also understood here in the "geometric" topology, see Lemma 5.2.4.)

To prove this, we fix $\check{\theta} \in \mathfrak{c}_X^-$, which will not appear in the notation, and work with $B(\mathfrak{o})$ orbits, defining spaces whose central fibers satisfy the following set-theoretic analogies:

ing spaces whose central fibers satisfy the following set-theoretic analogy
$$Z^{\check{\lambda}-\check{\nu},\check{\lambda}} \qquad \overline{X^{\bullet}(\mathfrak{o})}_{B:\check{\lambda}-\check{\nu}} \overset{B(\mathfrak{o})}{\times} (T(\mathfrak{o})t^{\check{\nu}}N(F) \cap \overline{Kt^{\check{\theta}}K})/B(\mathfrak{o})$$

$$Z^{\check{\lambda}-\check{\nu},\check{\lambda},\check{\theta}',\check{\eta}} \qquad \overline{X^{\bullet}(\mathfrak{o})}_{G:\check{\theta}'} \overset{B(\mathfrak{o})}{\times} (T(\mathfrak{o})t^{\check{\nu}}N(F) \cap Kt^{\check{\eta}}K)/B(\mathfrak{o})$$

Here, $\check{\eta} \geq \check{\theta}$ (and antidominant), so that the Cartan double coset $Kt^{\check{\eta}}K$ belongs (in the affine Grassmannian) to the closure of $Kt^{\check{\theta}}K$. Both of the spaces above are subspaces of the preimage of $X(\mathfrak{o})_{B:\check{\lambda}}$ in $\overline{X^{\bullet}(\mathfrak{o})} \times^K \overline{Kt^{\check{\theta}}K}$ — in fact, substacks, in the appropriate setting. Indeed, stratifying $Kt^{\check{\theta}}K$ by the "Mirković-Vilonen (MV) cycles" corresponding to its intersection with the horocycles $Kt^{\check{\nu}}N(F)$, we obtain a stratification of this space by

$$\left(\overline{X^{\bullet}(\mathfrak{o})} \times^K (Kt^{\check{\nu}}N(F) \cap \overline{Kt^{\check{\theta}}K})\right)/B(\mathfrak{o}) \cap \operatorname{act}^{-1}(X(\mathfrak{o})/B(\mathfrak{o}))_{B:\check{\lambda}}$$

(where act denotes the action map). Note that $Kt^{\check{\nu}}N(F)/B(\mathfrak{o}) = K \times^{B(\mathfrak{o})} T(\mathfrak{o})t^{\check{\nu}}N(F)/B(\mathfrak{o})$, hence the above can also be written

$$\left(\overline{X^{\bullet}(\mathfrak{o})}\times^{B(\mathfrak{o})}(T(\mathfrak{o})t^{\check{\nu}}N(F)\cap\overline{Kt^{\check{\theta}}K})\right)/B(\mathfrak{o})\cap\mathrm{act}^{-1}(X(\mathfrak{o})/B(\mathfrak{o}))_{B:\check{\lambda}}.$$

An element of $\overline{X^{\bullet}(\mathfrak{o})}$, multiplied by $t^{\check{\nu}}N(F)$, lands in $(X(\mathfrak{o})/B(\mathfrak{o}))_{B:\check{\lambda}}$ if and only if it belongs to $\overline{X^{\bullet}(\mathfrak{o})}_{B:\check{\lambda}-\check{\nu}}$, showing that the sets corresponding to $Z^{\check{\lambda}-\check{\nu},\check{\lambda}}$ (and also to $Z^{\check{\lambda}-\check{\nu},\check{\lambda},\check{\theta}',\check{\eta}}$, after a further stratification by G-valuations) are indeed subspaces of $\overline{X^{\bullet}(\mathfrak{o})}\times^K \overline{Kt^{\check{\theta}}K}$.

Here comes the important geometric input: the topology on the affine Grassmannian, resp. on the arc space of X (or rather, its global models), implies that $Z^{\check{\lambda}-\check{\theta},\check{\lambda},0,\check{\theta}}$ is open and dense. This is a combination of the statements that $X^{\bullet}(\mathfrak{o})$ is (tautologically) open dense in $\overline{X^{\bullet}(\mathfrak{o})}$ (allowing us to take $\check{\theta}'=0$), that the Cartan double coset $Kt^{\check{\theta}}K$ is (again tautologically) open dense in its closure, and that the MV cycle $Kt^{\check{\theta}}K\cap Kt^{\check{\theta}}N(F)$ is open in it.

The open MV stratum on the affine Grassmannian $K\backslash G(F)$ meets the coset $K\backslash Kt^{\check{\theta}}K$ along the $N(\mathfrak{o})$ -orbit $K\backslash Kt^{\check{\theta}}N(\mathfrak{o})$. Therefore, the quotient

$$(T(\mathfrak{o})t^{\check{\theta}}N(F)\cap Kt^{\check{\theta}}K)/B(\mathfrak{o})$$

is just a point ($\check{\theta}$ is anti-dominant). Thus,

$$X^{\bullet}(\mathfrak{o})_{B:\check{\lambda}-\check{\theta}}\times^{B(\mathfrak{o})}(T(\mathfrak{o})t^{\check{\theta}}N(F)\cap Kt^{\check{\theta}}K)/B(\mathfrak{o})=(X^{\bullet}(\mathfrak{o})/B(\mathfrak{o}))_{B:\check{\lambda}-\check{\theta}},$$

which, on the other hand, can be identified as an open dense subset of $(X(\mathfrak{o})/B(\mathfrak{o}))_{G:\check{\theta}}$. Thus, in the geometric setting, the action map (5.8), restricted to the subsets with B-valuation equal to some (any) $\check{\lambda}$, is indeed birational. On the other hand, the map $\mathcal{Y}^{\check{\lambda}} \to \mathcal{M}_X$ is smooth for large $\check{\lambda}$ (Corollary 3.5.2), proving the birationality of (the geometric version of) (5.8).

5.5.2. We now turn to the actual proof of Theorem 5.1.1.

Since $\overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$ lives over $C^{\check{\Theta}},$ we can factor $\mathrm{act}_{\mathcal{M}}$ into

(5.9)
$$\mathcal{M}_{X} \tilde{\times} \overline{\mathrm{Gr}_{G,C^{\check{\Theta}}}} \stackrel{\mathrm{act}_{C^{\check{\Theta}}}}{\longrightarrow} \mathcal{M}_{X} \times C^{\check{\Theta}} \stackrel{\mathrm{pr}_{1}}{\longrightarrow} \mathcal{M}_{X}$$

where $\operatorname{act}_{C\check{\Theta}}$ is the naturally induced map over $C^{\check{\Theta}}$.

Lemma 5.5.3. Let $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$.

- (i) The preimage of $\mathcal{M}_X^{\check{\Theta}}$ in $\mathcal{M}_X \times \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$ under $\mathrm{act}_{\mathcal{M}}$ is contained in the open substack $\mathrm{Bun}_H \times \mathrm{Gr}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$.
- (ii) The image $\operatorname{act}_{\mathcal{M}}(\operatorname{Bun}_{H} \tilde{\times} \overline{\operatorname{Gr}}_{G,\mathring{C}^{\Theta}}) \subset \mathcal{M}_{X}$ contains $\mathcal{M}_{X}^{\check{\Theta}}$ as an open dense substack.

Proof. Statement (i) follows from the description of $\operatorname{act}_{\mathcal{M}}$ on k-points, Lemma 5.2.4, and the fact that no element of $\mathfrak{c}_X^- - 0$ is ≤ 0 .

To simplify notation, we give the proof of (ii) in the case when $\check{\Theta} = [\check{\theta}], \check{\theta} \neq 0$ is a singleton partition. The general case is entirely analogous.

Now we are considering $\operatorname{act}_C: \mathcal{M}_X \times \overline{\operatorname{Gr}}_{G,C}^{\check{\theta}} \to \mathcal{M}_X \times C$. Recall that we have a locally closed embedding $\mathcal{M}_X^{\check{\theta}} \hookrightarrow \mathcal{M}_X \times C$. First we show that $M := \operatorname{act}_C(\operatorname{Bun}_H \tilde{\times} \overline{\operatorname{Gr}}_{G,C}^{\check{\theta}}) \subset \mathcal{M}_X \times C$ contains $\mathcal{M}_X^{\check{\theta}}$. Fix $v \in |C|$. Take an arbitrary H-bundle $\mathcal{P}_H \in \operatorname{Bun}_H(k)$, and choose a trivialization of $\mathcal{P}_H|_{\operatorname{Spec}\,\mathfrak{o}_v}$, which induces a trivialization $\tau_0: (\mathcal{P}_H \times^H G)|_{\operatorname{Spec}\,\mathfrak{o}_v} \cong \mathcal{P}_G^0|_{\operatorname{Spec}\,\mathfrak{o}_v}$. Then $(\mathcal{P}_H, v, \tau_0)$ defines a point of $\widehat{\mathcal{M}}_X$. We also have the point $t^{\check{\theta}} \in \operatorname{Gr}_{G,v}^{\check{\theta}}$. The image of $(\mathcal{P}_H, v, \tau_0, t^{\check{\theta}})$ gives a point in $\mathcal{M}_X \tilde{\times} \operatorname{Gr}_{G,v}^{\check{\theta}}$, and by construction act_C will send this point to the stratum $\mathcal{M}_X^{\check{\theta}}$.

Thus we have shown that M contains a point of \mathcal{M}_X^{θ} over every point $v \in |C|$. It follows from Lemma A.4.7 that $M \subset \mathcal{M}_X \times C$ is stable under generic-Hecke modifications away from the marked point in C. Since these generic-Hecke modifications act transitively on the stratum $\mathcal{M}_X^{\check{\theta}}$ (Proposition A.4.5), we deduce that M contains all of $\mathcal{M}_X^{\check{\theta}}$.

The previous paragraph and Lemma 5.2.4 imply that

$$\mathfrak{M}_X^{\check{\theta}} = M - \bigcup_{\check{\theta}' \succ \check{\theta}} \operatorname{act}_C(\mathfrak{M}_X \times \overline{\operatorname{Gr}}_G^{\check{\theta}'}),$$

which shows that \mathcal{M}_X^{θ} is open in M.

Since $\overline{\operatorname{Gr}}_G^{\theta}$ is irreducible, for every connected component U of Bun_H , the image $M_U :=$ $\operatorname{act}_C(U \times \overline{\operatorname{Gr}}_{G,C}^{\theta})$ is irreducible. It is easy to see that its intersection with $\mathcal{M}_X^{\check{\theta}}$ is nonempty, hence dense.

As a corollary, we can now prove Theorem 5.1.1(i):

Proof of Theorem 5.1.1(i). Let M denote an irreducible component of $\overline{\mathcal{M}}_X^0$. Then M':= $\operatorname{act}_{\mathfrak{M}}(M \times \operatorname{\overline{Gr}}_{G,C}^{\Theta})$ is an irreducible closed substack of \mathfrak{M}_X , and since M contains a connected component of Bun_H , Lemma 5.5.3(ii) implies that $M' \cap \mathcal{M}_X^{\check{\Theta}}$ is dense in M'. It follows that $\mathcal{M}_X^{\check{\Theta}}$ is dense in $\operatorname{act}_{\mathcal{M}}(\overline{\mathcal{M}}_{X}^{0} \times \overline{\operatorname{Gr}}_{G,C}^{\theta})$

5.5.4. Base change to Zastava model. Fix $\check{\theta} \in \mathfrak{c}_X^- - 0$ and a point $v \in |C|$. Consider the restriction of $\operatorname{act}_{\mathcal{M}}$ to $\mathcal{M}_X \times \overline{\operatorname{Gr}}_{G,v}^{\theta} \to \mathcal{M}_X$, which is the fiber of act_C over $v \to C = C^{[\check{\theta}]}$.

Let us consider for $\check{\lambda} \in \mathfrak{c}_X$ the fiber product diagram

(5.10)
$$Z_{v}^{?,\check{\lambda}} \xrightarrow{\operatorname{act}_{y}} \mathcal{Y}^{\check{\lambda}}$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\overline{\mathcal{M}}_{X}^{0} \tilde{\times} \overline{\operatorname{Gr}}_{G,v}^{\check{\theta}} \xrightarrow{\operatorname{act}_{\mathcal{M}}} \mathcal{M}_{X}$$

An S-point of $Z_v^{?,\check{\lambda}}$ consists of the data $(\sigma, \mathcal{P}_G, \mathcal{P}_B', v, \tau)$ where

- $$\begin{split} \bullet \ \ & (\sigma, \mathcal{P}_G) \in \overline{\mathbb{M}}_X^0, \\ \bullet \ & \mathcal{P}_B' \in \operatorname{Bun}_B^{-\check{\lambda}} \text{ is a B-structure on a G-bundle } \mathcal{P}_G' := G \times^B \mathcal{P}_B', \end{split}$$
- $\tau: \mathcal{P}'_G|_{(C-v)\times S}\cong \mathcal{P}_G|_{(C-v)\times S}$ is a modification inducing a point in $\overline{\mathrm{Gr}}_G^{\check{\theta}}$ such that $\tau^{-1}\circ\sigma$ generically lands in $X^\circ\times^B\mathcal{P}'_B$.

Let $\widehat{\mathcal{Y}} \to \mathcal{Y}$ denote the Zariski locally trivial L⁺B-torsor parametrizing $(\sigma, \mathcal{P}_B) \in \mathcal{Y}$ and a trivialization $\mathcal{P}_B|_{\widehat{C}'_v} \cong \mathcal{P}_B^0|_{\widehat{C}'_v}$, and let $\widehat{\mathcal{Y}}_0 \to \mathcal{Y}_0$ be the restriction to $\overline{\mathcal{M}}_X^0$. (We will use the index 0 for the same purpose on the strata of y.)

Proposition-Construction 5.5.5.

(i) The fiber product $Z_v^{?,\lambda}$ admits a stratification by

$$Z_v^{\check{\lambda}-\check{\nu},\check{\lambda}}:=\widehat{\mathcal{Y}}_0\overset{\mathsf{L}^+B}{\times}(\mathsf{L}^+T\cdot t^{\check{\nu}}\cdot \mathsf{L}N\cap \overline{\mathsf{L}^+G\cdot t^{\check{\theta}}\cdot \mathsf{L}^+G})/\mathsf{L}^+B$$

where $\check{\nu}$ ranges over the weights of the irreducible \check{G} -module $V^{\check{\theta}}$

(ii) We have an isomorphism at the level of reduced schemes

$$Z_v^{\check{\lambda}-\check{\nu},\check{\lambda}} \cong \mathcal{Y}_0^{\check{\lambda}-\check{\nu}} \, \tilde{\times} (\mathsf{S}^{\check{\nu}} \cap \overline{\mathrm{Gr}}_G^{\check{\theta}}).$$

where $y_0^{\check{\lambda}-\check{\nu}} \tilde{\times} - means \, \hat{y}_0 \times^{\mathsf{L}^+B} - .$

(iii) The open stratum corresponds to $\check{\nu} = \check{\theta}$. More precisely, we have

$$Z_v^{\check{\lambda}-\check{\theta},\check{\lambda}} \cong \mathcal{Y}_0^{\check{\lambda}-\check{\theta}} \times v,$$

where v corresponds to the embedding $\{t^{\check{\theta}}\} \hookrightarrow \operatorname{Gr}_{G,v}^{\check{\theta}}$.

Proof. Recall that $\overline{\mathsf{L}^+G \cdot \mathsf{L}^{\check{\theta}} \cdot \mathsf{L}^+G}$ has a stratification by intersecting with $\mathsf{L}^+G \cdot \mathsf{L}^{\check{\nu}} \cdot \mathsf{L}N$. If we take the quotient on the left by L^+G , then under the identification $\mathsf{L}^+G \backslash \mathsf{L}G \cong \mathrm{Gr}_G : g \mapsto g^{-1}$, we have described above the stratification of $\overline{\mathrm{Gr}}_G^{-\check{\theta}}$ by MV cycles $\overline{\mathrm{Gr}}_G^{-\check{\theta}} \cap \mathsf{S}^{-\check{\nu}}$, where $-\check{\theta}$ is dominant. It is known by [MV07, Theorem 3.2] that $\overline{\mathrm{Gr}}_G^{-\check{\theta}} \cap \mathsf{S}^{-\check{\nu}}$ is non-empty precisely when $-\check{\nu}$ is a weight of $V^{-\check{\theta}}$, and the open stratum corresponds to when $-\check{\nu}$ equals $-\check{\theta}$.

Now let $(\sigma, \mathcal{P}_G, \mathcal{P}'_B, \tau)$ be an S-point of $Z_v^{?,\check{\lambda}}$. The restriction of $(\mathcal{P}_G, \mathcal{P}'_B, \tau)$ to \widehat{C}'_v gives a point in $\mathsf{L}^+G \setminus \mathsf{L}^+G \cdot \mathsf{L}^{\check{\theta}} \cdot \mathsf{L}^+G/\mathsf{L}^+B$. Therefore, by the previous paragraph, we can stratify $Z_v^{?,\check{\lambda}}$ by the preimages of

$$\mathsf{L}^+ G \backslash (\overline{\mathsf{L}^+ G \cdot t^{\check{\theta}} \cdot \mathsf{L}^+ G} \cap \mathsf{L}^+ G \cdot t^{\check{\nu}} \cdot \mathsf{L} N) / \mathsf{L}^+ B.$$

Suppose our S-point lies in such a stratum corresponding to $\check{\nu}$. In particular, τ corresponds to a point in $\mathsf{L}^+B\backslash(\mathrm{Gr}_B^{\check{\nu}})_{\mathrm{red}}$, which means that there exists a B-structure $\mathfrak{P}_B|_{\widehat{C}'_v}$ on $\mathfrak{P}_G|_{\widehat{C}'_v}$ such that τ gives an isomorphism of generic B-bundles $\tau: \mathfrak{P}'_B|_{\widehat{C}_v} \cong \mathfrak{P}_B|_{\widehat{C}_v}$. By Beauville–Laszlo's theorem, the datum $(\mathfrak{P}'_B|_{C-v}, \mathfrak{P}_B|_{\widehat{C}'_v}, \tau)$ descends to a B-structure \mathfrak{P}_B on \mathfrak{P}_G such that $\mathfrak{P}'_B|_{C-v} \cong \mathfrak{P}_B|_{C-v}$. Then $(\sigma, \mathfrak{P}_B) \in \mathfrak{P}_0^{\check{\lambda}-\check{\nu}}(S)$ and $(\sigma, \mathfrak{P}_B, \mathfrak{P}'_B, v, \tau) \in Z_v^{\check{\lambda}-\check{\nu},\check{\lambda}}(S)$. The procedure above can be reversed to see that $Z_v^{\check{\lambda}-\check{\nu},\check{\lambda}}$ is equal to the entire stratum. This shows (i).

Observe that $\mathsf{L}^+T \cdot t^{\check{\nu}} \cdot \mathsf{L}N = \mathsf{L}N \cdot t^{\check{\nu}} \cdot \mathsf{L}^+B$ and there is a map from $(\mathsf{L}N \cdot t^{\check{\nu}} \cdot \mathsf{L}^+B \cap \overline{\mathsf{L}^+G \cdot t^{\check{\theta}} \cdot \mathsf{L}^+G})/\mathsf{L}^+B$ to $\mathsf{S}^{\check{\nu}} \cap \overline{\mathrm{Gr}}_G^{\check{\theta}}$ which is an isomorphism at the level of reduced schemes. Now (ii) follows from (i).

The open stratum $Z_v^{\check{\lambda}-\check{\theta},\check{\lambda}}$ corresponds to the open stratum $\overline{\mathrm{Gr}}_G^{-\check{\theta}}\cap \mathsf{S}^{-\check{\theta}}$, when $\check{\nu}=\check{\theta}$. By [MV07, (3.6)], we have $\mathsf{S}^{\check{\theta}}\cap \overline{\mathrm{Gr}}_G^{\check{\theta}}=\{t^{\check{\theta}}\}$ so (iii) is a special case of (ii).

Observe that $\overline{\mathcal{M}}_X^0 \times \overline{\mathrm{Gr}}_{G,v}^{\check{\theta}}$ has a stratification by $\mathcal{M}_X^{\check{\Theta}'} \times \overline{\mathrm{Gr}}_{G,v}^{\check{\eta}}$ for those $\check{\Theta}' \in \mathrm{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$ such that the stratum $\mathcal{M}_X^{\check{\Theta}'}$ belongs to $\overline{\mathcal{M}}_X^0$ — still to be determined, see Lemma 5.6.3 — and those $\check{\eta} \in \check{\Lambda}_G^-$ such that $\check{\eta} \geq \check{\theta}$ (equivalently, $\check{\eta}$ is a weight of $V^{\check{\theta}}$). Let $Z_v^{?,\check{\lambda},\check{\Theta}',\check{\eta}}$ denote the preimage of the corresponding stratum in $Z_v^{?,\check{\lambda}}$, so we have a Cartesian square

$$Z_{v}^{?,\check{\lambda},\check{\Theta}',\check{\eta}} \xrightarrow{\operatorname{act}_{y}} \mathcal{Y}_{X}^{\check{\lambda}}$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\mathcal{M}_{X}^{\check{\Theta}'} \tilde{\times} \operatorname{Gr}_{G,v}^{\check{\eta}} \xrightarrow{\operatorname{act}_{\mathcal{M}}} \mathcal{M}_{X}$$

This diagram now has no dependence on $\check{\theta}$ and is defined entirely with respect to $\check{\eta}$. Proposition 5.5.5 implies that there is a stratification

$$Z_v^{?,\check{\lambda},\check{\Theta}',\check{\eta}} = \bigcup_{\check{\nu}} Z_v^{\check{\lambda}-\check{\nu},\check{\lambda},\check{\Theta}',\check{\eta}}$$

where $\check{\nu}$ runs through the weights of $V^{\check{\eta}}$, and $Z^{\check{\lambda}-\check{\nu},\check{\lambda},\check{\Theta}',\check{\eta}}$ admits a map to $\mathcal{Y}^{\check{\lambda}-\check{\nu},\check{\Theta}'}$. The open stratum is

$$Z_v^{\check{\lambda}-\check{\eta},\check{\lambda},\check{\Theta}',\check{\eta}} \cong \mathcal{Y}^{\check{\lambda}-\check{\eta},\check{\Theta}'} \times \{t^{\check{\eta}}\}.$$

We will make special use of the case $\check{\Theta}'=0$ and $\check{\eta}=\check{\theta}$ because Lemma 5.5.3 implies that $\mathcal{Y}^{\check{\lambda},\check{\theta}}$ is contained in the images of $Z_v^{?,\check{\lambda},0,\check{\theta}}\to\mathcal{Y}^{\check{\lambda}}$ over all $v\in |C|$. Note that $Z_v^{\check{\lambda}-\check{\theta},\check{\lambda},0,\check{\theta}}=\mathcal{Y}^{\check{\lambda}-\check{\theta},0}$. Recall that $\mathfrak{C}_X^{\mathcal{D}}$ denotes the monoid generated by $\check{\nu}_D$ for $D\in\mathcal{D}$.

Corollary 5.5.6. Let $\check{\theta} \in \mathfrak{c}_X^- - 0$.

- (i) The scheme $\mathcal{Y}^{\check{\lambda},\check{\theta}}$ is nonempty only if $\check{\lambda} \succeq \check{\theta}$.
- (ii) We have an isomorphism $(\mathcal{Y}^{\check{\theta},\check{\theta}})_{\mathrm{red}} \cong C$ over the diagonal $C \hookrightarrow \mathcal{A}^{\check{\theta}}$.
- (iii) The single point in the central fiber $Y^{\check{\theta},\check{\theta}}$ corresponds to $t^{\check{\theta}} \in S^{\check{\theta}} \cap Gr_{G}^{\check{\theta}}$

Proof. As we remarked above, every point of $\mathcal{Y}^{\check{\lambda},\check{\theta}}$ is contained in the image of $Z^{?,\check{\lambda},0,\check{\theta}}_v$ for some $v \in |C|$. By the description of the stratification of $Z^{?,\check{\lambda},0,\check{\theta}}_v$, the latter is nonempty only if $\mathcal{Y}^{\check{\lambda}-\check{\nu},0}$ is nonempty for some $\check{\nu}$. This implies that $\check{\lambda} \succeq \check{\nu}$ by Lemma 5.4.1. Since $\check{\nu} \geq \check{\theta}$, we deduce (i).

If $\check{\lambda} = \check{\theta}$, then $\mathcal{Y}^{\check{\theta} - \check{\nu}, 0}$ is only nonempty when $\check{\nu} = \check{\theta}$, in which case $\mathcal{Y}^0 = \mathsf{pt}$ and $Z^{0,\check{\theta},0,\check{\theta}}_v = \{t^{\check{\theta}}\}$. By moving the point v around, we get a surjection $C \to (\mathcal{Y}^{\check{\theta},\check{\theta}})_{\mathrm{red}}$ and it must be an isomorphism (on reduced schemes) since the composition with $\pi : \mathcal{Y}^{\check{\theta},\check{\theta}} \to \mathcal{A}^{\check{\theta}}$ gives the diagonal embedding.

Let $y^{\check{\lambda}-\check{\theta},0} \mathring{\times} v$ denote the disjoint locus $y^{\check{\lambda}-\check{\theta},0} \mathring{\times} Y_v^{\check{\theta},\check{\theta}}$. Observe that the composition

$$\mathcal{Y}^{\check{\lambda}-\check{\theta}}\overset{\circ}{\times}v\hookrightarrow\mathcal{Y}^{\check{\lambda}-\check{\theta}}\times v=Z_v^{\check{\lambda}-\check{\theta},\check{\lambda}}\overset{\mathrm{act}}{\longrightarrow}\mathcal{Y}^{\check{\lambda}}$$

coincides with the composition $\mathcal{Y}^{\check{\lambda}-\check{\theta}}\mathring{\times}v\hookrightarrow\mathcal{Y}^{\check{\lambda}-\check{\theta}}\mathring{\times}\mathcal{Y}^{\check{\theta},\check{\theta}}\to\mathcal{Y}^{\check{\lambda}}$ where the second map is given by the graded factorization property.

Lemma 5.5.7. Let $\check{\theta} \in \mathfrak{c}_X^- - 0$. Then:

- (i) The open embedding $\mathcal{Y}^{\check{\lambda}-\check{\theta},0}\mathring{\times}\mathcal{Y}^{\check{\theta},\check{\theta}}\hookrightarrow\mathcal{Y}^{\check{\lambda},\check{\theta}}$ given by the graded factorization property is dense.
- (ii) For $\check{\lambda}$ large enough, the open stratum $\mathcal{Y}^{\check{\lambda}-\check{\theta},0} \times v \hookrightarrow Z_v^{?,\check{\lambda},0,\check{\theta}}$ is dense.

Proof. First observe that by Lemma 3.5.4 we may always assume that $\check{\lambda}$ is large enough so that the conditions of Corollary 3.5.2 hold: namely, $\mathcal{Y}^{\check{\lambda}} \to \mathcal{M}_X$ is smooth with connected fibers.

By definition, $\mathcal{Y}^{\check{\lambda},\check{\theta}}$ is the preimage of $\mathcal{M}_X^{\check{\theta}}$. Now let V be a connected component of $\mathcal{M}_X^{\check{\theta}}$ that intersects the image of $\mathcal{Y}^{\check{\lambda}}$. Then $\mathcal{Y}_V^{\check{\lambda},\check{\theta}}:=\mathcal{Y}^{\check{\lambda}}\times_{\mathcal{M}_X}V$ is a nonempty connected component of the smooth scheme $\mathcal{Y}^{\check{\lambda},\check{\theta}}$. Lemma 5.5.3 implies that V lies in $\mathrm{act}_{\mathcal{M}}(U\,\check{\times}\,\mathrm{Gr}_{G,C}^{\check{\theta}})$, where U is a connected component of Bun_H . Consider the fiber product diagram

(5.12)
$$Z^{?,\check{\lambda},0,\check{\theta}} \longrightarrow \mathcal{Y}^{\check{\lambda}} \qquad \qquad \downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\operatorname{Bun}_{H} \tilde{\times} \operatorname{Gr}_{G,C}^{\check{\theta}} \longrightarrow \mathcal{M}_{X}$$

which is the analog of (5.11) where we allow v to vary. The fibers of $\mathcal{Y}^{\check{\lambda}} \to \mathcal{M}_X$ are irreducible, so $Z_U^{?,\check{\lambda},0,\check{\theta}} := Z^{?,\check{\lambda},0,\check{\theta}} \times_{\operatorname{Bun}_H} U$ is irreducible and the image of

$$Z_U^{?,\check{\lambda},0,\check{\theta}} \to \mathcal{Y}^{\check{\lambda}}$$

contains $\mathcal{Y}_{V}^{\check{\lambda},\check{\theta}}$ as a dense open. Proposition 5.5.5 (twisted by C) implies that $Z_{U}^{?,\check{\lambda},0,\check{\theta}}$ has a stratification $\bigcup_{\check{\nu}} Z_{U}^{\check{\lambda}-\check{\nu},\check{\lambda},0,\check{\theta}}$ over the weights $\check{\nu}$ of $V^{\check{\theta}}$, and $Z_{U}^{\check{\lambda}-\check{\nu},\check{\lambda},0,\check{\theta}}$ maps to $\mathcal{Y}^{\check{\lambda}-\check{\nu},0} \times_{\operatorname{Bun}_{H}} U$. In particular, $\mathcal{Y}_{U}^{\check{\lambda}-\check{\nu},0} := \mathcal{Y}^{\check{\lambda}-\check{\nu},0} \times_{\operatorname{Bun}_{H}} U$ is nonempty for some $\check{\nu}$. Since $\check{\nu} \geq \check{\theta}$, Corollary 3.5.2(ii) implies that $\mathcal{Y}_{U}^{\check{\lambda}-\check{\theta},0}$ is also nonempty. Therefore, the open stratum $Z_{U}^{\check{\lambda}-\check{\theta},\check{\lambda},0,\check{\theta}} \cong \mathcal{Y}_{U}^{\check{\lambda}-\check{\theta},0} \times C$ is nonempty. By Corollary 5.5.6, we can identify $C \cong (\mathcal{Y}^{\check{\theta},\check{\theta}})_{\mathrm{red}}$, and the map

$$\mathcal{Y}^{\check{\lambda}-\check{\theta},0}\mathring{\times}C\hookrightarrow Z^{?,\check{\lambda},0,\check{\theta}}\to\mathcal{Y}^{\check{\lambda}}$$

coincides with the open embedding $\mathcal{Y}^{\check{\lambda}-\check{\theta},0}\mathring{\times}\mathcal{Y}^{\check{\theta},\check{\theta}}\hookrightarrow\mathcal{Y}^{\check{\lambda},\check{\theta}}$ given by graded factorization. Then $\mathcal{Y}_U^{\check{\lambda}-\check{\theta}}\mathring{\times}^c C$ is a nonempty open subscheme of the irreducible component $\mathcal{Y}_U^{\check{\lambda},\check{\theta}}$, so it must be dense.

To show (ii), note that a priori the preimage of $\mathcal{Y}_{V}^{\check{\lambda},\check{\theta}}$ in $Z^{?,\check{\lambda},0,\check{\theta}}$ is contained in a finite union of $Z_{U}^{?,\check{\lambda},0,\check{\theta}}$ for connected components U of Bun_{H} . However we saw above that the open stratum $\mathcal{Y}_{U}^{\check{\lambda}-\check{\theta},0}\times C$ is nonempty, and therefore dense in $Z_{U}^{?,\check{\lambda},0,\check{\theta}}$, for each such U.

In the proof above, we have essentially shown Theorem 5.1.1(ii) along the way:

Proof of Theorem 5.1.1(ii). To simplify notation, we only show the case $\check{\Theta} = [\check{\theta}]$. The multipoint version is proved in exactly the same way. We continue to use the notation from the previous proof of Lemma 5.5.7. By Corollary 3.5.2 and the base change diagram (5.12), it is enough to show, by restricting to open dense subsets, that

$$Z^{?,\check{\lambda},0,\check{\theta}} o \mathcal{Y}^{\check{\lambda}} \underset{\mathcal{M}_X}{ imes} \overline{\mathcal{M}}_X^{\check{\theta}}$$

is birational for $\check{\lambda}$ large enough. It follows from Lemma 5.5.7 that $\mathcal{Y}^{\check{\lambda}-\check{\theta},0}\mathring{\times}C$ is a dense open in both the target and source.

Proof of Theorem 5.1.1(iii). Again to simplify notation we only show the case $\check{\Theta} = [\check{\theta}]$. By Zariski's Main Theorem, it suffices to show that the restriction of $\operatorname{act}_{\mathfrak{M}}$ to

$$(5.13) \operatorname{act}_{\mathcal{M}}^{-1}(\mathcal{M}_{X}^{\check{\theta}}) \cap (\operatorname{Bun}_{H} \tilde{\times} \operatorname{Gr}_{G,C}^{\check{\theta}}) \to \mathcal{M}_{X}^{\check{\theta}}$$

is a bijection on k-points. By Lemma A.4.7, this map is equivariant with respect to generic-Hecke modifications away from the marked point in C. Fix a marked point $v \in C$ and let $\mathcal{M}_{X,v}^{\tilde{\theta}}$ denote the preimage of v under the projection $\mathcal{M}_{X}^{\tilde{\theta}} \to C$. By Proposition A.4.5, all the k-points of $\mathcal{M}_{X,v}^{\tilde{\theta}}$ are equivalent under the equivalence relation generated by generic-Hecke correspondences. Thus it suffices to determine the fiber of (5.13) at the single point $\{t_v^{\tilde{\theta}}\} = \mathsf{Y}^{\tilde{\theta},\tilde{\theta}}$. By Proposition 5.5.5, the fiber of $t_v^{\tilde{\theta}}$ has a stratification by $\mathsf{Y}^{\tilde{\theta}-\tilde{\nu},0} \tilde{\times} (\mathsf{S}^{\tilde{\nu}} \cap \overline{\mathsf{Gr}}_G^{\tilde{\theta}})$ for $\tilde{\nu}$ a weight of $V^{\tilde{\theta}}$. On the other hand, Corollary 5.5.6 implies that $\mathsf{Y}^{\tilde{\theta}-\tilde{\nu},0}$ is nonempty only if $\tilde{\nu}=\tilde{\theta}$. In this case $\mathsf{Y}^{0,0} \times (\mathsf{S}^{\tilde{\theta}} \cap \overline{\mathsf{Gr}}_G^{\tilde{\theta}})$ is a point.

5.5.8. Recall that for an arbitrary algebraic group H, the algebraic fundamental group $\pi_1(H)$ is defined as the quotient of the coweight lattice by the coroot lattice of the reductive group $H/H_{\rm u}$, where $H_{\rm u}$ is the unipotent radical of H. For $\check{\mu} \in \pi_1(H)$, let ${\rm Bun}_H^{\check{\mu}}$ denote the corresponding connected component of ${\rm Bun}_H$ and let ${}^{\check{\mu}}\overline{\mathbb{M}}_X^0$ be its closure in \mathbb{M}_X .

Corollary 5.5.9. Let $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$. Then the set of irreducible components of $\overline{\mathcal{M}}_X^{\check{\Theta}}$ is in bijection with $\pi_1(H)$, where $\check{\mu} \in \pi_1(H)$ corresponds to

$${}^{\check{\mu}}\overline{\mathfrak{M}}_{X}^{\check{\Theta}} := \operatorname{act}_{\mathfrak{M}}({}^{\check{\mu}}\overline{\mathfrak{M}}_{X}^{0} \, \tilde{\times} \, \overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}).$$

Proof. It follows from Theorem 5.1.1 that the irreducible components of $\overline{\mathbb{M}}_X^{\check{\Theta}}$ are in bijection with the irreducible components of $\operatorname{Bun}_H \tilde{\times} \overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$. Since $\overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$ is irreducible, the latter are in bijection with $\pi_0(\operatorname{Bun}_H) = \pi_1(H)$.

We will let ${}^{\check{\mu}}\mathcal{M}_X^{\check{\Theta}} := {}^{\check{\mu}}\overline{\mathcal{M}}_X^{\check{\Theta}} \cap \mathcal{M}_X^{\check{\Theta}}$ denote the corresponding connected component of $\mathcal{M}_X^{\check{\Theta}}$ (which is smooth).

5.5.10. $Mirković-Vilonen\ cycles$. We finish this subsection by proving a result that will be used in the following sections. The goal here is to show that the Mirković-Vilonen cycles in the $\check{\theta}$ -stratum of the affine Grassmannian map, generically, to the $\check{\theta}$ -stratum of the global model of X, under the action map.

Fix $v \in |C|$ and $\check{\theta} \in \mathfrak{c}_X^- - 0$. Consider the restriction of $\operatorname{act}_{\mathfrak{M},v}$ to

(5.14)
$$\operatorname{pt} \times \overline{\operatorname{Gr}_{G,v}^{\check{\theta}}} \to \operatorname{Bun}_{H} \tilde{\times} \overline{\operatorname{Gr}_{G,v}^{\check{\theta}}} \to \overline{\mathcal{M}}_{X}^{\check{\theta}},$$

where $pt \to Bun_H$ corresponds to the trivial H-bundle. Note that the map (5.14) above can be extended to a map

$$(5.15) \operatorname{act}_{v} : \operatorname{Gr}_{G} \underset{\mathsf{L}X/\mathsf{L}^{+}G}{\times} (\mathsf{L}^{+}X/\mathsf{L}^{+}G) \to \mathfrak{M}_{X}$$

using Beauville–Laszlo's theorem: a point of the left hand side consists of a G-bundle \mathcal{P}_G and a trivialization $\tau: \mathcal{P}_G|_{C-v} \cong \mathcal{P}_G^0|_{C-v}$ such that $\tau^{-1} \circ x_0: C-v \to X \times^G \mathcal{P}_G$ is regular when localized at \mathfrak{o}_v . Here $x_0: C \to X \times^G \mathcal{P}_G^0 = X \times C$ denotes the section corresponding to the base point $x_0 \in X^\circ$. Thus, $\operatorname{act}_v(\mathcal{P}_G, \tau) := (\mathcal{P}_G, \tau^{-1} \circ x_0) \in \mathcal{M}_X^{\check{\theta}}$ is well-defined.

Define $\operatorname{Gr}_G^{\check{e}} \subset \operatorname{Gr}_G^{\check{e}}$ to be the open subscheme equal to the preimage of the stratum $\mathcal{M}_X^{\check{e}}$ under (5.14). We can also identify $\operatorname{Gr}_G^{\check{e}} = \operatorname{Gr}_{G L X/L+G}^{\check{e}} \times (\mathsf{L}^{\check{e}} X/\mathsf{L}^+ G)$.

Taking central fibers with respect to v, the restriction of act_v to a semi-infinite orbit factors through

$$\mathsf{S}^{\check{\lambda}} \cap \overline{\mathrm{Gr}}_{G}^{\check{\theta}} \hookrightarrow \mathsf{Y}^{\check{\lambda}} \underset{\mathcal{M}_{X}}{\times} \overline{\mathcal{M}}_{X}^{\check{\theta}},$$

where $\check{\lambda}$ is a weight of $V^{\check{\theta}}$ and we consider $(\mathsf{Y}^{\check{\lambda}})_{\mathrm{red}}$ as a subscheme of $\mathsf{S}^{\check{\lambda}}$ via Lemma 4.3.2. Note that $\mathsf{S}^{\check{\lambda}} \cap \overline{\mathrm{Gr}}_G^{\check{\theta}}$ is isomorphic to the *closed* stratum $Z_v^{0,\check{\lambda}}$ from Proposition 5.5.5. Under this identification (5.16) coincides with the restriction of the map $\mathrm{act}_{\mathcal{Y}}: Z_v^{?,\check{\lambda}} \to \mathcal{Y}^{\check{\lambda}}$ from (5.10).

Observe that $\mathsf{S}^{\check{\theta}} \cap \overline{\mathrm{Gr}}_G^{\check{\theta}} = \{t^{\check{\theta}}\}\$ is contained in $\mathrm{Gr}_G^{\check{\theta}}$. We will use this to deduce:

Lemma 5.5.11. Let $\check{\lambda}, \check{\theta}$ as above. Then $S^{\check{\lambda}} \cap \overset{\circ}{\mathrm{Gr}}_{G}^{\check{\theta}}$ intersects every irreducible component of $S^{\check{\lambda}} \cap \overline{\mathrm{Gr}}_{G}^{\check{\theta}}$.

Proof. Let Z denote an irreducible component of $S^{\check{\lambda}} \cap \overline{\mathrm{Gr}}_{G}^{\check{\theta}}$, which must be of dimension $d = \langle \rho_{G}, \check{\lambda} - \check{\theta} \rangle$. Let \overline{Z} denote its closure in $\overline{S}^{\check{\lambda}} \cap \overline{\mathrm{Gr}}_{G}^{\check{\theta}}$. Then the proof of [MV07, Theorem 3.2], which we briefly recall in the next paragraph, shows that \overline{Z} contains $t^{\check{\theta}} \in \mathring{\mathrm{Gr}}_{G}^{\check{\theta}}$. Thus, $\overline{Z} \cap \mathring{\mathrm{Gr}}_{G}^{\check{\theta}}$ is open and nonempty, hence dense in \overline{Z} .

Since the boundary of $S^{\check{\lambda}}$ in $\overline{S}^{\check{\lambda}}$ is a hyperplane section (Proposition 4.4.1), $\overline{Z} - Z$ contains an irreducible component of dimension d-1 inside $S^{\check{\lambda}_1} \cap \overline{\operatorname{Gr}}_G^{\check{\theta}}$ for $\check{\lambda}_1 < \check{\lambda}$. In this way we produce a sequence $\check{\lambda} = \check{\lambda}_0, \ldots, \check{\lambda}_d$ and irreducible components of $S^{\check{\lambda}_i} \cap \overline{Z}$ of dimension d-i. The only weight $\check{\lambda}_d$ of $V^{\check{\theta}}$ such that $\langle \rho_G, \check{\lambda} - \check{\lambda}_d \rangle \geq d$ is $\check{\lambda}_d = \check{\theta}$, so \overline{Z} must contain $S^{\check{\theta}} \cap \overline{\operatorname{Gr}}_G^{\check{\theta}} = \{t^{\check{\theta}}\}$. \square

5.6. Closure relations and components in the global model. Let $\check{\Theta}, \check{\Theta}' \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X}^{-}-0)$. Consider $\check{\Theta}' - \check{\Theta}$ as a formal sum in $\bigoplus_{\check{\theta} \in \mathfrak{c}_{X}^{-}-0} \mathbb{Z}[\check{\theta}]$. We say that

$$\check{\Theta}' \succ \check{\Theta}$$

if $\check{\Theta}' - \check{\Theta}$ can be written as a sum of formal differences $[\check{\theta}'] - [\check{\theta}]$ where $\check{\theta}, \check{\theta}' \in \mathfrak{c}_X^-$ and $\check{\theta}' \succeq \check{\theta}$. Note that we allow $\check{\theta} = 0$ (in which case [0] = 0), and for general $\check{\theta}, \check{\theta}'$ as above, it is not necessarily the case that $\check{\theta} - \check{\theta}' \in \mathcal{V}$.

Proposition 5.6.1. Let $\check{\Theta}, \check{\Theta}' \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X}^{-} - 0)$. We have that $\mathfrak{M}_{X}^{\check{\Theta}'}$ lies in the closure of $\mathfrak{M}_{X}^{\check{\Theta}}$ if and only if there exists $\check{\Theta}'' \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X}^{-} - 0)$ such that $\check{\Theta}$ refines $\check{\Theta}''$ and $\check{\Theta}' \succeq \check{\Theta}''$.

Remark 5.6.2. We warn that our notation in Corollary 5.5.9 is unfortunately not compatible with the closure relations in the following sense: it is possible that ${}^{\check{\mu}}\overline{\mathcal{M}}_X^{\check{\Theta}}\cap \mathcal{M}_X^{\check{\Theta}'} \neq {}^{\check{\mu}}\mathcal{M}_X^{\check{\Theta}'}$ for $\check{\Theta}' \succ \check{\Theta} \text{ and } \check{\mu} \in \pi_1(H).$

The reader may want to skip the proof of this proposition at first reading, and focus on the corollaries that follow. Note that the proof will use Zastava models, despite the fact that the statement is about the global model. The proof of the proposition starts with the following special case:

Lemma 5.6.3. The closure of \mathfrak{M}_X^0 in \mathfrak{M}_X intersects $\mathfrak{M}_X^{\check{\Theta}}$ only if $\check{\Theta} \succeq 0$.

Proof. Let $\check{\Theta}$ correspond to a stratum such that $\mathcal{M}_X^{\check{\Theta}}$ intersects $\overline{\mathcal{M}}_X^0$. Then by Corollary 3.5.2 there exists $\check{\lambda}$ such that $\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ intersects the closure of $\mathcal{Y}^{\check{\lambda},0}$ in $\mathcal{Y}^{\check{\lambda}}$. Now consider the torsor $H'\backslash G'\to X^{\bullet}$ from Lemma 5.3.3. Let $X'=\operatorname{Spec} k[H'\backslash G']$ considered as an affine G'-variety. The corresponding compactified Zastava model $\overline{\mathcal{Y}}_{X'}^D \to \mathcal{A}_{X'}^D$ is indexed by $D \in \mathfrak{c}_{X'} = \mathbb{N}^D$. Since $X'^{\bullet}/B' \cong X^{\bullet}/B$ as stacks, $\mathcal{Y}_{X}^{\check{\lambda},0}$ is a disjoint union of $\mathcal{Y}_{X^{\bullet}}^{D} = \mathcal{Y}_{X'}^{D,0}$ ranging over all $D \in \mathbb{N}^{D}$ such that $\varrho_{X}(D) = \check{\lambda}$ (see §5.4). Choose $D \in \mathbb{N}^{D}$ such that the closure of $\mathcal{Y}_{X^{\bullet}}^{D}$ in $\mathcal{Y}_{X}^{\check{\lambda}}$ intersects the stratum $\mathcal{Y}_{X}^{\check{\lambda},\check{\Theta}}$. The map

$$\overline{\mathcal{Y}}_{X'}^{D} \to \overline{\mathcal{Y}}_{X}^{\check{\lambda}}$$

is proper because $\overline{\mathcal{Y}}_{X'}^D, \overline{\mathcal{Y}}_X^{\check{\lambda}}$ are proper over $\mathcal{A}_{X'}^D, \mathcal{A}_X^{\check{\lambda}}$, respectively, and the natural map $\mathcal{A}_{X'}^D \to \mathcal{A}_X^{\check{\lambda}}$ is proper. Therefore, the closure of $\mathcal{Y}_{X\bullet}^D$ in $\mathcal{Y}_X^{\check{\lambda}}$ is contained in the image of (5.17). Note that the stratification of $\mathcal{M}_{X'}$ is indexed by $\check{\Theta}' \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X'}^- - 0)$. In particular, $\check{\Theta}' \succeq 0$ since $\mathfrak{c}_{X'} = \mathbb{N}^{\mathcal{D}}$. Therefore, $\overline{\mathcal{Y}}_{X'}^{D}$ is a union of $\overline{\mathcal{Y}}_{X'}^{D} \times_{\mathfrak{M}_{X'}} \mathfrak{M}_{X'}^{\check{\Theta}'}$ for $\check{\Theta}' \succeq 0$, which implies its image in $\overline{\mathcal{Y}}_X^{\check{\Lambda}}$ is contained in the union of $\overline{\mathcal{Y}}_X^{\check{\Lambda}} \times_{\mathcal{M}_X} \mathcal{M}_X^{\check{\Theta}}$ for $\check{\Theta} \succeq 0$.

Proof of Proposition 5.6.1. By Theorem 5.1.1(i), the closure of $\mathcal{M}_X^{\check{\Theta}}$ is equal to the image of $\overline{\mathcal{M}}_{X}^{0} \tilde{\times} \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\Theta}$, so we will consider the latter. Note that $C^{\check{\Theta}}$ is stratified by disjoint loci $\mathring{C}^{\check{\Theta}''}$ for all partitions $\check{\Theta}''$ such that $\check{\Theta}$ refines $\check{\Theta}''$. By the description of the fibers of $\mathrm{Gr}_{C,C\check{\Theta}}^{\check{\Theta}} \to C^{\check{\Theta}}$ in (5.3), we have an identification

$$\overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}} \underset{C^{\check{\Theta}}}{\times} \mathring{C}^{\check{\Theta}''} = \overline{\mathrm{Gr}}_{G,\mathring{C}^{\check{\Theta}''}}^{\check{\Theta}''}$$

at the level of reduced schemes. Therefore, replacing $\check{\Theta}$ by $\check{\Theta}''$, it suffices to show that the image of $\overline{\mathcal{M}}_X^0 \times \overline{\mathrm{Gr}}_{G,\mathring{C}}^{\check{\Theta}}$ contains $\mathcal{M}_X^{\check{\Theta}'}$ if and only if $\check{\Theta}' \succeq \check{\Theta}$. The "only if" direction follows from the description of $\mathrm{act}_{\mathcal{M}}$ on k-points and Lemmas 5.2.4

and 5.6.3.

We will show the "if" direction only in the case when $\check{\Theta} = [\check{\theta}], \check{\Theta}' = [\check{\theta}']$ are singleton (so $\check{\theta}' \succeq \check{\theta}$) to lessen notation (allowing $\check{\theta} = 0$), but the multi-point version is proved in exactly the same way. Fix $v \in |C|$. We have a distinguished point in $\mathcal{M}_X^{\check{\theta}'}$ degenerate at v: the image under act_v of $t^{\check{\theta}'} \in \mathsf{Y}^{\check{\theta}',\check{\theta}'}$. We will show that $\overline{\mathcal{M}}_X^{\check{\theta}}$ contains this point. Then, stability of $\overline{\mathcal{M}}_X^{\check{\theta}}$ under generic-Hecke correspondences (Theorem 5.1.1(i) and Lemma A.4.7) and Proposition A.4.5 imply that $\overline{\mathcal{M}}_X^{\check{\theta}}$ contains all of $\mathcal{M}_{X_v}^{\check{\theta}'}$.

Since $\check{\theta}' \succeq \check{\theta}$, we can decompose $\check{\theta}' - \check{\theta} = \sum_{j=1}^d \check{\nu}_j$ for (not necessarily distinct) $\check{\nu}_j$ equal to valuations of colors $D_j \in \mathcal{D}$ (in case D_j is not uniquely determined by its valuation, the choice of D_j is arbitrary). The graded factorization property gives a map $\mathring{\mathbb{C}} := \mathcal{Y}^{\check{\theta},\check{\theta}} \mathring{\times} \mathring{\prod}_j \mathcal{Y}_{X^{\bullet}}^{D_j} \to \mathcal{Y}^{\check{\theta}'}$. We claim that the image of $\mathring{\mathbb{C}}$ contains $C = \mathcal{Y}_{\mathrm{red}}^{\check{\theta}',\check{\theta}'}$ in its closure. (Note that $\mathring{\mathbb{C}}_{\mathrm{red}} = \mathring{\mathbb{C}}^{d+1}$.) This will produce an irreducible variety whose generic point maps to $\mathcal{M}_X^{\check{\theta}}$ while a special point maps to $\mathrm{act}_v(t^{\check{\theta}'}) \in \mathcal{M}_X^{\check{\theta}'}$.

Consider the Beilinson–Drinfeld affine Grassmannian $\operatorname{Gr}_{B,C^{d+1}}$, whose fiber over d+1 pairwise distinct points $(v_0,\ldots,v_d)\in \mathring{C}^{d+1}$ is $\prod_{j=0}^d\operatorname{Gr}_B$ while the fiber over $v_0=\ldots=v_d$ is Gr_B . By Lemma 4.3.2, we have an isomorphism $\mathsf{Y}^{\check{\lambda}}\cong\operatorname{Gr}_B^{\check{\lambda}}\times_{\mathsf{L}X/\mathsf{L}^+B}(\mathsf{L}^+X/\mathsf{L}^+B)$, which depends on a fixed base point $x_0\in X$. Now a reduction to $\mathbb{G}_m\backslash\operatorname{GL}_2$ (see proof of Lemma 5.4.2 and Example 4.3.3) shows that the central fiber $\mathsf{Y}_{X^{\bullet}}^{D_j}=\mathsf{pt}$ is contained in $\mathsf{L}^+N\cdot t^{\check{\nu}_j}\subset\operatorname{Gr}_B^{\nu_j}$. Therefore, $\mathring{\mathbb{C}}$ is contained in the orbit of the multi-point jet space $(\mathcal{L}^+N)_{C^{d+1}}$ acting on the closed subscheme $C^{d+1}\subset\operatorname{Gr}_{T,C^{d+1}}\subset\operatorname{Gr}_{B,C^{d+1}}$ given by $(v_0,\ldots,v_d)\mapsto t_{v_0}^{\check{\nu}}\prod_j t_{v_j}^{\check{\nu}_j}$, where the v_j 's are allowed to collide. The orbit of $(v_0,\ldots,v_d)\in \mathring{C}^{d+1}$ is

$$\{t_{v_0}^{\check{\theta}}\} \times \prod_{j=1}^d (\mathsf{L}^+ N \cdot t_{v_j}^{\check{\nu}_j}) \subset \mathrm{Gr}_{B,v_0}^{\check{\theta}} \times \prod_{j=1}^d \mathrm{Gr}_{B,v_j}^{\check{\nu}_j},$$

while the orbit of the diagonal $v=v_0=\ldots=v_d$ is $\mathsf{L}^+N\cdot t_v^{\check\theta'}=\{t_v^{\check\theta'}\}\subset \mathrm{Gr}_{B,v}^{\check\theta'}$ since $\check\theta'$ is antidominant.

Now assume that $C=\mathbb{A}^1$, which is justified by Proposition 4.2.3. Then we can identify $(\mathcal{L}^+N)_C=\mathsf{L}^+N\times C$. Let $\mathsf{Y}_{X^\bullet}^{D_j}$ correspond to the point $n_jt^{\check{\nu}_j}\in\mathsf{Gr}_B$ for $n_j\in\mathsf{L}^+N(k)$. For any pairwise distinct $v_0,\ldots,v_d\in\mathbb{A}^1$ we have a line $\mathbb{A}^1\to C^{d+1}:a\mapsto (av_0,\ldots,av_d)$ contracting all points to 0. Multiplication defines a map $m:(\mathsf{L}^+N\times C)^d\to(\mathcal{L}^+N)_{C^d}$. Letting $m(n_1,\ldots,n_d)$ act on the point $t_{av_0}^{\check{\theta}}\prod_j t_{av_j}^{\check{\nu}_j}\in\mathsf{Gr}_{T,C^{d+1}}$ as $a\to 0$, we get a curve connecting $\{t_{v_0}^{\check{\theta}}\}\times\prod\mathsf{Y}_{X^\bullet,v_j}^{D_j}$ to $t_0^{\check{\theta}'}$. Hence the closure of $\mathring{\mathbb{C}}$ in $\mathbb{Y}^{\check{\theta}'}\times_{\mathcal{A}^{\check{\theta}'}}C^{d+1}\subset\mathsf{Gr}_{B,C^{d+1}}$ contains $\mathbb{Y}^{\check{\theta}',\check{\theta}'}_{\mathrm{red}}$. Since the map $C^{d+1}\to\mathcal{A}^{\check{\theta}'}$ is finite and $\mathring{\mathbb{C}}$ is irreducible, we have proved the claim.

Now we draw some corollaries from Proposition 5.6.1.

Corollary 5.6.4. The open substack $\mathfrak{M}_X^0 = \operatorname{Bun}_H$ is dense in $\mathfrak{M}_{X^{\operatorname{can}}}$ iff $\mathfrak{c}_X^{\mathcal{D}} \cap \mathcal{V} = \mathfrak{c}_{X^{\operatorname{can}}}^-$.

This is an analog of [BG02, Proposition 1.2.3], which says that Bun_B is dense in $\overline{\text{Bun}}_B$ if [G, G] is simply connected.

Proof. Indeed, $\mathfrak{c}_X^{\mathcal{D}} \cap \mathcal{V} = \mathfrak{c}_{X^{\mathrm{can}}}^-$ is precisely the condition that every $\check{\Theta} \in \mathrm{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$ is $\succeq 0$.

Define $\mathcal{D}_{\mathrm{sat}}^G(X)$ to be the set of primitive elements in $\mathrm{Prim}(\mathfrak{c}_X^-)$ that cannot be decomposed as a sum $\check{\theta}+\check{\lambda}$ where $\check{\theta},\check{\lambda}$ are both nonzero, $\check{\theta}\in\mathfrak{c}_X^-$ and $\check{\lambda}\succeq 0$ (see §3.1.4 for the definition of primitive). Note that $\mathcal{D}_{\mathrm{sat}}^G(X)$ contains $\varrho_X(\mathcal{D}(X)-\mathcal{D})$, the valuations of the G-stable prime divisors, but the containment may be strict.

Corollary 5.6.5. There is a bijection between the set of irreducible components of \mathfrak{M}_X and $\pi_1(H) \times \operatorname{Sym}^{\infty}(\mathcal{D}_{\operatorname{out}}^G(X)),$

such that $\check{\mu} \in \pi_1(H)$, $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathcal{D}^G_{\operatorname{sat}}(X))$ corresponds to $\check{\mu}\overline{\mathcal{M}}_X^{\check{\Theta}}$.

Proof. For any $\check{\Theta}' \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$, let $\check{\Theta}'' \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$ be a minimal element with respect to the ordering \preceq such that $\check{\Theta}'' \preceq \check{\Theta}'$. Then $\check{\Theta}''$ can be refined to an element $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathcal{D}^G_{\operatorname{sat}}(X))$. Therefore, the closure relations from Proposition 5.6.1 tell us that any stratum is contained in the closure of $\mathfrak{M}_X^{\check{\Theta}}$ for a partition $\check{\Theta}$ as above. By definition if $\check{\theta}_1, \check{\theta}_2 \in \mathcal{D}^G_{\operatorname{sat}}(X)$ satisfy $\check{\theta}_1 \succeq \check{\theta}_2$ then they must be equal. From this one deduces that $\mathfrak{M}_X^{\check{\Theta}}$ is not contained in the closure of any other stratum.

Thus, the closure of each $\mathcal{M}_X^{\check{\Theta}}$, $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathcal{D}_{\operatorname{sat}}^G(X))$, is a union of irreducible components, and no irreducible component is contained in two different such closures. The Corollary now follows from the description of irreducible components of $\mathcal{M}_X^{\check{\Theta}}$ by Corollary 5.5.9.

Lemma 5.6.6. Let $X_1 \to X_2$ be a G-equivariant map of affine spherical varieties with $X_1^{\bullet} = X_2^{\bullet} = H \setminus G$. Then the induced map $\mathfrak{M}_{X_1} \to \mathfrak{M}_{X_2}$ is a closed embedding.

Proof. Let S be a test scheme and let $(\mathcal{P}_G, \sigma_2) \in \mathcal{M}_{X_2}(S)$, where $\sigma_2 : C \times S \to X_2 \times^G \mathcal{P}_G$ is a section. By Corollary 3.5.2 there exists a B-structure \mathcal{P}_B on \mathcal{P}_G after a suitable surjective étale base change $S' \to S$ such that $(\mathcal{P}_B, \sigma_2) \in \mathcal{Y}_{X_2}(S')$. Then in particular there exists a relative effective divisor $D \subset C \times S'$ such that $\sigma_2(C \times S' - D) \subset X_2^{\circ} \times^B \mathcal{P}_B$, cf. §3.7.1. Since $X_1^{\bullet} = X_2^{\bullet}$, we have $X_1^{\circ} = X_2^{\circ}$. By Lemma 3.7.3 the condition that $\sigma_2|_{C \times S' - D}$ extends to a section $C \times S' \to X_1 \times^B \mathcal{P}_B = X_1 \times^G \mathcal{P}_G$ is closed in S'. Therefore, $S' \times_{\mathcal{M}_{X_2}} \mathcal{M}_{X_1} \to S'$ is a closed embedding.

Lemma 5.6.6 implies that $\mathcal{M}_{X^{\operatorname{can}}}$ is a closed substack of \mathcal{M}_X containing $\overline{\mathcal{M}}_X^0$, and when the condition of Corollary 5.6.4 is satisfied, $\mathcal{M}_{X^{\operatorname{can}}} = \overline{\mathcal{M}}_X^0$.

5.7. Closure relations and components in the Zastava model. In order to extend the results above to strata of \mathcal{Y} , we will need the following result:

Lemma 5.7.1. Let $y \in \mathcal{Y}^{\check{\lambda}}(k)$ for $\check{\lambda} \in \mathfrak{c}_X$. For any simple root α with $\mathcal{D}(\alpha) = \{D_{\alpha}^+, D_{\alpha}^-\}$ and any $N \gg 0$, there exists a k-point

$$y' \in \mathcal{Y}^{\check{\lambda}} \mathring{\times} C^{(N)} \mathring{\times} C^{(N)} \subset \mathcal{Y}^{\check{\lambda}} \mathring{\times} \mathcal{Y}^{N\check{\nu}_{\alpha}^{+}} \mathring{\times} \mathcal{Y}^{N\check{\nu}_{\alpha}^{-}}$$

such that the first coordinate is y and the image of y' under the composition

$$y^{\check{\lambda}} \mathring{\times} y^{N\check{\nu}_{\alpha}^{+}} \mathring{\times} y^{N\check{\nu}_{\alpha}^{-}} \to y^{\check{\lambda}+N\check{\alpha}} \to \mathfrak{M}_{X}$$

coincides with the image of y.

Above we are using the fact, from Lemma 5.4.2, that $\mathcal{Y}^{N\check{\nu}_{\alpha}^{\pm}}$ contains $C^{(N)}$ if $\check{\nu}_{\alpha}^{+} \neq \check{\nu}_{\alpha}^{-}$ or $C^{(N)} \sqcup C^{(N)}$ if $\check{\nu}_{\alpha}^{+} = \check{\nu}_{\alpha}^{-}$. In the latter case, Lemma 5.7.1 is picking out one of these components (depending on the point y).

Proof. The point y is equivalent to a datum (\mathcal{P}_B, σ) where \mathcal{P}_B is a B-bundle on C and σ : $C \to X \times^B \mathcal{P}_B$ is a section. Let $\mathcal{P}_G = G \times^B \mathcal{P}_B$ denote the induced G-bundle. The image of y in \mathcal{M}_X corresponds to (\mathcal{P}_G, σ) . Set $\mathbb{k} := k(C)$. Then $\sigma|_{\operatorname{Spec}\,\mathbb{k}}$ defines a trivialization of $\mathcal{P}_G|_{\operatorname{Spec}\,\mathbb{k}}$, which we fix. With respect to this trivialization, B-structures on \mathcal{P}_G are in bijection with sections $\operatorname{Spec}\,\mathbb{k} \to G/B$, and \mathcal{P}_B corresponds to $1B \in (G/B)(\mathbb{k})$. The preimage of (\mathcal{P}_G, σ) in \mathcal{Y} identifies with the orbit $H(\mathbb{k}) \cdot 1B \subset (G/B)(\mathbb{k})$. Since $X \circ P_\alpha/\mathfrak{R}(P_\alpha) = \mathbb{G}_m \backslash \operatorname{PGL}_2$, the orbit of $H \cap P_\alpha$ on $1 \in P_\alpha/B = \mathbb{P}^1$ is \mathbb{G}_m . Let $r \in \mathbb{G}_m(\mathbb{k}) = k(C)^\times$ be a rational function on the curve C. Then the principal divisor defined by r is of the form $v_{-} = v_{-} =$

both contained in $\sigma^{-1}(X^{\circ} \times^{B} \mathcal{P}_{B})$. Under the isomorphism $X^{\circ}P_{\alpha}/B \cong \mathbb{G}_{m} \setminus \mathbb{P}^{1}$, let us identify D_{α}^{+} with $0 \in \mathbb{P}^{1}$ and D_{α}^{-} with $\infty \in \mathbb{P}^{1}$. Then the preimage of $(\mathcal{P}_{G}, \sigma)$ in \mathcal{Y} corresponding to $r \in \mathbb{P}^{1}(\mathbb{k})$ has the desired property.

Recall from Corollary 5.5.9 that the irreducible components of $\mathcal{M}_{X}^{\check{\Theta}}$ are denoted by ${}^{\check{\mu}}\mathcal{M}_{X}^{\check{\Theta}}$, with $\check{\mu} \in \pi_{1}(H)$. For any $\check{\lambda} \in \mathfrak{c}_{X}$, define ${}^{\check{\mu}}\mathcal{Y}^{\check{\lambda},\check{\Theta}} = \mathcal{Y}^{\check{\lambda}} \times_{\mathcal{M}_{X}} {}^{\check{\mu}}\mathcal{M}_{X}^{\check{\Theta}}$ and ${}^{\check{\mu}}\mathcal{Y}^{\check{\lambda},\succeq\check{\Theta}} := \mathcal{Y}^{\check{\lambda}} \times_{\mathcal{M}_{X}} {}^{\check{\mu}}\overline{\mathcal{M}}_{X}^{\check{\Theta}}$.

Corollary 5.7.2. For any $\check{\lambda} \in \mathfrak{c}_X$, $\check{\mu} \in \pi_1(H)$, $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$, the scheme $\check{\mu} \mathcal{Y}^{\check{\lambda},\succeq \check{\Theta}}$ is irreducible (if nonempty), and $\check{\mu} \mathcal{Y}^{\check{\lambda},\check{\Theta}}$ is open dense in it.

Proof. For $\check{\lambda}$ large enough, the claim immediately follows from Corollaries 3.5.2 and 5.6.5. Now, for arbitrary $\check{\lambda}$, consider $\check{\lambda}' = \check{\lambda} + \sum_{\alpha \in \Delta_G} n_\alpha \check{\alpha}$ large enough. By the graded factorization property and Lemma 5.7.1, there exists an étale map

$$(5.18) \qquad \qquad {}^{\check{\mu}}\mathcal{Y}^{\check{\lambda},\succeq\check{\Theta}}\overset{\circ}{\times}\mathcal{C}\to{}^{\check{\mu}}\mathcal{Y}^{\check{\lambda}',\succeq\check{\Theta}},$$

where $\mathcal{C} = \prod_{\Delta_G} C^{(n_\alpha)} \mathring{\times} C^{(n_\alpha)}$.

By the validity of the proposition for large $\check{\lambda}'$, we know that ${}^{\check{\mu}}\mathcal{Y}^{\check{\lambda}',\check{\Phi}}$ is connected and dense in ${}^{\check{\mu}}\mathcal{Y}^{\check{\lambda}',\succeq\check{\Phi}}$. Therefore, if ${}^{\check{\mu}}\mathcal{Y}^{\check{\lambda},\succeq\check{\Phi}}$ is not irreducible, there exist y_1,y_2 in disjoint connected components of ${}^{\check{\mu}}\mathcal{Y}^{\check{\lambda},\check{\Phi}}$ and $c_1,c_2\in\mathcal{C}$ such that (y_1,c_1) and (y_2,c_2) have the same image in ${}^{\check{\mu}}\mathcal{Y}^{\check{\lambda}',\check{\Phi}}$. Let $|c_1|,|c_2|\subset|C|$ denote the support of c_1,c_2 as divisors. From the definition of the factorization map (5.18), if (y_1,c_1) and (y_2,c_2) have the same image, we have

$$y' = y_1|_{C-(|c_1|\cup|c_2|)} = y_2|_{C-(|c_1|\cup|c_2|)} : C-(|c_1|\cup|c_2|) \to X/B.$$

We can extend y' to an element of $\mathcal{Y}^{\check{\lambda}-\check{\nu},\check{\Theta}}$ with $|c_1|\cup |c_2|$ in its B-nondegenerate locus, for some $\check{\nu}\succ 0$. By replacing \mathcal{C} by a possibly different irreducible scheme \mathcal{C}' , we may assume that $|c_1|$ and $|c_2|$ are disjoint. Now we must have $(y',c_2)\mapsto y_1,\ (y',c_1)\mapsto y_2$ under the factorization map $\mathcal{Y}^{\check{\lambda}-\check{\nu},\check{\Theta}}\mathring{\times}\mathcal{C}'\to\mathcal{Y}^{\check{\lambda},\check{\Theta}}$. This implies that y_1,y_2 were originally in the same connected component.

Corollary 5.7.3. For every $\check{\lambda} \in \mathfrak{c}_X$, there is an injection from the set of irreducible components of $\mathcal{Y}^{\check{\lambda}}$ to $\pi_1(H) \times \operatorname{Sym}^{\infty}(\mathcal{D}^G_{\operatorname{sat}}(X))$.

Proof. By Corollaries 5.7.2 and 5.6.5, the irreducible components $\mathcal{Y}^{\check{\lambda}}$ are those $\check{\mu}\mathcal{Y}^{\check{\lambda},\succeq\check{\Theta}}$ that are nonempty, for $\check{\Theta}\in \mathrm{Sym}^{\infty}(\mathcal{D}^G_{\mathrm{sat}}(X))$.

In §6.2, and in particular Corollary 6.2.2, we will see a different description of the irreducible components of \mathcal{Y}_X , based on the partition of $\mathcal{Y}_{X^{\bullet}} = \mathcal{Y}_X^{?,0}$ into the subschemes $\mathcal{Y}_{X^{\bullet}}^D$ of §5.4.

6. Stratified semi-smallness

We keep the assumptions of $\S 5$, i.e., B acts simply transitively on X° and every simple root of G is a spherical root of type T.

The main result of this section is the following, which will be proved in §6.3.

Theorem 6.0.1. Under the assumptions above, $\bar{\pi}_!(IC_{\bar{y}^{\bar{\lambda}}})$ is perverse and constructible with respect to the stratification $\mathcal{A}^{\check{\lambda}} = \bigcup_{\deg(\mathfrak{P}) = \check{\lambda}} \mathring{C}^{\mathfrak{P}}$ of Proposition 3.2.3.

6.1. Upper bounds on dimension. Define

$$\begin{split} \overline{\mathcal{Y}}^{\check{\lambda},\succeq\check{\theta}} &= \overline{\mathcal{Y}}^{\check{\lambda}} \underset{\mathcal{M}_X}{\times} \overline{\mathcal{M}}_X^{\check{\theta}} \subset \overline{\mathcal{Y}}^{\check{\lambda}} \\ \overline{\mathbf{Y}}^{\check{\lambda},\succeq\check{\theta}} &= \overline{\mathcal{Y}}^{\check{\lambda},\succeq\check{\theta}} \cap \overline{\mathbf{Y}}^{\check{\lambda}} \subset \overline{\mathbf{S}}^{\check{\lambda}} \end{split}$$

and let $\mathcal{Y}^{\check{\lambda},\succeq\check{\theta}}, \mathsf{Y}^{\check{\lambda},\succeq\check{\theta}}$ denote the corresponding intersections with $\mathcal{Y}^{\check{\lambda}}$ (the notation is justified by Proposition 5.6.1). From the stratification $\overline{\mathcal{Y}}^{\check{\lambda}} = \bigcup_{\check{\nu}} \overline{\mathcal{Y}}^{\check{\lambda}}$ we deduce that

$$\overline{Y}^{\check{\lambda},\succeq\check{\theta}}=\bigcup_{\check{\lambda}'\leq\check{\lambda}}Y^{\check{\lambda}',\succeq\check{\theta}}.$$

Proposition 6.1.1. Let $\check{\lambda} \in \mathfrak{c}_X - 0$ and $\check{\theta} \in \mathfrak{c}_X^-$. For any connected component \mathscr{Y} of $\mathcal{Y}^{\check{\lambda},\check{\theta}}$, we have

(6.1)
$$\dim(\mathsf{Y}^{\check{\lambda}} \cap \mathscr{Y}) \le \frac{1}{2}(\dim(\mathscr{Y}) - 1).$$

Moreover, whenever the inequality above is an equality for an irreducible component Y of $Y^{\check{\lambda}} \cap \mathscr{Y}$, the same holds for the inequality of Proposition 4.4.2; that is, the closure \overline{Y} in the affine Grassmannian Gr_G meets a semi-infinite orbit $S^{\check{\lambda}'}$ with $\langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle = \dim \overline{Y}$.

Proof. Let $\overline{\mathscr{Y}}$ denote the closure of \mathscr{Y} in $\overline{\mathbb{Y}}^{\check{\lambda}}$. Recall from Corollary 5.7.2 that $\mathscr{Y} = \mathbb{Y}^{\check{\lambda}} \times_{\mathcal{M}_X} \mathscr{M}$ where \mathscr{M} is a connected component of $\mathcal{M}_X^{\check{\theta}}$. Then we have $\overline{\mathscr{Y}} \subset \overline{\mathbb{Y}}^{\check{\lambda}} \times_{\mathcal{M}_X} \overline{\mathscr{M}}$ where $\overline{\mathscr{M}}$ is the closure of \mathscr{M} in $\overline{\mathcal{M}}_X^{\check{\theta}}$.

Fix an irreducible component Y of $Y^{\check{\lambda}} \cap \mathscr{Y}$ and let \overline{Y} be its closure in $\overline{Y}^{\check{\lambda}} \cap \overline{\mathscr{Y}}$. From Proposition 4.4.2 we know, by intersecting \overline{Y} with semi-infinite orbits in the affine Grassmannian, that there is a $\check{\lambda}' \leq \check{\lambda}$ with $Y' := \overline{Y} \cap S^{\check{\lambda}'}$ nonempty of dimension zero, and $d := \dim Y \leq \langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle$. Note that $Y' \subset Y^{\check{\lambda}'} \cap \overline{\mathscr{Y}} \subset Y^{\check{\lambda}'} \times_{\mathcal{M}_X} \overline{\mathscr{M}}$.

Note that $\mathsf{Y}' \subset \mathcal{Y}^{\check{\lambda}'} \cap \overline{\mathscr{Y}} \subset \mathcal{Y}^{\check{\lambda}'} \times_{\mathfrak{M}_X} \overline{\mathscr{M}}$. Since $\check{\lambda} - \check{\lambda}' \in \check{\Lambda}_G^{\mathrm{pos}}$, we can decompose $\check{\lambda} - \check{\lambda}' = \sum_{\alpha \in \Delta_G} n_{\alpha} (\check{\nu}_{D_{\alpha}^+} + \check{\nu}_{D_{\alpha}^-})$ where the sum is over simple roots and $\sum n_{\alpha} = \langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle$. By the graded factorization property and Lemma 5.7.1, there is an étale map

$$(6.2) \qquad (\mathcal{Y}^{\check{\lambda}'} \underset{\mathcal{M}_X}{\times} \overline{\mathscr{M}}) \overset{\circ}{\times} \prod_{\alpha \in \Delta_G} (C^{(n_{\alpha})} \overset{\circ}{\times} C^{(n_{\alpha})}) \to \mathcal{Y}^{\check{\lambda}} \underset{\mathcal{M}_X}{\times} \overline{\mathscr{M}}$$

over $\overline{\mathcal{M}}$.

If $\check{\lambda}' \neq 0$, then $\check{\mathcal{Y}}^{\check{\lambda}'} \times_{\mathcal{M}_X} \overline{\mathscr{M}}$ contains $\mathsf{Y}' \check{\times} C$ and hence (6.2) implies that $\dim(\mathscr{Y}) \geq 1 + 2\langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle$. Therefore,

(6.3)
$$d \leq \langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle \leq \frac{1}{2} (\dim(\mathscr{Y}) - 1)$$

if $\check{\lambda}' \neq 0$. This proves the claim in the case $\check{\lambda}' \neq 0$.

There remains to consider when $d = \langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle$ and $\check{\lambda}' = 0$. In particular, $\mathsf{Y}^0 \times_{\mathcal{M}_X} \overline{\mathcal{M}}_X^\theta$ is nonempty, which can only be the case if $\check{\theta} = 0$ since $\mathsf{Y}^0 = \mathsf{Y}^{0,0} = \mathsf{pt}$. When considering $\mathcal{Y}^{\check{\lambda},0}$, we may use §5.3 to reduce to assuming that $\mathfrak{c}_X = \mathbb{N}^{\mathcal{D}}$. In this case again by Proposition 4.4.2, there is a simple root α and an irreducible component Y_{d-1} of $\overline{\mathsf{Y}} \cap \mathsf{S}^{\check{\alpha}}$ of dimension 1 such that $\overline{\mathsf{Y}}_{d-1}$ contains $\mathsf{Y}^{0,0} = \mathsf{pt}$. This implies that Y_{d-1} is an irreducible component of $\mathsf{Y}^{\check{\alpha},0}$ of dimension 1. Since \mathfrak{c}_X is now the free monoid, $\check{\alpha} = \check{\nu}_{D_{\alpha}^+} + \check{\nu}_{D_{\alpha}^-}$ for $\mathcal{D}(\alpha) = \{D_{\alpha}^+, D_{\alpha}^-\}$. Lemma 5.4.2 implies that $\mathsf{Y}^{\check{\alpha},0}$ is empty, a contradiction.

6.2. Connected components of open Zastava. Recall from §5.4 that $\mathcal{Y}_{X}^{\gamma,0} = \mathcal{Y}_{X^{\bullet}}$ is a disjoint union of subschemes $\mathcal{Y}_{X^{\bullet}}^{D}$, indexed by $D \in \mathbb{N}^{\mathcal{D}}$, i.e., multisets of colors. Therefore, we have a map $\pi_{0}(\mathcal{Y}_{X^{\bullet}}) \to \mathbb{N}^{\mathcal{D}}$. On the other hand, Corollary 5.7.3 gives an injection $\pi_{0}(\mathcal{Y}_{X^{\bullet}}) \to \pi_{1}(H) \times \mathfrak{c}_{X}$ where $\check{\mu}\mathcal{Y}^{\check{\lambda},0} \mapsto (\check{\mu},\check{\lambda})$ when the former is nonempty.

Observe that $\pi_1(H) \otimes \mathbb{Q} \cong \mathcal{X}(H) \otimes \mathbb{Q}$. Then tensoring (5.5) by \mathbb{Q} , we get an injection

$$(6.4) \mathbb{Z}^{\mathcal{D}} \hookrightarrow \mathbb{Q}^{\mathcal{D}} \hookrightarrow (\pi_1(H) \times \check{\Lambda}_X) \otimes \mathbb{Q}.$$

One can check that the maps above fit into a commutative diagram

(6.5)
$$\pi_0(\mathcal{Y}_X \bullet) \hookrightarrow \pi_1(H) \times \mathfrak{c}_X$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\mathbb{N}^{\mathcal{D}} \hookrightarrow^{(6.4)} (\pi_1(H) \times \check{\Lambda}_X) \otimes \mathbb{Q}$$

We now show that the left vertical arrow is a bijection.

Lemma 6.2.1. For $D \in \mathbb{N}^{\mathcal{D}}$, the smooth scheme $\mathcal{Y}_{X^{\bullet}}^{D}$ is connected of dimension len(D).

Proof. As discussed in §5.4, we may assume that $X = X^{\operatorname{can}}$ and $\mathfrak{c}_X = \mathbb{N}^{\mathcal{D}}$. Now the claim is equivalent to showing that, under those assumptions, $\mathbb{Y}^{\check{\lambda},0}$ is connected for $\check{\lambda} = \sum n_{D'}\check{\nu}_{D'}$. One deduces from the graded factorization property of \mathbb{Y} that if $\mathbb{Y}^{\check{\lambda},0}$ is not connected, there must exist a possibly different $\check{\lambda}$ with a connected component \mathscr{Y} contained entirely in the preimage of the diagonal $\mathbb{Y}^{\check{\lambda},0} \times_{\mathcal{A}^{\check{\lambda}},\delta\check{\lambda}} C$. Then $\dim(\mathbb{Y}^{\check{\lambda}} \cap \mathscr{Y}) = \dim \mathscr{Y} - 1$, and the dimension inequality of Proposition 6.1.1 can only hold if $\dim \mathscr{Y} = 1$. By Corollary 5.7.2, the component \mathscr{Y} must be of the form

$$\mathscr{Y}={}^{\check{\mu}}\mathscr{Y}^{\check{\lambda},0}=\operatorname{Bun}_{H}^{\check{\mu}}\underset{\operatorname{Bun}_{G}}{\times}\operatorname{Bun}_{B}^{-\check{\lambda}}$$

for some $\check{\mu} \in \pi_1(H)$. Now by the graded factorization property and Lemma 5.7.1, we have an étale map

$$\check{\mu} y \check{\lambda}, 0 \overset{\circ}{\times} \prod_{\alpha \in \Lambda_C}^{\circ} C^{(N_{\alpha})} \overset{\circ}{\times} C^{(N_{\alpha})} \to \check{\mu} y \check{\lambda}', 0$$

for $\check{\lambda}'=\check{\lambda}+\sum N_{\alpha}\check{\alpha}$ and any N_{α} large enough. Thus, $\dim {}^{\check{\mu}}\mathcal{Y}^{\check{\lambda}',0}=1+2\sum N_{\alpha}$. By Lemma 3.5.1 we may assume that $\mathrm{Bun}_B^{-\check{\lambda}'}\to \mathrm{Bun}_G$ is smooth. Note that $\dim \mathrm{Bun}_H^{\check{\mu}}$ only depends on the image of $\check{\mu}$ in $\pi_1(H)\otimes \mathbb{Q}$. On the other hand, the commutative diagram (6.5) says that this image is determined by $\check{\lambda}'$. These observations imply that $\mathcal{Y}^{\check{\lambda}',0}=\mathrm{Bun}_H \underset{\mathrm{Bun}_G}{\times} \mathrm{Bun}_B^{-\check{\lambda}'}$ is equidi-

mensional. However we know that $\mathcal{Y}^{\check{\lambda}',0}$ has a connected component birational to $\mathcal{A}^{\check{\lambda}'}$, which is of dimension $\sum n_{D'} + 2 \sum N_{\alpha}$. The equality

$$\dim {}^{\check{\mu}}\mathcal{Y}^{\check{\lambda}',0} = 1 + 2\sum N_{\alpha} = \sum n_{D'} + 2\sum N_{\alpha}$$

forces $\check{\lambda} = \check{\nu}_{D'}$ for some color $D' \in \mathcal{D}$, hence D = D'. Lemma 5.4.2 now implies that $\mathcal{Y}^{\check{\nu}_D,0}$ is connected.

Corollary 6.2.2. For every $\check{\lambda} \in \mathfrak{c}_X$, $\check{\theta} \in \mathfrak{c}_X^-$, the connected components of $\forall^{\check{\lambda},\check{\theta}}$ are in bijection with the closures of

$$\mathcal{Y}_{X\bullet}^{D}\mathring{\times}\mathcal{Y}^{\check{\theta},\check{\theta}}\hookrightarrow\mathcal{Y}^{\check{\lambda},\check{\theta}}$$

for $D \in \mathbb{N}^{\mathcal{D}}$ such that $\varrho_X(D) = \check{\lambda} - \check{\theta}$.

Proof. Immediate from Lemmas 5.5.7(i) and 6.2.1.

6.3. Stratified semi-smallness. Following [MV07, §4], we will use the notion of a stratified semi-small map, which we now review. Let $f: Y \to A$ be a proper map between two stratified spaces (Y, \mathbb{S}) and (A, \mathbb{T}) . Suppose all strata are smooth and connected and each $f(S), S \in \mathbb{S}$ is a union of strata $S' \in \mathbb{T}$. We say f is étale-locally trivial (in the stratified sense) if whenever $S' \subset f(S)$, the restriction of f to $S \cap f^{-1}(S') \to S'$ is étale-locally a trivial fibration. We say that f is stratified semi-small if it is étale-locally trivial and for any $S \in \mathbb{S}$ and any $S' \in \mathbb{T}$ such that $S' \subset f(S)$ we have

(6.6)
$$\dim(f^{-1}(a) \cap S) \le \frac{1}{2}(\dim S - \dim S')$$

for any (and thus all) $a \in S'$.

The notion of stratified semi-smallness is relevant due to the observation below, which follows from dimension counting and the definition of the perverse t-structure:

Lemma 6.3.1 ([MV07, Lemma 4.3]). If f is a stratified semi-small map then $f_*(\mathfrak{F}) \in P_{\mathfrak{T}}(A)$ for all $\mathfrak{F} \in P_{\mathfrak{F}}(Y)$.

Note that the lemma holds even if the stratifications are not Whitney. In this case we simply define $P_{\mathcal{S}}(Y) := P(Y) \cap D^b_{\mathcal{S}}(Y)$ to be the subcategory of perverse sheaves that are S-constructible, i.e., the $\mathcal{F} \in P(Y)$ such that $H^i(\mathcal{F})|_S$ is a local system of finite rank for all $i \in \mathbb{Z}$ and $S \in \mathcal{S}$.

6.3.2. Let us return to our situation: consider the proper map $\bar{\pi}: \bar{\overline{y}}^{\check{\lambda}} \to \mathcal{A}^{\check{\lambda}}$. We have the smooth stratification defined in Proposition 4.2.1,

$$\overline{\mathcal{Y}}^{\check{\lambda}} = \bigcup_{\check{\nu}} {}_{\check{\Phi}} \overline{\mathcal{Y}}^{\check{\lambda}, \check{\Theta}}, \qquad {}_{\check{\nu}} \overline{\mathcal{Y}}^{\check{\lambda}, \check{\Theta}} \cong C_{\check{\nu}} \times \mathcal{Y}^{\check{\lambda} - \check{\nu}, \check{\Theta}}$$

for $\check{\nu}\in\check{\Lambda}_G^{\mathrm{pos}}$, $\check{\Theta}\in\mathrm{Sym}^\infty(\mathfrak{c}_X^--0)$. Let $(\overline{\mathcal{Y}}^{\check{\lambda}},\mathcal{S})$ denote the stratification by the connected components of ${}_{\check{\nu}}\mathcal{Y}^{\check{\lambda},\check{\Theta}}$. We will not show that \mathcal{S} is a Whitney stratification; for our purposes the following suffices:

Lemma 6.3.3. For any $\check{\lambda} \in \mathfrak{c}_X$, the IC complex of $\overline{\mathbb{y}}^{\check{\lambda}}$ is S-constructible, i.e.,

$$\mathrm{IC}_{\overline{y}^{\check{\lambda}}} \in \mathrm{P}_{\mathcal{S}}(\overline{\mathcal{Y}}^{\check{\lambda}}).$$

Proof. Let $\check{\nu} \in \check{\Lambda}_G^{\mathrm{pos}}$ be such that $\overline{\mathcal{Y}}^{\check{\lambda}} = \underline{\langle} \check{\nu} \overline{\mathcal{Y}}^{\check{\lambda}}$, which exists since $\overline{\mathcal{Y}}^{\check{\lambda}}$ is of finite type. For any $\check{\mu} \in \check{\Lambda}_G^{\mathrm{pos}}$ large enough, we have a smooth correspondence preserving stratifications

$$\overline{\mathcal{Y}}^{\check{\lambda}} \leftarrow \overline{\mathcal{Y}}^{\check{\lambda}} \mathring{\times} \mathcal{Y}^{\check{\mu},0} \rightarrow \langle_{\check{\nu}} \overline{\mathcal{Y}}^{\check{\lambda}+\check{\mu}} =: Y$$

where the right arrow comes from the graded factorization property of \overline{y} . Thus, it suffices to check that IC_Y is S-constructible. By (4.11), we have an isomorphism

$$\operatorname{IC}_Y \cong \left(\operatorname{IC}_{\mathfrak{M}_X} \underset{\operatorname{Bun}_G}{\boxtimes} \operatorname{IC}_{\overline{\operatorname{Bun}}_B^{-\check{\lambda}-\check{\mu}}} \right) \Big|_Y.$$

Now Proposition 3.1.7 implies that $IC_{\mathcal{M}_X}$ is constructible with respect to the fine stratification on \mathcal{M}_X , and Theorem 4.5.3 implies that $IC_{\overline{\mathrm{Bun}}_B}$ is constructible with respect to the stratification by defect. Thus, it follows that IC_Y is S-constructible.

Let $(\mathcal{A}^{\check{\lambda}}, \mathfrak{T})$ denote the stratification from Proposition 3.2.3,

$$\mathcal{A}^{\check{\lambda}} = \bigcup_{\mathfrak{P}} \mathring{C}^{\mathfrak{P}}$$

for $\mathfrak{P} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X - 0)$ such that $\operatorname{deg}(\mathfrak{P}) = \check{\lambda}$.

Now Theorem 6.0.1 follows from Lemmas 6.3.1, 6.3.3 and the following theorem:

Theorem 6.3.4. The map $\bar{\pi}: (\bar{\mathbb{y}}^{\check{\lambda}}, \mathbb{S}) \to (\mathcal{A}^{\check{\lambda}}, \mathbb{T})$ is stratified semi-small.

Proof. Fix $\mathfrak{P} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X} - 0)$ such that $\check{\lambda} = \operatorname{deg}(\mathfrak{P})$. Let $I = \{1, \dots, |\mathfrak{P}|\}$ be the finite set of cardinality $|\mathfrak{P}|$. We fix an ordering $\mathfrak{P} = \sum_{i \in I} [\check{\lambda}_{i}]$ on our partition \mathfrak{P} , so each $\check{\lambda}_{i} \in \mathfrak{c}_{X}$. Let \mathring{C}^{I} denote the I-fold product C^{I} with the diagonal divisor removed. Then the map $\mathring{C}^{I} \to \mathring{C}^{\mathfrak{P}}$ corresponds to choosing an ordering on an unordered multiset of points in C.

We leave it to the reader to check that $\bar{\pi}(S)$ is a union of strata for $S \in \mathcal{S}$. It is also a consequence of the graded factorization property that for $\check{\nu} \in \check{\Lambda}_G^{\mathrm{pos}}, \check{\Theta} \in \mathrm{Sym}^{\infty}(\mathfrak{c}_X - 0)$, the fiber product

(6.7)
$${}_{\check{\nu}}\overline{\mathcal{J}}^{\check{\lambda},\check{\Theta}}\underset{\mathcal{A}\check{\lambda}}{\times}{}_{C}^{I} = \bigsqcup_{i \mapsto \check{\nu}_{i},\check{\lambda}'_{i},\check{\theta}_{i}} \prod_{i \in I}^{\circ} (\mathsf{Y}^{\check{\lambda}'_{i},\check{\theta}_{i}} \tilde{\times} C)$$

is equal to a disjoint union of open and closed subschemes, running over all assignments $i \in I \mapsto \check{\nu}_i \in \check{\Lambda}_G^{\mathrm{pos}}, \check{\lambda}_i' \in \mathfrak{c}_X, \check{\theta}_i \in \mathfrak{c}_X^-$ (including zero) such that $\check{\nu}_i + \check{\lambda}_i' = \check{\lambda}_i$ and $\sum_I [\check{\theta}_i] = \check{\Theta} + (|I| - |\check{\Theta}|)[0] \in \mathrm{Sym}^{\infty}(\mathfrak{c}_X)$ as a multiset with zero. Since $\mathring{C}_I^I \to \mathring{C}^{\mathfrak{P}}$ is finite étale, (and everything is of finite type) we deduce that the restriction

Since $\mathring{C}^I \to \mathring{C}^{\mathfrak{P}}$ is finite étale, (and everything is of finite type) we deduce that the restriction of $\bar{\pi}$ to ${}_{\check{\nu}}\bar{\mathcal{Y}}^{\check{\lambda},\check{\Theta}}\cap\bar{\pi}^{-1}(\mathring{C}^{\mathfrak{P}})\to\mathring{C}^{\mathfrak{P}}$ is étale-locally a trivial fibration.

It remains to check the dimension inequality (6.6). Let $\mathfrak{P}, \check{\nu}, \check{\Theta}$ be as before, and fix a connected component S of ${}_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda},\check{\Theta}}$. We deduce from (6.7) that for any $a \in \mathring{C}^{\mathfrak{P}}$, the restricted fiber $\bar{\pi}^{-1}(a) \cap S$ is contained in a union of $\mathring{\prod}_{I}(\mathsf{Y}^{\check{\lambda}'_{i},\check{\theta}_{i}} \cap S_{i})$ for $\check{\nu}_{i}, \check{\lambda}'_{i}, \check{\theta}_{i}$ as above and S_{i} some connected component of $\mathcal{Y}^{\check{\lambda}'_{i},\check{\theta}_{i}}$. By Proposition 6.1.1, we have

(6.8)
$$\dim(\mathsf{Y}^{\check{\lambda}_i',\check{\theta}_i}\cap S_i) \leq \frac{1}{2}(\dim(S_i) - 1).$$

The image of the composition

$$\prod_{i\in I}^{\circ}(C_{\nu_i}\times S_i)\to \prod_{i\in I}^{\circ}_{\check{\nu}_i}\overline{\mathcal{Y}}^{\check{\lambda}_i,\check{\theta}_i}\to_{\check{\nu}}\overline{\mathcal{Y}}^{\check{\lambda},\check{\Theta}}$$

is connected, so it must be contained in S. Thus, $\dim(S) \geq \langle \rho_G, \check{\nu} \rangle + \sum_I \dim(S_i)$. Summing (6.8) over I, we get

$$\dim(\bar{\pi}^{-1}(a) \cap S) \le \frac{1}{2}(\dim(S) - \dim(\mathring{C}^{\mathfrak{P}}) - \langle \rho_G, \check{\nu} \rangle),$$

which establishes the inequality (6.6).

6.4. **Euler product.** Let us explain how we can combine the graded factorization property of \overline{y} with Theorem 6.3.4 to deduce that $\overline{\pi}_!(IC_{\overline{y}})$ "looks like an Euler product".

In the expression that we are about to obtain, a special role will be played by those strata $\mathcal{Y}^{\check{\lambda},\check{\theta}}$ of $\mathcal{Y}^{\check{\lambda}}$ which correspond to elements $\check{\theta} \in \mathcal{D}^G_{\mathrm{sat}}(X) \cup \{0\}$. Recall, by Corollary 5.7.2, that the closures of those strata are unions of irreducible components of \mathcal{Y} . Let

$$\mathfrak{B}_{X,\check{\lambda}} = \bigcup_{\check{\theta} \in \mathcal{D}^G_{\mathrm{sat}}(X) \cup \{0\}} \bigcup_{\mathscr{Y}} \mathfrak{B}_{\mathscr{Y}},$$

where \mathscr{Y} runs over all irreducible components of $\mathscr{Y}^{\check{\lambda},\check{\theta}}$, and $\mathfrak{B}_{\mathscr{Y}}$ is the set of those irreducible components b of the central fiber $Y^{\check{\lambda}} \cap \mathscr{Y}$ for which the inequality of (6.1) is an equality, that

is,

(6.10)
$$\dim \mathsf{b} = \frac{1}{2}(\dim \mathscr{Y} - 1).$$

Such components will be said to be of *critical dimension*; we will explicate this dimension in Proposition 6.5.1. The sets $\mathfrak{B}_{X,\check{\lambda}}$ will define the *crystal of X* in Section 7. For now, we treat them as a black box.

We let $V_{X,\check{\lambda}}$ denote the free vector space on $\mathfrak{B}_{X,\check{\lambda}}$, that is,

$$(6.11) V_{X,\check{\lambda}} = \bigoplus_{\mathfrak{B}_{X,\check{\lambda}}} \overline{\mathbb{Q}}_{\ell}.$$

Note that, when X is defined over a finite field \mathbb{F} , and k is its algebraic closure, the (geometric) Frobenius morphism induces a dimension-preserving bijection between the sets $\mathfrak{B}_{X,\check{\lambda}}$ and $\mathfrak{B}_{X,\mathrm{Fr}\check{\lambda}}$, for every $\check{\lambda} \in \check{\Lambda}$. Hence, Fr acts naturally on the sum of vector spaces $\bigoplus_{\check{\lambda} \in \mathfrak{c}_X} V_{X,\check{\lambda}}$. For a partition $\mathfrak{R} = \sum_{\check{\mu} \in \mathfrak{c}_X - 0} N_{\check{\mu}}[\check{\mu}] \in \mathrm{Sym}^{\infty}(\mathfrak{c}_X - 0)$, let $\iota^{\mathfrak{R}} : \mathring{C}^{\mathfrak{R}} := \mathring{\prod} \mathring{C}^{(N_{\check{\mu}})} \hookrightarrow \mathcal{A}^{\check{\lambda}}$ denote

For a partition $\mathfrak{R} = \sum_{\check{\mu} \in \mathfrak{c}_X = 0} N_{\check{\mu}}[\check{\mu}] \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X = 0)$, let $\iota^{\mathfrak{R}} : \mathring{C}^{\mathfrak{R}} := \mathring{\prod} \mathring{C}^{(N_{\check{\mu}})} \hookrightarrow \mathcal{A}^{\check{\lambda}}$ denote the locally closed embedding, with $\check{\lambda} = \deg(\mathfrak{R})$. This extends to a finite map $\bar{\iota}^{\mathfrak{R}} : C^{\mathfrak{R}} := \prod C^{(N_{\check{\mu}})} \to \mathcal{A}^{\check{\lambda}}$ which is the normalization of the closure of $\mathring{C}^{\mathfrak{R}}$ in $\mathcal{A}^{\check{\lambda}}$.

Proposition 6.4.1. For $\check{\lambda} \in \mathfrak{c}_X$, there exists a canonical isomorphism

$$(6.12) \bar{\pi}_{!}(\mathrm{IC}_{\overline{\mathbb{Y}}^{\check{\lambda}}}) \cong \bigoplus_{\deg(\mathfrak{R}) = \check{\lambda}} \left(\bigotimes_{\check{\mu}} \mathrm{Sym}^{N_{\check{\mu}}}(V_{X,\check{\mu}}) \right) \otimes \bar{\iota}_{!}^{\mathfrak{R}}(\mathrm{IC}_{C^{\mathfrak{R}}})$$

where $\Re = \sum_{\check{\mu} \in \mathfrak{c}_X - 0} N_{\check{\mu}}[\check{\mu}]$ and the spaces $V_{X,\check{\mu}}$ are defined by (6.11).

When X is defined over a finite field \mathbb{F} , and k is its algebraic closure, this isomorphism is Galois-equivariant.

The implied Galois action on the right hand side of (6.12) is the one obtained by the action of Frobenius on the sum of spaces $V_{X,\check{\mu}}$ (by permuting their basis elements), and the standard Weil structure $\overline{\mathbb{Q}}_{\ell}(\frac{|\mathfrak{R}|}{2})[|\mathfrak{R}|]$ on $\mathrm{IC}_{C^{\mathfrak{R}}}$.

Note that if $\mathfrak{R} = [\check{\lambda}]$ is the singleton partition, then $C^{[\check{\lambda}]} = C$ and $\iota^{[\check{\lambda}]} = \delta^{\check{\lambda}} : C \hookrightarrow \mathcal{A}^{\check{\lambda}}$ is the diagonal embedding. The corresponding summand of $\bar{\pi}_!(\mathrm{IC}_{\bar{y}^{\check{\lambda}}})$ above is $V_{X,\check{\lambda}} \otimes \mathrm{IC}_C$. We call this the diagonal contribution of $\bar{\pi}_!(\mathrm{IC}_{\bar{u}^{\check{\lambda}}})$.

Proof. The proof follows the same logic as [BFGM02, §5.4, 5.11]. Theorem 6.0.1 implies that $\bar{\pi}_!(IC_{\overline{y}\check{\lambda}})$ is perverse, and the decomposition theorem ([BBDG18, Théorème 6.2.5]) implies that it is semisimple. Since $\bar{\pi}_!(IC_{\overline{y}\check{\lambda}})$ is constructible with respect to the stratification by $\iota^{\mathfrak{R}}: \mathring{C}^{\mathfrak{R}} \hookrightarrow \mathcal{A}^{\check{\lambda}}$ for $\mathfrak{R} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X - 0)$, $\deg(\mathfrak{R}) = \check{\lambda}$, we deduce that there exists a *canonical* decomposition

(6.13)
$$\bar{\pi}_{!}(\mathrm{IC}_{\bar{y}^{\check{\lambda}}}) \cong \bigoplus_{\deg(\mathfrak{R})=\check{\lambda}} \iota_{!*}^{\mathfrak{R}}(\mathcal{L}^{\mathfrak{R}})[|\mathfrak{R}|]$$

where $\iota_{!*}^{\mathfrak{R}}$ denotes the middle extension functor along the locally closed embedding, and $\mathcal{L}^{\mathfrak{R}}$ is a local system on $\mathring{\mathcal{C}}^{\mathfrak{R}}$.

Now consider the singleton partition $[\check{\lambda}]$. For $v \in |C|$ let $\delta_v^{\check{\lambda}}: v \to \mathcal{A}^{\check{\lambda}}$ denote the composition of $v \to C$ with $\delta^{\check{\lambda}}: C \to \mathcal{A}^{\check{\lambda}}$. Recall from §4.3.4 that $\overline{\mathcal{Y}}^{\check{\lambda}} \times_{\mathcal{A}^{\check{\lambda}}, \delta^{\check{\lambda}}} C \cong \overline{\mathsf{Y}}^{\check{\lambda}} \times^{\operatorname{Aut} k[\![t]\!]} C^{\wedge}$. Taking the *-pullback along $\delta_v^{\check{\lambda}}$ of (6.13), we have

$$R\Gamma(\overline{\mathbf{Y}}^{\check{\lambda}},\mathrm{IC}_{\overline{\mathcal{Y}}^{\check{\lambda}}}|_{\overline{\mathbf{Y}}^{\check{\lambda}}}^*) \cong \bigoplus_{\deg(\mathfrak{R}) = \check{\lambda}} (\delta_v^{\check{\lambda}})^* \iota_{!*}^{\mathfrak{R}}(\mathcal{L}^{\mathfrak{R}})[|\mathfrak{R}|].$$

For $\mathfrak{R}=[\check{\lambda}]$, we have $\mathcal{L}^{[\check{\lambda}]}$ is a local system on C, so $(\delta_v^{\check{\lambda}})^*\iota_{!*}^{[\check{\lambda}]}(\mathcal{L}^{[\check{\lambda}]})[1]$ lives in cohomological degree -1. For $\mathfrak{R}\neq [\check{\lambda}]$, we have $\dim C^{\mathfrak{R}}>1$ so the uniqueness property of middle extension implies that $(\delta^{\check{\lambda}})^*\iota_{!*}^{\mathfrak{R}}(\mathcal{L}^{\mathfrak{R}})[|\mathfrak{R}|]$ has lisse cohomology sheaves on C and lives in usual cohomological degrees <-1. Therefore, we deduce that

$$\mathcal{L}^{[\check{\lambda}]}|_{v\to C}^* = H_c^{-1}(\overline{\mathsf{Y}}^{\check{\lambda}}, \mathrm{IC}_{\overline{\mathsf{Y}}^{\check{\lambda}}}|_{\overline{\mathsf{Y}}^{\check{\lambda}}}^*),$$

which is the top cohomological degree.

Recall from Corollary 5.7.3 that the irreducible components of $\mathcal{Y}^{\check{\lambda}}$ are naturally parametrized by a subset of $\pi_1(H) \times \operatorname{Sym}^{\infty}(\mathcal{D}^G_{\operatorname{sat}}(X))$. Let $\mathcal{Y}^{\check{\lambda},\circ}$ denote the union of $\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ for all $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathcal{D}^G_{\operatorname{sat}}(X))$. Then $\mathcal{Y}^{\check{\lambda},\circ}$ is a disjoint union of smooth connected components in bijection with the irreducible components of $\mathcal{Y}^{\check{\lambda}}$, by Corollary 5.7.2. Again, the uniqueness property of the IC complex implies that $\operatorname{IC}_{\bar{\mathcal{Y}}^{\check{\lambda}}}|_{\bar{\mathcal{Y}}^{\check{\lambda}}-\mathcal{Y}^{\check{\lambda},\circ}}^*$ lives in strictly negative perverse cohomological degrees. Since $\bar{\pi}_!$ is perverse t-exact by Theorem 6.3.4, we deduce that $\bar{\pi}_!(\operatorname{IC}_{\bar{\mathcal{Y}}^{\check{\lambda}}}|_{\bar{\mathcal{Y}}^{\check{\lambda}}-\mathcal{Y}^{\check{\lambda},\circ}}^*)$ is constructible and lives in strictly negative perverse degrees. This in turn implies that $(\delta_v^{\check{\lambda}})^*\bar{\pi}_!(\operatorname{IC}_{\bar{\mathcal{Y}}^{\check{\lambda}}}|_{\bar{\mathcal{Y}}^{\check{\lambda}}-\mathcal{Y}^{\check{\lambda},\circ}}^*)$ lives in (usual=perverse) degrees <-1. We conclude that

$$\mathcal{L}^{[\check{\lambda}]}|_{v\to C}^* = H_c^{-1}(\mathsf{Y}^{\check{\lambda}}, (\mathrm{IC}_{\overline{\mathcal{Y}}^{\check{\lambda}}}|_{\mathcal{Y}^{\check{\lambda},\circ}})|_{\mathsf{Y}^{\check{\lambda}}}^*) = \bigoplus_{\mathscr{Y}} H_c^{\dim \mathscr{Y}-1}(\mathsf{Y}^{\check{\lambda}}\cap \mathscr{Y}, \overline{\mathbb{Q}}_{\ell}(\tfrac{\dim \mathscr{Y}}{2})),$$

with the sum running over all irreducible components of $\mathcal{Y}^{\check{\lambda},\circ}$. Note that $\mathsf{Y}^{\check{\lambda}}\cap\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ is empty unless $\check{\Theta}=[\check{\theta}]$ is singleton, for $\check{\theta}\in\mathcal{D}^G_{\mathrm{sat}}(X)\cup\{0\}$. Moreover, the right hand side consists of only top cohomological degrees, by Proposition 6.1.1, so $H_c^{\dim\mathscr{Y}-1}(\mathsf{Y}^{\check{\lambda}}\cap\mathscr{Y},\overline{\mathbb{Q}}_{\ell}(\frac{\dim\mathscr{Y}}{2}))$ is equal to the sum

$$\bigoplus_{\mathfrak{B}_{\mathscr{A}}}\overline{\mathbb{Q}}_{\ell}(\tfrac{1}{2}),$$

where $\mathfrak{B}_{\mathscr{Y}}$ is the set of irreducible components of $\mathsf{Y}^{\check{\lambda}} \cap \mathscr{Y}$ of dimension $\frac{\dim \mathscr{Y} - 1}{2}$. In particular, $\mathcal{L}^{[\check{\lambda}]}|_{v \to C}^*$ has trivial monodromy under Aut $k[\![t]\!]$, so we deduce that $\mathcal{L}^{[\check{\lambda}]} \cong V_{X,\check{\lambda}} \otimes \mathrm{IC}_C$ where $V_{X,\check{\lambda}}$ is defined by (6.11) (with the $(\frac{1}{2})$ -twist absorbed by IC_C).

Next, consider an arbitrary partition $\mathfrak{R} = \sum_{\check{\mu} \in \mathfrak{c}_X - 0} N_{\check{\mu}}[\check{\mu}]$. We defined $\mathring{C}^{\mathfrak{R}} = \mathring{\prod}_{\check{\mu}} \mathring{C}^{(N_{\check{\mu}})}$. By the graded factorization property, we have a diagram with Cartesian squares

$$\overset{\circ}{\prod}_{\check{\mu}} (\overline{\mathbf{Y}}^{\check{\mu}} \overset{\sim}{\times} C)^{\mathring{\times}N_{\check{\mu}}} \overset{\longleftarrow}{\longleftrightarrow} \overset{\circ}{\prod}_{\check{\mu}} (\overline{\mathcal{Y}}^{\check{\mu}})^{\mathring{\times}N_{\check{\mu}}} \overset{\text{\'etale}}{\longrightarrow} \overline{\mathcal{Y}}^{\check{\lambda}} \\
\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow \\
\overset{\circ}{\prod} \mathring{\mathbf{C}}^{N_{\check{\mu}}} \overset{\longleftarrow}{\longleftrightarrow} \overset{\circ}{\prod} \overset{\circ}{\prod} (\mathcal{A}^{\check{\mu}})^{\mathring{\times}N_{\check{\mu}}} \overset{\text{\'etale}}{\longleftrightarrow} \mathcal{A}^{\check{\lambda}}$$

and the composition of the bottom arrows factors through the $(\prod_{\check{\mu}} \mathfrak{S}_{N_{\check{\mu}}})$ -torsor

$$\mathring{\prod} \mathring{C}^{N_{\check{\mu}}} \to \mathring{\prod} \mathring{C}^{(N_{\check{\mu}})} = \mathring{C}^{\mathfrak{R}},$$

where \mathfrak{S}_N denotes the symmetric group on N elements. By induction, we deduce that

(6.14)
$$\mathcal{L}^{\mathfrak{R}}[|\mathfrak{R}|]|_{\mathring{\Pi}\mathring{C}^{N_{\tilde{\mu}}}}^{*} \cong (\bigotimes_{\tilde{\mu}}(V_{X,\tilde{\mu}}\otimes \mathrm{IC}_{C})^{\boxtimes N_{\tilde{\mu}}})|_{\mathring{\Pi}\mathring{C}^{N_{\tilde{\mu}}}}^{*}.$$

There is a natural $\mathfrak{S}_{N_{\check{\mu}}}$ -equivariant structure on $(V_{X,\check{\mu}} \otimes \mathrm{IC}_C)^{\boxtimes N_{\check{\mu}}}$ compatible with the $\mathfrak{S}_{N_{\check{\mu}}}$ -action on $C^{N_{\check{\mu}}}$. On the other hand, we have the map

$$(\overline{\mathbf{Y}}^{\check{\boldsymbol{\mu}}} \, \widetilde{\times} \, C)^{\mathring{\times} N_{\check{\boldsymbol{\mu}}}} \to \mathcal{Y}^{N_{\check{\boldsymbol{\mu}}} \cdot \check{\boldsymbol{\mu}}} \underset{\mathcal{A}^{N_{\check{\boldsymbol{\mu}}} \cdot \check{\boldsymbol{\mu}}}}{\times} \mathring{C}^{(N_{\check{\boldsymbol{\mu}}})}$$

which is a $\mathfrak{S}_{N_{\tilde{\mu}}}$ -torsor, where $\mathfrak{S}_{N_{\tilde{\mu}}}$ acts on the left hand side in the natural way. Now from the definition of $V_{X,\tilde{\mu}}$ we deduce that the isomorphism (6.14) must intertwine the $\prod \mathfrak{S}_{N_{\tilde{\mu}}}$ -structures. By Galois descent we conclude that $\mathcal{L}^{\mathfrak{R}}[|\mathfrak{R}|] \cong (\bigotimes_{\tilde{\mu}} \operatorname{Sym}^{N_{\tilde{\mu}}}(V_{X,\tilde{\mu}})) \otimes \operatorname{IC}_{\mathring{C}^{\mathfrak{R}}}$. Since $\bar{\iota}^{\mathfrak{R}} : C^{\mathfrak{R}} \to \mathcal{A}^{\check{\lambda}}$ is the normalization of the closure of the stratum $\mathring{C}^{\mathfrak{R}}$ in $\mathcal{A}^{\check{\lambda}}$, the middle extension of $\operatorname{IC}_{\mathring{C}^{\mathfrak{R}}}$ is isomorphic to $\bar{\iota}^{\mathfrak{R}}_{1}(\operatorname{IC}_{C^{\mathfrak{R}}})$.

When X is defined over a finite field \mathbb{F} and k is the algebraic closure, the Galois group of k over \mathbb{F} acts naturally on the set of components $\mathfrak{B}_X^+ := \bigcup_{\tilde{\mu} \in \mathfrak{c}_X} \mathfrak{B}_{X,\tilde{\mu}}$, and the isomorphisms above are clearly equivariant, taking into account that $\mathrm{IC}_{C^{\mathfrak{R}}}$ is the constant sheaf $\overline{\mathbb{Q}}_{\ell}(\frac{|\mathfrak{R}|}{2})[|\mathfrak{R}|]$. \square

6.5. **Critical dimension.** We can now give a more precise description of the diagonal contribution $V_{X,\check{\lambda}}$ defined in (6.11), that is, the free vector space on the set $\mathfrak{B}_{X,\check{\lambda}}$ of components of critical dimension in the "open strata" of the central fiber $\mathsf{Y}^{\check{\lambda}}$.

Proposition 6.5.1. For $\check{\lambda} \in \mathfrak{c}_X - 0$, the set $\mathfrak{B}_{X,\check{\lambda}}$ of components of critical dimension on the central fiber $\mathsf{Y}^{\check{\lambda}}$ consists of

- (i) the irreducible components of $\mathbb{Y}_{X^{\bullet}}^{D} \cap \mathsf{Y}^{\check{\lambda}}$ of dimension $\frac{1}{2}(\operatorname{len}(D) 1)$, for $D \in \mathbb{N}^{\mathcal{D}}$ with $\varrho_{X}(D) = \check{\lambda}$;
- (ii) the irreducible components of $S^{\check{\lambda}} \cap Gr_G^{\check{\theta}}$, for $\check{\theta} \in \mathcal{D}^G_{\operatorname{sat}}(X)$, embedded in $Y^{\check{\lambda}}$ via (5.16).

Remark 6.5.2. We have MV cycles for every $\check{\theta} \in \mathfrak{c}_X^-$, but only those belonging to $\mathcal{D}^G_{\operatorname{sat}}(X)$ contribute to $V_{X,\check{\lambda}}$ since those correspond to $\mathcal{Y}^{\check{\lambda},\check{\theta}}$ which are connected components of $\mathcal{Y}^{\check{\lambda}}$. We will reserve the term *critical dimension* of $\mathcal{Y}^{\check{\lambda}}$ for the maximal dimensions in the two cases above.

Proof. By definition, an element of $\mathfrak{B}_{X,\check{\lambda}}$ is a component b of the central fiber $Y^{\check{\lambda}} \cap \mathscr{Y}$, where \mathscr{Y} is an irreducible component of the smooth stratum $\mathcal{Y}^{\check{\lambda},\check{\theta}}$, for some $\check{\theta} \in \mathcal{D}^G_{\mathrm{sat}}(X) \cup \{0\}$, such that

$$\dim(\mathsf{b}) = \frac{1}{2}(\dim(\mathscr{Y}) - 1).$$

If $\check{\theta} = 0$, then $\mathcal{Y}^{\check{\lambda},0}$ is the disjoint union of connected components $\mathcal{Y}^D_{X\bullet}$ for $D \in \mathbb{N}^D$ with $\varrho_X(D) = \check{\lambda}$, by Lemma 6.2.1. Then (6.15) becomes $\dim(\mathsf{b}) = \frac{1}{2}(\operatorname{len}(D) - 1)$.

If $\check{\theta} \neq 0$, by Corollary 6.2.2 the connected components of $\mathcal{Y}^{\check{\lambda},\check{\theta}}$ are in bijection with the closures of

$$y_{X\bullet}^D \mathring{\times} y^{\check{\theta},\check{\theta}} \hookrightarrow y^{\check{\lambda},\check{\theta}}$$

for $D \in \mathbb{N}^{\mathcal{D}}$ such that $\varrho_X(D) = \check{\lambda} - \check{\theta}$. The statement now follows from the following lemma, which we write separately, for later use, because it applies to arbitrary $\check{\theta} \neq 0$.

Lemma 6.5.3. For any $\check{\theta} \in \mathfrak{c}_X^- - 0$, $D \in \mathbb{N}^{\mathcal{D}}$ and $\check{\lambda} = \varrho_X(D) + \check{\theta}$, if we denote by \mathscr{Y} the closure of the image of

$$y_{\mathbf{Y}\bullet}^{D} \mathring{\times} y^{\check{\theta},\check{\theta}} \hookrightarrow y^{\check{\lambda},\check{\theta}},$$

then we have $\dim(Y^{\check{\lambda}} \cap \mathscr{Y}) \leq \frac{1}{2} \operatorname{len}(D) = \frac{1}{2} (\dim \mathscr{Y} - 1)$. The irreducible components of $Y^{\check{\lambda},\check{\theta}}$ for which this is an equality are precisely the MV cycles in $S^{\check{\lambda}} \cap \overline{\operatorname{Gr}}_{G}^{\check{\theta}}$, embedded in $Y^{\check{\lambda}}$ via (5.16).

Proof. Proposition 6.1.1 implies that $\dim(\mathsf{Y}^{\check{\lambda}}\cap\mathscr{Y})\leq\frac{1}{2}(\dim\mathscr{Y}-1)$. Since $\check{\theta}\neq0$, we have $\mathscr{Y}_{\mathrm{red}}^{\check{\theta},\check{\theta}}=C$, and by Lemma 6.2.1 this inequality translates to $\dim(\mathsf{Y}^{\check{\lambda}}\cap\mathscr{Y})\leq\frac{1}{2}\ln(D)$.

If $v \in |C|$ is the point we are taking central fibers with respect to, then $\mathsf{Y}^{\check{\lambda}}$ maps to the substack $\mathcal{M}_{X,v} \subset \mathcal{M}_X$ of maps that are only G-degenerate at v. Recall from Theorem 5.1.1 that $\mathcal{M}_{X,v}^{\check{\theta}}$ is contained in the image of

$$\operatorname{act}_{\mathfrak{M},v}:\operatorname{Bun}_H\tilde{\times}\operatorname{Gr}_G^{\check{\theta}}\to\mathfrak{M}_{X,v}$$

and the map is birational onto its image. We deduce from (5.11) and Proposition 5.5.5 that the fiber product $(\operatorname{Bun}_H \tilde{\times} \operatorname{Gr}_G^{\tilde{\theta}}) \times_{\mathcal{M}_X} \mathsf{Y}^{\check{\lambda}}$ has a stratification by

$$\bigcup_{\check{\nu}}\mathsf{Y}^{\check{\lambda}-\check{\nu},0}\,\tilde{\times}(\mathsf{S}^{\check{\nu}}\cap\mathrm{Gr}_G^{\check{\theta}})$$

where $\check{\nu}$ ranges over the weights of $V^{\check{\theta}}$.

Observe that we have an embedding $\check{\Lambda}_G^{\text{pos}} \hookrightarrow \mathbb{N}^{\mathcal{D}}$ by sending a simple coroot $\check{\alpha} \mapsto D_{\alpha}^+ + D_{\alpha}^-$. By restricting to $X^{\circ}P_{\alpha}$ we can deduce that the image of $\check{\alpha}$ in $\pi_1(H) \otimes \mathbb{Q}$ under (6.4) is zero. Thus, the commutativity of diagram (6.5) ensures that if we restrict to the connected component of Bun_H corresponding to $\mathcal{Y}_{X\bullet}^D$, then the stratification above becomes

$$\bigcup_{\check{\chi}} (\mathsf{Y}^{\check{\lambda}-\check{\nu}}\cap \mathfrak{Y}_{X^{\bullet}}^{D-(\check{\nu}-\check{\theta})})\, \check{\times} (\mathsf{S}^{\check{\nu}}\cap \mathrm{Gr}_{G}^{\check{\theta}}),$$

where $\check{\nu} - \check{\theta} \in \check{\Lambda}_G^{\mathrm{pos}} \subset \mathbb{N}^{\mathcal{D}}$. In particular, the dimension of the stratum corresponding to $\check{\nu}$ is

$$\leq \frac{1}{2} \Bigl(\mathrm{len}(D - (\check{\nu} - \check{\theta})) - 1 \Bigr) + \langle \rho_G, \check{\nu} - \check{\theta} \rangle \leq \frac{1}{2} (\mathrm{len}(D) - 1)$$

by Proposition 6.1.1 unless $D = \check{\nu} - \check{\theta}$. Thus, in order for $\dim \mathsf{Y}^{\check{\lambda},\check{\theta}} = \frac{1}{2} \operatorname{len}(\check{\lambda} - \check{\theta})$, we must have $\check{\lambda} = \check{\nu}$ is a weight of $V^{\check{\theta}}$ and $\mathsf{Y}^{\check{\lambda},\check{\theta}}$ is birational to an irreducible component of $\operatorname{\mathsf{pt}} \check{\times} (\mathsf{S}^{\check{\lambda}} \cap \operatorname{Gr}_G^{\check{\theta}})$, i.e., a Mirković-Vilonen cycle. By Lemma 5.5.11, this latter case always occurs.

Remark 6.5.4. By Proposition 6.5.1, the irreducible components of central Zastava fibers of critical dimension, which give rise to the "new" contributions $V_{X,\tilde{\lambda}}$ to the pushforward of the IC sheaf by Proposition 6.4.1, are of two different kinds: those associated to the Zastava space of the open G-orbit X^{\bullet} , and those associated to certain strata of the affine Grassmannian. On the other hand, Theorem 5.1.5 gives a similar description of the intersection complex of the global model in terms of the Hecke action on the intersection complex of the global model for X^{\bullet} . These two descriptions "match" under the nearby cycles functor of Theorem 8.3.6 and the Hecke action on Drinfeld's compactification $\overline{\mathrm{Bun}}_{N^-}$ (cf. [BG02]).

7. The Crystal of a spherical variety

We keep the assumptions of §5–6. In this section, we study the irreducible components of central Zastava fibers of critical dimension (Proposition 6.5.1) which give rise to the "new" contributions $V_{X,\tilde{\lambda}}$ to the pushforward of the IC sheaf by Proposition 6.4.1. Our main result is that these components give rise to a crystal, in the sense of Kashiwara, if we formally attach to them their "negatives". These components are, by Proposition 6.5.1, of two different kinds, namely those associated to the Zastava space of the open G-orbit X^{\bullet} and those associated to certain strata of the affine Grassmannian. Since the relation of the latter to crystals is well-known by [BG01, BFG06], the problem quickly reduces to the study of the crystal associated to X^{\bullet} .

7.1. The crystal \mathfrak{B}_X .

7.1.1. Definition of crystal. We review the definition of crystal, in the sense of Kashiwara [Kas93], over the Langlands dual Lie algebra $\check{\mathfrak{g}}$. We refer the reader to [Kas94, Kas95, BS17, HK02] for further details on crystals, which can be associated to any Kac-Moody algebra.

Let I denote the set of vertices of the Dynkin diagram associated to G, so $\{\alpha_i\}_{i\in I} = \Delta_G$ is the set of simple roots of G.

A crystal $\mathfrak B$ over $\check{\mathfrak g}$ is a set with the following data:

$$\begin{split} \operatorname{wt} : \mathfrak{B} &\to \check{\Lambda}_G \\ \varepsilon_i, \varphi_i : \mathfrak{B} &\to \mathbb{Z} \sqcup \{-\infty\} \\ \tilde{e}_i, \tilde{f}_i : \mathfrak{B} &\to \mathfrak{B} \sqcup \{0\} \end{split} \qquad \qquad \text{for } i \in I, \end{split}$$

satisfying the following axioms:

- (1) $\varphi_i(\mathsf{b}) = \varepsilon_i(\mathsf{b}) + \langle \alpha_i, \operatorname{wt}(\mathsf{b}) \rangle$ for $\mathsf{b} \in \mathfrak{B}, i \in I$.
- (2) If $b \in \mathfrak{B}$ and $\tilde{e}_i b \neq 0$, then

$$\operatorname{wt}(\tilde{e}_i \mathsf{b}) = \operatorname{wt}(\mathsf{b}) + \check{\alpha}_i, \ \varepsilon_i(\tilde{e}_i \mathsf{b}) = \varepsilon_i(\mathsf{b}) - 1, \ \varphi_i(\tilde{e}_i \mathsf{b}) = \varphi_i(\mathsf{b}) + 1.$$

(3) If $b \in \mathfrak{B}$ and $\tilde{f}_i \neq 0$, then

$$\operatorname{wt}(\tilde{f}_i\mathsf{b}) = \operatorname{wt}(\mathsf{b}) - \check{\alpha}_i, \ \varepsilon(\tilde{f}_i\mathsf{b}) + 1, \ \varphi_i(\tilde{f}_i\mathsf{b}) = \varphi_i(\mathsf{b}) - 1.$$

- (4) For $b_1, b_2 \in \mathfrak{B}$, $b_2 = \tilde{f}_i b_1$ if and only if $b_1 = \tilde{e}_i b_2$.
- (5) If $\varphi_i(\mathbf{b}) = -\infty$, then $\tilde{e}_i \mathbf{b} = \tilde{f}_i \mathbf{b} = 0$.

A crystal \mathfrak{B} is called $seminormal^{17}$ if

$$\varepsilon_i(\mathsf{b}) = \max\{n \geq 0 \mid \tilde{e}_i^n \mathsf{b} \in \mathfrak{B}\} \in \mathbb{N}, \quad \varphi_i(\mathsf{b}) = \max\{n \geq 0 \mid \tilde{f}_i^n \mathsf{b} \in \mathfrak{B}\} \in \mathbb{N}$$

for all $b \in \mathfrak{B}, i \in I$. From now on we will only consider seminormal crystals, so the maps ε_i, φ_i are uniquely determined by wt, \tilde{e}_i, \tilde{f}_i .

Kashiwara showed the existence and uniqueness of crystal bases for any integrable module of the quantized enveloping algebra $U_q(\check{\mathfrak{g}})$. The crystal basis of an integrable $U_q(\check{\mathfrak{g}})$ -module is the limit at q=0 of Lusztig's canonical basis ([Lus90, GL93]). A crystal \mathfrak{B} is called *normal* if it is isomorphic to the crystal basis of an integrable $U_q(\check{\mathfrak{g}})$ -module.

For any subset $J \subset I$, let $\check{\mathfrak{g}}_J$ denote the corresponding Levi subalgebra. For a crystal \mathfrak{B} of $\check{\mathfrak{g}}$, let $\Phi_J(\mathfrak{B})$ denote \mathfrak{B} regarded as a crystal over $\check{\mathfrak{g}}_J$. Then saying that \mathfrak{B} is seminormal is equivalent to saying that $\Phi_{\{i\}}(\mathfrak{B})$ is isomorphic to the crystal basis of an integrable $U_q(\check{\mathfrak{g}}_{\{i\}})$ -module. One can check the normality of a crystal by restricting to every pair of vertices in the Dynkin diagram:

Proposition 7.1.2 ([KKM⁺92, Proposition 2.4.4], [BS17, Theorem 5.21]). Let \mathfrak{B} be a finite crystal over $\check{\mathfrak{g}}$ such that for every subset $\{i,j\} \subset I$, the crystal $\Phi_{\{i,j\}}(\mathfrak{B})$ is isomorphic to the crystal basis of a finite-dimensional $U_q(\check{\mathfrak{g}}_{\{i,j\}})$ -module. Then \mathfrak{B} is normal.

For a crystal \mathfrak{B} one can construct an oriented *crystal graph* with vertex set \mathfrak{B} and edges given by the \tilde{f}_i . We can decompose \mathfrak{B} into a disjoint union of crystals corresponding to the connected components of the crystal graph. We will call these the connected components of \mathfrak{B} .

For $\check{\lambda} \in \check{\Lambda}_G^+$, there is a unique crystal basis $\mathfrak{B}_{\check{\mathfrak{g}}}^{\lambda}$ for the irreducible highest weight module V^{λ} of $U_q(\check{\mathfrak{g}})$. (We will abuse notation and use $V^{\check{\lambda}}$ to denote both the representation of the quantized enveloping algebra and its classical limit at q=1, which is the corresponding irreducible $\check{\mathfrak{g}}$ -module.) In other words, there is a unique normal connected crystal with highest weight vector

 $^{^{17}}$ This is the terminology of [Kas94, Kas95]. In [Kas93] the term normal was used for what we call seminormal.

of weight $\check{\lambda}$. However, we warn that in general there may be many seminormal connected crystals with the same property.

Given a crystal \mathfrak{B} , we can define a crystal \mathfrak{B}^{\vee} by "reversing the arrows": the set $\mathfrak{B}^{\vee} = \{b^{\vee} \mid b \in \mathfrak{B}\}$ is formally the same as \mathfrak{B} , and wt(b^{\vee}) = -wt(b). The roles of \tilde{e}_i , \tilde{f}_i are swapped. The crystal $(\mathfrak{B}_{\tilde{\mathfrak{g}}}^{\check{\lambda}})^{\vee}$ is isomorphic to the crystal basis of the irreducible $U_q(\tilde{\mathfrak{g}})$ -module of lowest weight $-\check{\lambda}$, which we also denote by $V^{-\check{\lambda}}$.

7.1.3. Let us mention an important consequence of the structure of a seminormal crystal \mathfrak{B} . Let \widetilde{W} be the free group generated by $\{s_i \mid i \in I\}$ with the relation $s_i^2 = 1$. The Weyl group W is the quotient of \widetilde{W} by the braid relations.

It follows from the classification of integrable $U_q(\mathfrak{sl}_2)$ -modules that we have a natural action of \widetilde{W} on \mathfrak{B} defined by

$$s_i(\mathsf{b}) = \begin{cases} \tilde{f}_i^{\langle \alpha_i, \operatorname{wt}(\mathsf{b}) \rangle}(\mathsf{b}) & \text{if } \langle \alpha_i, \operatorname{wt}(\mathsf{b}) \rangle \ge 0\\ \tilde{e}_i^{-\langle \alpha_i, \operatorname{wt}(\mathsf{b}) \rangle}(\mathsf{b}) & \text{if } \langle \alpha_i, \operatorname{wt}(\mathsf{b}) \rangle \le 0 \end{cases}$$

for $b \in \mathfrak{B}_{X^{\bullet}}$. For $\tilde{w} \in \widetilde{W}$ we have $\operatorname{wt}(\tilde{w}b) = \tilde{w}(\operatorname{wt}(b))$, where \widetilde{W} acts on $\check{\Lambda}_G = \check{\Lambda}_X$ through W. In other words, we have isomorphisms

$$(7.1) \tilde{w}: \mathfrak{B}_{\check{\lambda}} \stackrel{\sim}{\to} \mathfrak{B}_{\check{w}\check{\lambda}}$$

a priori depending on $\widetilde{w} \in \widetilde{W}$ for all $\check{\lambda} \in \check{\Lambda}_G$.

If \mathfrak{B} is normal, the \widetilde{W} -action on \mathfrak{B} factors through W.

7.1.4. Definition of \mathfrak{B}_X . For $\check{\lambda} \in \mathfrak{c}_X$, we have defined the set $\mathfrak{B}_{X,\check{\lambda}}$ to consist of the irreducible components of $\mathsf{Y}^{\check{\lambda}}$ of critical dimension (Proposition 6.5.1), that is:

- if $\check{\lambda} \in \mathfrak{c}_X^{\mathcal{D}}$, the irreducible components of $\mathsf{Y}^{\check{\lambda}}$ (or equivalently, of $\mathsf{Y}_{X^{\bullet}}^{\check{\lambda}} = \mathsf{Y}^{\check{\lambda},0}$) of dimension $\frac{1}{2}(\operatorname{len}(\check{\lambda}) 1)$;
- the irreducible components of $S^{\check{\lambda}} \cap Gr_G^{\check{\theta}}$ of dimension $\langle \rho_G, \check{\lambda} \check{\theta} \rangle$, for $\check{\theta} \in \mathcal{D}^G_{\text{sat}}(X)$, identified with their image in $Y^{\check{\lambda}}$ through the action map (5.16).

Note that $\check{\lambda} = 0$ never satisfies the conditions above.

Define $\mathfrak{B}_{X,-\check{\lambda}}:=\mathfrak{B}_{X,\check{\lambda}}$, which is well-defined since $\mathcal{C}_0(X)$ is strictly convex. Let

$$\mathfrak{B}_X^+ = \bigcup_{\check{\lambda} \in \mathfrak{c}_X} \mathfrak{B}_{X,\check{\lambda}}, \qquad \mathfrak{B}_X^- = \bigcup_{\check{\lambda} \in \mathfrak{c}_X} \mathfrak{B}_{X,-\check{\lambda}}.$$

In other words \mathfrak{B}_X^+ is the set of all irreducible components of the central fiber of \mathcal{Y} of the maximal dimensions satisfying the semi-smallness equality.

We (rather artificially) define $\mathfrak{B}_X = \mathfrak{B}_X^+ \sqcup \mathfrak{B}_X^-$. Let wt : $\mathfrak{B}_X \to \mathfrak{c}_X$ be the map sending $\mathfrak{B}_{X,\check{\lambda}}$ to $\check{\lambda}$.

Theorem 7.1.5. The set \mathfrak{B}_X has the structure of a semi-normal crystal over $\check{\mathfrak{g}}$ such that the defining bijection $\mathfrak{B}_X^+ \leftrightarrow \mathfrak{B}_X^-$ is an isomorphism of crystals $\mathfrak{B}_X \cong \mathfrak{B}_X^{\vee}$.

The statement of the theorem above is not optimal, of course, as it does not specify all the data that give rise to the structure of a crystal, such as the operations \tilde{e}_i , \tilde{f}_i . To do so, we will need to introduce a process of "reduction to a Levi", in particular, a Levi of semisimple rank one, that will provide these operators. We will define these operators, giving the structure of a self-dual semi-normal crystal to \mathfrak{B}_X , in §7.2.5.

Conjecture 7.1.6. The crystal \mathfrak{B}_X is isomorphic to the unique crystal basis of a finite-dimensional \check{G} -module ρ_X .

In Remark 7.2.6 we explain that it suffices to prove Conjecture 7.1.6 when G has semisimple rank 2, where there are finitely many cases (corresponding to the wonderful varieties in [Was96] with only spherical roots of type T).

7.1.7. Reduction to X^{\bullet} . Theorem 7.1.5 and Conjecture 7.1.6 immediately reduce to the study of the irreducible components of critical dimension in the central fiber of $\mathcal{Y}_{X^{\bullet}}$:

Lemma 7.1.8. Let $\mathfrak{B}_{X\bullet}^+ = \mathfrak{B}_X \times_{\check{\Lambda}_X} \mathfrak{c}_X^{\mathcal{D}}$, hence $\mathfrak{B}_{X\bullet}^+$ is the set of irreducible components of central fibers of $\mathfrak{Y}_{X\bullet}$ of critical dimension. Define $\mathfrak{B}_{X\bullet}^-$ and $\mathfrak{B}_{X\bullet} = \mathfrak{B}_{X\bullet}^+ \sqcup \mathfrak{B}_{X\bullet}^-$ as before. Then, Theorem 7.1.5 and Conjecture 7.1.6 hold if they hold for $\mathfrak{B}_{X\bullet}$, with a decomposition of the crystal \mathfrak{B}_X into a disjoint union of crystals:

(7.2)
$$\mathfrak{B}_{X} = \mathfrak{B}_{X^{\bullet}} \sqcup \bigsqcup_{\tilde{\theta} \in \pm \mathcal{D}_{\text{sat}}^{G}(X)} \mathfrak{B}_{\tilde{\mathfrak{g}}}^{\tilde{\theta}},$$

where $\mathfrak{B}^{\check{\theta}}_{\check{\mathfrak{g}}}$ is the crystal associated to the irreducible \check{G} -module $V^{\check{\theta}}$ of lowest weight $\check{\theta}$, if $\check{\theta} \in \mathcal{D}^G_{\mathrm{sat}}(X) \subset \check{\Lambda}^-_G$, or highest weight $\check{\theta}$ if $-\check{\theta} \in \mathcal{D}^G_{\mathrm{sat}}(X)$.

Proof. Indeed, by Proposition 6.5.1, the elements of \mathfrak{B}_X^+ consist of elements of $\mathfrak{B}_{X^{\bullet}}^+$ and irreducible components of $\mathsf{S}^{\check{\lambda}} \cap \mathrm{Gr}_G^{\check{\theta}}$ of dimension $\langle \rho_G, \check{\lambda} - \check{\theta} \rangle$, for $\check{\theta} \in \mathcal{D}_{\mathrm{sat}}^G(X)$. As explained in [BG01], the latter can be identified with the elements of the crystal basis in the $\check{\lambda}$ -eigenspace of the irreducible \check{G} -module $V^{\check{\theta}}$ of lowest weight $\check{\theta}$.

While the methods of this paper are insufficient to prove Conjecture 7.1.6 for $\mathfrak{B}_{X^{\bullet}}$, we do show that it must satisfy the following properties in §7.3.

Theorem 7.1.9. The crystal $\mathfrak{B}_{X\bullet}$ has the following properties:

- (i) The set $\operatorname{wt}(\mathfrak{B}_{X^{\bullet}})$ is equal¹⁸ to the set of weights of $\bigoplus_{\check{\lambda}\in\check{\Lambda}^+_G\cap W\varrho_X(\mathcal{D})}V^{\check{\lambda}}$, where $\check{\Lambda}^+_G\cap W\varrho_X(\mathcal{D})$ denotes the dominant Weyl translates of valuations of colors.
- (ii) If $b \in \mathfrak{B}_{X^{\bullet}}^+$, then there is a sequence of lowering operators \tilde{f}_{i_j} sending b to an element of $\mathfrak{B}_{X^{\bullet},\check{\nu}_D}$ for some color $D \in \mathcal{D}$.
- (iii) For $\check{\lambda} \in W \varrho_X(\mathcal{D})$, the cardinality of $\mathfrak{B}_{X^{\bullet},\check{\lambda}}$ is equal to 1, unless $\check{\lambda} = \frac{\check{\gamma}}{2}$ for some (not necessarily simple) coroot $\check{\gamma}$, in which case the cardinality is 2.

Remark 7.1.10. The "multiplicity 2" case appears when two colors have the same valuation, e.g., $X = \mathbb{G}_m \backslash PGL_2$; see §2.1.

Remark 7.1.11. Note that if Conjecture 7.1.6 is true, then properties (i)–(iii) of Theorem 7.1.9 uniquely determine the \check{G} -module ρ_X : it must be isomorphic to

(7.3)
$$\bigoplus_{\check{\lambda}\in\check{\Lambda}_{G}^{+}\cap W\varrho_{X}(\mathcal{D})} (V^{\check{\lambda}})^{\oplus|\mathfrak{B}_{X^{\bullet},\check{\lambda}}|} \oplus \bigoplus_{\check{\theta}\in\pm\mathcal{D}_{\mathrm{sat}}^{G}(X)} V^{\check{\theta}},$$

where the cardinality of $\mathfrak{B}_{X^{\bullet},\check{\lambda}}$ is specified by property (iii).

Corollary 7.1.12. If all coweights in $\check{\Lambda}_{G}^{+} \cap W \varrho_{X}(\mathcal{D})$ are minuscule, then Conjecture 7.1.6 holds, i.e., \mathfrak{B}_{X} is the crystal basis of the \check{G} -module given by (7.3).

Proof. This is immediate from Theorems 7.1.5, 7.1.9 and §7.1.3 after we make the assumption $\mathfrak{c}_{X^{\bullet}} = \mathbb{N}^{\mathcal{D}}$, which is allowed by (5.6).

We also show in Corollary 7.3.4 that if X is affine homogeneous (equivalently, H is reductive), then all coweights in $\check{\Lambda}_G^+ \cap W \varrho_X(\mathcal{D})$ must be minuscule.

problem: what if two colors belong to same weightmultiplicity-free irrep?

 $^{^{18}}$ Here we only describe an equality of sets counted without multiplicities.

7.2. Reduction to Levi. From now on, having reduced the problem to giving a crystal structure to the set $\mathfrak{B}_{X^{\bullet}}$, we may (by (5.6)) and will assume, unless otherwise specified, that $X = X^{\operatorname{can}}$ and $\mathfrak{c}_X \cong \mathbb{N}^{\mathcal{D}}$. Under this assumption, $\mathfrak{Y}^{\check{\lambda},0}$ is dense in $\mathfrak{Y}^{\check{\lambda}}$ by Corollary 5.7.2, and $\mathfrak{B}_X = \mathfrak{B}_{X^{\bullet}}$. Moreover, any $\check{\lambda} \succeq 0$ is an element of $\mathbb{N}^{\mathcal{D}}$, so the length function len is a function of $\check{\lambda}$.

Let P be a standard parabolic subgroup of G, i.e., $P \supset B$. Let N_P denote its unipotent radical and $M = P/N_P$ the Levi quotient. Observe that the map $X \to X /\!\!/ N$ factors through $X \to X /\!\!/ N_P \to X /\!\!/ N$. Set

$$X_M := X /\!\!/ N_P = \operatorname{Spec} k[X]^{N_P}.$$

Then X_M is an affine spherical M-variety and the map $X \to X_M$ is M-equivariant. However, note that even if $X = X^{\operatorname{can}}$, it will not in general be true that X_M is the canonical embedding of X_M^{\bullet} . We will use this to our advantage later, using the crystals of Lemma 7.1.8 to produce the \tilde{e}_i , \tilde{f}_i operations, when M is taken to have semisimple rank one.

For now, we work with a general parabolic P. Let B_M denote the image of the Borel subgroup B in M. We have $k[X]^{(B)} = k[X_M]^{(B_M)}$, therefore $\mathfrak{c}_{X_M} = \mathfrak{c}_X$. On the other hand, $\mathfrak{c}_{X_M}^- = \mathfrak{c}_{X_M} \cap \check{\Lambda}_M^-$ is, in general, larger than \mathfrak{c}_X^- . The open P-orbit $X^{\circ}P$ maps to the open M-orbit X_M^{\bullet} , and we have

Lemma 7.2.1. The preimage of X_M^{\bullet} under the quotient map $X \to X_M$ coincides with the open P-orbit $X^{\circ}P$, and the quotient stacks $X^{\circ}P/P$ and X_M^{\bullet}/M are isomorphic.

Proof. A color $D \in \mathcal{D}$ belongs to the open P-orbit $X^{\circ}P$ if and only if $D \in \mathcal{D}(\alpha)$ for some $\alpha \in \Delta_M$; otherwise, it is P-stable, and induces an M-stable valuation on $k(X_M)$, which is the function field of $k[X]^{N_P}$. This valuation is nontrivial (because it is nontrivial on $k[X]^{(B)} = k[X_M]^{(B_M)}$), therefore the image of D cannot belong to X_M^{\bullet} .

Since N acts freely on X° , the subgroup N_P acts freely on $X^{\circ}P$, and therefore $X^{\circ}P/P = (X^{\circ}P/N_P)/M = X^{\bullet}_M/M$.

Define the parabolic Zastava model

$$\mathcal{Y}_{X,P} := \operatorname{Maps}_{\operatorname{gen}}(C, X/P \supset X^{\circ}P/P) \subset \mathcal{M}_X \underset{\operatorname{Bun}_G}{\times} \operatorname{Bun}_P,$$

which naturally maps to Bun_P . The Cartesian diagram

$$X/B \longrightarrow X/P$$

$$\downarrow \qquad \qquad \downarrow$$

$$X_M/B_M \longrightarrow X_M/M$$

gives rise to a diagram

$$\begin{array}{cccc}
\mathbf{Y}_{X}^{\check{\lambda}} & \longrightarrow & \mathcal{Y}_{X} & \longrightarrow & \mathcal{Y}_{X,P} \\
\downarrow & & \downarrow & & \downarrow \\
\mathbf{Y}_{X_{M}}^{\check{\lambda}} & \longrightarrow & \mathcal{Y}_{X_{M}} & \longrightarrow & \mathcal{M}_{X_{M}}
\end{array}$$
(7.4)

with all squares Cartesian. Central fibers are taken with respect to a fixed point $v \in |C|$.

Our goal is to study the components of critical dimension of Y_X^λ in terms of $\mathsf{Y}_{X_M}^\lambda$ and the fibers of the map $\pi_{X,P}$. At this point, it will be critical to distinguish the stratum $\mathsf{Y}_{X_M}^{\check{\lambda},\check{\theta}}$ where the image of the generic point of a component $\mathsf{b} \in \mathfrak{B}_{X,\check{\lambda}}$ lies, that is, the $M(\mathfrak{o}_v)$ -orbit of its image in $X_M(\mathfrak{o}_v)$ (after trivialization in a formal neighborhood of v). The reason is, as we are

about to see, that this stratum will completely determine the fiber of the map $\pi_{X,P}$ over the image.

Indeed, recall from §4.3 that lifting a point from \mathcal{M}_{X_M} to $\mathsf{Y}_{X_M}^{\check{\lambda}}$ induces a trivialization of the corresponding G-bundle away from v (depending on a fixed choice of base point $x_0 \in X^{\circ}$), which identifies the central fiber $\mathsf{Y}_{X_M}^{\check{\lambda}}$ with a subscheme of the affine Grassmannian Gr_M . Moreover, the map $\mathsf{Y}_{X_M}^{\check{\lambda}} \to \mathcal{M}_{X_M}$ factors through the map

$$\operatorname{Gr}_M \underset{\mathsf{L}X_M/\mathsf{L}^+M}{\times} (\mathsf{L}^+X_M/\mathsf{L}^+M) \to \mathfrak{M}_{X_M}.$$

as defined in (5.15). Similarly, the map $\mathsf{Y}_X^{\check{\lambda}} \to \mathcal{Y}_{X,P}$ factors through

$$\operatorname{Gr}_P \underset{\mathsf{L}X/\mathsf{L}^+P}{\times} (\mathsf{L}^+X/\mathsf{L}^+P) \to \mathcal{Y}_{X,P}.$$

Hence we have a commutative diagram

$$(7.5) \qquad \begin{array}{c} Y_X^{\check{\lambda}} \stackrel{\iota_{X,P}}{\longrightarrow} \operatorname{Gr}_P \underset{\mathsf{L}X/\mathsf{L}^+P}{\times} (\mathsf{L}^+X/\mathsf{L}^+P) \stackrel{\operatorname{act}_v}{\longrightarrow} \mathfrak{Y}_{X,P} \\ \downarrow^q \qquad \qquad \qquad \downarrow \qquad \qquad \downarrow^{\pi_{X,P}} \\ Y_{X_M}^{\check{\lambda}} \stackrel{\iota_{X_M}}{\longrightarrow} \operatorname{Gr}_M \underset{\mathsf{L}X_M/\mathsf{L}^+M}{\times} (\mathsf{L}^+X_M/\mathsf{L}^+M) \stackrel{\operatorname{act}_v}{\longrightarrow} \mathfrak{M}_{X_M} \end{array}$$

with all squares Cartesian.

Let H_M be the stabilizer in P of the base point x_0 . By Lemma 7.2.1, it is isomorphic to the stabilizer in M of the image of x_0 in X_M . We first note:

Lemma 7.2.2. For any $\check{\theta} \in \mathfrak{c}_{X_M}^-$, the fibers of the map of ind-schemes

(7.6)
$$\operatorname{Gr}_{P} \underset{\mathsf{L}X/\mathsf{L}^{+}P}{\times} (\mathsf{L}^{+}X/\mathsf{L}^{+}P) \to \operatorname{Gr}_{M} \underset{\mathsf{L}X_{M}/\mathsf{L}^{+}M}{\times} (\mathsf{L}^{+}X_{M}/\mathsf{L}^{+}M)$$

over the stratum $\mathsf{L}^{\check{\theta}}X_M/\mathsf{L}^+M$ are isomorphic under the $\mathsf{L}H_M$ -action, and this action gives rise to a canonical bijection between the irreducible components of any two fibers.

More precisely, all fibers are isomorphic to $\{t^{\check{\theta}}\}\times_{\mathcal{M}_{X_M}}\mathcal{Y}_{X,P}$ and of dimension $\leq \frac{1}{2}(\operatorname{len}(\check{\theta})-1)$, unless $\check{\theta}=0$, in which case the restriction of (7.6) to the $\check{\theta}$ -stratum is an isomorphism.

Notice that, under our assumption that $\mathfrak{c}_X \cong \mathbb{N}^{\mathcal{D}}$ since the beginning of this subsection, len $(\check{\theta})$ makes sense.

Proof. Since LH_M acts transitively on $Gr_M \times_{LX_M/L^+M} (L^{\check{\theta}}X_M/L^+M)$, the fibers of (7.6) over the $\check{\theta}$ -stratum are all isomorphic.

We may now choose the point $t^{\check{\theta}} \in \mathsf{Y}_{X_M}^{\check{\theta}}$, whose image in $\mathsf{L}^+X_M/\mathsf{L}^+M$ lies in the $\check{\theta}$ -stratum — in fact, by Corollary 5.5.6, $\mathsf{Y}_{X_M}^{\check{\theta},\check{\theta}} = \{t^{\check{\theta}}\}$. By Corollary 2.3.13, the stabilizer in $\mathsf{L}H_M$ of its image $t^{\check{\theta}} \in \mathsf{Gr}_M$ is connected. Therefore, the action of $\mathsf{L}H_M$ induces a *canonical* bijection between irreducible components of the fibers.

Finally, if $\check{\theta} \neq 0$, the dimension of $\mathsf{Y}_X^{\check{\theta}}$ is $\leq \frac{1}{2}(\operatorname{len}(\check{\theta}) - 1)$, as explained in Proposition 6.5.1, and therefore so is, *a fortiori*, the dimension of the fiber over $\mathsf{Y}_{X_M}^{\check{\theta},\check{\theta}} = \{t^{\check{\theta}}\}$. For $\check{\theta} = 0$, we observe that

$$\mathcal{Y}_{X,P} \underset{\mathcal{M}_{X_M}}{\times} \mathcal{M}_{X_M}^0 = \mathrm{Maps}(C, X^{\circ}P/P) = \mathrm{Maps}(C, X_M^{\bullet}/M) = \mathcal{M}_{X_M}^0,$$

by Lemma 7.2.1, so the fibers are singletons.

Remark 7.2.3. At this point, we would like to emphasize a fine point in the arguments that follow: Consider the decomposition of $\mathfrak{B}_{X_M}^+$ according to (7.2) (restricted to the +-part):

$$\mathfrak{B}_{X_M}^+=\mathfrak{B}_{X_M^\bullet}^+\sqcup\bigsqcup_{\check{\theta}\in\mathcal{D}_{\mathrm{sat}}^M(X_M)}\mathfrak{B}_{\check{\mathfrak{m}}}^{\check{\theta}}$$

(where we have denoted $\mathfrak{B}^{\check{\theta}}$ by $\mathfrak{B}_{\check{\mathfrak{m}}}^{\check{\theta}}$, to emphasize that it corresponds to the $\check{\theta}$ -lowest weight crystal of a $\check{\mathfrak{m}}$ -module). We will not claim that the map q of (7.4) induces a map from \mathfrak{B}_X^+ to $\mathfrak{B}_{X_M}^+$. Indeed, the generic fiber of a $\mathsf{b} \in \mathfrak{B}_X^+$ may map to the image of $\mathsf{S}_M^{\check{\lambda}} \cap \mathsf{Gr}_M^{\check{\theta}} \to \mathsf{Y}_{X_M}^{\check{\lambda}}$ (where $\mathsf{S}_M^{\check{\lambda}}$ denotes the semi-infinite orbit corresponding to $\check{\lambda}$ in Gr_M), for some $\check{\theta} \in \mathfrak{c}_{X_M}^-$ that is not an element of $\mathcal{D}_{\mathrm{sat}}^M(X_M)$. These are the MV cycles that were discussed in Remark 6.5.2, which are not "of critical dimension" in terms of X_M . Representation-theoretically, if we believe that \mathfrak{B}_X corresponds to a representation of \check{G} (as predicted by Conjecture 7.1.6), this just says that the \check{M} -lowest weights of the spans of some vectors do not need to be extremal in $\mathfrak{c}_{X_M}^-$; however, in §7.3 we will see that there are weight-lowering operators \check{f}_i , possibly corresponding to roots not in \check{M} , which eventually lower such weights to the weight of a color.

For that reason, for the following proposition, which is the main technical result of this subsection, we denote by $\mathfrak{B}_{\tilde{\mathfrak{m}}}^{\check{\theta}}$ the crystal corresponding to the representation of $\check{\mathfrak{m}}$ of lowest weight $\check{\theta}$, that is, the set of irreducible components of $\mathsf{S}_{M}^{\check{\lambda}}\cap \mathrm{Gr}_{M}^{\check{\theta}}$, for any $\check{\theta}\in\mathfrak{c}_{X_{M}}^{-}$.

Proposition 7.2.4. For any $\check{\lambda} \in \mathfrak{c}_X^{\mathcal{D}}$, the diagram (7.4) induces a canonical decomposition

$$\mathfrak{B}_{X^{\bullet},\check{\lambda}} = \mathfrak{B}_{X^{\bullet}_{M},\check{\lambda}} \sqcup \bigsqcup_{\check{\theta} \in \mathfrak{c}_{X,\ell}^{-} - 0} \mathfrak{B}_{X,\check{\theta}}^{P} \times \mathfrak{B}_{\check{\mathfrak{m}},\check{\lambda}}^{\check{\theta}},$$

where $\mathfrak{B}_{X,\check{\theta}}^P$ denotes the set of irreducible components of $\{t^{\check{\theta}}\}\times_{\mathfrak{M}_{X_M}}\mathfrak{Y}_{X,P}$ of dimension $\frac{1}{2}(\operatorname{len}(\check{\theta})-1)$.

Taking the union over all such $\check{\lambda}$, we get

$$\mathfrak{B}_{X^{\bullet}}^{+} = \mathfrak{B}_{X_{M}^{\bullet}}^{+} \sqcup \bigsqcup_{\check{\theta} \in \mathfrak{c}_{X_{M}}^{-} - 0} \mathfrak{B}_{X,\check{\theta}}^{P} \times \mathfrak{B}_{\check{\mathfrak{m}}}^{\check{\theta}}.$$

The set $\mathfrak{B}_{X,\check{\theta}}^P$ should be thought of as the multiplicity space for the irreducible representation with basis $\mathfrak{B}_{\check{\mathfrak{m}}}^{\check{\mathfrak{g}}}$, and is something of a "black box" to us.

Proof. The dimension yoga here goes as follows: Let $b \in \mathfrak{B}_{X^{\bullet},\check{\lambda}}$, and suppose that its generic point lands in the stratum $\mathsf{Y}_{X_M}^{\check{\lambda},\check{\theta}}$ under the map q of (7.5).

If $\check{\theta}=0$, then $\check{\lambda}\succeq_{X_M^{\bullet}}0$, i.e., it belongs to the positive span of colors in X_M , hence $\operatorname{len}(\check{\lambda})$ is the same, whether we define it with respect to X or with respect to X_M . By Lemma 7.2.2 the irreducible components of critical dimension $\frac{1}{2}(\operatorname{len}(\check{\lambda})-1)$ of Y_X and $\mathsf{Y}_{X_M^{\bullet}}$ are in bijection, hence the set of $\mathsf{b}\in\mathfrak{B}_{X,\check{\lambda}}$ which map generically to $\mathsf{Y}_{X_M}^{\check{\lambda},0}$ is identified with $\mathfrak{B}_{X_M^{\bullet},\check{\lambda}}$.

If $\check{\theta} \neq 0$, then q sends b to $\mathsf{Y}_{X_M}^{\check{\lambda},\succeq\check{\theta}}$, which has dimension $\leq \frac{1}{2}\operatorname{len}(\check{\lambda}-\check{\theta})$ by Lemma 6.5.3. On the other hand, Lemma 7.2.2 states that the dimension of the corresponding fibers of $\pi_{X,P}$ is $\leq \frac{1}{2}(\operatorname{len}(\check{\theta})-1)$. Thus, the only way that b is of critical dimension $\frac{1}{2}(\operatorname{len}(\check{\lambda})-1)$ is if both inequalities are equalities. In this case Lemma 6.5.3 implies that $\check{\lambda} \geq \check{\theta}$ and the generic point of b is sent under the map $(\operatorname{act}_v \circ \iota_X, \iota_{X_M} \circ q)$ to an element of $\mathfrak{B}_{X,\check{\theta}}^P \times \mathfrak{B}_{\check{\mathfrak{m}},\check{\lambda}}^{\check{\theta}}$. Vice versa, for any irreducible component (MV cycle) of $\mathsf{S}_M^{\check{\lambda}} \cap \operatorname{Gr}_M^{\check{\theta}}$ (i.e., every element of $\mathfrak{B}_{\check{\mathfrak{m}},\check{\lambda}}^{\check{\theta}}$), Lemma 5.5.11

guarantees that it corresponds to a component b' of $\mathsf{Y}_{X_M}^{\check{\lambda},\check{\theta}}$ of the same dimension, and Lemma 7.2.2 ensures that the components of $\mathsf{Y}_X^{\check{\lambda}}$ of critical dimension in the preimage of b' are in canonical bijection with $\mathfrak{B}_{X|\check{\theta}}^P$.

7.2.5. Kashiwara operations \tilde{e}_i , \tilde{f}_i . For $i \in I$ let $P_i = P_{\alpha_i}$ denote the corresponding parabolic subgroup of semisimple rank one. Let M_i denote the Levi factor. Then the Langlands dual Lie algebra $\check{\mathfrak{m}}_i$ equals $\check{\mathfrak{g}}_{\{i\}}$ in our previous notation. Applying Proposition 7.2.4 to M_i we get the disjoint union

$$\mathfrak{B}_{X^{\bullet}}^{+} = \mathfrak{B}_{X_{M_{i}}^{\bullet}}^{+} \sqcup \bigsqcup_{\check{\theta} \in \mathfrak{c}_{i}^{-} - 0} \mathfrak{B}_{X,\check{\theta}}^{P_{i}} \times \mathfrak{B}_{\check{\mathfrak{m}}_{i}}^{\check{\theta}},$$

where $\mathfrak{c}_i^- = \mathfrak{c}_{X_{M_i}}^-$.

Now, by our "type T" assumption (see §2.1.1), $X_{M_i}^{\bullet}/B_{M_i} = \mathbb{G}_m \backslash \mathbb{P}^1$ as stacks. Therefore, $\mathcal{Y}_{X_{M_i}^{\bullet}} = \operatorname{Sym} C \overset{\circ}{\times} \operatorname{Sym} C$ (see Example 3.3.1) and $\mathfrak{B}_{X_{M_i}^{\bullet}}^{+}$ consists of two elements, which can be identified with their images $\check{\nu}_i^{\pm}$ in $\check{\Lambda}_X$, $\check{\nu}_i^{\pm} = \varrho_X(D_{\alpha_i}^{\pm})$ are the valuations of the two colors in $\mathcal{D}(\alpha_i)$. Therefore, we have a bijection of sets

$$\mathfrak{B}_{X^{\bullet}}=\mathfrak{B}_{X^{\bullet}}^{+}\cup\mathfrak{B}_{X^{\bullet}}^{-}=\big\{\check{\nu}_{i}^{+},\check{\nu}_{i}^{-},-\check{\nu}_{i}^{+},-\check{\nu}_{i}^{-}\big\}\sqcup\bigsqcup_{\check{\theta}\in\mathfrak{c}_{i}^{-}-0}\mathfrak{B}_{X,\check{\theta}}^{P}\times\big(\mathfrak{B}_{\check{\mathfrak{m}}_{i}}^{\check{\theta}}\sqcup(\mathfrak{B}_{\check{\mathfrak{m}}_{i}}^{\check{\theta}})^{\vee}\big).$$

Observe that $\{\check{\nu}_i^+, -\check{\nu}_i^-\}$ is in bijection with the normal crystal $\mathfrak{B}_{\check{\mathfrak{m}}_i}^{\check{\nu}_i^+}$ since $\langle \alpha_i, \check{\nu}_i^+ \rangle = 1$ and $\check{\nu}_i^+ - \check{\alpha}_i = -\check{\nu}_i^-$. We also observe that $\{\check{\nu}_i^-, -\check{\nu}_i^+\} = \mathfrak{B}_{\check{\mathfrak{m}}_i}^{\check{\nu}_i^-} = (\mathfrak{B}_{\check{\mathfrak{m}}_i}^{\check{\nu}_i^+})^{\vee}$ as sets. Now we simply define the operations \tilde{e}_i, \tilde{f}_i such that

$$(7.9) \qquad \Phi_{\{i\}}(\mathfrak{B}_{X^{\bullet}}) = \mathfrak{B}_{\tilde{\mathfrak{m}}_{i}}^{\check{\nu}_{i}^{+}} \sqcup \mathfrak{B}_{\tilde{\mathfrak{m}}_{i}}^{\check{\nu}_{i}^{-}} \sqcup \bigsqcup_{\check{\theta} \in \mathfrak{c}_{i}^{-} = 0} \mathfrak{B}_{X,\check{\theta}}^{P} \times \left(\mathfrak{B}_{\tilde{\mathfrak{m}}_{i}}^{\check{\theta}} \sqcup (\mathfrak{B}_{\tilde{\mathfrak{m}}_{i}}^{\check{\theta}})^{\vee}\right)$$

as normal crystals over $\check{\mathfrak{m}}_i$, where $\mathfrak{B}_{X,\check{\theta}}^P$ is treated as an abstract set. This gives the structure of a seminormal crystal over $\check{\mathfrak{g}}$ to \mathfrak{B}_X , such that the bijection $\mathfrak{B}_X^+ \leftrightarrow \mathfrak{B}_X^-$ identifies it with its dual. This completes the proof of Theorem 7.1.5.

Remark 7.2.6. The decomposition (7.7) gives a decomposition into crystals over $\check{\mathfrak{m}}$. Therefore, if we consider Proposition 7.2.4 for all standard parabolics corresponding to $\{i,j\}\subset I$, then Proposition 7.1.2 implies that \mathfrak{B}_X is normal (i.e., Conjecture 7.1.6 holds) if $\mathfrak{B}_{X_M^{\bullet}}$ is normal for all M of semisimple rank 2.

The discussion of $\S7.2.5$ also leads to the following observation:

Lemma 7.2.7. The W-orbit of $\varrho_X(\mathcal{D})$ is contained in $\mathfrak{c}_X^{\mathcal{D}} \sqcup -\mathfrak{c}_X^{\mathcal{D}}$. If $\check{\lambda} \in \operatorname{wt}(\mathfrak{B}_{X^{\bullet}}^+)$ is not in $W\varrho_X(\mathcal{D})$, the entire W-orbit $W\check{\lambda}$ is contained in the monoid $\mathfrak{c}_X^{\mathcal{D}}$.

Proof. Let $b \in \mathfrak{B}_{X^{\bullet}}^+$ with $\check{\lambda} = \operatorname{wt}(b)$. The decomposition of $\Phi_{\{i\}}(\mathfrak{B}_{X^{\bullet}}^+)$ from §7.2.5 shows that if $\check{\lambda} \notin \{\check{\nu}_i^{\pm}\}$, we have $s_i b \in \mathfrak{B}_{X^{\bullet}}^+$, so $s_i \check{\lambda} \in \mathfrak{c}_X^{\mathcal{D}}$. Since $s_i \check{\nu}_i^{\pm} = -\check{\nu}_i^{\mp}$, we can iteratively apply simple reflections to deduce the claims.

7.3. Lowering operators via hyperplane intersections. In this subsection we prove Theorem 7.1.9. Property (iii) follows from Lemma 5.4.2 and the \widetilde{W} -action on \mathfrak{B}_X given by seminormality of the crystal (§7.1.3).

To prove properties (i)–(ii) we will need a geometric interpretation of the weight-lowering operators \tilde{f}_i . This interpretation is already hiding behind the crystal structure of $\mathfrak{B}_{\tilde{\mathfrak{m}}}^{\tilde{\theta}}$ (in the notation of Proposition 7.2.4), and has to do with closure relations of semi-infinite orbits in the

affine Grassmannian. To bring such closure relations into our discussion, we need to extend the considerations of §7.2 to the compactified Zastava models.

Proposition 7.3.1. For $\check{\lambda} \in \mathfrak{c}_X^{\mathcal{D}}$, let $b \in \mathfrak{B}_{X^{\bullet},\check{\lambda}}$ be an irreducible component of critical dimension, and let \bar{b} be its closure in \bar{Y}^{λ} . For $i \in I$, consider the intersection

$$\overline{b} \cap Y^{\check{\lambda} - \check{\alpha}_i} \subset \overline{Y}^{\check{\lambda}}.$$

- (i) If the intersection above is non-empty, then $\tilde{f}_i b \neq 0$ and it corresponds to an irreducible component of dimension $\dim(b) - 1$ of $\bar{b} \cap Y^{\check{\lambda} - \check{\alpha}_i}$. Vice versa, if $\tilde{f}_i b \neq 0$ then the intersection above is non-empty, unless $\check{\lambda} = \check{\nu}_i^{\pm}$ is a color, in which case $b \subset Y^{\check{\lambda}}$ is a
- (ii) The intersection $\bar{b} \cap Y^{\check{\lambda}-\check{\alpha}_i}$ is empty only if either $\check{\lambda}=\check{\nu}_i^{\pm}$ or $\langle \alpha_i,\check{\lambda}\rangle \leq 0$.

We remark that it may be possible for $\overline{\mathsf{b}} \cap \mathsf{Y}_X^{\check{\lambda} - \check{\alpha}_i}$ to be reducible (cf. [BFG06, Proposition 19.2], which replaces the erroneous Proposition 15.2 of loc. cit.).

The proof of this proposition will be given at the end of this section. We first use the proposition to prove Theorem 7.1.9. Both properties (i)-(ii) of the theorem rely on the following

Lemma 7.3.2. For $\check{\lambda} \in \mathfrak{c}_X^{\mathcal{D}}$ and $b \in \mathfrak{B}_{X^{\bullet},\check{\lambda}}$ there is a sequence $\alpha_1, \ldots, \alpha_d$ of simple roots (possibly with repetitions), where $d = \dim b$, such that we have

- b_j := f̃_{α_j} ··· f̃_{α₁}(b) ≠ 0 for 0 ≤ j ≤ d,
 the intersection b̄_{j-1} ∩ S^{×-ᾱ₁-···-ᾱ_j} is nonempty of dimension d − j for 1 ≤ j ≤ d,
 Ă − ∑_{j=1}^d ᾱ_j = ν̄_D for some color D ∈ D.

In particular, $\check{\lambda} \geq \check{\nu}_D$.

Proof. By definition, **b** is of critical dimension $d = \frac{1}{2}(\operatorname{len}(\check{\lambda}) - 1)$. Proposition 4.4.2 shows that there exists a $\check{\lambda}' \leq \check{\lambda}$ such that $\langle \rho_G, \check{\lambda} - \check{\lambda}' \rangle \geq \frac{1}{2} (\operatorname{len}(\check{\lambda}) - 1)$, and $\bar{b} \cap S^{\check{\lambda}'}$ is nonempty of dimension zero. Proposition 6.1.1 states that the dimension inequality should be an equality, in which case Proposition 4.4.2 again provides the sequence of simple roots as in the statement. In that case, $\operatorname{len}(\check{\lambda}') = 1$, hence $\check{\lambda}' = \check{\nu}_D$ for some color $D \in \mathcal{D}$. Then Proposition 7.3.1(i) applied inductively shows that $f_{\alpha_i} \cdots f_{\alpha_1}(b) \neq 0$ satisfies the claim.

Proof of Theorem 7.1.9(ii). Immediate from Lemma 7.3.2.

Proof of Theorem 7.1.9(i). We assume as in §7.2 that $\mathfrak{c}_X = \mathbb{N}^{\mathcal{D}}$. First we show that the weights of $\mathfrak{B}_{X^{\bullet}}$ are contained in the weights of $V^{\check{\lambda}}$ for $\check{\lambda} \in \check{\Lambda}_{G}^{+} \cap W \varrho_{X}(\mathcal{D})$. Let $\check{\theta} \in \operatorname{wt}(\mathfrak{B}_{X^{\bullet}}^{+})$. By (7.1) and Lemma 7.2.7, we may assume that $\check{\theta} \in \check{\Lambda}_{G}^{-}$. Now Lemma 7.3.2 gives some color $D \in \mathcal{D}$ such that $\check{\nu}_D \leq \check{\theta}$. Since $\check{\theta}$ is antidominant, it must be a weight of V^{λ} where $\check{\lambda}$ is the unique dominant coweight in the W-orbit of ν_D . If α is a simple root such that $\mathcal{D}(\alpha) = \{D, D'\}$, then $s_{\alpha}(\check{\nu}_{D'}) = -\check{\nu}_{D}$. Thus, we see that $W\varrho_{X}(\mathcal{D}) = -W\varrho_{X}(\mathcal{D})$. Hence all of wt $(\mathfrak{B}_{X^{\bullet}}^{-})$ is also contained in the weights of the claimed representations.

Next suppose that $\check{\mu}$ is a weight of $V^{\check{\lambda}}$ for $\check{\lambda} \in \check{\Lambda}^+_G \cap W\varrho_X(\mathcal{D}) \subset \mathfrak{c}^{\mathcal{D}}_X$. We will show that $\mathfrak{B}_{X^{\bullet},\check{\mu}}$ is nonempty. By Theorem 7.1.9(iii), there exists an element $b \in \mathfrak{B}_{X,\check{\lambda}}$, and by (7.1) we may assume that $\check{\mu} \in \check{\Lambda}_G^+$ is also dominant. By Lemma 7.3.3 below, we can find a sequence of simple coroots $\check{\alpha}_1,\ldots,\check{\alpha}_d$ (possibly with repetitions), where $d=\langle \rho_G,\check{\lambda}-\check{\mu}\rangle$, such that $\check{\lambda}_j:=\check{\lambda}-\check{\alpha}_1-\cdots-\check{\alpha}_j$ satisfies $\check{\lambda}_j+\check{\rho}_G\in\check{\Lambda}_G^+$ for $j=1,\ldots,d$ and $\check{\lambda}_d=\check{\mu}$. In particular, this means that $\langle \alpha_{j+1},\check{\lambda}_j-\check{\alpha}_{j+1}\rangle=\langle \alpha_{j+1},\check{\lambda}_{j+1}\rangle\geq -1$ for all $0\leq j< d$. Equivalently, $\langle \alpha_{j+1}, \check{\lambda}_j \rangle > 0$ for all $0 \leq j < d$. Now Proposition 7.3.1(ii) implies that $\tilde{f}_{\alpha_d} \cdots \tilde{f}_{\alpha_1}(\mathsf{b}) \in \mathfrak{B}_{X^{\bullet},\check{\mu}}$ is nonzero.

Lemma 7.3.3. Let $\check{\mu} \in \check{\Lambda}_G^+$ and $\check{\lambda} \in \check{\Lambda}_G$ such that $\check{\lambda} \geq \check{\mu}$ and $\check{\lambda} + \check{\rho}_G \in \check{\Lambda}_G^+$. Then $\check{\lambda} - \check{\mu} = \check{\alpha}_1 + \cdots + \check{\alpha}_d$ for a sequence of simple coroots $\check{\alpha}_j$ (possibly with repetitions) where $d = \langle \rho_G, \check{\lambda} - \check{\mu} \rangle$ such that $\check{\lambda} - \check{\alpha}_1 - \cdots - \check{\alpha}_j + \check{\rho}_G \in \check{\Lambda}_G^+$ for $j = 1, \ldots, d$.

Proof. Since $\check{\lambda} \geq \check{\mu}$ we can decompose $\check{\lambda} - \check{\mu} = \check{\alpha}_1 + \cdots + \check{\alpha}_d$ into a sum of simple coroots (possibly with repetitions), where $d = \langle \rho_G, \check{\lambda} - \check{\mu} \rangle$. To prove the lemma, it suffices, by induction on $\check{\lambda}$, to show that there exists an $\check{\alpha}_j$ such that $\check{\lambda} - \check{\alpha}_j + \check{\rho}_G \in \check{\Lambda}_G^+$ for some $1 \leq j \leq d$. We claim that if $\check{\lambda} \neq \check{\mu}$, then there exists some $\check{\alpha}_j$ with $\langle \alpha_j, \check{\lambda} \rangle \geq 1$. If not, then $\langle \alpha_j, \check{\lambda} - \check{\mu} \rangle \leq \langle \alpha_j, \check{\lambda} \rangle \leq 0$ for all j. This implies that $\check{\lambda} - \check{\mu}$ has non-positive norm with respect to an appropriate inner product on \mathfrak{t} , and hence $\check{\lambda} = \check{\mu}$. Therefore if d > 0, then we have an $\check{\alpha}_j$ such that $\langle \alpha_j, \check{\lambda} \rangle \geq 1$. Equivalently, $\langle \alpha_j, \check{\lambda} + \check{\rho}_G \rangle \geq 2$ and $\langle \alpha_j, \check{\lambda} - \check{\alpha}_j + \check{\rho}_G \rangle \geq 0$. This implies that $\check{\lambda} - \check{\alpha}_j + \check{\rho}_G \in \check{\Lambda}_G^+$. \square

Corollary 7.3.4. If $X^{\bullet} = H \setminus G$ is affine, then all weights in $\operatorname{wt}(\mathfrak{B}_{X^{\bullet}})$ are Weyl translates of colors $\varrho_X(\mathcal{D})$. The multiplicities of these weights are determined by Theorem 7.1.9(iii).

Proof. Assume X^{\bullet} is affine, so $\mathfrak{c}_{X^{\bullet}}^{-}=0$. If there exists a non-minuscule coweight in $\check{\Lambda}_{G}^{+}\cap W\varrho_{X}(\mathcal{D})$, then Theorem 7.1.9(i) implies that $\mathfrak{B}_{X^{\bullet},\check{\theta}}$ is non-empty for some $\check{\theta}\in\mathfrak{c}_{X^{\bullet}}-0$ not in $W\varrho_{X}(\mathcal{D})$. Lemma 7.2.7 and (7.1) allow us to assume $\check{\theta}\in\mathfrak{c}_{X^{\bullet}}^{-}=0$, which gives a contradiction.

problem: we don't show crystal is normal anymore

7.3.5. The rest of this section is devoted to the proof of Proposition 7.3.1. We use the notation from §7.2.5. For brevity we write $X_i = X_{M_i}$, $H_i = H_{M_i}$, $B_i = B_{M_i}$ and $N_i = N_{B_i}$. We say $\check{\mu} \leq_i \check{\lambda}$ if $\check{\lambda} - \check{\mu} \in \mathbb{N}\check{\alpha}_i$.

We would like to embed the left Cartesian square of (7.5) into a Cartesian square involving compactified Zastava spaces. For that purpose, consider the extension of the map $\iota_{X_M} = \iota_{X_i}$ of that diagram to $\overline{\mathsf{Y}}_{X_i}^{\check{\lambda}}$, and define $\mathsf{Y}_{X_i}^{\leq i\check{\lambda}}$ by the Cartesian diagram

$$(7.10) \qquad \begin{array}{c} \mathsf{Y}_{X}^{\leq_{i}\check{\lambda}} \xrightarrow{\iota_{X,P_{i}}} \mathrm{Gr}_{P_{i}} \underset{\mathsf{L}X/\mathsf{L}^{+}P_{i}}{\times} (\mathsf{L}^{+}X/\mathsf{L}^{+}P_{i}) \xrightarrow{\mathrm{act}_{v}} \mathscr{Y}_{X,P_{i}} \\ \downarrow^{q} & \downarrow & \downarrow^{\pi_{X,P_{i}}} \\ \overline{\mathsf{Y}}_{X_{i}}^{\check{\lambda}} \xrightarrow{\iota_{X_{i}}} \mathrm{Gr}_{M_{i}} \underset{\mathsf{L}X_{i}/\mathsf{L}^{+}M_{i}}{\times} (\mathsf{L}^{+}X_{i}/\mathsf{L}^{+}M_{i}) \xrightarrow{\mathrm{act}_{v}} \mathscr{M}_{X_{i}} \end{array}$$

Lemma 7.3.6. The scheme $(Y_X^{\leq_i\check{\lambda}})_{\mathrm{red}}$ is naturally a locally closed subscheme of $\overline{Y}_X^{\check{\lambda}}$, equal to the union of strata

$$\mathsf{Y}_X^{\leq_i\check{\lambda}} = \bigcup_{n>0} \mathsf{Y}_X^{\check{\lambda}-n\check{\alpha}_i}.$$

Proof. From the Cartesian diagram (7.4), we see that the stratification $\overline{\mathsf{Y}}_{X_i}^{\check{\lambda}} = \bigcup_{n\geq 0} \mathsf{Y}_{X_i}^{\check{\lambda}-n\check{\alpha}_i}$ induces a stratification of $\mathsf{Y}_X^{\leq_i\check{\lambda}}$ by $\bigcup_{n\geq 0} \mathsf{Y}_X^{\check{\lambda}-n\check{\alpha}_i}$.

A point of $\mathsf{Y}_X^{\leq_i\check{\lambda}}$ is a map of stacks

$$y: C \to X \overset{P_i}{\times} \overline{M_i/N_i}/T$$

such that C-v is sent to the open substack $\mathsf{pt} = X^{\circ}/B$. In particular, y defines a $P_i \times T$ -bundle together with a trivialization on C-v, i.e., a point of $\mathrm{Gr}_{P_i} \times \mathrm{Gr}_T$. This gives a map

$$(7.11) Y_X^{\leq_i,\check{\lambda}} \to \operatorname{Gr}_{P_i} \times \operatorname{Gr}_T^{\check{\lambda}}.$$

Since $X \times \overline{M_i/N_i}$ is affine and $X \times \overline{M_i/N_i}/(P_i \times T) \supset (X \times M_i)/(P_i \times N_i T) = X/B$ contains pt as an open substack, Lemma 3.7.3 implies that (7.11) is a closed embedding (the argument is the same as the proof of Lemma 4.1.2). Therefore, at the level of reduced schemes we have a closed embedding

$$(\mathsf{Y}_X^{\leq_i\check{\lambda}})_{\mathrm{red}} \hookrightarrow \mathrm{Gr}_{P_i}.$$

We deduce that $(\mathsf{Y}_X^{\leq_i\check{\lambda}})_{\mathrm{red}}$ is a locally closed subscheme of $(\overline{\mathsf{Y}}_X^{\check{\lambda}})_{\mathrm{red}}$, since the (reduced) connected components of Gr_{P_i} are locally closed subschemes of Gr_G .

We are now ready to prove Proposition 7.3.1:

Proof of Proposition 7.3.1. Let $b \in \mathfrak{B}_{X^{\bullet},\check{\lambda}}$. The generic point of $b \subset \mathsf{Y}_X^{\check{\lambda}}$ maps to $\mathsf{Y}_{X_i}^{\check{\lambda},\check{\theta}}$ for some $\check{\theta} \in \mathfrak{c}_{X_i}^-$.

If $\check{\check{\theta}}=0$, then in the decomposition (7.7) we have that $\mathsf{b}\in\mathfrak{B}_{X_i^\bullet,\check{\lambda}}$ and $\check{\lambda}=\check{\nu}_i^\pm$ is the valuation attached to a color of X_i^\bullet . In that case, $\check{\nu}_i^\pm-\check{\alpha}_i\notin\mathfrak{c}_X$, so $\mathsf{Y}_{X_i}^{\check{\lambda}-\check{\alpha}_i,0}=\emptyset$. At the same time, b is a point, as recalled in §7.2.5.

Now assume $\check{\theta} \neq 0$. As in the proof of Proposition 7.2.4, the composition $\iota_{X_i} \circ q$ sends b into $\mathsf{S}_{M_i}^{\check{\Lambda}} \cap \overline{\mathsf{Gr}}_{M_i}^{\check{\theta}}$; let b_i be the image. By construction, $\tilde{f}_i \mathsf{b}$ either

- is zero iff $\check{\lambda}=\check{\theta},$ which happens precisely when $\mathsf{S}_{M_i}^{\check{\lambda}}\cap\overline{\mathsf{Gr}}_{M_i}^{\check{\theta}}=\{t^{\check{\theta}}\}$ is closed in $\overline{\mathsf{Gr}}_{M_i}^{\check{\theta}}$;
- or has image (under $\iota_{X_i} \circ q$) equal to an irreducible component $\tilde{f}_i \mathsf{b}_i$ of $\overline{\mathsf{b}}_i \cap \mathsf{S}_{M_i}^{\check{\lambda} \check{\alpha}_i}$. Indeed, this is a property of the crystal structure on MV cycles by [BG01].

In either case, the closure relations "downstairs" lift to closure relations "upstairs" under (7.10), as b and \tilde{f}_i b get identified with the lifts of b_i and \tilde{f}_ib_i corresponding to the same element of $\mathfrak{B}_{X,\check{\theta}}^{P_i}$ under the identification of fibers afforded by Lemma 7.2.2. Therefore, in the first case the closure of b in $\mathsf{Y}_X^{\leq_i\check{\lambda}}$ is entirely contained in $\mathsf{Y}^{\check{\lambda}}$, which by Lemma 7.3.6 means that $\overline{b}\cap\mathsf{Y}^{\check{\lambda}-\check{\alpha}_i}$ is empty; while in the second case $\tilde{f}_i b$ corresponds to an irreducible component of $\overline{b}\cap\mathsf{Y}^{\check{\lambda}-\check{\alpha}_i}$. \square

8. Nearby cycles

8.1. **Principal degeneration.** In this section we will consider the principal degeneration $\mathfrak{X} \to \mathbb{A}^1$ degenerating X to a horospherical variety. This is the base change of the affine family $\mathscr{X} \to \overline{T_{X,\mathrm{ss}}}$ from §2.2 along a certain line in the base. The principal degeneration was studied by [Pop86] in characteristic 0 and [Gro92] in positive characteristic.

We fix a choice of a regular dominant coweight $\check{\varrho} \in \check{\Lambda}_G^+ \cap \check{\Lambda}_G^{\mathrm{pos}}$ once and for all; none of our results depend on this choice. The image of $\check{\varrho}$ lies in $-\mathcal{V}$ and thus induces a map $\mathbb{A}^1 \to \overline{T_{X,\mathrm{ss}}}$. We define the *principal degeneration*

$$\mathfrak{X} := \mathscr{X} \underset{T_{X,\mathrm{ss}},\check{\varrho}}{\times} \mathbb{A}^1.$$

We have a $G \times \mathbb{G}_m$ -morphism $\mathfrak{X} \to \mathbb{A}^1$, where G acts trivially on \mathbb{A}^1 . This forms an affine flat family ([Pop86, Proposition 9]). The fiber over $1 \in \mathbb{A}^1$ is canonically isomorphic to X and the fiber over $0 \in \mathbb{A}^1$ is $X_{\emptyset} := \operatorname{Spec}(\operatorname{gr} k[X])$. The \mathbb{G}_m -action on \mathfrak{X} induces an isomorphism $X \times \mathbb{G}_m \cong \mathfrak{X} \times_{\mathbb{A}^1} \mathbb{G}_m$. We also have a canonical isomorphism $\mathfrak{X}/\!\!/N = (X/\!\!/N) \times \mathbb{A}^1$.

For k of arbitrary characteristic, Grosshans showed that there is an injection

$$(8.1) k[X_{\emptyset}] \to (k[X//N] \otimes k[N^{-}\backslash G])^{T}$$

of G-algebras, which is an isomorphism if and only if k[X] admits a G-module filtration with subquotients isomorphic to dual Weyl modules of G ([Gro92, Theorems 8, 16]). Of course this always holds in characteristic 0; in positive characteristic we will assume that (8.1) is an isomorphism in what follows.

Let \mathfrak{X}° denote the preimage of $T_X \times \mathbb{A}^1 \subset X/\!\!/ N \times \mathbb{A}^1$ under the map $\mathfrak{X} \to \mathfrak{X}/\!\!/ N$. This is the open subvariety which specializes over each fiber above \mathbb{A}^1 to the dense open B-orbit of that fiber. Let $\mathfrak{X}^{\bullet} \subset \mathfrak{X}$ denote the open subvariety which specializes over each fiber to the open G-orbit of that fiber.

The isomorphism (8.1) implies that X_{\emptyset} has a natural left T_X -action, and the orbit of the coset N^-1 gives an embedding $T_X \hookrightarrow X_{\emptyset}$. We will temporarily denote its closure by $\overline{T_X}$.

Lemma 8.1.1. The composition $\overline{T_X} \hookrightarrow X_\emptyset \to X_\emptyset /\!\!/ N = X/\!\!/ N$ is an isomorphism. In other words, we have a section $X/\!\!/ N \hookrightarrow X_\emptyset$.

Proof. This is essentially a special case of [AT05, Proposition 7]. For $\lambda \in \Lambda_G^+$, the isotypic component $k[N^- \backslash G]^{(\lambda)}$ is the dual Weyl module of highest weight λ . The embedding $T \hookrightarrow N^- \backslash G$ corresponds to the algebra map $k[N^- \backslash G] \to k[T]$ that sends $k[N^- \backslash G]^{(\lambda)} \to ke^{\lambda}$ (explicitly it sends all T-eigenvectors not of highest weight to zero). Thus, using (8.1), the map $k[X_{\emptyset}] \to k[T_X]$ sends $k[X_{\emptyset}]^{(\lambda)} \to ke^{\lambda}$ for $\lambda \in \mathfrak{c}_X^{\vee}$.

8.1.2. Contracting action. Let $s: X/\!\!/N \hookrightarrow \mathfrak{X}$ denote the composition of the section given by Lemma 8.1.1 and the embedding $X_{\emptyset} \hookrightarrow \mathfrak{X}$ as the zero fiber. We will construct a \mathbb{G}_m -action on \mathfrak{X} that contracts \mathfrak{X} to the section s.

Recall the coweight $\check{\varrho}: \mathbb{G}_m \to T$ used to define \mathfrak{X} . Let \mathbb{G}_m act on \mathfrak{X} via the group homomorphism $\mathbb{G}_m \to G \times \mathbb{G}_m : a \mapsto (\check{\varrho}(a^{-1}), a)$ and the natural $G \times \mathbb{G}_m$ -action on \mathfrak{X} .

Lemma 8.1.3. The action map $\mathbb{G}_m \times \mathfrak{X} \to \mathfrak{X}$ extends to a regular map $\mathbb{A}^1 \times \mathfrak{X} \to \mathfrak{X}$ such that the composition $0 \times \mathfrak{X} \to \mathbb{A}^1 \times \mathfrak{X} \to \mathfrak{X}$ coincides with the composition $\mathfrak{X} \to X /\!\!/ N \xrightarrow{s} \mathfrak{X}$.

Proof. The action map can be described as the map of rings

$$k[\mathfrak{X}] \to k[\mathbb{G}_m] \otimes k[\mathfrak{X}] : f_{\mu}e^n \mapsto e^{n-\langle \mu, \check{\varrho} \rangle} \otimes f_{\mu}e^n$$

for a *T*-eigenvector $f_{\mu} \in k[X]_{(\lambda)}$ of weight μ . We have $\langle \mu, \check{\varrho} \rangle \leq \langle \lambda, \check{\varrho} \rangle \leq n$, so the image of the co-action lies in the subalgebra $k[\mathbb{A}^1] \otimes k[\mathfrak{X}]$.

Moreover, since $\check{\varrho}$ is regular dominant, observe that $n - \langle \mu, \check{\varrho} \rangle = 0$ if and only if f_{μ} is a highest weight vector and $n = \langle \lambda, \check{\varrho} \rangle$, which implies that the composition $0 \times \mathfrak{X} \to \mathbb{A}^1 \times \mathfrak{X} \to \mathfrak{X}$ factors through $\mathfrak{X} \to X /\!\!/ N \stackrel{s}{\to} \mathfrak{X}$.

8.2. **Grinberg–Kazhdan theorem in families.** The map $\mathfrak{X} \to \mathbb{A}^1$ induces a map of formal arc spaces $L^+\mathfrak{X} \to L^+\mathbb{A}^1$. Define $L^+_{\mathbb{A}^1}\mathfrak{X} = L^+\mathfrak{X} \underset{L^+\mathbb{A}^1}{\times} \mathbb{A}^1$ where $\mathbb{A}^1 \hookrightarrow L^+\mathbb{A}^1$ is the map of constant

arcs. Then $\mathsf{L}_{\mathbb{A}^1}^+\mathfrak{X}$ is an affine scheme over \mathbb{A}^1 with fiber over \mathbb{G}_m isomorphic to L^+X and zero fiber isomorphic to L^+X_\emptyset . While there does not currently exist a theory of nearby cycles for infinite type schemes, we explain below how the nearby cycles of the "IC complex" of L^+X can be modeled by nearby cycles on the global or Zastava model.

8.2.1. From now on we return to assuming that B acts simply transitively on X° , so $T_X = T$. Note that now (8.1) implies that X_{\emptyset} is an affine embedding of $N^- \backslash G$.

First, we define the analogous models in families: let

$$\mathcal{M}_{\mathfrak{X}} = \operatorname{Maps}_{\text{gen}}(C, \mathfrak{X}/G \supset \mathfrak{X}^{\bullet}/G), \quad \mathcal{Y}_{\mathfrak{X}} = \operatorname{Maps}_{\text{gen}}(C, \mathfrak{X}/B \supset \mathfrak{X}^{\circ}/B),$$

where $\mathfrak{X}^{\circ}/B = \mathbb{A}^1$. Since C is proper, $\mathfrak{M}_{\mathfrak{X}}$ maps to \mathbb{A}^1 with generic fiber isomorphic to \mathfrak{M}_X and special fiber $\mathfrak{M}_{X_{\emptyset}}$. The arguments of §3 easily generalize to show that $\mathfrak{M}_{\mathfrak{X}}$ is an algebraic stack locally of finite type over \mathbb{A}^1

The *B*-equivariant map $\mathfrak{X} \to X/\!\!/ N \times \mathbb{A}^1$ induces a map $\mathcal{Y}_{\mathfrak{X}} \to \mathcal{A} \times \mathbb{A}^1$. We think of $\mathcal{Y}_{\mathfrak{X}} \to \mathbb{A}^1$ as a family degenerating $\mathcal{Y} = \mathcal{Y}_X$ to $\mathcal{Y}_{\emptyset} := \mathcal{Y}_{X_{\emptyset}}$. For $\check{\lambda} \in \mathfrak{c}_X$, let $\mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}}$ denote the preimage of $\mathcal{A}^{\check{\lambda}}$ under $\pi_{\mathfrak{X}} : \mathcal{Y}_{\mathfrak{X}} \to \mathcal{A}$. We summarize the properties of $\mathcal{Y}_{\mathfrak{X}}$ below; the proofs are easy generalizations of those for \mathcal{Y} .

- $\mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}}$ is representable by a finite type scheme,
- $\mathcal{Y}_{\mathfrak{X}}$ satisfies the graded factorization property in families, in the sense that there is a natural isomorphism

$$\mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}}\underset{\mathcal{A}^{\check{\lambda}}}{\times} (\mathcal{A}^{\check{\lambda}_{1}} \mathring{\times} \mathcal{A}^{\check{\lambda}_{2}}) \cong (\mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}_{1}} \underset{\mathbb{A}^{1}}{\times} \mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}_{2}})|_{\mathcal{A}^{\check{\lambda}_{1}} \mathring{\times} \mathcal{A}^{\check{\lambda}_{2}}}$$

when $\check{\lambda}_1 + \check{\lambda}_2 = \check{\lambda}$.

• There exists a closed embedding $\mathcal{Y}_{\mathfrak{X}} \hookrightarrow \operatorname{Gr}_{B,\operatorname{Sym} C} \times \mathbb{A}^1$.

The models $\mathcal{M}_{\mathfrak{X}}$, $\mathcal{Y}_{\mathfrak{X}}$ are smooth-locally isomorphic as families, by the following generalization of Lemma 3.5.4: Let $\mathcal{Y}_{\mathfrak{X}^{\bullet}} = \operatorname{Maps}_{\operatorname{gen}}(C, \mathfrak{X}^{\bullet}/B \supset \mathfrak{X}^{\circ}/B)$.

Lemma 8.2.2. For fixed $\check{\lambda} \in \mathfrak{c}_X$ and any $\check{\mu} \in \check{\Lambda}_G^{\mathrm{pos}}$ large enough, there is a correspondence

$$(8.2) \mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}} \leftarrow \mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}} \stackrel{\circ}{\underset{\mathbb{A}^1}{\times}} \mathcal{Y}_{\mathfrak{X}^{\bullet}}^{\check{\mu}} \to \mathcal{M}_{\mathfrak{X}}$$

over \mathbb{A}^1 , where the left arrow is smooth surjective and the right arrow is smooth.

The proof is the same as in *loc cit.*, together with the observation that $\operatorname{Maps}(C, \mathfrak{X}^{\bullet}/G)$ is smooth because \mathfrak{X}^{\bullet}/G is the classifying stack of a smooth group scheme over \mathbb{A}^1 .

8.2.3. Fix an arc $\gamma_0 \in X_{\emptyset}(k[\![t]\!]) \cap X_{\emptyset}^{\bullet}(k(\!(t)\!))$ and consider γ_0 as a point in $\mathsf{L}_{\mathbb{A}^1}^+\mathfrak{X}(k)$.

Theorem 8.2.4 (Grinberg–Kazhdan, Drinfeld). There exists a point $y \in \mathcal{Y}_{\emptyset}(k) \subset \mathcal{Y}_{\mathfrak{X}}(k)$ such that the formal neighborhood $(\widehat{\mathsf{L}_{\pm 1}^{\perp}\mathfrak{X}})_{\gamma_0}$ is isomorphic to $\widehat{\mathbb{A}}^{\infty} \times \widehat{\mathcal{Y}}_{\mathfrak{X},y}$.

Proof. Fix a point $v \in |C|$ and an identification $\mathfrak{o}_v \cong k[\![t]\!]$. Note that each orbit in $X_\emptyset^\bullet(F_v)/G(\mathfrak{o}_v)$ has a representative in $X_\emptyset^\circ(F_v)$. Thus, by $G(\mathfrak{o}_v)$ -translation we may assume that $\gamma_0 \in X_\emptyset(\mathfrak{o}_v) \cap X_\emptyset^\circ(F_v)$. We may consider γ_0 as a section Spec $\mathfrak{o}_v \to X_\emptyset \times^B \hat{\mathcal{P}}_B^0$ where $\hat{\mathcal{P}}_B^0$ is the trivial B-bundle on Spec \mathfrak{o}_v . By Lemma 3.7.7, this is equivalent to a map $y: C \to X_\emptyset/B$ with $y(C-v) = \mathsf{pt}$. This is the point $y \in \mathcal{Y}_\emptyset(k)$ we will use.

Next, we define a map of formal schemes

$$(\widehat{\mathsf{L}_{\mathbb{A}^1}^+\mathfrak{X}})_{\gamma_0} \to \widehat{\mathcal{Y}}_{\mathfrak{X},y}$$

as follows: Formal schemes are determined by their R-points where R is a local commutative k-algebra with residue field k whose maximal ideal \mathfrak{m} is nilpotent. Let $\gamma:\operatorname{Spec} R[\![t]\!]\to \mathfrak{X}$ be an R-point of $(\widehat{\mathsf{L}_{\mathbb{A}^1}^+}\mathfrak{X})_{\gamma_0}$. The reduction modulo \mathfrak{m} of $\gamma|_{\operatorname{Spec} R((t))}$ equals $\gamma_0|_{\operatorname{Spec} k((t))}\in \mathfrak{X}^\circ(k((t)))$. Since k((t)) is the unique closed point of R((t)), we deduce that $\gamma|_{\operatorname{Spec} R((t))}$ has image contained in \mathfrak{X}° . Consider γ as a section $\operatorname{Spec}(R \widehat{\otimes} \mathfrak{o}_v) \to \mathfrak{X} \times^B \widehat{\mathcal{P}}^0_B$ where $\widehat{\mathcal{P}}^0_B$ is the trivial B-bundle. By an easy

generalization of Lemma 3.7.7, the pair $(\hat{\mathbb{P}}_B^0, \gamma)$ is equivalent to a map $\tilde{y}: C \times \operatorname{Spec}(R) \to \mathfrak{X}/B$ such that $\tilde{y}|_{(C-v)\times\operatorname{Spec} R}$ factors through $\operatorname{Spec}(R) \to \mathbb{A}^1 = \mathfrak{X}^\circ/B$. We define (8.3) on R-points by sending $\gamma \mapsto \tilde{y}$.

Since $(\widehat{\mathsf{L}^+B})_1$ is non-canonically isomorphic to $\widehat{\mathbb{A}}^{\infty}$, the theorem follows from the proposition below.

Proposition 8.2.5. Let $y: C \to X_{\emptyset}/B$ satisfy y(C-v) = pt and $y|_{\text{Spec }\mathfrak{o}_v}$ corresponds to $(\hat{\mathcal{P}}_B^0, \gamma_0)$. Then the map (8.3) is a $(\widehat{\mathsf{L}}^+B)_1$ -torsor.

Proof. Let (R, \mathfrak{m}) as above. By a generalization of Lemma 3.7.7, an R-point of $\widehat{\mathcal{Y}}_{\mathfrak{X},y}$ is equivalent to a map $\bar{\gamma}: \operatorname{Spec} R[\![t]\!] \to \mathfrak{X}/B$ such that $\bar{\gamma}|_{\operatorname{Spec} R((t))}$ factors through $\operatorname{Spec} R \to \mathbb{A}^1$. The fiber of (8.3) over $\bar{\gamma}$ parametrizes maps $\gamma: \operatorname{Spec} R[\![t]\!] \to \mathfrak{X}$ that induce $\bar{\gamma}$ and whose reduction modulo \mathfrak{m} equals γ_0 . Since any B-bundle on $\operatorname{Spec} R[\![t]\!]$ can be trivialized, we see that $(\widehat{\mathsf{L}^+B})_1(R)$ acts simply transitively on this fiber, since $\gamma_0 \in X_0^{\circ}(k((t))) \cong B(k((t)))$.

- 8.3. **Results on nearby cycles.** We will now consider nearby cycles on the global and Zastava models.
- 8.3.1. Fix $\check{\lambda} \in \mathfrak{c}_X$ and consider the family $f: \mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}} \to \mathbb{A}^1$. We have complementary embeddings

$$\mathcal{Y}_{\emptyset}^{\check{\lambda}} \stackrel{i}{\hookrightarrow} \mathcal{Y}_{\mathfrak{X}}^{\check{\lambda}} \stackrel{j}{\hookleftarrow} \mathcal{Y}^{\check{\lambda}} \times \mathbb{G}_{m}$$

where i, j correspond to $f^{-1}(0), f^{-1}(\mathbb{G}_m)$, respectively. We let

$$\Psi_{\mathcal{Y}}: \mathcal{D}_{c}^{b}(\mathcal{Y}^{\check{\lambda}} \times \mathbb{G}_{m}) \to \mathcal{D}_{c}^{b}(\mathcal{Y}_{\emptyset}^{\check{\lambda}})$$

denote the nearby cycles functor defined in [SGA73, Exposé XIII], shifted by 1 so that Ψ_{y} is perverse t-exact.

Let Ψ^u_{y} denote the direct summand where the monodromy operator acts unipotently.

Remark 8.3.2. Since \mathfrak{X} has a \mathbb{G}_m -action making f equivariant and $\mathfrak{F}\boxtimes \mathrm{IC}_{\mathbb{G}_m}$ is \mathbb{G}_m -equivariant for any sheaf \mathfrak{F} on $\mathfrak{Y}^{\check{\lambda}}$, a standard argument ([AB09, Remark 14]) shows that $\Psi_{\mathfrak{Y}}(\mathfrak{F}\boxtimes \mathrm{IC}_{\mathbb{G}_m}) = \Psi^u_{\mathfrak{Y}}(\mathfrak{F}\boxtimes \mathrm{IC}_{\mathbb{G}_m})$, i.e., the monodromy is unipotent.

By t-exactness, $\Psi_{\mathcal{Y}}(\mathrm{IC}_{\mathcal{Y}^{\check{\lambda}}\times\mathbb{G}_m})$ is also perverse. In this section we will compute the image of $\Psi_{\mathcal{Y}}(\mathrm{IC}_{\mathcal{Y}^{\check{\lambda}}\times\mathbb{G}_m})$ in the Grothendieck group of $\mathrm{D}^b_c(\mathcal{Y}^{\check{\lambda}}_{\emptyset})$ (equivalently, that of $\mathrm{P}(\mathcal{Y}^{\check{\lambda}}_{\emptyset})$); this determines the semisimplification of $\Psi_{\mathcal{Y}}(\mathrm{IC}_{\mathcal{Y}^{\check{\lambda}}\times\mathbb{G}_m})$ in the Artinian category $\mathrm{P}(\mathcal{Y}^{\check{\lambda}}_{\emptyset})$ over the algebraically closed field k.

There is an analogous global family $\mathcal{M}_{\mathfrak{X}} \to \mathbb{A}^1$ and complementary embeddings

$$\mathcal{M}_{X_a} \hookrightarrow \mathcal{M}_{\mathfrak{X}} \hookleftarrow \mathcal{M}_X \times \mathbb{G}_m$$
.

Denote the associated unipotent nearby cycles functor by $\Psi_{\mathfrak{M}}: \mathrm{D}^{b}_{c}(\mathfrak{M}_{X} \times \mathbb{G}_{m}) \to \mathrm{D}^{b}_{c}(\mathfrak{M}_{X_{\emptyset}})$. We will simultaneously compute $[\Psi_{\mathfrak{M}}(\mathrm{IC}_{\mathfrak{M}_{X} \times \mathbb{G}_{m}})]$.

8.3.3. Stratifications in the horospherical case. Before proceeding, we give a more concrete description of the stratifications on \mathcal{Y}_{\emptyset} , $\mathcal{M}_{X_{\emptyset}}$ using (8.1).

Since $T_X = T$, we have $X_{\emptyset}^{\bullet} = N^- \backslash G$ and $\mathfrak{M}_{X_{\emptyset}^{\bullet}} = \operatorname{Bun}_{N^-}$. The isomorphism (8.1) induces a map of affine schemes $X /\!\!/ N \times^T \overline{N^- \backslash G} \to X_{\emptyset}$, which in turn induces a map of stacks

$$\bar{\iota}_{\mathfrak{M}}: \mathcal{A} \underset{\mathrm{Bun}_{T}}{\times} \overline{\mathrm{Bun}}_{B^{-}} \to \mathfrak{M}_{X_{\emptyset}}.$$

Let $\iota_{\mathfrak{M}}: \mathcal{A} \underset{\mathrm{Bun}_{T}}{\times} \mathrm{Bun}_{B^{-}} \hookrightarrow \mathfrak{M}_{X_{\emptyset}}$ denote the restriction. For $\check{\lambda} \in \mathfrak{c}_{X}$, let $\iota_{\mathfrak{M}}^{\check{\lambda}}, \bar{\iota}_{\mathfrak{M}}^{\check{\lambda}}$ denote the maps corresponding to $\mathcal{A}^{\check{\lambda}}$. The following is a variant of [BG02, Proposition 1.2.7], whose proof we leave to the reader.

Proposition 8.3.4. The map $\bar{\iota}_{\mathcal{M}}^{\check{\lambda}}$ is finite, and its restriction $\iota_{\mathcal{M}}^{\check{\lambda}}$ is a locally closed embedding.

The subschemes $\mathcal{M}_{X_{\emptyset}}^{(\check{\lambda})} = \iota_{\mathcal{M}}^{\check{\lambda}}(\mathcal{A}^{\check{\lambda}} \times_{\operatorname{Bun}_{T}^{-\check{\lambda}}} \operatorname{Bun}_{B^{-}}^{-\check{\lambda}})$ for $\check{\lambda} \in \mathfrak{c}_{X}$ form a (possibly non-smooth) stratification of $\mathcal{M}_{X_{\emptyset}}$. This is a coarser stratification than the one defined in §3.1.5 in the following sense: Note that $\mathcal{V}(X_{\emptyset}) = \mathfrak{t}_{X}$ so $\mathfrak{c}_{X_{\emptyset}}^{-} = \mathfrak{c}_{X}$. For a partition $\mathfrak{P} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X} - 0)$, there is a locally closed embedding $\mathring{C}^{\mathfrak{P}} \hookrightarrow \mathcal{A}^{\operatorname{deg}(\mathfrak{P})}$, and the collection of these subschemes ranging over all partitions \mathfrak{P} with $\operatorname{deg}(\mathfrak{P}) = \check{\lambda}$ forms a smooth stratification of $\mathcal{A}^{\check{\lambda}}$. It follows from the constructions that the strata from §3.1.5 are given by

$$\mathcal{M}_{X_{\emptyset}}^{\mathfrak{P}} \cong \mathring{C}^{\mathfrak{P}} \underset{\operatorname{Bun}_{T}}{\times} \operatorname{Bun}_{B^{-}} \cong \mathring{C}^{\mathfrak{P}} \underset{A^{\deg(\mathfrak{P})}}{\times} \mathcal{M}_{X_{\emptyset}}^{(\deg(\mathfrak{P}))}$$

indexed over all $\mathfrak{P} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X - 0)$.

Next if we consider the corresponding Zastava model, we have

$$\mathcal{Y}_{X_{\mathfrak{a}}^{\bullet}} = \mathcal{Z}^{?,0} = \operatorname{Maps}(C, N^{-} \backslash G/B \supset \mathsf{pt})$$

is the open Zastava space of Finkelberg–Mirković. The Zastava space¹⁹ is in turn defined by $\mathcal{Z} = \operatorname{Maps}_{\operatorname{gen}}(C, N^- \backslash \overline{G/N}/T \supset \operatorname{pt})$. The geometry of the Zastava space has been extensively studied in [FM99, FFKM99, BFGM02]. The components of \mathcal{Z} are indexed by $\check{\Lambda}_G^{\operatorname{pos}}$.

We can also define the relative open Zastava space $\mathcal{Z}_{\operatorname{Bun}_T}^{?,0} = \operatorname{Maps}_{\operatorname{gen}}(C, B^- \backslash G/B \supset \operatorname{pt}/T)$. The spaces $\mathcal{Z}^{?,0}$ and $\mathcal{Z}_{\operatorname{Bun}_T}^{?,0}$ are smooth locally isomorphic (cf. [BFGM02, §3.1]). For $\check{\lambda} \in \check{\Lambda}_G^{\operatorname{pos}}$, let $\mathcal{Z}_{\operatorname{Bun}_T}^{\check{\lambda},0}$ denote the preimages of $\operatorname{Bun}_{B^-}^{\check{\mu}} \times \operatorname{Bun}_{B^-}^{\check{\mu}-\check{\lambda}}$ running over all $\check{\mu} \in \check{\Lambda}_G$.

Since \mathcal{Y}_{\emptyset} is open in $\mathcal{M}_{X_{\emptyset}} \times_{\operatorname{Bun}_G} \operatorname{Bun}_B$, we deduce by base change that for any $\check{\lambda} \in \mathfrak{c}_X$, $\check{\mu} \in \check{\Lambda}_G^{\operatorname{pos}}$, there is a locally closed embedding

$$\iota_{y_{\emptyset}}^{\check{\lambda},\check{\mu}}:\mathcal{A}^{\check{\lambda}}\underset{\mathrm{Bun}_{T}}{\times}\mathfrak{Z}_{\mathrm{Bun}_{T}}^{\check{\mu},0}\hookrightarrow y_{\emptyset}^{\check{\lambda}+\check{\mu}}$$

where we are mapping $\mathcal{Z}_{\mathrm{Bun}_T}^{?,0} \to \mathrm{Bun}_{B^-} \to \mathrm{Bun}_T$. Note that $\mathcal{Z}_{\mathrm{Bun}_T}^{0,0} = \mathrm{Bun}_T$, so $\iota_{\mathcal{Y}_{\emptyset}}^{\check{\lambda},0}$ defines a map $\mathcal{A}^{\check{\lambda}} \hookrightarrow \mathcal{Y}_{\emptyset}^{\check{\lambda}}$. One can check that this map corresponds to applying $\mathrm{Maps}(C, T \setminus ?)$ to the section $s_{\emptyset} : X /\!\!/ N \hookrightarrow X_{\emptyset}$ from Lemma 8.1.1. Therefore,

$$\mathfrak{s}_{\emptyset}^{\check{\lambda}} := \iota_{\mathfrak{Y}_{\emptyset}}^{\check{\lambda},0} : \mathcal{A}^{\check{\lambda}} \hookrightarrow \mathfrak{Y}_{\emptyset}^{\check{\lambda}}$$

is a section of the projection $\pi_{\emptyset}: \mathcal{Y}_{\emptyset}^{\check{\lambda}} \to \mathcal{A}^{\check{\lambda}}$.

8.3.5. We compute the *-restriction of nearby cycles to the strata above, which suffices to determine nearby cycles in the Grothendieck group. Let $\Omega(\check{\mathfrak{n}}_C)^{-\check{\nu}}=\mathbb{D}(\Upsilon(\check{\mathfrak{n}}_C)^{\check{\nu}})$ denote the Verdier dual of the factorization algebra defined in §4.5. Recall that we defined a convolution product \star on $\mathcal{D}^b_c(\mathcal{A})$ in §4.5.7. The statement of our main result is:

¹⁹The Finkelberg–Mirković Zastava space is the Zastava model for $\overline{G/N}$. In this paper we made a slight distinction in semantics between 'model' and 'space' to avoid confusion, but the two terms are interchangeable.

Theorem 8.3.6. We have equalities in the Grothendieck group

$$\begin{split} &[\Psi_{\mathfrak{M}}(\mathrm{IC}_{\mathfrak{M}_{X}\times\mathbb{G}_{m}})] = \sum_{\check{\lambda}\in\mathfrak{c}_{X}}\sum_{\check{\nu}\geq0} \left[\iota_{\mathfrak{M},!}^{\check{\lambda}}\left(\left(\mathfrak{i}_{\mathcal{A},\check{\nu},!}(\Omega(\check{\mathfrak{n}}_{C})^{-\check{\nu}}\right)\star\bar{\pi}_{!}(\mathrm{IC}_{\overline{y}\check{\lambda}-\check{\nu}})\right)\underset{\mathrm{Bun}_{T}}{\boxtimes}\mathrm{IC}_{\mathrm{Bun}_{B^{-}}}\right)\right] \\ &[\Psi_{\mathfrak{Y}}(\mathrm{IC}_{\mathfrak{Y}\check{\mu}\times\mathbb{G}_{m}})] = \sum_{\check{\lambda}\in\mathfrak{c}_{X}}\sum_{\check{\nu}\geq0} \left[\iota_{\mathfrak{Y},!}^{\check{\lambda},\check{\mu}-\check{\lambda}}\left(\left(\mathfrak{i}_{\mathcal{A},\check{\nu},!}(\Omega(\check{\mathfrak{n}}_{C})^{-\check{\nu}}\right)\star\bar{\pi}_{!}(\mathrm{IC}_{\overline{y}\check{\lambda}-\check{\nu}})\right)\underset{\mathrm{Bun}_{T}}{\boxtimes}\mathrm{IC}_{\mathcal{Z}_{\mathrm{Bun}_{T}}^{\check{\mu}-\check{\lambda},0}}\right)\right] \end{split}$$

for any $\check{\mu} \in \mathfrak{c}_X$ in the second equality.

We point out that the description of $\bar{\pi}_! IC_{y\bar{\lambda}}$ is the main content of the previous sections of this paper. In particular it has the format (6.12).

Recall that $\mathcal{F} \boxtimes_{\operatorname{Bun}_T} \mathcal{G}$ denotes the *-restriction of $\mathcal{F} \boxtimes \mathcal{G}$ to the corresponding fiber product over Bun_T , shifted by $[-\dim \operatorname{Bun}_T]$. Note that Bun_{B^-} and $\mathcal{Z}^{?,0}_{\operatorname{Bun}_T}$ are smooth stacks, so the respective IC complexes are shifted constant sheaves.

Proof. Theorem 8.3.6 follows by combining Corollary 4.5.9 with Lemma 8.3.7 and Theorem 8.3.8 below. $\hfill\Box$

First, a well-known argument using some Zastava-to-global yoga allows us to reduce from computing restrictions to all strata to only computing $\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi_{\mathcal{Y}}(\mathrm{IC}_{\check{\mathcal{Y}}\check{\lambda}\times\mathbb{G}_m})$ for all $\check{\lambda}\in\mathfrak{c}_X$.

Lemma 8.3.7. We have equalities in the Grothendieck group²⁰

$$[\iota_{\mathcal{M}}^{\check{\lambda},*}\Psi_{\mathcal{M}}(\mathrm{IC}_{\mathcal{M}_X\times\mathbb{G}_m})] = [\mathfrak{s}_{\emptyset}^{\check{\lambda},*}(\Psi_{\mathcal{Y}}(\mathrm{IC}_{\mathcal{Y}^{\check{\lambda}}\times\mathbb{G}_m})) \underset{\mathrm{Bun}_T}{\boxtimes} \mathrm{IC}_{\mathrm{Bun}_{B^-}}]$$

$$[\iota_{y_{\emptyset}}^{\check{\lambda},\check{\nu},*}\Psi_{y}(\mathrm{IC}_{y_{\check{\lambda}+\check{\nu}}\times\mathbb{G}_{m}})] = [\mathfrak{s}_{\emptyset}^{\check{\lambda},*}(\Psi_{y}(\mathrm{IC}_{y_{\check{\lambda}}\times\mathbb{G}_{m}})) \underset{\mathrm{Bun}_{T}}{\boxtimes} \mathrm{IC}_{2_{\mathrm{Bun}_{T}}^{\check{\nu},0}}],$$

where $\check{\lambda} \in \mathfrak{c}_X$, $\check{\nu} \in \check{\Lambda}_C^{\text{pos}}$.

Proof. The argument is the same as [BFGM02, §3.1], [BFGM04, §8(1)] and [BG08, Proof of Proposition 4.4]. The strategy is that we first show (8.4) and then use it to show (8.5).

Fix $\check{\lambda}' \in \mathfrak{c}_X$ and $\check{\mu} \in \check{\Lambda}_G^{\mathrm{pos}}$ large enough as in Lemma 8.2.2 and consider the correspondence (8.2). The fiber of (8.2) over $0 \in \mathbb{A}^1$ gives the correspondence

$$\mathcal{Y}_{\emptyset}^{\check{\lambda}'} \leftarrow \mathcal{Y}_{\emptyset}^{\check{\lambda}'} \overset{\circ}{\times} \mathcal{Z}^{\check{\mu},0} \to \mathcal{M}_{X_{\emptyset}}.$$

Since nearby cycles commutes with smooth base change of the family over \mathbb{A}^1 , we deduce that there is an isomorphism

(8.6)
$$\Psi_{\mathcal{Y}}(\mathrm{IC}_{\mathcal{Y}^{\check{\lambda}'} \times \mathbb{G}_{m}})|_{\mathcal{Y}^{\check{\lambda}'} \overset{\circ}{\times} \mathcal{Z}^{\check{\mu},0}}^{!*} \cong \Psi_{\mathcal{M}}(\mathrm{IC}_{\mathcal{M}_{X} \times \mathbb{G}_{m}})|_{\mathcal{Y}^{\check{\lambda}'} \overset{\circ}{\times} \mathcal{Z}^{\check{\mu},0}}^{!*},$$

We now restrict to strata: for $\check{\lambda} \in \mathfrak{c}_X$ observe that there is a commutative diagram where both squares are Cartesian

$$(8.7) \qquad \begin{array}{c} \mathcal{A}^{\check{\lambda}} \underset{\operatorname{Bun}_{T}}{\times} \mathcal{Z}_{\operatorname{Bun}_{T}}^{\check{\lambda}' - \check{\lambda}, 0} \longleftarrow (\mathcal{A}^{\check{\lambda}} \underset{\operatorname{Bun}_{T}}{\times} \mathcal{Z}_{\operatorname{Bun}_{T}}^{\check{\lambda}' - \check{\lambda}, 0}) \stackrel{\circ}{\times} \mathcal{Z}^{\check{\mu}, 0} \longrightarrow \mathcal{A}^{\check{\lambda}} \underset{\operatorname{Bun}_{T}}{\times} \operatorname{Bun}_{B^{-1}} \\ \downarrow \iota_{\mathfrak{I}}^{\check{\lambda}, \check{\lambda}' - \check{\lambda}} \downarrow \qquad \qquad \qquad \downarrow \iota_{\mathfrak{I}}^{\check{\lambda}'} \\ \qquad \qquad \qquad \qquad \qquad \downarrow \iota_{\mathfrak{I}}^{\check{\lambda}'} \longleftarrow \qquad \qquad \mathcal{I}^{\check{\lambda}'} \underset{\mathfrak{I}}{\times} \mathcal{I}^{\check{\mu}, 0} \longrightarrow \mathcal{I}^{\check{\lambda}} \underset{\mathfrak{I}}{\times} \mathcal{I}^{\check{\mu}, 0} \longrightarrow \mathcal{I}^{\check{\lambda}} \\ \qquad \qquad \qquad \qquad \downarrow \iota_{\mathfrak{I}}^{\check{\lambda}} \end{array}$$

 $^{^{20}}$ In fact the isomorphisms hold in the derived category, but we omit the proof as it uses generic-Hecke equivariance to show no twist exists on the Bun_B- factor.

By the argument of [BFGM02, §8(1)], every point of $\mathcal{A}^{\check{\lambda}} \times_{\operatorname{Bun}_T} \operatorname{Bun}_{B^-}$ is in the image of $\mathcal{A}^{\check{\lambda}} \stackrel{\circ}{\times} \mathcal{Z}^{\check{\mu},0}$ for some $\check{\mu}$ large enough, i.e., we only need to consider the diagram (8.7) when $\check{\lambda}' = \check{\lambda}$. Note that $\mathcal{A}^{\check{\lambda}} \times_{\operatorname{Bun}_T} \operatorname{Bun}_{B^-}$ is of finite type, so there exists a single $\check{\mu}$ such that the map $\mathcal{Y}_0^{\check{\lambda}} \stackrel{\circ}{\times} \mathcal{Z}^{\check{\mu},0} \to \mathcal{M}_{X_0}$ has geometrically irreducible fibers and contains $\mathcal{A}^{\check{\lambda}} \times_{\operatorname{Bun}_T} \operatorname{Bun}_{B^-}$ in its image (Corollary 3.5.2). In particular, pullback along this map is fully faithful on perverse sheaves.

By restricting the isomorphism (8.6) to the stratum $\mathcal{A}^{\check{\lambda}} \stackrel{\circ}{\times} \mathcal{Z}^{\check{\mu},0}$, we get

$$\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi_{\mathcal{Y}}(\mathrm{IC}_{\mathcal{Y}^{\check{\lambda}}\times\mathbb{G}_{m}})|_{\mathcal{A}^{\check{\lambda}}\overset{\circ}{\times}\mathbb{Z}^{\check{\mu},0}}^{!*}\cong\iota_{\mathcal{M}}^{\check{\lambda},*}\Psi_{\mathcal{M}}(\mathrm{IC}_{\mathcal{M}_{X}\times\mathbb{G}_{m}})|_{\mathcal{A}^{\check{\lambda}}\overset{\circ}{\times}\mathbb{Z}^{\check{\mu},0}}^{!*}.$$

This establishes the equality (8.4) in the Grothendieck group by fully faithfulness of the pullback.

The equality (8.5) follows from (8.4) and (8.6) in the same fashion by considering the diagram (8.7) with $\check{\lambda}' = \check{\lambda} + \check{\nu}$.

Define the functor $\Psi: \mathcal{D}^b_c(\mathcal{Y}^{\check{\lambda}}) \to \mathcal{D}^b_c(\mathcal{Y}^{\check{\lambda}})$ by $\Psi(\mathcal{F}) := \Psi_{\mathcal{Y}}(\mathcal{F} \boxtimes \mathcal{IC}_{\mathbb{G}_m}) = \Psi^u_{\mathcal{Y}}(\mathcal{F} \boxtimes \mathcal{IC}_{\mathbb{G}_m})$. The crucial fact that will allow us to do our computations is the following:

Theorem 8.3.8. There are natural isomorphisms of functors $D_c^b(\mathcal{Y}^{\check{\lambda}}) \to D_c^b(\mathcal{A}^{\check{\lambda}})$:

$$\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi \cong \pi_{\emptyset,*}\Psi \cong \pi_{*}$$
 $\mathfrak{s}_{\emptyset}^{\check{\lambda},!}\Psi \cong \pi_{\emptyset,!}\Psi \cong \pi_{!}.$

The theorem will be proved using a standard argument involving the contraction principle.

8.4. Contraction principle. Set $\mathfrak{s}^{\check{\lambda}} = i \circ \mathfrak{s}^{\check{\lambda}}_{\emptyset} : \mathcal{A}^{\check{\lambda}} \hookrightarrow \mathcal{Y}^{\check{\lambda}}_{\mathfrak{X}}$. We drop the superscript to denote the section $\mathfrak{s} : \mathcal{A} \hookrightarrow \mathcal{Y}_{\mathfrak{X}}$ on all components. Recall that \mathfrak{s} corresponds to the map induced by the embedding $s : X/\!\!/N \to \mathfrak{X}$, and Lemma 8.1.3 defines an action of \mathbb{A}^1 on \mathfrak{X} that contracts to s. The action of \mathbb{A}^1 commutes with that of G, so we get an action $\mathbb{A}^1 \times \mathcal{Y}_{\mathfrak{X}} \to \mathcal{Y}_{\mathfrak{X}}$ such that $0 \times \mathcal{Y}_{\mathfrak{X}} \to \mathcal{Y}_{\mathfrak{X}}$ coincides with $\mathfrak{s} \circ \pi_{\mathfrak{X}}$. In this situation, the contraction principle ([BFGM02, Lemma 5.3], [Laf, Lemme 2.2], which is closely related to Braden's theorem [Bra03]) says that there is a natural isomorphism of functors $\pi_{\mathfrak{X},*} \cong \mathfrak{s}^* : \mathcal{D}^b_c(\mathcal{X}_{\mathfrak{X}}) \to \mathcal{D}^b_c(\mathcal{A})$.

Proof of Theorem 8.3.8. We will prove the first line of isomorphisms; the second line follows from the first by Verdier duality. If we apply the contraction principle to $\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi = \mathfrak{s}^{\check{\lambda},*}i_{*}\Psi$, we immediately get the first isomorphism $\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi \cong \pi_{\emptyset,*}\Psi$.

Next, we will show the isomorphism $\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi \cong \pi_*$. Recall that [Bei87b] gives an equivalence $D^b\mathrm{P}(\mathcal{Y}^{\check{\lambda}})\cong D_c^b(\mathcal{Y}^{\check{\lambda}})$, so we only need to define the isomorphism on perverse sheaves. Let $\mathcal{F}\in\mathrm{P}(\mathcal{Y}^{\check{\lambda}})$. For any $a\geq 1$, let \mathcal{L}_a denote the local system on \mathbb{G}_m whose monodromy is a unipotent Jordan block of rank a. There are canonical injections $\mathcal{L}_a\to\mathcal{L}_{a+1}$. Beilinson's construction of the unipotent nearby cycles functor (cf. [Bei87a, 2.3]) gives an isomorphism

$$\Psi(\mathcal{F}) \cong \underset{a>1}{\operatorname{colim}} i^* j_* (\mathcal{F} \boxtimes \mathcal{L}_a).$$

We can further apply $\mathfrak{s}_{\emptyset}^{\check{\lambda},*}$ to get an isomorphism $\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi(\mathcal{F})\cong\operatorname{colim}\mathfrak{s}^{\check{\lambda},*}j_{*}(\mathcal{F}\boxtimes\mathcal{L}_{a})$. Applying the contraction principle, we get an isomorphism

$$\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi(\mathfrak{F}) \cong \underset{a>1}{\operatorname{colim}} (\pi_{\mathfrak{X}} \circ j)_{*}(\mathfrak{F} \boxtimes \mathcal{L}_{a}).$$

Note that $\pi_{\mathfrak{X}} \circ j : \mathcal{Y} \times \mathbb{G}_m \to \mathcal{A}$ is equal to the composition of the first projection $\mathcal{Y} \times \mathbb{G}_m \to \mathcal{Y}$ and $\pi : \mathcal{Y} \to \mathcal{A}$. Therefore,

$$(\pi_{\mathfrak{X}} \circ j)_*(\mathfrak{F} \boxtimes \mathcal{L}_a) = \pi_*(\mathfrak{F}) \otimes H^*(\mathbb{G}_m, \mathcal{L}_a).$$

Since $\operatorname{colim}_{a\geq 1} H^*(\mathbb{G}_m, \mathcal{L}_a) = \overline{\mathbb{Q}}_{\ell}$, we conclude that $\mathfrak{s}_{\emptyset}^{\check{\lambda},*}\Psi(\mathfrak{F}) \cong \pi_*(\mathfrak{F})$.

9. Function-theoretic corollaries

9.1. **Pushforward of the basic function.** When k is the algebraic closure of a finite field \mathbb{F} , and X is defined over \mathbb{F} , satisfying the assumptions of §2.2, the action of the geometric Frobenius Fr morphism on $\bar{\pi}_!(\mathrm{IC}_{\bar{y}^{\check{\lambda}}})$ is described, up to a yet unknown permutation action on the set $\mathfrak{B}_X^+ = \bigcup_{\check{\lambda}} \mathfrak{B}_{X,\check{\lambda}}$ of central components of critical dimension, by Proposition 6.4.1. We use this to prove Theorems 1.1.2 and 1.1.4 from the introduction.

Proof of Theorems 1.1.2 and 1.1.4. Recall that \mathfrak{o} , in the context of these theorems, denotes the ring $\mathbb{F}[\![t]\!]$, where \mathbb{F} is the finite field of definition of X, and F is its fraction field.

We need to recall the definition of the IC function Φ_0 from [BNS16]: It is a function on $(X(\mathfrak{o}) \cap X^{\bullet}(F))/G(\mathfrak{o})$ which, in our case, is parametrized by the set $(\mathfrak{c}_X^-)^{\operatorname{Fr}}$ of elements of \mathfrak{c}_X^- that are fixed under the Galois group. To define it, choose an arc γ in the coset of such an element $\check{\theta}$, and consider a finite-dimensional formal model \widehat{Y}_y of the formal neighborhood of γ in the arc space L⁺X (Definition 3.8.1). In our case, we can take $Y = \mathcal{Y}^{\check{\theta}}$ and y = 0 the point $t^{\check{\theta}}$ on the central fiber $Y^{\check{\theta},\check{\theta}}$, by Theorem 3.8.2. Then, the value of Φ_0 on $\check{\theta}$ is equal to the trace of geometric Frobenius on the stalk of the intersection complex IC_Y at y, where the intersection complex is normalized to be constant (without Tate or cohomological twists) on the smooth locus of Y.

In our setting, this means that for every component \mathscr{Y} as in Proposition 6.4.1, the Tate and cohomological twist on $IC_{\mathscr{Y}}$ should be modified from $\overline{\mathbb{Q}}_{\ell}(\frac{\dim \mathscr{Y}}{2})[\dim \mathscr{Y}]$ to $\overline{\mathbb{Q}}_{\ell}$, and (6.11) should be replaced by the space

$$(9.1) \qquad \bigoplus_{\mathscr{Y}} \bigoplus_{\mathfrak{B}_{\mathscr{Y}}} \overline{\mathbb{Q}}_{\ell}(-\tfrac{\dim \mathscr{Y}}{2})[-\dim \mathscr{Y}] = \bigoplus_{\mathsf{b} \in \mathfrak{B}_{X,\bar{\lambda}}} \overline{\mathbb{Q}}_{\ell}(-\dim \mathsf{b} - \tfrac{1}{2})[-2\dim \mathsf{b} - 1].$$

To calculate the value at $\check{\lambda}(t)$ of the integral of the basic function that was denoted by $\pi_!\Phi_0$ in the introduction, for $\check{\lambda}$ fixed by Frobenius, we need to calculate the (alternating) trace of Frobenius on the fiber of $\bar{\pi}_!(\mathrm{IC}_{\bar{y}\check{\lambda}})$ over an \mathbb{F} -point of the diagonal $C \hookrightarrow \mathcal{A}^{\check{\lambda}}$, and then divide by the factor $tr_{\check{T}}(\mathrm{Fr},\mathrm{Sym}^{\bullet}(\check{\mathfrak{n}}(1)))$ in order to account for the difference between $\mathrm{IC}_{\bar{y}}$ and $\mathrm{IC}_{\bar{y}}$ (Corollary 4.5.9).

Taking into account the twists in the intersection complexes of the $C^{\mathfrak{R}}$'s in (6.12), we deduce that, with this normalization of the IC sheaves, Frobenius acts on that fiber as on

$$\bigoplus_{\deg(\mathfrak{R})=\check{\lambda}} \Bigl(\bigotimes_{\check{\mu}} \operatorname{Sym}^{N_{\check{\mu}}} \bigl(\bigoplus_{\mathsf{b} \in \mathfrak{B}_{X,\check{\mu}}} \overline{\mathbb{Q}}_{\ell} (-\dim \mathsf{b})[2\dim \mathsf{b}])\Bigr).$$

Thus, in the notation of Theorems 1.1.2 and 1.1.4, we have

$$\pi_!\Phi_0=tr_{\check{T}}(\mathrm{Fr},\mathrm{Sym}^{\bullet}(\check{\mathfrak{n}}(1)))^{-1}\cdot tr_{\check{T}}(\mathrm{Fr},\mathrm{Sym}^{\bullet}(\bigoplus_{\mathsf{b}\in\mathfrak{B}_X^+}\overline{\mathbb{Q}}_{\ell}(-\dim\mathsf{b}))),$$

where we remind that $\mathfrak{B}_X^+ = \bigcup_{\check{\mu} \in \mathfrak{c}_X} \mathfrak{B}_{X,\check{\mu}}$.

The dimensions dim b are given by Proposition 6.5.1: they are either equal to $\frac{1}{2}(\text{len}(D)-1)$, if b meets $\mathcal{Y}_{X\bullet}^{D_{\bullet}}$, or $\langle \rho_G, \check{\lambda} - \check{\theta} \rangle$, if b meets $\mathcal{Y}_{\text{sat}}^{\check{\Lambda},\check{\theta}}$ for $\check{\theta} \in \mathcal{D}_{\text{sat}}^G(X)$.

Both theorems assume the existence of a G-eigen-volume form on X^{\bullet} , which we fix, with eigencharacter \mathfrak{h} . We denote the absolute value of this eigencharacter by η . By Remark 5.4.4, for any $D \in \mathbb{N}^{\mathcal{D}}$ with $\varrho_X(D) = \check{\lambda}$, we have $\operatorname{len}(D) = \langle \mathfrak{h} + 2\rho_G, \check{\lambda} \rangle$. The effect of multiplying by $(\eta \delta)^{\frac{1}{2}}(t)$ will be to replace $\overline{\mathbb{Q}}_{\ell}(-\dim \mathfrak{b})$, in the expression above, by $\overline{\mathbb{Q}}_{\ell}(\frac{1}{2})$, for those \mathfrak{b} in $\mathcal{Y}_{X\bullet}^D$; this proves Theorem 1.1.2(i),(iv),(v).

For the rest of the components b, taking into account that $\check{\lambda} \geq \check{\theta}$ and \mathfrak{h} is trivial on the roots, therefore $\langle \mathfrak{h}, \check{\lambda} \rangle = \langle \mathfrak{h}, \check{\theta} \rangle$ and $\langle 2\rho_G, \check{\lambda} \rangle$ has the same parity as $\langle 2\rho_G, \check{\theta} \rangle$, the effect of multiplying by $(\eta \delta)^{\frac{1}{2}}(t)$ will be to replace $\overline{\mathbb{Q}}_{\ell}(-\dim \mathfrak{b})$ by $\overline{\mathbb{Q}}_{\ell}(\frac{\langle \mathfrak{h}+2\rho_G, \check{\theta} \rangle}{2})$, proving Theorem 1.1.4. The remaining parts, (ii), (iii) of Theorem 1.1.2 follow directly from Theorem 7.1.9.

9.2. **Asymptotics and the basic function.** We explain the proof of Corollary 1.2.1, computing the basic function and its asymptotics.

Proof of Corollary 1.2.1. The (function-theoretic) Radon transform $\pi_{\emptyset!}$ on X_{\emptyset}^{\bullet} is well-understood; in particular, the function $1_{X_{\alpha}^{\bullet}(\mathfrak{o})}$ maps to

$$\delta^{-\frac{1}{2}} tr_{\check{T}}(\operatorname{Fr}, \operatorname{Sym}^{\bullet}(\check{\mathfrak{n}})) \cdot tr_{\check{T}}(\operatorname{Fr}, \operatorname{Sym}^{\bullet}(\check{\mathfrak{n}}(1)))^{-1},$$

in the notation of (1.3). Moreover, $\pi_{\emptyset!}$ is equivariant for the action of the torus T(F). This implies the formula (1.9) for $e_{\emptyset}^*\Phi_0$.

This, in turn, implies the formula (1.10) for Φ_0 , by [Sak18, Corollary 5.5], where it is proven that, for an appropriate parametrization of the $G(\mathfrak{o})$ -orbits, any $G(\mathfrak{o})$ -invariant function Φ on $X^{\bullet}(F)$ is equal to its asymptotic $e_{\emptyset}^*\Phi$, restricted to the antidominant lattice $\check{\Lambda}^{-,\operatorname{Fr}}$. Note that that paper was written under the assumption of G being split (and X being "wavefront", which is automatic in our setting), but the proof of the result that we are quoting is valid without this assumption. (For example, "quasi-split" is enough for the Casselman–Shalika method, which is the basis of that argument, and for the asymptotics theory developed in [SV17].)

9.3. Euler factors of global integrals. Finally, to demonstrate how our results apply directly to compute Euler factors of global integrals of automorphic forms, let us return to the setting of Examples 1.1.3, 1.2.2, in order to discuss the Euler factorization of the pertinent global period integral. This will not directly invoke the nearby cycles functor (hence, is based on results of previous sections only), but it makes use of the asymptotics map e_{\emptyset}^* that we just recalled.

To recall, the group is $G = (\mathbb{G}_m \times \mathrm{SL}_2^n)/\mu_2$, and X is the affine closure of $H \setminus G$, where H is the product of the group H_0 of (1.1.3) with a copy of \mathbb{G}_m embedded diagonally into G as

$$(9.2) a \mod \mu_2 \mapsto \left(a, \begin{pmatrix} a \\ a^{-1} \end{pmatrix}^n\right) \mod \mu_2.$$

We let $\Phi = \prod_v \Phi_v$ be a function on $X^{\bullet}(\mathbb{A})$ as in Example 1.2.2.

Let $\phi \in \pi$ be a cusp form, where the restriction of π to \mathbb{G}_m is a character of the form $\chi_0| \bullet |^s$ for χ_0 unitary and some $s \gg 0$. Fix a nontrivial character ψ of \mathbb{A}/\mathbb{k} , identified as a character of the quotient N^-/H_0 , where N^- is the lower triangular unipotent subgroup (mapping to the additive group through summation of the entries). The standard method of writing ϕ in terms of its Whittaker function W_{ϕ} with respect to (N^-, ψ) shows that the integral

$$\int_{G(\Bbbk)\backslash G(\mathbb{A})} \phi(g) \sum_{\gamma \in X^{\bullet}(\Bbbk)} \Phi(\gamma g) dg = \int_{X^{\bullet}(\mathbb{A})} \Phi(g) \int_{H(\Bbbk)\backslash H(\mathbb{A})} \phi(hg) dh dg$$

"unfolds" to the Eulerian integral

$$\int_{H_0\backslash G(\mathbb{A})} W_{\phi}(g)\Phi(g)dg.$$

We can write the Whittaker function as $W_{\Phi}(g) = \prod_{v} W_{\phi,v}(g_v)$, so that $W_{\phi,v}(1) = 1$ for almost all v, and we can factorize the invariant measure on $H_0 \setminus G$ so that for almost all v, $\mu_v(H_0 \setminus G(\mathfrak{o}_v)) = 1$.

Proposition 9.3.1. At places v where $W_{\phi,v}(1)$ is $G(\mathfrak{o}_v)$ -invariant with $W_{\phi,v}(1)=1$ (in particular, π_v is unramified), $\Phi_v=\Phi_{0,v}$ (the IC function), $\mu_v(H_0\backslash G(\mathfrak{o}_v))=1$, and the conductor of $\psi|_{\mathbb{K}_v}$ is the ring of integers, we have

(9.3)
$$\int_{H_0\backslash G(\mathbb{K}_v)} W_{\phi,v}(g_v) \Phi_v(g_v) dg_v = L(\pi_v, \otimes, 1 - \frac{n}{2}).$$

Proof. The proof can be obtained by direct computation from the explicit formula (1.10) for the basic function. However, we would like to sketch a "pure thought" argument, which applies to other cases as well, without providing all details. We drop the index v, denoting the place under consideration, and write F for \mathbb{k}_v .

First of all, the square of the absolute value of (9.3) follows immediately from Plancherel-theoretic considerations. Indeed, we can write the local integral as

$$Z(\Phi, \pi) = \int_{N^- \backslash G(F)} W_{\phi}(g) W_{\Phi}(g) dg,$$

where

(9.4)
$$W_{\Phi}(g) = \int_{H_0 \backslash N^-(F)} \Phi(ng) \psi(n) dn.$$

We identify the abelianization G^{ab} with \mathbb{G}_m , so that the composite $\mathbb{G}_m \to G \to \mathbb{G}_m$ is the square character. Let η be the character $a \mapsto |a|^{1-n}$ on G — it is the character by which it acts on the SL_2^n -invariant measure on X^{\bullet} . Fix that measure giving volume 1 to $X^{\bullet}(\mathfrak{o})$. It is known from [SV17, Theorem 9.5.9] that the "unfolding" map $\Phi \mapsto W_{\Phi}$ extends to an L^2 -isometry

$$L^2(X^{\bullet}(F)) \xrightarrow{\sim} L^2(N^- \backslash G(F), \psi^{-1}),$$

where the Haar measure on $N^-\backslash G(F)$ is fixed so that the volume of $N^-\backslash G(\mathfrak{o})$ is 1.

The local zeta integrals appear in the Plancherel decomposition of Whittaker functions,

(9.5)
$$\|\Phi\|_{L^2(X^{\bullet}(F))}^2 = \|W_{\Phi}\|_{L^2(N(F)\backslash G(F),\psi^{-1})}^2 = \int |Z(\Phi,\pi)|^2 d\pi,$$

where π ranges over the unitary unramified dual of G(F), which can be identified with the set of semisimple conjugacy classes in the compact form of the complex dual group $\check{G}(\mathbb{C})$, and the Plancherel measure $d\pi$ is the Weyl measure $|(1-e^{\check{\alpha}})|^2 d\chi$ (when we identify the unramified dual with the quotient \check{T}^1/W , where \check{T}^1 is the group of unitary unramified characters — the maximal compact subgroup of the complex points of the dual Cartan \check{T}).

On the other hand, the Plancherel decomposition of Φ can also be expressed in terms of its asymptotics:

$$\|\Phi\|_{L^2(X^{\bullet}(F))}^2 = \frac{1}{|W|} \|e_{\emptyset}^* \Phi\|_{L^2(X_{\emptyset}^{\bullet}(F))}^2,$$

where we have applied [SV17, Theorem 7.3.1], together with the following observations: 1) our calculation of $e_{\emptyset}^*\Phi$ shows that it already lies in $L^2(X_{\emptyset}^{\bullet}(F))$, so there is no difference between that and what is denoted by $\iota_{\emptyset}^*\Phi$ in loc.cit.; 2) there are no unramified functions Φ on $X^{\bullet}(F)$ with $e_{\emptyset}^*\Phi = 0$; this is essentially [Sak08, Theorem 6.2.1], and it implies, together with the previous point, that the summands of [SV17, Theorem 7.3.1] with $\Theta \neq \emptyset$ do not contribute to the Plancherel decomposition. (This is not an essential point for this proof; the argument would go through even if there were contributions outside of the most continuous spectrum.)

By our formula (1.4), as specialized in Example 1.1.3, and the commutation of asymptotics with Radon transform, (1.8), we can write the Plancherel decomposition of the basic function as

$$(9.6) \|\Phi\|_{L^{2}(X^{\bullet}(F))}^{2} = \frac{1}{|W|} \int_{\check{T}^{1}} \left| \frac{\prod_{\check{\alpha} \in \check{\Phi}^{+}} (1 - e^{\check{\alpha}}(\chi))}{\prod_{\check{\lambda} \in \mathfrak{B}^{+}} (1 - q^{-\frac{1}{2}} e^{\check{\lambda}}(\chi))} \right|^{2} d\chi = \int L(\pi, \otimes, \frac{1}{2}) L(\tilde{\pi}, \otimes, \frac{1}{2}) d\pi.$$

Comparing (9.5) and (9.6), we get that $|Z(\Phi,\pi)|^2 = L(\pi,\otimes,\frac{1}{2})L(\tilde{\pi},\otimes,\frac{1}{2})$ for π unitary. This is what we get "for free" from the L^2 -decomposition.

To upgrade it to a formula on $Z(\Phi, \pi)$, we utilize the following information about the image W_{Φ} under the unfolding map. Note that this is a $G(\mathfrak{o})$ -invariant Whittaker function, hence can be identified as a function on the *antidominant* coweights, by evaluating it on the elements $t^{\check{\theta}}$. The facts that we need are:

(i) The support of W_{Φ} lies in the intersection of the antidominant lattice with the cone generated by \mathfrak{c}_X^- and the coweight

$$\check{\theta}_0 := -\frac{\check{\alpha}_1 + \check{\alpha}_2 + \dots + \check{\alpha}_n - \check{m}}{2}.$$

Its value at $1 = t^0$ is 1.

Indeed, the support of Φ lies in $H\xi t^{\mathfrak{c}_{\overline{X}}}G(\mathfrak{o})$, where ξ is the diagonal image of $\begin{pmatrix} 1 & 0 \\ 1 & 1 \end{pmatrix}$, and from the definition (9.4) of the unfolding map, the support of W_{Φ} will lie in $H\xi N^-t^{\mathfrak{c}_{\overline{X}}}G(\mathfrak{o}) = N^-\mathbb{G}_m t^{\mathfrak{c}_{\overline{X}}}G(\mathfrak{o})$, where \mathbb{G}_m is embedded as in (9.2). But a $G(\mathfrak{o})$ -invariant Whittaker function with respect to a character of N^- whose conductor is \mathfrak{o} can only be supported on antidominant elements.

When g = 1, the integrand of (9.4) only lies in the support of Φ when $n \in H_0 \backslash N^-(\mathfrak{o})$.

- (ii) The value of W_{Φ} at any $t^{\check{\theta}}$ is a polynomial in $q^{-\frac{1}{2}}$.
 - This fact, which can be seen as expressing the "motivic" nature of this function, requires some explanation. First of all, the statement is true if W_{Φ} is replaced by the pushforward $\pi_!\Phi$, by Theorem 1.1.2 (see also §9.1). Secondly, the unfolding integral (9.4) of the characteristic function of each $G(\mathfrak{o})$ -orbit has this property; this can be seen by direct calculation, or by a similar geometric argument, and we omit the details.
- (iii) If $\varphi_{\check{\theta}}$ denotes the $G(\mathfrak{o})$ -invariant Whittaker function supported on the coset $N^-t^{\check{\theta}}G(\mathfrak{o})$ (with $\check{\theta}$ antidominant) and equal to 1 on $\check{\theta}$, its asymptotic $e_{\check{\theta}}^*\varphi_{\check{\theta}}$ is supported on the union of a finite number of cosets $N^-t^{\check{\theta}'}G(\mathfrak{o})$, with $\check{\theta}'-\check{\theta}$ in the positive root monoid, and is a polynomial in $q^{-\frac{1}{2}}$.

Here, the asymptotics map is for the Whittaker model, but we denote it by the same symbol. It takes values in $N^-\backslash G(F)$, without a character on N^- . This fact is a corollary of the Casselman–Shalika formula. (See [Sak18, Theorem 6.8 and Example 6.4] for an interpretation of the Casseman–Shalika formula in terms of asymptotics.)

Granted, now, the facts above, we can represent $e_{\emptyset}^*W_{\Phi}$ as a formal series in the elements $e^{\check{\theta}}$, which stand for the characteristic functions of the sets $N^-t^{\check{\theta}}G(\mathfrak{o})$, multiplied by $\delta^{-\frac{1}{2}}(t^{\check{\theta}}) = q^{\langle \rho_G,\check{\theta} \rangle}$, with coefficients in $\mathbb{C}[q^{-\frac{1}{2}}]$:

$$e_{\emptyset}^* W_{\Phi} = \sum_{i,\check{\theta}} c_{i,\check{\theta}} q^{-\frac{i}{2}} e^{\check{\theta}}.$$

According to the first and the third facts above, the support of this function lies in the set of $\hat{\theta}$'s which are translates, by the positive coroot monoid, of the intersection

$$(\mathfrak{c}_X^- + \operatorname{span}(\check{\theta}_0)) \cap \check{\Lambda}^-.$$

The description of colors of X in Example 1.1.3 allows us to conclude that this monoid only includes coweights for which the "determinant" character

$$\frac{\kappa\check{\mu} + \sum_{i} \kappa_{i}\check{\alpha}_{i}}{2} \mapsto \kappa$$

is nonnegative; moreover, the restriction to the kernel of the determinant cocharacter is simply equal to the asymptotics of the "basic Whittaker function" φ_0 , which by [Sak18, Theorem 6.8 and Example 6.4] is equal to

$$\prod_{\check{\alpha}\in\check{\Phi}^+}(1-e^{\check{\alpha}}).$$

Thus, if we "divide" the function $e_{\emptyset}^*W_{\Phi}$ by the product above (this corresponds to acting on it by a series in the Hecke algebra of the torus T), we obtain another series

$$\prod_{\check{\alpha}\in\check{\Phi}^+} (1 - e^{\check{\alpha}})^{-1} e_{\emptyset}^* W_{\Phi} = 1 \cdot e^0 + \sum_{j=1}^{\infty} \sum_{i=0}^{\infty} q^{-\frac{i}{2}} e^{j\frac{\check{\mu}}{2}} C_{i,j},$$

where the $C_{i,j}$'s are finite linear combinations of the elements e^{λ} , where $\check{\lambda}$ now ranges over half-multiples of elements in the coroot lattice. (The finiteness of the linear combination also follows from [Sak18, Theorem 6.8 and Example 6.4]: the asymptotics of every $\varphi_{\tilde{\theta}}$ is a "multiple" of the factor $\prod_{\check{\alpha}\in\check{\Phi}^+}(1-e^{\check{\alpha}})$.) Note that $\frac{\check{\mu}}{2}$ is not, by itself, a coweight of G, so one has to expand this series to make sense of it as a linear combination of the elements $e^{\hat{\theta}}$ with $\check{\theta} \in \check{\Lambda}$.

Now we invoke the Plancherel formula: the fact that the unfolding map is an isometry tells us that the Plancherel density of W_{Φ} is also given by (9.6). On the other hand, this Plancherel density can be expressed in terms of the asymptotics $e_{\emptyset}^*W_{\Phi}$ as

$$||W_{\Phi}||^{2} = \frac{1}{|W|} ||e_{\emptyset}^{*}W_{\Phi}||^{2} = \frac{1}{|W|} \int_{\tilde{T}^{1}} \left| 1 \cdot e^{0} + \sum_{j=1}^{\infty} \sum_{i=0}^{\infty} q^{-\frac{i}{2}} e^{j\frac{\tilde{\mu}}{2}} C_{i,j} \right|^{2} \left| \prod_{\check{\alpha} \in \check{\Phi}^{+}} (1 - e^{\check{\alpha}}) \right|^{2} d\chi =$$

$$= \frac{1}{|W|} \int_{\check{T}^{1}} \left(1 \cdot e^{0} + \sum_{j=1}^{\infty} \sum_{i=0}^{\infty} q^{-\frac{i}{2}} e^{j\frac{\tilde{\mu}}{2}} C_{i,j} \right) \left(1 \cdot e^{0} + \sum_{j=1}^{\infty} \sum_{i=0}^{\infty} q^{-\frac{i}{2}} e^{-j\frac{\tilde{\mu}}{2}} C_{i,j}^{*} \right) \prod_{\check{\alpha} \in \check{\Phi}^{+}} \left| (1 - e^{\check{\alpha}}) \right|^{2} d\chi,$$

where we now identify the elements $e^{\check{\theta}}$ with characters of the dual torus \check{T} , use the fact that for a unitary character $\overline{e^{\check{\theta}}(\chi)} = e^{-\check{\theta}}(\chi)$, and set $C^*_{i,j} = \sum_{\check{\lambda}} \overline{c_{\check{\lambda}}} e^{-\check{\lambda}}$ if $C_{i,j} = \sum_{\check{\lambda}} c_{\check{\lambda}} e^{\check{\lambda}}$. In other words, we have expressed the Plancherel density

$$L(\pi, \otimes, \frac{1}{2})L(\tilde{\pi}, \otimes, \frac{1}{2})$$

as the product

$$\left(1 \cdot e^0 + \sum_{j=1}^{\infty} \sum_{i=0}^{\infty} q^{-\frac{i}{2}} e^{j\frac{\check{\mu}}{2}} C_{i,j}\right) \left(1 \cdot e^0 + \sum_{j=1}^{\infty} \sum_{i=0}^{\infty} q^{-\frac{i}{2}} e^{-j\frac{\check{\mu}}{2}} C_{i,j}^*\right).$$

We will be done if we can identify the first factor of the former with the first factor of the latter. But, viewed as series in the elements $q^{-\frac{i}{2}}e^{-j\frac{\mu}{2}}$ (with coefficients in polynomials in the group ring of the half-root lattice), the first factor of the former is the restriction of the series to the elements of the form $q^{-\frac{i}{2}}e^{i\frac{\tilde{\mu}}{2}}$, while the product $L(\pi, \otimes, \frac{1}{2})L(\tilde{\pi}, \otimes, \frac{1}{2})$ is supported on elements of the form $q^{-\frac{i}{2}}e^{j\frac{\tilde{\mu}}{2}}$ with $j \leq i$. It follows that the series $1 \cdot e^0 + \sum_{j=1}^{\infty} \sum_{i=0}^{\infty} q^{-\frac{i}{2}}e^{j\frac{\tilde{\mu}}{2}}C_{i,j}$ is also supported on elements of the form $q^{-\frac{i}{2}}e^{j\frac{\tilde{\mu}}{2}}$ with $j \leq i$, coincides with $L(\pi, \otimes, \frac{1}{2})$ on elements with i=j, and a simple inductive argument in the variable i-j shows that it coincides with it everywhere.

APPENDIX A. PROPERTIES OF THE GLOBAL STRATIFICATION

A.1. The factorizable space of formal loops. We briskly review the definitions of multipoint versions of the spaces of formal arcs and formal loops. We refer the reader to [KV04], [Zhu17, §3.1] for a more complete account.

Let C be a smooth curve over k. For any $N \in \mathbb{N}$, we have the Nth symmetric product $C^{(N)}$ of C, which identifies with the Hilbert scheme $\operatorname{Hilb}^{N}(C)$ parametrizing relative effective divisors in C of degree N.

Recall that if S is an affine scheme and $D \subset C \times S$ is a closed affine subscheme, we denote by \widehat{C}'_D the spectrum of the ring of regular functions on the formal completion of $C \times S$ along D (so \widehat{C}'_D is a true scheme, not merely a formal scheme). Let $\widehat{C}^{\circ}_D := \widehat{C}'_D - D$ denote the open subscheme.

For any k-scheme X, we define the global space of formal arcs by the functor

$$(\mathcal{L}^+X)_{C^{(N)}}(S) = \{ D \in C^{(N)}(S), \gamma \in X(\widehat{C}'_D) \}.$$

for affine test schemes S. By [KV04, Proposition 2.4.1], the functor $(\mathcal{L}^+X)_{C^{(N)}}$ is representable by a scheme of infinite type over $C^{(N)}$. If X is affine, then $(\mathcal{L}^+X)_{C^{(N)}}$ is affine over $C^{(N)}$, cf. [KV04, 2.4.3]. More specifically, define the space of n-jets $(\mathcal{L}_n^+X)_{C^{(N)}}$ by

$$(\mathcal{L}_{n}^{+}X)_{C^{(N)}}(S) = \{D \in C^{(N)}(S), \gamma \in X(\widehat{C}_{D}^{n})\},$$

where \widehat{C}_D^n denotes the *n*th infinitesimal neighborhood of D in $C \times S$. Then $(\mathcal{L}_n^+ X)_{C^{(N)}}$ is representable by a scheme over $C^{(N)}$, which is affine (resp. of finite type) if X is. As n varies the schemes $(\mathcal{L}_n^+ X)_{C^{(N)}}$ form a projective system of schemes with affine transition maps, and $(\mathcal{L}_n^+ X)_{C^{(N)}}$ is equal to the projective limit of this system. If X is smooth, the schemes $(\mathcal{L}_n^+ X)_{C^{(N)}}$ are smooth over $C^{(N)}$ with smooth surjective transition maps (cf. [Ras, Lemma 2.5.1]).

We can also define the functor for the global loop space by

$$(\mathcal{L}X)_{C^{(N)}}(S) = \{D \in C^{(N)}(S), \gamma \in X(\widehat{C}_D^{\circ})\}.$$

If X is affine, then $(\mathcal{L}X)_{C^{(N)}}$ is representable by an ind-scheme ind-affine over $C^{(N)}$, cf. [KV04, Proposition 2.5.2]. We have a closed embedding $(\mathcal{L}^+X)_{C^{(N)}} \hookrightarrow (\mathcal{L}X)_{C^{(N)}}$.

A.1.1. Let $\operatorname{Aut}^0(k\llbracket t \rrbracket)$ denote the functor sending a k-algebra R to the group of R-algebra automorphisms of $R\llbracket t \rrbracket$ that reduce to the identity map mod t. Then $\operatorname{Aut}^0(k\llbracket t \rrbracket)$ is representable by the group scheme Spec $k[a_1^{\pm 1}, a_2, a_3, \ldots]$, cf. [Zhu17, (1.3.13)].

There is an $\operatorname{Aut}^0(k[\![t]\!])$ -torsor $\operatorname{Coord}^0(C) \to C$ classifying $v \in C$ together with an isomorphism $k[\![t]\!] \cong \mathfrak{o}_v$ that sends t to a uniformizer, cf. [Zhu17, (3.1.10)]. We can think of $(\mathcal{L}^+X)_C$, $(\mathcal{L}X)_C$ as twisted products

$$(\mathcal{L}^+X)_C = C \,\tilde{\times}\, \mathsf{L}^+X, \qquad (\mathcal{L}X)_C = C \,\tilde{\times}\, \mathsf{L}X,$$

where $C \times \mathsf{L}^+ X := \mathrm{Coord}^0(C) \times^{\mathrm{Aut}^0(k[\![t]\!])} \mathsf{L}^+ X$.

Remark A.1.2. The space $\mathcal{L}X$ really lives over the Ran space of C. Essentially this just means $\mathcal{L}X$ only cares about the support of the divisor D and not its multiplicities. More specifically for any finite set I we have a map $C^I \to C^{(|I|)}$ where |I| denotes the cardinality of I. Then the spaces $(\mathcal{L}X)_{C^I} := C^I \times_{C^{(|I|)}} (\mathcal{L}X)_{C^{(|I|)}}$ have a factorization monoid structure as defined in [KV04, Definition 2.2.1], and we can think of the collection of these spaces as $(\mathcal{L}X)_{\operatorname{Ran}_C}$. This is certainly the more philosophically correct approach to considering the loop space, but for technical simplicity it will suffice for our study of arc spaces to work with \mathcal{L}^+X over $\operatorname{Sym} C$.

A.1.3. We can apply the constructions above to the algebraic group G. Since G is smooth, \mathcal{L}^+G is a group scheme formally smooth over Sym C.

Consider the (factorizable) Beilinson–Drinfeld affine Grassmannian $\operatorname{Gr}_{G,\operatorname{Sym} C}$ defined in §3.7. By Beauville–Laszlo's theorem (see [BD96, Remark 2.3.7], [Zhu17, Proposition 3.1.9]), we have an isomorphism $\operatorname{Gr}_{G,\operatorname{Sym} C} \cong \mathcal{L}G/\mathcal{L}^+G$.

A.1.4. Let X be an affine spherical G-variety. Define

$$\mathcal{L}^{\bullet}X := \mathcal{L}X - \mathcal{L}(X - X^{\bullet}),$$

which admits an open embedding into $\mathcal{L}X$. The G-action on X induces a natural action of $\mathcal{L}G$ on $\mathcal{L}X$ and the subspace $\mathcal{L}^{\bullet}X$ (resp. $\mathcal{L}^{+}X$) is stable under the action of $\mathcal{L}G$ (resp. $\mathcal{L}^{+}G$).

We will primarily be concerned with the global space of non-degenerate arcs

$$(\mathcal{L}^+X)^{\bullet} := \mathcal{L}^+X \times_{\mathcal{L}X} \mathcal{L}^{\bullet}X.$$

The study of the loop space $\mathcal{L}^{\bullet}X$ is beyond the scope of this paper.

A.2. **Multi-point orbits.** We now consider the \mathcal{L}^+G -orbits on $(\mathcal{L}^+X)^{\bullet}$, and we will prove a multi-point version of Proposition 2.3.7.

A.2.1. What is going on at the level of k-points. A k-point of Ran_C is a nonempty finite subset $\{v_i \in |C|\}_{i \in I}$ of points on C. Then a k-point of $(\mathcal{L}^+X)^{\bullet}_{\operatorname{Ran}_C}$ over $\{v_i\}$ consists of points $x_i \in X(\mathfrak{o}_{v_i}) \cap X^{\bullet}(F_{v_i})$. Each x_i belongs to an orbit $X^{\bullet}(F_{v_i})_{G:\check{\theta}_i}$ for a unique $\check{\theta}_i \in \mathfrak{c}_X^-$ by Theorem 2.3.5(ii). The collection $\check{\theta}_i$, $i \in I$ forms a multiset in \mathfrak{c}_X^- . Note that $\check{\theta}_i$ may be zero. Therefore, the tuple $(x_i)_{i \in I}$ is a k-point in the product of orbits $\prod_{v \in I} \mathsf{L}_{v_i}^{\check{\theta}_i} X$.

The idea for what follows is that this product of orbits only depends on the unordered multiset $\{\check{\theta}_i\}$, counted with multiplicity. Moreover since \mathcal{C}_0 is a strictly convex cone, the set of formal sums $\sum_I \check{\theta}_i \cdot v_i$ admits a positive grading.

A.2.2. Construction. Let $\check{\Theta}_0$ denote an (unordered) multiset in \mathfrak{c}_X^- , by which we mean a formal sum

$$\sum_{\check{\theta} \in \mathfrak{c}_X^-} N_{\check{\theta}} [\check{\theta}] \in \operatorname{Sym}^\infty(\mathfrak{c}_X^-)$$

where all but finitely many $N_{\tilde{\theta}} = 0$. Note that we include the case $\tilde{\theta} = 0$ and N_0 may be nonzero. See §3.1.4 for notation.

We have an étale map

$$\mathring{C}^{|\check{\Theta}_0|} = \prod_{\check{\theta} \in \mathfrak{c}_X^-} \mathring{C}^{N_{\check{\theta}}} \to \prod_{\check{\theta} \in \mathfrak{c}_X^-} \mathring{C}^{(N_{\check{\theta}})} = \mathring{C}^{\check{\Theta}_0}.$$

By the factorization property, the base change $\mathring{C}^{|\check{\Theta}_0|} \times_{\operatorname{Sym} C} \mathcal{L}^+ X$ identifies with the $|\check{\Theta}_0|$ -fold disjoint product of $C \times \mathsf{L}^+ X$. By Proposition 2.3.7, we have the locally closed subscheme

(A.1)
$$\prod_{\check{\theta} \in \mathfrak{c}_{*}^{-}} (C \, \check{\times} \, \mathsf{L}^{\check{\theta}} X)^{\check{\times} N_{\check{\theta}}} \hookrightarrow \mathring{C}^{|\check{\Theta}_{0}|} \underset{\operatorname{Sym}}{\times}_{\mathcal{C}} \mathcal{L}^{+} X.$$

This is stable under the action of $\prod_{\check{\theta}} \mathfrak{S}_{N_{\check{\theta}}}$, and therefore (A.1) descends to a locally closed subscheme

$$\mathcal{L}^{\check{\Theta}_0}X \hookrightarrow \mathring{C}^{\check{\Theta}_0} \underset{\operatorname{Sym} C}{\times} \mathcal{L}^+X =: (\mathcal{L}^+X)_{\mathring{C}^{\check{\Theta}_0}}.$$

Proposition A.2.3. The scheme $\mathcal{L}^{\check{\Theta}_0}X$ is formally smooth over $\mathring{C}^{\check{\Theta}_0}$, and second projection induces a locally closed embedding $\mathcal{L}^{\check{\Theta}_0}X \hookrightarrow \mathcal{L}^+X$.

Proof. Formal smoothness follows from Proposition 2.3.7. It remains to show that the second projection $\operatorname{pr}_2: \mathcal{L}^{\Theta_0}X \to \mathcal{L}^+X$ is a locally closed embedding. Note that pr_2 is the composition of the finite étale map

$$\operatorname{pr}_2': (\mathcal{L}^+X)_{\mathring{C}^{\check{\Theta}_0}} \to (\mathcal{L}^+X)_{\mathring{C}^{(|\check{\Theta}_0|)}} := \mathring{C}^{(|\check{\Theta}_0|)} \underset{\operatorname{Sym} C}{\times} \mathcal{L}^+X$$

and the open embedding $(\mathcal{L}^+X)_{\mathring{C}(|\Theta_0|)} \hookrightarrow \mathcal{L}^+X$. Hence it suffices to show that the restriction $\mathcal{L}^{\check{\Theta}_0}X \subset (\mathcal{L}^+X)_{\mathring{C}^{\check{\Theta}_0}} \to (\mathcal{L}^+X)_{\mathring{C}(|\Theta_0|)}$ is locally closed. Let Y be the closure of $\mathcal{L}^{\check{\Theta}_0}X$ in $(\mathcal{L}^+X)_{\mathring{C}^{\check{\Theta}_0}}$, so $Y' = \operatorname{pr}_2'(Y)$ is a closed subscheme of $(\mathcal{L}^+X)_{\mathring{C}(|\Theta_0|)}$ and $\operatorname{pr}_2'(\mathcal{L}^{\check{\Theta}_0}X)$ is open in Y'. Observe that the induced map

(A.2)
$$\mathcal{L}^{\check{\Theta}_0} X \to \operatorname{pr}'_2(\mathcal{L}^{\check{\Theta}_0} X)$$

is a bijection on geometric points: a point in the right hand side consists of an unordered set of points $\{v_i\} \subset C$ and $x_i \in \mathsf{L}^+_{v_i}X$ such that if $\check{\theta}_i$ denotes the G-valuation of x_i , then $\sum_i [\check{\theta}_i] = \check{\Theta}_0$. There is a unique way to partition the v_i 's according to distinct values of $\check{\theta}_i$'s to get a point in $\mathcal{L}^{\check{\Theta}_0}X$. Therefore (A.2) is étale and a bijection on geometric points, hence an isomorphism. \square

A.3. **Proof of Lemma 3.1.6.** Assume that C is complete. Let $(\mathcal{M}_X \times \operatorname{Sym} C)^{\bullet}$ denote the substack of $\mathcal{M}_X \times \operatorname{Sym} C$ consisting of those pairs (f, D) where $f(C - D) \subset X^{\bullet}/G$, i.e., C - D is contained in the non-degenerate locus. This is an open substack since C is complete. Define the map

$$(A.3) \qquad (\mathcal{M}_X \times \operatorname{Sym} C)^{\bullet} \to \mathcal{L}^+ X / \mathcal{L}^+ G$$

over Sym C by sending (f, D) to $(f|_{\widehat{C}'_D}, D)$. Here we are using the fact that an \mathcal{L}^+G -torsor on an affine scheme S is the same as a G-torsor on \widehat{C}'_D by formal lifting.

Recall that we defined a partition $\check{\Theta}$, or unordered multiset without zero, in \mathfrak{c}_X^- , to mean an element of $\operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$. To such a partition $\check{\Theta}$, we have a corresponding locally closed subscheme $\mathcal{L}^{\check{\Theta}}X \hookrightarrow \mathcal{L}^+X$ by Proposition A.2.3. Then we can define $\mathfrak{M}_X^{\check{\Theta}}$ to be the preimage of $\mathcal{L}^{\check{\Theta}}X/\mathcal{L}^+G$ under the map (A.3). One can check from the construction that this definition gives the description on k-points from §3.1.5. By base change $\mathfrak{M}_X^{\check{\Theta}}$ is a locally closed subscheme of $\mathfrak{M}_X \times C^{(|\check{\Phi}|)}$. In particular, $\mathfrak{M}_X^{\check{\Theta}}$ is an algebraic stack locally of finite type over k.

Lemma A.3.1. The natural map $\mathcal{M}_X^{\check{\Theta}} \to \mathcal{L}^{\check{\Theta}}X/\mathcal{L}^+G$ is formally smooth.

Proof. Let $S \hookrightarrow S'$ be a nilpotent thickening of affine schemes. Let $(f,D) \in \mathcal{M}_X^{\check{\Theta}}(S)$. This maps to the point $(f|_{\widehat{C}'_D},D) \in \mathcal{L}^{\check{\Theta}}X(S)$. Suppose that we have a lift of this point to $(\hat{f},D') \in \mathcal{L}^{\check{\Theta}}X(S)$.

 $\mathcal{L}^{\check{\Theta}}X(S')$, where $D' \subset C \times S'$ is a relative effective Cartier divisor and $\hat{f}: \widehat{C}'_{D'} \to X/G$ is equivalent to the datum of a G-bundle $\hat{\mathcal{P}}'_{G}$ on $\widehat{C}'_{D'}$ and a section $\hat{\sigma}': \widehat{C}'_{D'} \to X \times^G \hat{\mathcal{P}}'_{G}$.

We would like to lift (f,D) to an S'-point of $\mathcal{M}_X^{\hat{\Theta}}$ that maps to (\hat{f},D') . The map $f:C\times S\to X/G$ consists of the datum of a G-bundle \mathcal{P}_G on $C\times S$ and a section $\sigma:C\times S\to X\times^G\mathcal{P}_G$ satisfying the condition that $\sigma(C\times S-D)\subset X^{\bullet}\times^G\mathcal{P}_G=(H\backslash G)\times^G\mathcal{P}_G$. The restriction $\sigma|_{C\times S-D}$ gives a reduction of $\mathcal{P}_G|_{C\times S-D}\cong G\times^H\mathcal{P}_H$ to an H-bundle \mathcal{P}_H on $C\times S-D$ such that $\sigma|_{C\times S-D}$ identifies with the canonical section

$$C \times S - D \cong H \backslash \mathcal{P}_H \hookrightarrow (H \backslash G) \overset{H}{\times} \mathcal{P}_H \cong (H \backslash G) \overset{G}{\times} \mathcal{P}_G|_{C \times S - D}$$

corresponding to $H1 \in H \backslash G$.

The obstruction to lifting \mathcal{P}_H to an H-bundle \mathcal{P}'_H on $C \times S' - D'$ is an element in $H^2(C \times S - D, \mathfrak{h}_{\mathcal{P}_H} \otimes_{\mathcal{O}_S} I)$ where I is the zero ideal of $S \hookrightarrow S'$ and $\mathfrak{h}_{\mathcal{P}_H}$ denotes the quasicoherent sheaf on $C \times S - D$ obtained by twisting the adjoint representation of H by \mathcal{P}_H . This obstruction vanishes since $C \times S - D$ has relative dimension 1 over S and we can compute cohomology over the Zariski site. Thus, we obtained an H-bundle \mathcal{P}'_H over $C \times S' - D'$. Let $\sigma' : C \times S' - D' \to X \times^G (G \times^H \mathcal{P}'_H)$ denote the corresponding section.

We know that after base change to S, there exists an isomorphism

$$\tau: \hat{\mathcal{P}}'_{G}|_{\widehat{C}^{\circ}_{D}} \cong \mathcal{P}_{G}|_{\widehat{C}^{\circ}_{D}} \cong (G \overset{H}{\times} \mathcal{P}'_{H})|_{\widehat{C}^{\circ}_{D}}$$

such that $\tau \circ \hat{\sigma}'|_{\widehat{C}_D^{\circ}} = \sigma'|_{\widehat{C}_D^{\circ}}$. This is equivalent to a section $\beta : \widehat{C}_D^{\circ} \to \mathcal{P}'_H \times^H G \times^G \widehat{\mathcal{P}}'_G$ such that β is sent under

to the restriction $\hat{\sigma}'|_{\widehat{C}_D^{\circ}}$. The map (A.4) is smooth since $G \to X^{\bullet} = H \backslash G$ is smooth. The scheme $\widehat{C}_{D'}^{\circ}$ is affine since S' is affine. The zero ideal of $\widehat{C}_D^{\circ} \hookrightarrow \widehat{C}_{D'}^{\circ}$ is still nilpotent, so by formal smoothness of (A.4), we can lift β to a section β' that maps to $\hat{\sigma}'|_{\widehat{C}_{D'}^{\circ}}$. Such a section β' is equivalent to an isomorphism $\tau': \widehat{\mathcal{P}}'_{G}|_{\widehat{C}_{D'}^{\circ}} \cong (G \times^H \mathcal{P}'_H)|_{\widehat{C}_{D'}^{\circ}}$ such that $\tau' \circ \hat{\sigma}'|_{\widehat{C}_{D'}^{\circ}} = \sigma'|_{\widehat{C}_{D'}^{\circ}}$. By Beauville–Laszlo's theorem (Lemma 3.7.7), the data $((\widehat{\mathcal{P}}'_G, \widehat{\sigma}'), (\mathcal{P}'_H, \sigma'), \tau')$ descends to a map $f': C \times S' \to X/G$. By construction, (f', D') is an S'-point of \mathcal{M}_X° lifting f.

Proof of Lemma 3.1.6. Lemma A.3.1 and Proposition A.2.3 together imply that $\mathcal{M}_X^{\check{\Theta}}$ is formally smooth over k. Since $\mathcal{M}_X^{\check{\Theta}}$ is locally of finite type, it is therefore smooth over k.

We claim that the first projection $\operatorname{pr}_1: \mathcal{M}_X \times C^{(|\check{\Theta}|)} \to \mathcal{M}_X$ induces a locally closed embedding $\mathcal{M}_X^{\check{\Theta}} \hookrightarrow \mathcal{M}_X$. Let $Z \subset \mathcal{M}_X \times C^{(|\check{\Theta}|)}$ denote the substack with S-points consisting of those (f,D) such that $f^{-1}(X^{\bullet}/G) \cap D = \emptyset$. Since D is faithfully flat over S, the image of $f^{-1}(X^{\bullet}/G) \cap D$ in S is open. Therefore, Z is a closed substack of $\mathcal{M}_X \times C^{(|\check{\Theta}|)}$. Since $\check{\Theta}$ is a multiset without zero, observe that $\mathcal{M}_X^{\check{\Theta}}$ embeds into Z. Now $(f,D) \in Z(k)$ satisfies the property that the support of $D \in C^{(|\check{\Theta}|)}$ is contained in the support of $C - f^{-1}(X^{\bullet}/G)$. We deduce that $\operatorname{pr}_1|_Z: Z \to \mathcal{M}_X$ is proper and quasifinite, hence finite. On the other hand, $\operatorname{pr}_1|_{\mathcal{M}_X^{\check{\Theta}}}$ is injective on k-points, and $(\operatorname{pr}_1|_Z)^{-1}(\operatorname{pr}_1(\mathcal{M}_X^{\check{\Theta}})) = \mathcal{M}_X^{\check{\Theta}}$. From this we deduce that $\operatorname{pr}_1|_{\mathcal{M}_X^{\check{\Theta}}}$ is a locally closed embedding.

A.4. Generic-Hecke modifications. We review the notion of generic-Hecke modifications between quasimaps introduced in [GN10, §2.2], applied to our situation. We assume that B acts simply transitively on X° and that H is connected.

A.4.1. Function-theoretic analog. We explain the idea behind the generic-Hecke modifications at the level of sets; this construction appears in the geometric Langlands program in the construction of Whittaker models, cf. [Gai15, §5.3.1].

We use the notation of §3.1.3. For any finite subset $\underline{v} \subset |C|$, let $\mathbb{A}^{\underline{v}} = \prod'_{v' \notin \underline{v}} F_{v'}$, $\mathbb{A}_{\underline{v}} = \prod_{v' \in v} F_{v'}$ and similarly for $\mathbb{O}^{\underline{v}}$, $\mathbb{O}_{\underline{v}}$. Then we can consider the set

(A.5)
$$G(\mathbb{A}^{\underline{v}})/G(\mathbb{O}^{\underline{v}}) \times H(\mathbb{A}_{v})/H(\mathbb{O}_{v})$$

which maps to $H(\Bbbk)\backslash G(\mathbb{A})/G(\mathbb{O})$ in two ways: (i) by projecting along the first factor to $(H(\Bbbk)\times_{H(\mathbb{A}_v)}H(\mathbb{O}_{\underline{v}}))\backslash G(\mathbb{A}^{\underline{v}})/G(\mathbb{O}^{\underline{v}})\subset H(\Bbbk)\backslash G(\mathbb{A})/G(\mathbb{O})$ and (ii) by projecting to

$$H(\Bbbk)\backslash \left(G(\mathbb{A}^{\underline{v}})/G(\mathbb{O}^{\underline{v}})\times H(\mathbb{A}_{\underline{v}})/H(\mathbb{O}_{\underline{v}})\right)\subset H(\Bbbk)\backslash G(\mathbb{A})/G(\mathbb{O}).$$

Every meromorphic quasimap (element of $H(\mathbb{k})\backslash G(\mathbb{A})/G(\mathbb{O})$) belongs to the image of the second projection for some \underline{v} . If H is connected, then by weak approximation $H(\mathbb{A}_{\underline{v}}) = H(\mathbb{k})H(\mathbb{O}_{\underline{v}})$ so every quasimap also belongs to the image of the first projection. Thus, the union of (A.5) over all finite subsets v defines a groupoid acting on the set of quasimaps.

A.4.2. We define the ind-stack $\mathcal{H}_{H,\mathcal{M}_X}$ of generic-Hecke modifications to be the stack classifying data

$$(\mathfrak{P}_G^1, \mathfrak{P}_G^2, \sigma_1, \sigma_2; \underline{v}, \tau)$$

where $(\mathcal{P}_{G}^{i}, \sigma_{i}) \in \mathcal{M}_{X}$, $\underline{v} \in \operatorname{Sym} C$ is a divisor with support \underline{v} contained in the non-degenerate locus $\sigma_{i}^{-1}(X^{\bullet} \times^{G} \mathcal{P}_{G}^{i})$ for both i = 1, 2 and τ is an isomorphism of G-bundles

$$\tau: \mathcal{P}_G^1|_{C-\underline{v}} \cong \mathcal{P}_G^2|_{C-\underline{v}}$$

such that the following diagram commutes

$$C - \underline{v} \xrightarrow{\sigma_1} \mathcal{P}_G^1 \overset{G}{\times} X|_{C - \underline{v}}$$

$$\downarrow^{\tau}$$

$$\mathcal{P}_G^2 \overset{G}{\times} X|_{C - v}.$$

Note that the definition only depends on the support of \underline{v} and not its multiplicities.

We call a generic-Hecke modification trivial if the isomorphism τ extends to an isomorphism over all of C. We have the natural projections

$$\mathfrak{M}_X \stackrel{h^{\leftarrow}}{\leftarrow} \mathfrak{H}_{H,\mathfrak{M}_X} \stackrel{h^{\rightarrow}}{\rightarrow} \mathfrak{M}_X,$$

and $\mathcal{H}_{H,\mathcal{M}_X} \to \operatorname{Sym} C$. By definition, the generic-Hecke correspondence preserves the strata $\mathcal{M}_X^{\check{\Theta}}$.

Define a smooth generic-Hecke correspondence to be any stack U equipped with smooth maps

$$\mathfrak{M}_X \stackrel{h_U^{\leftarrow}}{\leftarrow} U \stackrel{h_U^{\rightarrow}}{\rightarrow} \mathfrak{M}_X$$

such that there exists a map $U \to \mathcal{H}_{H,\mathcal{M}_X}$ such that the following diagram commutes

$$\mathcal{M}_{X} \xleftarrow{h^{\leftarrow}} \mathcal{H}_{H,\mathcal{M}_{X}} \xrightarrow{h^{\rightarrow}} \mathcal{M}_{X}$$

Call a smooth generic-Hecke correspondence U trivial if the image of $U \to \mathcal{H}_{H,\mathcal{M}_X}$ consists of trivial generic-Hecke modifications.

Define a morphism of smooth generic-Hecke correspondences to be a map $p: U_1 \to U_2$ such that $h_{U_2}^{\leftarrow} \circ p = h_{U_1}^{\leftarrow}$ and $h_{U_2}^{\rightarrow} \circ p = h_{U_1}^{\rightarrow}$.

A.4.3. Generic-Hecke equivariant sheaves. We define a generic-Hecke equivariant perverse sheaf on \mathcal{M}_X to be an object $\mathcal{F} \in \mathcal{P}(\mathcal{M}_X)$ of the category of perverse sheaves on \mathcal{M}_X equipped with isomorphisms

$$\phi_U: h_U^{\leftarrow *}(\mathfrak{F}) \stackrel{\sim}{\to} h_U^{\rightarrow *}(\mathfrak{F})$$

for every smooth generic-Hecke correspondence U, satisfying some natural conditions (see [GN10, §2.3] for details).

For any $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$, the uniqueness of the IC complex endows $\operatorname{IC}_{\widetilde{\mathfrak{M}}_X^{\check{\Theta}}}$ with the structure of a generic-Hecke equivariant perverse sheaf. (In general, when H is connected, the condition of generic-Hecke equivariance is a *property*, not additional structure, of a perverse sheaf on \mathfrak{M}_X by [GN10, Proposition 3.5.2].)

A.4.4. Fix $\check{\theta} \in \mathfrak{c}_X^- - 0$. Recall that by definition $\mathfrak{M}_X^{\check{\theta}}$ is a substack of $\mathfrak{M}_X \times C$. For a fixed $v_0 \in |C|$, define $\mathfrak{M}_{X,v_0}^{\check{\theta}} := \mathfrak{M}_X^{\check{\theta}} \times_C v_0$ to be the *based* stratum consisting of maps $C \to X/G$ such that $C - v_0$ maps to $H \setminus \mathsf{pt}$ and the G-valuation at v_0 is $\check{\theta}$. Observe that the generic-Hecke modifications preserve the substack $\mathfrak{M}_{X,v_0}^{\check{\theta}}$.

Proposition A.4.5 ([GN10, Proposition 3.5.1]). If H is connected, then all geometric points of $\mathcal{M}_{X,v_0}^{\theta} \subset \mathcal{M}_X$ are equivalent under the equivalence relation generated by the generic-Hecke correspondences.

The preimage of $C \subset \operatorname{Sym} C$ in $\mathcal{H}_{H,\mathcal{M}_X}$ may be realized as the twisted product of an open substack of $\mathcal{M}_X \times C$ with the affine Grassmannian Gr_H . From this it is not hard to deduce that all geometric points of $\mathcal{M}_{X,v_0}^{\check{\theta}}$ are equivalent under *smooth* generic-Hecke correspondences.

A.4.6. Recall the G-Hecke action defined in Proposition 5.2.3. By construction, this action commutes with the generic H-Hecke modifications (in our set-theoretic description above, we are considering the action of right $G(F_{v'})$ -translation at some point $v' \neq v_0$):

Lemma A.4.7. The map $\operatorname{act}_C: \mathcal{M}_X \times \operatorname{\overline{Gr}}_{G,C}^{\check{\theta}} \to \mathcal{M}_X \times C$ is equivariant with respect to the generic-Hecke modifications away from the marked point in C.

A.4.8. *Proof of Proposition 3.1.7.* Proposition A.4.5 allows us to consider the based strata with generic-Hecke modifications as an analog of the stratification by orbits of a group action. This will be the idea behind the proof of Proposition 3.1.7, together with the following fact:

Theorem A.4.9 ([Kal05, Theorem 2]). Let S be a smooth stratum in an algebraically stratified scheme of finite type over $k = \mathbb{C}$. Then the subset of points in S that do not satisfy Whitney's condition B form a constructible subset of dimension strictly lower than dim S.

Thus, if we show that any two points in a given stratum have neighborhoods in \mathcal{M}_X that are smooth locally isomorphic and compatible with the stratification, Theorem A.4.9 will imply that every point must satisfy Whitney's condition B.

Fix a connected component of $\mathcal{M}_X^{\check{\Theta}}$ for some $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X - 0)$. Since Whitney's condition B is local in the smooth topology, it suffices by Corollary 3.5.2 to show that every point in $\mathcal{Y}^{\check{\lambda},\check{\Theta}}$ satisfies Whitney's condition for all $\check{\lambda} \in \mathfrak{c}_X$. Now the graded factorization property of \mathcal{Y} allows us to to reduce to the case where $\check{\Theta} = [\check{\theta}]$ is singleton. By Proposition 4.2.3, we may replace our curve C with $\mathbb{A}^1 = \mathbb{P}^1 - \infty$, i.e., $\mathcal{Y}^{\check{\lambda}} = \mathcal{Y}^{\check{\lambda}}(\mathbb{A}^1)$. Now \mathbb{A}^1 acts on itself by translation, which induces an \mathbb{A}^1 -action by automorphisms on $\mathcal{Y}^{\check{\lambda}}(\mathbb{A}^1)$. This action allows us to move the

degenerate point of any $y \in \mathcal{Y}^{\check{\lambda},\check{\theta}}(\mathbb{A}^1)$ to $v_0 = 0 \in \mathbb{A}^1$, i.e., the map $y : \mathbb{A}^1 \to X/B$ sends $\mathbb{A}^1 - 0 \to X^{\bullet}/B$ and the G-valuation at v_0 equals $\check{\theta}$. Thus, we are reduced to showing that any point in $\mathcal{Y}^{\check{\lambda},\check{\theta}}$ with degenerate point at v_0 satisfies Whitney's condition.

Embed $\mathbb{A}^1 = \mathbb{P}^1 - \infty \subset \mathbb{P}^1$ so we also have an open embedding $\mathcal{Y}^{\check{\lambda}}(\mathbb{A}^1) \subset \mathcal{Y}^{\check{\lambda}}(\mathbb{P}^1)$. Lemma 3.5.4 shows that $\mathcal{Y}^{\check{\lambda}}(\mathbb{P}^1)$ is smooth locally isomorphic to $\mathcal{M}_X = \mathcal{M}_X(\mathbb{P}^1)$ in a way that preserves strata and degenerate points. Therefore, we may reduce to checking that any point in $\mathcal{M}_{X,v_0}^{\check{\theta}}$ satisfies Whitney's condition B. Proposition A.4.5 implies that all such points are equivalent under smooth generic-Hecke correspondences (which preserve strata), so either they all satisfy or fail to satisfy Whitney's condition. By Theorem A.4.9, we deduce that every point satisfies Whitney's condition B.

The same argument as above also shows that the closure of any stratum in \mathcal{M}_X must equal a union of strata.

A.4.10. Arbitrary characteristic. Let the characteristic of k be arbitrary.

Proposition A.4.11. For any $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_X^- - 0)$, the complex $\operatorname{IC}_{\overline{\mathcal{M}}_X^{\check{\Theta}}}$ is constructible with respect to the fine stratification of \mathcal{M}_X .

Proof. The argument is exactly the same as in the proof of Proposition 3.1.7 above, except that we replace the use of Theorem A.4.9 with the fact that for any connected smooth stratum S in \mathcal{M}_X , there exists some open $U \subset S$ such that $H^i(\mathrm{IC}_{\overline{\mathcal{M}}_X^{\Theta}})|_U$ is a local system for all i. We implicitly use the uniqueness of the IC complex, which in particular implies that $\mathrm{IC}_{\overline{\mathcal{M}}_X^{\Theta}}$ is generic-Hecke equivariant.

A.5. **Proof of Theorem 5.1.5.** For $\check{\Theta} \in \operatorname{Sym}^{\infty}(\mathfrak{c}_{X}^{-} - 0)$, let $P_{\mathcal{L}^{+}G}(\overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}})$ denote the category of perverse sheaves on $\overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$ that are equivariant with respect to the action of $(\mathcal{L}^{+}G)_{C(|\check{\Theta}|)}$, considered as a group scheme over $\operatorname{Sym} C$ (defined in §A.1). For $\mathcal{F} \in P(\mathcal{M}_{X})$ and $\mathcal{G} \in P_{\mathcal{L}^{+}G}(\overline{\operatorname{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}})$, we can form the twisted external product

$$\mathfrak{F}\tilde{\boxtimes}\,\mathfrak{G}\in\mathrm{P}(\mathfrak{M}_X\,\tilde{\times}\,\overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}})$$

with respect to the projections of $\widehat{\mathcal{M}_X} \times^{\mathcal{L}^+ G} \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$ to \mathcal{M}_X and $\mathcal{L}^+ G \backslash \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$. Then we define the external convolution product by

$$\mathfrak{F} \star \mathfrak{G} := \operatorname{act}_{\mathfrak{M},!}(\mathfrak{F} \tilde{\boxtimes} \mathfrak{G}) \in \mathcal{D}_c^b(\mathfrak{M}_X).$$

We have introduced all the ingredients in the statement of Theorem 5.1.5. The remainder of this section will be devoted to its proof.

Observe that $IC_{\overline{\mathcal{M}}_X^0} \overset{\tilde{\boxtimes}}{\boxtimes} IC_{\overline{Gr}_{G,C}^{\Theta}} = \overset{\tilde{I}C_{\overline{\mathcal{M}}_X^0} \times \overline{Gr}_{G,C}^{\Theta}}{\boxtimes}$, which we will simply denote \mathfrak{IC} for brevity. We have a stratification

$$\overline{\mathbb{M}}_X^0 \, \tilde{\times} \, \overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}} = \bigcup_{\check{\Phi}',\check{\Phi}''} \overline{\mathbb{M}}_X^{\check{\Phi}'} \, \tilde{\times} \, \mathrm{Gr}_{G,\mathring{C}^{\check{\Phi}''}}^{\check{\Phi}''}$$

running over all $\check{\Theta}' \succeq 0$ and $\check{\Theta}'' \geq \check{\Theta}_1$ where $\check{\Theta}$ refines $\check{\Theta}_1$ (the definition of $\check{\Theta}'' \geq \check{\Theta}_1$ is analogous to that of \succeq). Lemma 5.5.3(i) implies that $\operatorname{act}_{\mathcal{M}}^{-1}(\mathcal{M}_X^{\check{\Theta}})$ is contained in the dense open stratum corresponding to $\check{\Theta}' = 0$ and $\check{\Theta}'' = \check{\Theta}$.

A.5.1. By Proposition 3.1.7 we know that $IC_{\overline{\mathcal{M}}_X^0}$ is constructible with respect to the fine stratification of \mathcal{M}_X . Therefore, \mathcal{IC} is constructible with respect to the stratification above, so if we let $L_{\mathrm{glob}}^{\check{\Theta}',\check{\Theta}''}$ denote the *-restriction of \mathcal{IC} to the stratum $\mathcal{M}_X^{\check{\Theta}'} \tilde{\times} \mathrm{Gr}_{G,\hat{C}\check{\Theta}''}^{\check{\Theta}''}$, we have that its cohomology sheaves are local systems. Since $L_{\mathrm{glob}}^{\check{\Theta}',\check{\Theta}''}$ is the restriction of an IC complex, it lives in perverse cohomological degrees ≤ 0 , and the inequality is strict unless $\check{\Theta}' = 0$ and $\check{\Theta}'' = \check{\Theta}$. Since its cohomology sheaves are local systems, this implies that $L_{\mathrm{glob}}^{\check{\Theta}',\check{\Theta}''}$ lives in usual cohomological degrees $\leq -\dim(\mathcal{M}_X^{\check{\Theta}'} \tilde{\times} \mathrm{Gr}_{G,\hat{C}\check{\Theta}''}^{\check{\Theta}''})$, and the inequality is strict unless $\check{\Theta}' = 0$ and $\check{\Theta}'' = \check{\Theta}$. We are abusing notation here: $\mathcal{M}_X^{\check{\Theta}'}$ may not be connected, in which case each connected component should be considered separately.

We abuse notation and also use $L_{\mathrm{glob}}^{\check{\Theta}',\check{\Theta}''}$ to denote its !-extension to $\mathfrak{M}_X \,\tilde{\times} \,\overline{\mathrm{Gr}}_{G,C^{\check{\Theta}}}^{\check{\Theta}}$. Then by the characterizing properties of the intermediate extension (and the fact that \mathfrak{IC} is Verdier self-dual), Theorem 5.1.5 is equivalent to the following assertion:

Proposition A.5.2. For $\check{\Theta}', \check{\Theta}''$ as above, consider the complex $\operatorname{act}_{\mathcal{M},!}(L_{\operatorname{glob}}^{\check{\Theta}',\check{\Theta}''})$. Then:

- (i) It lives in ${}^pD^{\leq -1}(\mathfrak{M}_X)$ unless $\check{\Theta}'=0$ and $\check{\Theta}''=\check{\Theta}$.
- (ii) The *-restriction of $\operatorname{act}_{\mathcal{M},!}(L_{\operatorname{glob}}^{0,\check{\Theta}})$ to $\overline{\mathcal{M}}_X^{\check{\Theta}}-\mathcal{M}_X^{\check{\Theta}}$ lives in perverse cohomological degrees <-1.
- (iii) There is a natural identification $\operatorname{act}_{\mathcal{M},!}(L_{\operatorname{glob}}^{0,\check{\Theta}})|_{\mathcal{M}_X^{\check{\Theta}}}^* \cong \operatorname{IC}_{\mathcal{M}_X^{\check{\Theta}}}$

Point (iii) follows immediately from Theorem 5.1.1(iii). Points (i)-(ii) concern the cohomological degrees of the *-restriction of $\operatorname{act}_{\mathcal{M},!}(L_{\operatorname{glob}}^{\check{\Theta}',\check{\Theta}''})$ to a stratum of $\overline{\mathcal{M}}_X^{\check{\Theta}}-\mathcal{M}_X^{\check{\Theta}}$. To simplify notation we only consider the restriction to a stratum of the form $\mathcal{M}_X^{\check{\eta}}$ where $\check{\eta} \in \mathfrak{c}_X^- - 0$. The general case is proved in the same way by considering multiple points on C simultaneously.

By Theorem 5.1.1 and Proposition 5.6.1, the preimage $\operatorname{act}_{\mathcal{M}}^{-1}(\mathcal{M}_X^{\check{\eta}})$ intersects $\mathcal{M}_X^{\check{\Theta}'} \times \operatorname{Gr}_{G,\mathring{C}^{\check{\Theta}''}}^{\check{\Theta}''}$ only if $\check{\Theta}' = [\check{\theta}']$ and $\check{\Theta}'' = [\check{\theta}'']$ are singletons ($\check{\theta}'$ is allowed to be 0). Moreover we must have $\check{\theta}'' \geq \operatorname{deg}(\check{\Theta})$ and $\check{\eta} \succeq \check{\theta}' + \check{\theta}''$.

A.5.3. Fiber dimension. Fix a point $v \in |C|$ and consider the subscheme $\mathcal{M}_{X,v}$ of maps where there is only one G-degenerate point at v. Let $\mathcal{M}_{X,v}^{\check{\eta}}$ denote the substack of $\mathcal{M}_{X,v}$ where the G-valuation at this degenerate point is $\check{\eta}$. Then the preimage $\operatorname{act}_{\mathcal{M}}^{-1}(\mathcal{M}_{X,v}^{\check{\eta}}) \cap (\mathcal{M}_X^{\check{\theta}'} \times \operatorname{Gr}_{G,C}^{\check{\theta}''})$ is contained in $\mathcal{M}_X^{\check{\theta}'} \times \operatorname{Gr}_{G,v}^{\check{\theta}''}$. The restricted map

(A.6)
$$\operatorname{act}_{\mathcal{M}}: \mathcal{M}_{X}^{\check{\theta}'} \tilde{\times} \operatorname{Gr}_{G,v}^{\check{\theta}''} \to \mathcal{M}_{X,v}$$

is equivariant with respect to generic-Hecke modifications away from v by Lemma A.4.7. Meanwhile these modifications act transitively on the stratum $\mathcal{M}_{X,v}^{\tilde{\eta}}$. Therefore we deduce that the fibers of (A.6) over all points in $\mathcal{M}_{X,v}^{\tilde{\eta}}$ are isomorphic to one another. Let this fiber dimension be denoted d.

We have a special point $\{t^{\check{\eta}}\} = \mathsf{Y}_v^{\check{\eta},\check{\eta}} \to \mathfrak{M}_{X,v}^{\check{\eta}}$ by Corollary 5.5.6. We deduce from Proposition 5.5.5 that the preimage of $\mathsf{Y}_v^{\check{\eta}}$ under the map (A.6) has a stratification by

$$\mathsf{Y}^{\check{\eta}-\check{\nu},\check{\theta}'}\,\tilde{\times}(\mathsf{S}^{\check{\nu}}\cap\mathrm{Gr}_G^{\check{\theta}''})$$

where $\check{\nu}$ runs over the weights of $V^{\check{\theta}''}$. Therefore

(A.7)
$$d \leq \max_{\check{\nu}} (\dim \mathsf{Y}^{\check{\eta} - \check{\nu}, \check{\theta}'} + \langle \rho_G, \check{\nu} - \check{\theta}'' \rangle).$$

A.5.4. Passage to Zastava model. By Corollary 3.5.2, it suffices to prove Proposition A.5.2 after base change to the Zastava model. Consider the fiber product diagram

$$Z^{?,\check{\lambda},\check{\theta}',\check{\theta}''} \longrightarrow y^{\check{\lambda}} \qquad \qquad \downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow \\ M_X^{\check{\theta}'} \tilde{\times} \operatorname{Gr}_{G,C}^{\check{\theta}''} \xrightarrow{\operatorname{act}_{\mathfrak{M}}} \mathfrak{M}_X$$

which is the analog of (5.11) where we allow $v \in |C|$ to vary. We pull back along this diagram for all λ large enough. The corresponding diagram for strata is given by (5.11). We *-pullback with a shift by the fiber dimension of $\mathcal{Y}^{\check{\lambda}} \to \mathcal{M}_X$. With this shift, the discussion of §A.5.1 implies that $L_{\text{glob}}^{\check{\Theta}',\check{\eta}}$ goes to a complex

$$L_{\text{flag}}^{?,\check{\lambda},\check{\theta}',\check{\theta}''} \in \mathcal{D}_c^b(Z^{?,\check{\lambda},\check{\theta}',\check{\theta}''}),$$

which lives in usual cohomological degrees $\leq -\dim(Z^{?,\check{\lambda},\check{\theta}',\check{\theta}''})$, and the inequality is strict unless $\check{\theta}'=0$ and $\check{\theta}''=\check{\theta}$. The usual cohomology sheaves of $L^{?,\check{\lambda},\check{\theta}',\check{\theta}''}_{\mathrm{flag}}$ are local systems. To prove Proposition A.5.2 it is enough, by the definition of the perverse t-structure, to prove

the following:

Lemma A.5.5. Let $\check{\theta}', \check{\theta}'', \check{\eta}, \check{\lambda}, d$ be as above. Then we have

$$-\dim(Z^{?,\check{\lambda},\check{\theta}',\check{\theta}''}) + 2d \le -\dim \mathcal{A}^{\check{\lambda},\check{\eta}}$$

and the inequality is strict if $\check{\theta}' = 0$ and $\check{\theta}'' \neq \check{\eta}$.

Note that the statement of the lemma has nothing to do with $\dot{\theta}$.

Proof. By §5.3, we may assume that $\mathfrak{c}_{X^{\bullet}} = \mathbb{N}^{\mathcal{D}}$. Now it makes sense to talk about len($\check{\lambda}$) for $\check{\lambda} \succeq 0$, cf. §5.4. By Lemma 6.2.1 and Corollary 6.2.2, we have that $\dim \mathcal{Y}^{\check{\lambda},\check{\eta}} = \operatorname{len}(\check{\lambda} - \check{\eta}) + 1$.

For $\check{\nu}$ a weight of $V^{\check{\theta}''}$, we have $\dim \mathsf{Y}^{\check{\eta}-\check{\nu},\check{\theta}'} \leq \frac{1}{2}(\dim \mathcal{V}^{\check{\eta}-\check{\nu},\check{\theta}'}-1)$ by Proposition 6.1.1, unless $\check{\eta} = \check{\nu}$. If $\check{\theta}' = 0$, then dim $\mathcal{Y}^{\check{\eta} - \check{\nu}, 0} = \operatorname{len}(\check{\eta} - \check{\nu})$. Otherwise dim $\mathcal{Y}^{\check{\eta} - \check{\nu}, \check{\theta}'} = \operatorname{len}(\check{\eta} - \check{\nu}) + 1$. We also have $\langle \rho_G, \check{\nu} - \check{\theta}'' \rangle = \frac{1}{2} \operatorname{len}(\check{\nu} - \check{\theta}'')$. Therefore (A.7) implies that

(A.8)
$$d \le \frac{1}{2} \operatorname{len}(\check{\eta} - \check{\theta}' - \check{\theta}'')$$

and the inequality is strict if $\check{\theta}' = 0$ and $\check{\eta}$ is not a weight of $V^{\check{\theta}''}$.

In the case $\check{\theta}' = 0$ and $\check{\eta}$ is a weight of $V^{\check{\theta}''}$ not equal to $\check{\theta}''$, we claim the inequality (A.8) above is still strict. Indeed, equality can only hold if $\check{\eta} = \check{\nu}$ and an open subvariety of $\mathsf{Y}^{0,0} \, \check{\times} (\mathsf{S}^{\check{\eta}} \cap \mathrm{Gr}_G^{\check{\theta}''})$ is sent to the special point $t^{\check{\eta}}$ under (A.6). However we know from Lemma 5.5.11 that every irreducible component of $S^{\check{\eta}} \cap Gr_G^{\check{\theta}''}$ generically maps to the stratum $\mathcal{M}_G^{\check{\theta}''}$. Thus if $\check{\eta} \neq \check{\theta}''$, the inequality (A.8) is strict.

Thus to prove the lemma it suffices to show that

$$\operatorname{len}(\check{\lambda}-\check{\theta}'-\check{\theta}'')+1=\operatorname{len}(\check{\eta}-\check{\theta}'-\check{\theta}'')+\operatorname{len}(\check{\lambda}-\check{\eta})+1\leq \dim(Z^{?,\check{\lambda},\check{\theta}',\check{\theta}''}).$$

Proposition 5.5.5 (twisted by C) implies that $Z^{?,\check{\lambda},\check{\theta}',\check{\theta}''}$ has a stratification by

$$\mathcal{Y}^{\check{\lambda}-\check{\nu},\check{\theta}'}\,\tilde{\times}(\mathsf{S}^{\check{\nu}}\cap\mathrm{Gr}^{\check{\theta}''})\,\tilde{\times}\,C,$$

where $\check{\nu}$ ranges over weights of $V^{\check{\theta}^{\prime\prime}}$ (in particular, $\check{\nu} \geq \check{\theta}^{\prime\prime}$). By Corollary 3.5.2(ii), if $\mathcal{Y}^{\check{\lambda} - \check{\nu}, \check{\theta}^{\prime}}$ maps to a connected component M of $\mathcal{M}_X^{\check{\theta}^{\prime}}$, then $\mathcal{Y}^{\check{\lambda} - \check{\theta}^{\prime\prime}, \check{\theta}^{\prime}}$ must map to the same connected

component. Therefore $Z^{?,\check{\lambda},\check{\theta}',\check{\theta}''}$ is irreducible with dense open stratum $\mathcal{Y}^{\check{\lambda}-\check{\theta}'',\check{\theta}'}\times C$. Therefore $\dim Z^{?,\check{\lambda},\check{\theta}',\check{\theta}''}=\dim(\mathcal{Y}^{\check{\lambda}-\check{\theta}'',\check{\theta}'})+1\geq \ln(\check{\lambda}-\check{\theta}'-\check{\theta}'')+1$, as desired.

This completes the proof of Theorem 5.1.5.

References

- [AB09] Sergey Arkhipov and Roman Bezrukavnikov, Perverse sheaves on affine flags and Langlands dual group, Israel J. Math. 170 (2009), 135–183, With an appendix by Bezrukavnikov and Ivan Mirković. MR 2506322
- [ABB+05] S. Arkhipov, A. Braverman, R. Bezrukavnikov, D. Gaitsgory, and I. Mirković, Modules over the small quantum group and semi-infinite flag manifold, Transform. Groups 10 (2005), no. 3-4, 279– 362. MR 2183116
- [AT05] Ivan V. Arzhantsev and Dmitri A. Timashev, On the canonical embeddings of certain homogeneous spaces, Lie groups and invariant theory, Amer. Math. Soc. Transl. Ser. 2, vol. 213, Amer. Math. Soc., Providence, RI, 2005, pp. 63–83. MR 2140714
- [BBDG18] A. A. Beilinson, J. Bernstein, P. Deligne, and O. Gabber, Faisceaux pervers, Analysis and topology on singular spaces, I (Luminy, 1981), Astérisque, vol. 100, Soc. Math. France, Paris, 2018, pp. 5–171. MR 751966
- [BD96] A. A. Beilinson and V. G. Drinfeld, Quantization of Hitchin's fibration and Langlands' program, Algebraic and geometric methods in mathematical physics (Kaciveli, 1993), Math. Phys. Stud., vol. 19, Kluwer Acad. Publ., Dordrecht, 1996, pp. 3–7. MR 1385674
- [BD04] Alexander Beilinson and Vladimir Drinfeld, *Chiral algebras*, American Mathematical Society Colloquium Publications, vol. 51, American Mathematical Society, Providence, RI, 2004. MR 2058353
- [Bei87a] A. A. Beilinson, How to glue perverse sheaves, K-theory, arithmetic and geometry (Moscow, 1984–1986), Lecture Notes in Math., vol. 1289, Springer, Berlin, 1987, pp. 42–51. MR 923134
- [Bei87b] _____, On the derived category of perverse sheaves, K-theory, arithmetic and geometry (Moscow, 1984–1986), Lecture Notes in Math., vol. 1289, Springer, Berlin, 1987, pp. 27–41. MR 923133
- [BFG06] Alexander Braverman, Michael Finkelberg, and Dennis Gaitsgory, Uhlenbeck spaces via affine Lie algebras, The unity of mathematics, Progr. Math., vol. 244, Birkhäuser Boston, Boston, MA, 2006, Erratum added in https://arxiv.org/pdf/math/0301176v4.pdf, pp. 17–135. MR 2181803
- [BFGM02] A. Braverman, M. Finkelberg, D. Gaitsgory, and I. Mirković, Intersection cohomology of Drinfeld's compactifications, Selecta Math. (N.S.) 8 (2002), no. 3, 381–418. MR 1931170
- [BFGM04] ______, Erratum to: "Intersection cohomology of Drinfeld's compactifications", Selecta Math. (N.S.) 10 (2004), no. 3, 429–430. MR 2099075
- [BFO12] Roman Bezrukavnikov, Michael Finkelberg, and Victor Ostrik, Character D-modules via Drinfeld center of Harish-Chandra bimodules, Invent. Math. 188 (2012), no. 3, 589–620. MR 2917178
- [BG01] Alexander Braverman and Dennis Gaitsgory, Crystals via the affine Grassmannian, Duke Math. J. 107 (2001), no. 3, 561–575. MR 1828302
- [BG02] A. Braverman and D. Gaitsgory, Geometric Eisenstein series, Invent. Math. 150 (2002), no. 2, 287–384. MR 1933587
- [BG08] Alexander Braverman and Dennis Gaitsgory, Deformations of local systems and Eisenstein series, Geom. Funct. Anal. 17 (2008), no. 6, 1788–1850. MR 2399084
- [BL95] Arnaud Beauville and Yves Laszlo, Un lemme de descente, C. R. Acad. Sci. Paris Sér. I Math. 320 (1995), no. 3, 335–340. MR 1320381
- [BLV86] M. Brion, D. Luna, and Th. Vust, Espaces homogènes sphériques, Invent. Math. 84 (1986), no. 3, 617–632. MR 837530
- [BM20] Victor Batyrev and Anne Moreau, Satellites of spherical subgroups, Algebr. Geom. 7 (2020), no. 1, 86–112. MR 4038405
- [BNS16] A. Bouthier, B. C. Ngô, and Y. Sakellaridis, On the formal arc space of a reductive monoid, Amer. J. Math. 138 (2016), no. 1, 81–108. MR 3462881
- [BNS17] _____, Erratum to: "On the formal arc space of a reductive monoid", Amer. J. Math. 139 (2017), no. 1, 293-295. MR 3619916
- [Bra03] Tom Braden, Hyperbolic localization of intersection cohomology, Transform. Groups 8 (2003), no. 3, 209–216. MR 1996415
- [Bri07] Michel Brion, The total coordinate ring of a wonderful variety, J. Algebra 313 (2007), no. 1, 61–99.
 MR 2326138

- [BS17] Daniel Bump and Anne Schilling, Crystal bases, World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2017, Representations and combinatorics. MR 3642318
- [BZSV] David Ben-Zvi, Yiannis Sakellaridis, and Akshay Venkatesh, Duality in the relative Langlands program, In preparation.
- [Cam18] Justin Campbell, Nearby cycles of Whittaker sheaves on Drinfeld's compactification, Compos. Math. 154 (2018), no. 8, 1775–1800. MR 3830552
- [Cam19] _____, A resolution of singularities for Drinfeld's compactification by stable maps, J. Algebraic Geom. 28 (2019), no. 1, 153–167. MR 3875364
- [Cas80] W. Casselman, The unramified principal series of p-adic groups. I. The spherical function, Compositio Math. 40 (1980), no. 3, 387–406. MR 571057
- [CG97] N. Chriss and V. Ginzburg, Representation theory and complex geometry, Birkhäuser Boston, Inc., Boston, MA, 1997. MR 1433132
- [CS80] W. Casselman and J. Shalika, The unramified principal series of p-adic groups. II. The Whittaker function, Compositio Math. 41 (1980), no. 2, 207–231. MR 581582
- [Del77] P. Deligne, Théorèmes de finitude en cohomologie ℓ-adique, Cohomologie étale, Lecture Notes in Math., vol. 569, Springer, Berlin, 1977, pp. 233–261. MR 3727439
- [Dri18] V. Drinfeld, Grinberg-Kazhdan theorem and Newton groupoids, 2018, Available at https://arxiv. org/pdf/1801.01046v2.pdf.
- [DS95] V. G. Drinfeld and Carlos Simpson, B-structures on G-bundles and local triviality, Math. Res. Lett. 2 (1995), no. 6, 823–829. MR 1362973
- [FFKM99] Boris Feigin, Michael Finkelberg, Alexander Kuznetsov, and Ivan Mirković, Semi-infinite flags. II. Local and global intersection cohomology of quasimaps' spaces, Differential topology, infinite-dimensional Lie algebras, and applications, Amer. Math. Soc. Transl. Ser. 2, vol. 194, Amer. Math. Soc., Providence, RI, 1999, pp. 113–148. MR 1729361
- [FGI+05] Barbara Fantechi, Lothar Göttsche, Luc Illusie, Steven L. Kleiman, Nitin Nitsure, and Angelo Vistoli, Fundamental algebraic geometry, Mathematical Surveys and Monographs, vol. 123, American Mathematical Society, Providence, RI, 2005, Grothendieck's FGA explained. MR 2222646
- [FM99] Michael Finkelberg and Ivan Mirković, Semi-infinite flags. I. Case of global curve P¹, Differential topology, infinite-dimensional Lie algebras, and applications, Amer. Math. Soc. Transl. Ser. 2, vol. 194, Amer. Math. Soc., Providence, RI, 1999, pp. 81–112. MR 1729360
- [Gai] D. Gaitsgory, What acts on geometric Eisenstein series, Available at http://www.math.harvard.edu/~gaitsgde/GL/WhatActs.pdf.
- [Gai15] Dennis Gaitsgory, Outline of the proof of the geometric Langlands conjecture for GL₂, Astérisque (2015), no. 370, 1–112. MR 3364744
- [GL93] I. Grojnowski and G. Lusztig, A comparison of bases of quantized enveloping algebras, Linear algebraic groups and their representations (Los Angeles, CA, 1992), Contemp. Math., vol. 153, Amer. Math. Soc., Providence, RI, 1993, pp. 11–19. MR 1247495
- [GN10] Dennis Gaitsgory and David Nadler, Spherical varieties and Langlands duality, Mosc. Math. J. 10 (2010), no. 1, 65–137, 271. MR 2668830
- [Gro92] Frank D. Grosshans, Contractions of the actions of reductive algebraic groups in arbitrary characteristic, Invent. Math. 107 (1992), no. 1, 127–133. MR 1135467
- [HK02] Jin Hong and Seok-Jin Kang, Introduction to quantum groups and crystal bases, Graduate Studies in Mathematics, vol. 42, American Mathematical Society, Providence, RI, 2002, MR 1881971
- [Ive76] Birger Iversen, The geometry of algebraic groups, Advances in Math. 20 (1976), no. 1, 57–85. MR 399114
- [Kal05] V. Yu. Kaloshin, A geometric proof of the existence of Whitney stratifications, Mosc. Math. J. 5 (2005), no. 1, 125–133. MR 2153470
- [Kas93] Masaki Kashiwara, The crystal base and Littelmann's refined Demazure character formula, Duke Math. J. 71 (1993), no. 3, 839–858. MR 1240605
- [Kas94] _____, Crystal bases of modified quantized enveloping algebra, Duke Math. J. 73 (1994), no. 2, 383–413. MR 1262212
- [Kas95] _____, On crystal bases, Representations of groups (Banff, AB, 1994), CMS Conf. Proc., vol. 16, Amer. Math. Soc., Providence, RI, 1995, pp. 155–197. MR 1357199
- [KKLV89] Friedrich Knop, Hanspeter Kraft, Domingo Luna, and Thierry Vust, Local properties of algebraic group actions, Algebraische Transformationsgruppen und Invariantentheorie, DMV Sem., vol. 13, Birkhäuser, Basel, 1989, pp. 63–75. MR 1044585

- [KKM+92] Seok-Jin Kang, Masaki Kashiwara, Kailash C. Misra, Tetsuji Miwa, Toshiki Nakashima, and Atsushi Nakayashiki, Affine crystals and vertex models, Infinite analysis, Part A, B (Kyoto, 1991), Adv. Ser. Math. Phys., vol. 16, World Sci. Publ., River Edge, NJ, 1992, pp. 449–484. MR 1187560
- [Kno91] Friedrich Knop, The Luna-Vust theory of spherical embeddings, Proceedings of the Hyderabad Conference on Algebraic Groups (Hyderabad, 1989), Manoj Prakashan, Madras, 1991, pp. 225–249.
 MR 1131314
- [Kno94] _____, The asymptotic behavior of invariant collective motion, Invent. Math. 116 (1994), no. 1-3, 309–328. MR 1253195
- [Kno95] _____, On the set of orbits for a Borel subgroup, Comment. Math. Helv. 70 (1995), no. 2, 285–309.
 MR 1324631
- [Kno14] _____, Spherical roots of spherical varieties, Ann. Inst. Fourier (Grenoble) 64 (2014), no. 6, 2503–2526. MR 3331173
- [Kon95] Maxim Kontsevich, Enumeration of rational curves via torus actions, The moduli space of curves (Texel Island, 1994), Progr. Math., vol. 129, Birkhäuser Boston, Boston, MA, 1995, pp. 335–368. MR 1363062
- [KS17] F. Knop and B. Schalke, The dual group of a spherical variety, Trans. Moscow Math. Soc. 78 (2017), 187–216. MR 3738085
- [KV04] Mikhail Kapranov and Eric Vasserot, Vertex algebras and the formal loop space, Publ. Math. Inst. Hautes Études Sci. (2004), no. 100, 209–269. MR 2102701
- [Laf] Vincent Lafforgue, Quelques calculs reliés à la correspondance de Langlands géométrique pour \mathbb{P}^1 , Available at http://vlafforg.perso.math.cnrs.fr/files/geom.pdf.
- [Lun73] Domingo Luna, Slices étales, Sur les groupes algébriques, 1973, pp. 81–105. Bull. Soc. Math. France, Paris, Mémoire 33. MR 0342523
- [Lun97] D. Luna, Grosses cellules pour les variétés sphériques, Algebraic groups and Lie groups, Austral. Math. Soc. Lect. Ser., vol. 9, Cambridge Univ. Press, Cambridge, 1997, pp. 267–280. MR 1635686
- [Lus90] G. Lusztig, Canonical bases arising from quantized enveloping algebras. II, no. 102, 1990, Common trends in mathematics and quantum field theories (Kyoto, 1990), pp. 175–201 (1991). MR 1182165
- [LV83] D. Luna and Th. Vust, Plongements d'espaces homogènes, Comment. Math. Helv. 58 (1983), no. 2, 186–245. MR 705534
- [MV07] I. Mirković and K. Vilonen, Geometric Langlands duality and representations of algebraic groups over commutative rings, Ann. of Math. (2) 166 (2007), no. 1, 95–143. MR 2342692
- [Pop86] V. L. Popov, Contractions of actions of reductive algebraic groups, Mat. Sb. (N.S.) 130(172) (1986), no. 3, 310–334, 431. MR 865764
- [Ras] Sam Raskin, Chiral principal series categories II: the factorizable Whittaker category, Available at https://web.ma.utexas.edu/users/sraskin/cpsii.pdf.
- [Ric77] R. W. Richardson, Affine coset spaces of reductive algebraic groups, Bull. London Math. Soc. 9 (1977), no. 1, 38–41. MR 437549
- [Sak08] Yiannis Sakellaridis, On the unramified spectrum of spherical varieties over p-adic fields, Compos. Math. 144 (2008), no. 4, 978–1016. MR 2441254 (2010d:22026)
- [Sak12] _____, Spherical varieties and integral representations of L-functions, Algebra Number Theory 6 (2012), no. 4, 611–667, Updated version at https://arxiv.org/pdf/0905.4245.pdf. MR 2966713
- [Sak13] _____, Spherical functions on spherical varieties, Amer. J. Math. 135 (2013), no. 5, 1291–1381.
 MR 3117308
- [Sak18] _____, Inverse Satake Transforms, Geometric Aspects of the Trace Formula (Cham), Springer International Publishing, 2018, pp. 321–349.
- [Sch15] Simon Schieder, The Harder-Narasimhan stratification of the moduli stack of G-bundles via Drinfeld's compactifications, Selecta Math. (N.S.) 21 (2015), no. 3, 763–831. MR 3366920
- [Sch16] ______, Geometric Bernstein asymptotics and the Drinfeld-Lafforgue-Vinberg degeneration for arbitrary reductive groups, 2016.
- [Sch18] _____, Picard-Lefschetz oscillators for the Drinfeld-Lafforgue-Vinberg degeneration for SL₂, Duke Math. J. 167 (2018), no. 5, 835–921. MR 3782063
- [SGA73] Groupes de monodromie en géométrie algébrique. II, Lecture Notes in Mathematics, Vol. 340, Springer-Verlag, Berlin-New York, 1973, Séminaire de Géométrie Algébrique du Bois-Marie 1967– 1969 (SGA 7 II), Dirigé par P. Deligne et N. Katz. MR 0354657
- [Ste65] Robert Steinberg, Regular elements of semisimple algebraic groups, Inst. Hautes Études Sci. Publ. Math. (1965), no. 25, 49–80. MR 180554
- [SV17] Yiannis Sakellaridis and Akshay Venkatesh, Periods and harmonic analysis on spherical varieties, Astérisque (2017), no. 396, viii+360. MR 3764130

- [Tim11] Dmitry A. Timashev, Homogeneous spaces and equivariant embeddings, Encyclopaedia of Mathematical Sciences, vol. 138, Springer, Heidelberg, 2011, Invariant Theory and Algebraic Transformation Groups, 8. MR 2797018
- [VK78] È. B. Vinberg and B. N. Kimelfeld, Homogeneous domains on flag manifolds and spherical subgroups of semisimple Lie groups, Funktsional. Anal. i Prilozhen. 12 (1978), no. 3, 12–19, 96. MR 509380
- [Wan] J. Wang, Universal local acyclicity, Available at https://www.jonathanpwang.com/notes/ULA.pdf.
- [Was96] B. Wasserman, Wonderful varieties of rank two, Transform. Groups 1 (1996), no. 4, 375–403. MR 1424449
- [Zhu17] Xinwen Zhu, An introduction to affine Grassmannians and the geometric Satake equivalence, Geometry of moduli spaces and representation theory, IAS/Park City Math. Ser., vol. 24, Amer. Math. Soc., Providence, RI, 2017, pp. 59–154. MR 3752460